A Model for Developing Retrofitting Strategies for Office Buildings

Chenyang Li

A thesis submitted in fulfilment of the requirement for the degree of Doctor of Philosophy

> University of Technology Sydney Australia

> > November 2022

Certificate of Original Authorship

I certify that the work in this thesis has not previously been submitted for a degree nor has it been submitted as part of requirements for a degree except as fully acknowledged within the text.

I also certify that the thesis has been written by me. Any help that I have received in my research work and the preparation of the thesis itself has been acknowledged. In addition, I certify that all information sources and literature used are indicated in the thesis.

This research is supported by an Australian Government Research Training Program Scholarship.

Production Note: Signature of student: Signature removed prior to publication.

Date: 7 December 2022

Acknowledgements

There are many people who contributed to this thesis in various ways.

I would like to express my most sincere gratitude to my principal supervisor Associate Professor Grace Ding. Without her, I would not have started my academic career and had the opportunity to study for a PhD degree. During the past five years, Grace kept inspiring me with her passion for and commitment to work. She has always gone out of her way to impart knowledge of sustainable development and construction to me and her other students. Whenever I encountered difficulties in writing the thesis, she was always quick to find out what the problem was and helped me analyse the causes. Her supervision, guidance and support have made me who I am today.

I also would never have survived the past five years without my co-supervisor Professor Goran Runeson. He has a unique ability to transmit both his outstanding understanding of academic research and an open-minded attitude towards life. I remember the first time I met Goran as my co-supervisor. I was very impressed by his sense of humour and his sharp judgment. He can always come up with innovative ideas and think ahead about what to do in each stage. He likes to share his resources and is generous with his time in reading my writing and discussing problems. During the period of the COVID pandemic, he was willing to meet me face-to-face to discuss my research and thesis. I cannot think of any more words to express my gratitude and appreciation to him.

I would also like to express my gratitude to Professor Xiaoyu Ying of Zhejiang University City College who is also my co-supervisor. Since he is in China, we did not see each other often, but he was always there to help me when I had problems. When I decided to conduct a case study in China, he provided me with substantial background knowledge about sustainable retrofitting in China. When I established the mathematical models, he referred me to mathematical professionals to help with my questions. I give many thanks to him for his support and guidance during my PhD journey. I am also very thankful to all other participants in the research.

Last but not least, I must thank my father Heng Li, my mother Ailian Huang, my mother-inlaw Zhimin Yang, and my husband Shu Li for their continuous encouragement and support. The past five years of my PhD study have been like a dream. During these five years, I became a mother to my two lovely children, Myles and Liam. We experienced the terrible period of COVID together. My mother came to Australia twice from China to help me with the children so that I could focus on my research. My husband kept encouraging me to undertake the degree while he has made great sacrifices for me and our family. I express my deepest thanks to them for their patience, tolerance and endless support.

I owe a great debt to all of you.

Table of Contents

Certific	ate of Original Authorshipi
Acknow	vledgementsii
Table o	f Contentsiv
List of	Tablesxi
List of	Figures xv
List of .	Abbreviationsxvii
Abstra	ct xx
Chapte	r 1. Introduction1
1.1.	Introduction1
1.2.	Research background1
1.3.	Research significance
1.4.	Research gaps
1.5.	Research questions
1.6.	Research aim and objectives
1.7.	Research methodology
1.8.	Thesis structure
1.9.	Summary17
Chapte	r 2. Sustainable development of construction18
2.1.1	ntroduction
2.2.1	Definition of sustainable development18
2.3. S	Significance of sustainable construction
2.4. S	Sustainability dimensions

2.4.1. Three-dimensional model – triple-bottom line approach	
2.4.2. Relationship among the three sustainability pillars	
2.4.3. Weak, strong, and very strong view of sustainable development	
2.5. Challenges of developing sustainable construction	
2.5.1. High initial cost	
2.5.2. A lack of attention to the social dimension	
2.5.3. Difficulty of measuring trade-offs between different sustainability different sustainabil	mensions34
2.5.4. A lack of knowledge of sustainable construction	
2.6. Summary	
Chapter 3. Sustainable retrofitting of office buildings	
3.1. Introduction	
3.2. Current performance of existing office buildings	
3.2.1. Environmental performance	
3.2.2. Economic performance	
3.2.3. Social performance	
3.3. Sustainable retrofitting	
3.3.1. Definition of retrofitting	
3.3.2. Reasons for retrofitting	50
3.3.3. Benefits and barriers to retrofitting	
3.4. Common retrofitting activities for office buildings	55
3.4.1. Install and/or upgrade insulation of building envelopes (A1)	59
3.4.2. Install green roof (A2)	61
3.4.3. Upgrade windows (A3 & A4)	
3.4.4. Install building management control system (A5)	
3.4.5. Upgrade lifts to more energy efficient ones (A6)	65
3.4.6. Upgrade HVAC system (A7 & A8)	65
3.4.7. Upgrade lighting system (A9–A11)	67
3.4.8. Install solar/PV-assisted units (A12)	68
3.4.9. Upgrade water system (A13–A15)	
3.5. Interruption to building tenants	
3.6. Summary	

Chapter 4. Sustainability assessment methods for buildings	
4.1. Introduction	
4.2. Environmental impact assessment	
4.2.1. Life cycle assessment	73
4.2.2. Green building rating system	
4.3. Economic impact assessment	
4.3.1. Life cycle costing	
4.3.2. Cost benefit analysis	
4.4. Social impact assessment	
4.4.1. Social cost benefit analysis	
4.4.2. Social life cycle assessment	
4.5. Multi-dimensional sustainability model	
4.5.1. Integrate two pillars into sustainability assessment	
4.5.2. Life cycle sustainability assessment	
4.5.3. Muti-criteria decision making	
	110
4.6. Summary	
4.6. Summary Chapter 5. Research methodology	
 4.6. Summary Chapter 5. Research methodology 5.1. Introduction 	
 4.6. Summary Chapter 5. Research methodology 5.1. Introduction 5.2. The research process 	
 4.6. Summary Chapter 5. Research methodology 5.1. Introduction 5.2. The research process	
 4.6. Summary Chapter 5. Research methodology	
 4.6. Summary Chapter 5. Research methodology	
 4.6. Summary Chapter 5. Research methodology	
 4.6. Summary Chapter 5. Research methodology	
 4.6. Summary Chapter 5. Research methodology	
 4.6. Summary Chapter 5. Research methodology 5.1. Introduction 5.2. The research process 5.2.1. Epistemology and ontology 5.2.2. Theoretical perspectives 5.2.3. Research methodology 5.2.4. Data collection methods 5.3. Research design 5.3.1. Research inquiry approaches 5.3.2. Research process 	112 114 114 114 115 116 118 120 125 130 130 134
 4.6. Summary Chapter 5. Research methodology 5.1. Introduction 5.2. The research process 5.2.1. Epistemology and ontology 5.2.2. Theoretical perspectives 5.2.3. Research methodology 5.2.4. Data collection methods 5.3.1. Research design 5.3.1. Research inquiry approaches 5.3.2. Research process 	112 114 114 114 115 116 118 120 125 130 130 134 140
 4.6. Summary Chapter 5. Research methodology 5.1. Introduction 5.2. The research process 5.2.1. Epistemology and ontology 5.2.2. Theoretical perspectives 5.2.3. Research methodology 5.2.4. Data collection methods 5.3.1. Research design 5.3.2. Research process 5.4. Summary 	112 114 114 114 115 116 118 120 121 120 121 120 121 120 121 120 121 122 123 130 131 134 140 g strategies for

6.1. Introduction	142
6.2. Model development	142
6.3. Set retrofitting goals and objectives	146
6.4. Identify problems of the proposed building	147
6.5. Propose retrofitting activities	149
6.6. Develop assessment criteria	
6.7. Evaluate performance of retrofitting activities	
6.7.1. Evaluate performance and convert to numerical data	
6.7.2. Weight assessment criteria	
6.7.3. Normalise measurements for criteria aggregation	167
6.8. Develop retrofitting strategies	171
6.8.1. Ceteris paribus	171
6.8.2. A binary linear mathematical model for developing retrofitting strateg	gies 172
6.9. Conduct sensitivity analysis	
6.10. Choose the "best" solution	179
6.11. Summary	
Chapter 7. Case study – Data collection	
7.1. Introduction	
7.2. Rationale behind the case study	
7.3. Background information for the case study	
7.4. Data collection strategy for the case study	
7.4.1. Stage 1 – Questionnaire survey	
7.4.2. Stage 2 – Focus groups	
7.5. Estimate weights by AHP method	
7.5.1. Design a questionnaire form to collect individual judgment	
7.5.2. Combine individual judgment matrix to reach group decision	
7.5.3. Consistency test	
7.5.4. Outcome of weights generation	

7.6. Summary	
Chapter 8. Case study – Develop retrofitting strategies	
8.1. Introduction	
8.2. Developing retrofitting strategies based on the operating model	
8.2.1. Set retrofitting goals and identify problems of the case building	
8.2.2. Develop retrofitting activities	
8.2.3. Develop assessment criteria	
8.2.4. Evaluate performance of retrofitting activities	
8.2.5. Develop retrofitting strategies	
8.2.6. Conduct sensitivity analysis	
8.2.7. Determine the most preferred retrofitting strategy	
8.3. Outcomes and discussions	
8.3.1. Validity of the conceptual model	
8.3.2. Applicability and flexibility of the conceptual model	
8.4. Summary	
Chapter 9. Summary and conclusions	
9.1. Introduction	
9.2. Overview of aims and objectives	
9.2.1. Current performance of existing office buildings in triple-bottom l	ine aspects 283
9.2.2. Existing assessment methods for building retrofitting	
9.2.3. Identifying assessment criteria for sustainable retrofitting	
9.2.4. Developing a conceptual model for deciding retrofitting strategies	for office
buildings	
9.2.5. Verifying the conceptual model and illustrating the implementation	n of the model
9.2.6. Conclusion	
9.3. Summary of the research	
9.4. Contribution to knowledge	
9.5. Limitations of this research	

9.6. Recommendations for future research	
9.6.1. Develop a decision model to select suitable building materials for retrofi	tting 294
9.6.2. Include a post-occupancy evaluation in the model	
9.6.3. Adapt the model to be used for the public sector	
9.6.4. Extend the model	
9.7. Conclusion	
Appendices	298
Appendix A. Trade-off between three attributes	
Appendix B-1. Information letter and consent form for online questionnaire surve	ey 301
Appendix B-2. Questionnaire survey (English version)	304
Appendix B-3. Questionnaire survey (Chinese version)	309
Appendix B-4. Chi-Square test for association between region and commonly use	d
retrofitting activities	
Appendix B-5. Chi-Square test for association between importance level of enviro	onmental
assessment criteria and whether respondents have ever conducted projects involv	ing office
buildings	
Appendix B-6. Chi-Square test for association between importance level of social	ļ
assessment criteria and whether respondents have ever conducted projects involv	ving office
buildings	
Appendix C-1. Information letter for focus group discussions (English version)	
Appendix C-2. Consent form for focus group discussions (English version)	
Appendix D-1. AHP survey sample	
Appendix D-2. Establishing the matrix of group judgment	
Appendix D-3. Weight generation	350
Appendix E-1. Insulation performance	355
Appendix E-2. Environmental impact assessment	
Appendix E-3. Economic performance assessment	
Appendix E-4. Social performance assessment	383

Appendix E-5. Normalisation	
References	

List of Tables

Table 1.1. Thesis structure	16
Table 2.1. A summary of concepts and definitions of sustainable development	20
Table 2.2. Environmental pollution from the building sector	22
Table 2.3. Typical measures in triple-bottom line	26
Table 2.4. Studies about barriers to develop sustainable construction	36
Table 3.1. Common retrofitting alternatives for office buildings	57
Table 3.2. Limit U-value for buildings in category B in China	60
Table 3.3. Data for constructing a typical extensive roof	61
Table 3.4. Amount of substances released to the air per 1kg of polymer	62
Table 3.5. U-value, SHGC and cost of different options of glazing types and window framin Australia.	nes 63
Table 3.6. Disruption levels of retrofitting activities	71
Table 4.1. Functional unit options for LCA studies of buildings	74
Table 4.2. Available LDI databases	79
Table 4.3. Common building LCA tools	81
Table 4.4. Summary of four green building rating schemes	86
Table 4.5. The life expectancy of building materials and building systems	.94
Table 4.6. Characteristics of MODM and MADM1	10
Table 5.1 Relationship among the four elements of research process	16
Table 5.2. Summary of advantages and limitations of quantitative methods	22
Table 5.3. Characteristics of qualitative research and researchers	23
Table 5.4. Summary of advantages and limitations of qualitative methods	24
Table 5.5. Summary of advantages and limitations of quantitative data collection methods1	26
Table 5.6. Summary of advantages and limitations of qualitative data collection methods1	27
Table 5.6. Summary of advantages and limitations of qualitative data collection methods1 Table 5.7. Key differences between qualitative and quantitative research1	27 28

Table 6.2. Assessment criteria of retrofitting151
Table 6.3. Carbon emission coefficients
Table 6.4. Rating scale for subjective issues 162
Table 6.5. Pros and cons of common sensitivity analysis method
Table 7.1. Characteristics of the case building
Table 7.2. Respondents in different roles by gender
Table 7.3. Characteristics of respondents. 192
Table 7.4. Related experience of respondents
Table 7.5. Type of buildings ever retrofitted by respondents
Table 7.6. Chi-Square test between commonly used retrofitting activities and region196
Table 7.7. The parties who should take responsibility for improving sustainability
Table 7.8. Drivers of sustainable retrofitting by role group 199
Table 7.9. Importance rating of different aspects of sustainability by role group
Table 7.10. Importance rating scores on the three sustainability dimensions by age group201
Table 7.11. Importance rating of different criteria under environmental dimension by role
group
Table 7.12. Importance rating of different criteria under social dimension by role group203
Table 7.13. Assessment criteria by questionnaire survey
Table 7.14. Categories of participants in the focus group discussion 207
Table 7.15. Details about focus group discussions 207
Table 7.16. Summary of focus group discussions 209
Table 7.17. Modification of retrofitting activities and assessment criteria
Table 7.18. Rating score on the effectiveness of retrofitting activities
Table 7.19. Rating score on importance of assessment criteria
Table 7.20. Fundamental score of the AHP method
Table 7.21. Pairwise comparison matrix for individual judgement
Table 7.22. Group judgement matrix on the environmental assessment criteria in the retrofitting
stage
Table 7.23. Random index in AHP

Table 7.24. Consistency of the constructed matrixes of group judgement
Table 7.25. Weights of assessment criteria for the case study
Table 7.26. Weights of the three sustainability dimensions for the case study
Table 8.1. Categories of office buildings in Hangzhou
Table 8.2. Heat transfer coefficient of walls, roofs and windows of the case building227
Table 8.3. Test of background noise of the case building
Table 8.4. Proposed retrofitting activities for the case building
Table 8.5. Assessment criteria for the case study
Table 8.6. Materials data for retrofitting the case building
Table 8.7. Energy consumption and carbon emissions by operating construction equipment.
Table 8.8. Embodied energy (EE) and embodied carbon (EC) emissions of installed building
materials
Table 8.9. Evaluation results of annual energy consumption and carbon emissions of building
service system after retrofitting
Table 8.10. Recurrent embodied energy (REE) and embodied carbon (REC) emissions in the
operation stage
Table 8.11. Difference in energy consumption and carbon emission by retrofitting
activities
Table 8.12. Evaluation results of water saving
Table 8.13. Evaluation results on the use of reusable/recyclable materials
Table 8.14. Evaluation results on waste generation
Table 8.15. Parameters for interior daylight simulation in Ecotect
Table 8.16. Simulation results of interior daylight
Table 8.17. Assessment results of environmental impacts by proposed retrofitting activities
Table 8.18. Assessment of economic performance
Table 8.19. Equivalent annual cost or saving by retrofitting activities
Table 8.20. Operation cost saving by retrofitting activities
-

Table 8.21. Assessment of social impacts by retrofitting activities.	265
Table 8.22. Normalisation of environmental estimation in retrofitting stage	267
Table 8.23. The total weighted scores in the three sustainability dimensions	.268
Table 8.24. Establishing Equation 8.1 in Excel.	270
Table 8.25. The developed retrofitting strategy for the case building	271
Table 8.26. The assessment results based on the national green building standard	.272
Table 8.27. Results of sensitivity analysis.	274
Table 8.28. Alternative strategies by relaxing and tightening the project budget	276

List of Figures

Figure 2.1. Conceptual model of sustainable construction
Figure 2.2. The eight key areas covered by the concept of sustainable development25
Figure 2.3. Three dimensions of sustainable development
Figure 3.1. Extended life cycle of buildings by retrofit
Figure 3.2. Sinking stack theory
Figure 4.1. Modular information for different stages of the building assessment in LCA76
Figure 4.2. LCI process applied to a building system77
Figure 4.3. The framework of environmental LCIA80
Figure 4.4. S-LCA decision tree101
Figure 4.5. Assessment framework of S-LCA by reference scale approach102
Figure 4.6. Basic steps in MODM and MADM111
Figure 5.1. Basic elements of research process115
Figure 5.2. Process of research design
Figure 5.3. Research flowchart135
Figure 5.4. Explanatory sequential design137
Figure 6.1. Retrofit process at pre-retrofit stage143
Figure 6.2. The conceptual model of deciding retrofitting strategies for office buildings145
Figure 6.3. Common retrofitting activities for office buildings149
Figure 6.4. Decision diagram to guide the selection of sensitivity analysis methods177
Figure 7.1. Frequency of commonly undertaken retrofitting activities by respondents195
Figure 7.2. Frequency of using sustainable assessment tools
Figure 7.3. The flow of AHP method214
Figure 7.4. The hierarchy structure of assessment criteria
Figure 8.1. Allocation of electricity usage by the building
Figure 8.2. Aluminium sun shading slats
Figure 8.3. Layout of Level 2 of the case building.

Figure 8.4. Layout of Level 10 of the case building	253
Figure 8.5. Ecotect simulation results of Level 2	255
Figure 8.6. Ecotect simulation results of Level 10	255
Figure 8.7. An iterative process of information refining	280

List of Abbreviations

ABCB	Australia Building Codes Board
AC	Alternating current
AHP	Analytic Hierarchy Process
ANP	Analytic Network Process
ASGB	Assessment Standard of Green Building
ASHRAE	American Society of Heating Refrigerating and Air-Conditioning Engineers
BCR	Benefit/cost ratio
BEES	Building for Environmental and Economic Sustainability
BEEX	Building Energy Exchange
BIM	Building Information Modelling
BMCS	Building management control system
BMS	Building management system
BREEAM	Building Research Establishment Environmental Assessment Method
BSI	British Standards Institution
CBA	Cost benefit analysis
CFD	Computation fluid dynamics
CIB	International Council for Research and Innovation in Building and
	Construction
DC	Direct current
EAC	Equivalent annual cost
EC	Embodied carbon
ECBA	Environmental cost benefit analysis
EE	Embodied energy
eGRID	Emission & Generation Resource Integrated Database
ELCA	Environmental life cycle assessment
ELCC	Environmental life cycle costing

EPA	Environmental Protection Authority (U.S.)
EPD	Environmental product declarations
FU	Functional unit
GBRS	Green building rating system
GHG	Greenhouse gases
GP	Goal programming
HMD	Head-mounted display
HVAC	Heating, ventilation and air conditioning
I-O	Input-Output
ICE	Inventory of Carbon and Energy
IEA	International Energy Agency
IEQ	Indoor environmental quality
IESNA	Illuminating Engineering Society of North America
IRR	Internal rate of return
IVS	Immersive virtual simulation
LCA	Life cycle assessment
LCC	Life cycle costing
LCI	Life cycle inventory
LCIA	Life cycle impact assessment
LCSA	Life cycle sustainability assessment
LDPE	Low density polyethylene
LEED	Leadership in Energy and Environmental Design
MADM	Multi-attribute decision making
MCDM	Multi-criteria decision making
MLOP	Multiple linear objective programming
MNLOP	Multiple nonlinear objective programming
MODM	Multi-objective decision making

MOHURD	Ministry of Housing and Urban-Rural Development of the People's Republic of China
MRL	Machine-room-less technology
NCC	National Construction Code
NIST	National Institute of Standards and Technology (US)
NPV	Net present value
OAT	One-At-a-Time
PBP	Payback period
PMV	Predicted mean vote
PP	Semi-crystalline polypropylene
PRPs	Performance reference points
PV	Photovoltaic
RSP	Reference study period
SDSN	Sustainable Development Solutions Network
SETAC	Society of Environmental Toxicology and Chemistry
S-LCA	Social life cycle assessment
SLCA	Social cost benefit analysis
SHGC	Solar heat gain coefficient
SOTNBS	Survey Office of The National Bureau of Statistics in Zhejiang
TBL	Triple-bottom line
UN	United Nations
UNEP	United Nations Environment Programme
VAT	Value Added Tax
VAV	Variable air volume
WCED	World Commission on Environment and Development
WLC	Weighted linear combination
WWR	Window/wall ratio
ZPBS	Zhejiang Provincial Bureau of Statistics

Abstract

A considerable amount of material, water, energy and other natural resources are invested in the building sector. The poor performance in energy efficiency of most existing buildings and the relatively low rate of new and more efficient construction means that it will be a long time before new buildings with better performance can replace existing poor building stock. Compared to demolition and construction of new buildings, retrofitting of existing buildings may be a faster method to modernise the existing stock and mitigate unfavourable impacts on the natural environment from the building sector. However, most existing sustainability assessment methods and decision-making frameworks focus on the environmental and economic performance of buildings without much consideration of the social dimension. While retrofitting may be the best chance for existing buildings to achieve sustainability, it is necessary to consider all three sustainability dimensions when retrofitting existing buildings to achieve economic growth, protect the natural environment, and increase social wellbeing.

This study develops a model for deciding retrofitting strategies for office buildings to improve their sustainability performance. Different with most other existing decision-making models for retrofitting strategies which only consider the environmental and economic dimensions, the model developed in this study integrates all the environmental, economic and social dimensions into the decision-making process of retrofitting strategies. The retrofitting strategies developed by the model can maximise improvement of existing buildings in these three dimensions within project constraints and meet retrofitting goals at the same time. This is realised via a process from conceptualisation to operationalisation. First, a conceptual model for deciding retrofitting strategies for office buildings from a triple-bottom line perspective is developed based on literature review. Then, the conceptual model is converted to an operating model to suit local situations for sustainable retrofitting. A survey and focus group discussions are conducted to collect opinions about locally suitable retrofitting activities and assessment criteria from professionals in the construction and property management sectors as well as the key stakeholders of retrofitting. Finally, a case study is conducted in which the operating model is used to develop retrofitting strategies for the case building. With suitable potential retrofitting strategies developed for the case building, the validity of the conceptual model is verified. Meanwhile, the case study illustrates the process of quantifying and using the conceptual model to develop retrofitting strategies for an office building.

Chapter 1. Introduction

1.1. Introduction

Previous studies have indicated that building-related activities pose a serious threat to the natural environment, and most of the negative impacts are from the operation phase of buildings. Therefore, improving the sustainability performance of existing buildings is crucial to achieve sustainable development for the whole of society. This study aims to develop a model for deciding retrofitting strategies for office buildings using the triple-bottom line approach. The retrofitting strategies developed by the model are able to maximise the improvement of the environmental, economic and social dimensions within project constraints. Therefore, it is expected that the developed model can help improve the overall sustainability performance of the whole building sector. This chapter discusses the general research background, research significance, research gaps, research questions, aim and objectives, methodology, and thesis structure to provide a brief introduction to the whole research.

1.2. Research background

Existing buildings cannot be regarded only as a space for human activities. They are also a significant consumer of energy and resources while generating greenhouse gases (GHG) and waste. In the past decades, the rapid development of the building industry has caused many problems including the depletion of resources for construction and global warming by the emission of greenhouse gases from the operation and maintenance of existing buildings (Ardente et al. 2011; Chau, Tse & Chung 2010). Moreover, the poor performance of most existing buildings and the relatively low rate of new construction mean there is a long period before new buildings with better performance can replace existing poor building stock (Wilkinson 2012). Therefore, how to alleviate the unfavourable effects of existing buildings while not interrupting the services they provide has become a significant issue for all communities.

Existing buildings are potentially important in solving environmental problems because most of them have low efficiency in energy and resource use (Asadi et al. 2014; Che et al. 2019; Heo, Choudhary & Augenbroe 2012). Considerable quantities of materials, energy, water and

other natural resources are invested in the building sector annually with much waste and greenhouse gas output (Mikulić, Bakarić & Slijepčević 2016). Small and medium-sized commercial buildings in developed nations represent around 50% of energy consumed by all types of buildings (Juan, Gao & Wang 2010; Liang et al. 2018). Office buildings in particular are one of the largest energy consumers and greenhouse gas emitters (Krstić-Furundžić, Vujošević & Petrovski 2019) because the complex building systems of office buildings, including heating, ventilation and air conditioning (HVAC), lighting, and security systems, require extensive energy and natural resources to maintain their normal function (Luo et al. 2018). Moreover, different types of system failures may occur in office building services, increasing the energy used (Juan, Gao & Wang 2010). As for emissions, the greenhouse gases released from existing commercial buildings are 41% of total greenhouse gas emissions by buildings globally (Chidiac et al. 2011; Hong et al. 2020; Wu et al. 2014; Yuan, Nian & Su 2019). The data indicates that the environmental performance of existing buildings may not meet the demand for sustainability and is a serious threat to the natural environment.

Throughout the whole life span of buildings there are several stages, including design, construction, operation and end-of-life. Energy demands and adverse environmental impacts from buildings vary in different phases, but most impacts are concentrated in the operation stage (Ardente et al. 2011; Toosi et al. 2020). As discussed above, the operation stage is responsible for the majority of energy consumption and greenhouse gas emissions during a building's whole life span. While embodied energy becomes increasingly significant due to the continuous development of technology, reductions in embodied energy only occur in new buildings with advanced energy efficiency equipment and management strategies (Simonen 2014). For this reason, improving existing buildings' environmental performance is an opportunity to significantly reduce energy impacts.

Apart from environmental performance, the economic performance of existing buildings is also not ideal. Due to the large amount of energy consumed in the operation stage and high and increasing energy prices, the operation stage is regarded as the most critical stage due to its huge economic impacts during the whole life of buildings (Oregi, Hernandez & Hernandez 2017). Past studies have estimated that if a building's life is 50 years the operation and maintenance costs account for approximately 75%–80% of the total costs in a building's life cycle (Hauashdh, Jailani & Rahman 2022).

In addition, indoor environmental quality is also related to the economic performance of existing office buildings. Sound indoor environmental quality of existing office buildings in the UK can contribute to up to 20% improvement of occupant productivity, which is equivalent to £135 billion per year (Horr et al. 2016). In contrast, the poor indoor environmental quality of existing office buildings reduces maximum possible rents, and increases utility bills, resulting in a long payback period for construction investment (Oregi, Hernandez & Hernandez 2017). Due to the growing recognition of the importance of healthy living and work environmental quality in the US, especially for commercial buildings, has been increasing, and is estimated at hundreds of billions of dollars annually (Kats 2003; Kibert 2016).

Sustainability has been regarded as a significant marketing device by large companies because it can be used to describe the production methods of economic activities, and also to improve the quality of consumption and attributes of capital investment (Eichholtz, Kok & Quigley 2013). The environmental benefits of improving sustainability also contribute to economic growth for individuals and the public (Cetiner & Edis 2014; Mikulić, Bakarić & Slijepčević 2016). In contrast, poor environmental performance brings a weaker financial return. With higher energy prices expected in the future, the investment opportunities for energy-efficient retrofitting are likely to increase and are expected to become one of the main driving forces for owners to retrofit existing buildings (Amstalden et al. 2007; Pombo, Rivela & Neila 2016).

Regarding the social dimension, the social impacts of the building industry do not receive much consideration compared to environmental and economic aspects (Santos et al. 2017). However, the productivity of employees and corporate culture, two aspects under the social dimension, are affected by building performance (Zuo & Zhao 2014). For a long time, researchers have not had complete agreement on what social elements should be included in the context of sustainable buildings. Watson et al. (2016) believed that the social context can be expressed as building user group dynamics, which is an integration of institutional norms, culture and management. Wilkinson and Remøy (2017) and Parida et al. (2021) identified social contexts

for office buildings, which are work-related flow, comfort, wellbeing, and job satisfaction of occupants. Liu and Qian (2019) stated that the social context of a building project is supposed to meet the diverse requirements of multiple stakeholders involved in the whole project process including construction teams, suppliers, end users, and local communities. Indeed, construction activities generate various social impacts on individuals, communities, and even the whole of society. The wide coverage and the intangible characteristics of the social context make the social dimension one of the most challenging when aiming for sustainable buildings overall (Dendena & Corsi 2015; Zuo & Zhao 2014).

As an indirect way to preserve the cultural and societal assets that have been embodied in existing buildings, building retrofitting should not be carried out without social impact assessment (Jagarajan et al. 2017). The sound practice of social performance assessment should not only achieve effective engagement of different stakeholders, and increase understanding of change and capacities to respond to change, but also help to enhance the lives of vulnerable and disadvantaged people (Esteves, Franks & Vanclay 2012). Due to the increasing extent of retrofitting construction all over the world, social impact assessment for existing buildings is becoming more urgent, and more comprehensive social assessment models are required.

To solve the environmental, economic and social problems mentioned above, two solutions can be adopted: demolition or retrofitting. Buildings are a long-lasting product that need regular maintenance and renewal, but still cannot avoid becoming obsolete or redundant due to the continual change in demand and regulations. Eventually, obsolete buildings are demolished and new buildings are built complying with current standards and regulations, but the obsolescence stage can be delayed by retrofitting (Ongpenga et al. 2020; Solanki, Rastogi & Paul 2022). The reason for preferring to retrofit is that fewer resources and energy are required compared to demolition and new construction. Moreover, the embodied energy of existing buildings already exists. If buildings' performance can be enhanced by retrofitting, demolition and then new construction are a waste of both energy and time (Thomsen & Van der Flier 2011).

Considering the extensive number of existing office buildings around the world, the assessments for evaluating existing office buildings' performance and generating appropriate

retrofitting strategies are crucial for sustainable building development. However, current assessment systems for existing office buildings rarely cover all of the three sustainable pillars: environmental, economic and social dimensions (Phillips et al. 2020). The (perceived) high initial cost is one of the main obstacles to sustainable construction (Aghimien, Aigbavboa & Thwala 2019; Gunduz & Almuajebh 2020). In fact, the problem is that those assessment frames do not take a long-term perspective and are not able to balance different stakeholders' benefits so that the short-term payback and the risk of concerns by stakeholders mean retrofitting is not attractive (Maltz, Bi & Bateman 2018).

Due to the complex process of building retrofitting, it is difficult to develop appropriate retrofitting strategies that can embrace all the environmental, economic and social dimensions from a long-term perspective. This study aims to contribute a solution to define methods of improving the sustainability performance of existing office buildings, reducing the pollution, increasing economic growth, and enriching the social value. The research reviews current assessment methods for environmental, economic and social impacts of construction and buildings. Existing retrofitting activities that can reduce environmental damage and increase economic benefits and social value are also examined and analysed. The collected data and analysed information are then used to develop a conceptual model for deciding retrofitting strategies, and a case study is conducted to verify the validity of the conceptual model and illustrate how to use the model to develop retrofitting strategies for an office building.

1.3. Research significance

There is no doubt that building retrofitting can make a significant difference in dealing with environmental problems, but environmental concerns should not be the only focus of sustainable retrofitting (Ardente et al. 2011; Chidiac et al. 2011). Most of the existing assessment frameworks for upgrading existing buildings concentrate on the environmental and/or economic dimensions, but the social dimension is rarely considered (Ball 1999; Fatourehchi & Zarghami 2020; Mickaityte et al. 2008; Yung & Chan 2012).

Indeed, social sustainability has been recognised as the most difficult element of sustainability due to the wide range covered and its "soft" characteristics (Watson et al. 2016). Therefore, researchers prefer to study this dimension on its own instead of integrating it with the other two

pillars. As a result, using assessment tools to balance the environmental and economic performance is a common way to create a green built environment (Escrig-Olmedo et al. 2017; Fouche & Crawford 2017). However, retrofitting is the second, and probably the last, chance to pursue sustainability for existing buildings (Kohler et al. 2010). Therefore, the assessment process should include a broad context to account for all the life stages of a building's life cycle. The absence of the social dimension may hinder the understanding of the sustainability of buildings from a full life cycle perspective (Simonen 2014).

Numerous sustainable building assessment methods for new buildings have already been established and well developed due to the rapid growth of sustainable building development. However, it is still a great challenge to develop assessment models for existing buildings, especially for integrating the three sustainability dimensions into the assessment process (Andersen, Jensen & Ryberg 2021; Filippi & Sirombo 2015). This research aims to develop a model for deciding retrofitting strategies for office buildings. The developed model is able to address the environmental, economic and social issues of existing office buildings and balance the trade-offs between the three sustainability dimensions from a long-term perspective. In this way, the concerns and benefits of different stakeholders can be considered and balanced, so that the selected retrofitting strategy is more appropriate, and can increase interest in retrofitting buildings.

This study contributes to providing a new approach for assessing the value of existing office buildings and deciding suitable retrofitting decisions to reduce the environmental pollution, to increase financial return, and to enrich social value. Meanwhile, all stakeholders' awareness of sustainability can be improved through participating in the assessment process. Ultimately, the sustainable performance of the whole building industry can be improved.

1.4. Research gaps

Most effort has been paid to developing new buildings, and it is the reason why few assessment tools are available for building retrofitting. Currently, there are some assessment tools and regulations for assessing the performance of existing buildings, like the life cycle assessment (LCA) for assessing environmental impacts of buildings, the life cycle costing (LCC) tool for evaluating buildings' economic performance, and Europe Standard EN 16309 for assessing

buildings' social performance (BSI 2014). However, these assessment strategies do not adequately provide data or guidance for sustainability improvement by only focusing on one dimension. Filippi and Sirombo (2015) stated that it is time to deeply examine the sustainability performance of existing buildings and facilities.

Three major research problems need to be solved through this research:

- i) Current retrofitting assessment tools are ineffective and inefficient in dealing with sustainability issues due to their failure to integrate all the three sustainability dimensions by just focusing on the environmental and/or economic dimensions.
- ii) To obtain a suitable retrofitting strategy, different decision-making support tools need to be integrated with assessment models. The variation from integrating different tools may generate different results for the same case.
- iii) The process of retrofitting buildings is very complicated and various stakeholders may be involved at the same time, and the benefits and unfavourable impacts from retrofitting are generated at different stages of a building's whole life span. Therefore, how to balance benefits among different stakeholders, or maximise trade-offs between different sustainability dimensions, and evaluate the performance of existing buildings from a long-term perspective should be considered in an assessment framework.

Based on the identified research problems, this study aims to address the gaps by developing:

- i) a conceptual model for deciding retrofitting strategies, which can integrate the environmental, economic and social dimensions of sustainability
- ii) a way to maximise the trade-offs between the three sustainability dimensions from a long-term perspective.

1.5. Research questions

Based on the research problems and gaps discussed above, the research question is:

How can we develop retrofitting strategies that can effectively improve an office building's environmental, economic and social performance in a balanced manner?

To answer this research question, the following sub-questions are formulated:

- i) How can the environmental, economic and social dimensions be considered in retrofitting?
- ii) How can we use the resulting assessment outcomes of retrofitting activities to generate suitable retrofitting strategies?
- iii) How can we balance the improvement on the three sustainability dimensions in a balanced manner?

Through a deeper understanding of building sustainability, the study can determine whether this approach can contribute to improving the sustainable performance of existing office buildings.

1.6. Research aim and objectives

This research aims to develop a model for developing retrofitting strategies for office buildings to improve their sustainability performance. To accomplish this research aim, specific objectives are:

- i) To examine the current performance of existing office buildings in a triple-bottom line aspect that includes the environmental, economic and social dimensions
- ii) To investigate existing assessment methods for building retrofitting
- iii) To identify the process and significant criteria impacting building retrofitting on environmental, economic and social dimensions
- iv) To identify the trade-offs between the three sustainability dimensions, which reflect the conflicts of interest among different stakeholders
- v) To develop a conceptual model that can be used to develop retrofitting strategies based on the evaluation results of potential retrofitting activities
- vi) To verify the conceptual model by converting it to an operating model and quantifying the operating model based on a case study.

1.7. Research methodology

There are two major theoretical perspectives in social science: positivism and interpretivism. In the positivism perspective, positivists seek facts or causes of social phenomena in an

Chapter1. Introduction

objective way avoiding the involvement of individual judgment (which is hard to achieve). In the interpretivism perspective, interpretivists or phenomenologists attempt to understand social phenomena based on personal knowledge, experience and viewpoint. These two theoretical perspectives take different kinds of problems and seek different types of answers, so two different methodologies which are common in social science are developed: quantitative methodology and qualitative methodology (Cresswell 2012; Fellows & Liu 2015; Taylor, Bogdan & DeVault 2015). Quantitative methodology puts considerable trust in numerical data; in contrast, qualitative methodology attempts to use words and observations to describe reality (Amaratunga et al. 2002). Even though there is a clear difference between these two methodologies, qualitative and quantitative methods are not mutually exclusive, and qualitative methods can still be used by positivists to address their research problems (Hughes 2012; Taylor, Bogdan & DeVault 2015).

This study takes the positivism perspective to confirm that sustainable retrofitting can effectively improve existing buildings' environmental, economic and social performance. To solve research questions effectively and efficiently, the mixed method approach is adopted to collect both quantitative and qualitative data. Based on the above discussion, a research proposition can be identified that it is possible to improve sustainability while potentially allowing economic growth and improved social wellbeing by retrofitting existing buildings using the triple-bottom line approach. To test this research proposition, two main tasks need to be achieved in this study:

Task 1: Examine the interaction between existing office buildings and all sustainability dimensions. Understand reasons for sustainable retrofitting, existing assessment methods for building retrofitting, and other relevant background knowledge by reviewing the literature. Based on the knowledge and information from the literature review, develop a conceptual model for deciding retrofitting strategies for office buildings.

Task 2: Verify the conceptual model and illustrate the process of using the model to develop retrofitting strategies for an office building through a case study.

To accomplish the first task, related literature, including journal papers, books, reports and research studies, is reviewed. The topics of reviewed literature are not limited to sustainable buildings, but also theory and practice about how to achieve group consensus, normalisation methods, weighting methods, and so on. The literature review failed to locate any assessment system that enables the user to evaluate the environmental, economic and social dimensions for existing office buildings while determining suitable retrofitting strategies. Based on the literature review, a conceptual model for developing retrofitting strategies for office buildings was developed by using a triple-bottom line approach.

The second task is to verify the developed model and illustrate how to use the model to develop retrofitting strategies for an office building via a case study. The conceptual model is intended to be general, and it needs to be adapted to suit the local situation before applying it to an actual retrofitting project. A case study in China is conducted to illustrate the detailed process from converting the conceptual model to an operating model, and then using the operating model to develop retrofitting strategies for the case building. There are approximately 60 billion m² of floor space in existing buildings in China. Most of these existing buildings have poor operations and energy inefficiency, resulting in high energy consumption and greenhouse gas emissions (Liu, Tan & Li 2020). Based on the study by Guo et al. (2022), operations energy by existing buildings being the largest contributor. The Chinese government is also promoting sustainable retrofitting by launching various policies, which makes China a big market for retrofitting (Liu, Tan & Li 2020). Therefore, China is a suitable place to conduct the case study. Meanwhile, with diverse climate conditions in China, it is also a good opportunity to demonstrate the flexibility of the developed model.

In the case study, a two-stage data collection strategy is adopted to collect opinions about suitable retrofitting activities and assessment criteria for local retrofitting from the broad to the specific. First, a survey is conducted with professionals and key stakeholders in retrofitting in both northern and southern China to collect broad opinions. Following the survey, focus group discussions with local professionals and key stakeholders of retrofitting are organised to consolidate the results of the survey and further modify the retrofitting activities and assessment criteria to be suitable for local use. After finalising retrofitting activities and assessment criteria, the operating model is then implemented to develop retrofitting strategies

for the case building, to maximise the improvement in the environmental, economic and social dimensions while meeting the retrofitting goals within project constraints.

The process from conceptualisation to operationalisation can be regarded as a logically and methodologically correct framework for the work done and to be done in the future. By copying the process, the conceptual model can be adapted according to the specific situation of each retrofitting project and be used to develop retrofitting strategies.

1.8. Thesis structure

The structure of the thesis is represented in Table 1.1. The outline of each chapter is described as follows.

Chapter 1. Introduction

This chapter introduces the background information about this research. By discussing the research significance, research gaps, research problems, and research aim and objectives, this chapter introduces the reasons for undertaking this research – why a model for deciding retrofitting strategies needs to be developed. It also describes the research methodology including how to develop the conceptual model, and how to verify and illustrate the implementation of the developed model.

Chapter 2. Sustainable development of construction

This chapter reviews sustainable development of construction. First, based on existing studies about sustainable development and sustainable construction, the definition of sustainable retrofitting is given. It is a process of realising the reduction of cost of operating a building and increasing people's wellbeing in ways that reduce the deterioration of natural systems. Due to the diverse aspects, the context of sustainable construction can be categorised into different dimensions. The three-dimension model complying with the context of the triple-bottom line – the environmental, economic and social – is widely adopted to assess the sustainability performance of construction. This chapter also discusses the challenges facing sustainable construction that need to be overcome through this study to achieve sustainable retrofitting.

Chapter 3. Sustainable retrofitting of office buildings

This chapter reviews existing studies about sustainable retrofitting. First, the current environmental, economic and social performance of existing office buildings around the world is discussed. Due to the increasing demand for indoor thermal comfort and stricter regulations about building energy consumption and carbon emissions, the environmental and economic performance of existing office buildings is worse than for new buildings. The social dimension does not attract the same attention as the environmental and economic dimensions, leading to a lack of mature social assessment methods. To improve the sustainability performance of existing buildings, two solutions are retrofitting existing buildings or demolishing and replacing them with new buildings. This chapter discusses reasons why retrofitting is preferred over demolition. Retrofitting can deliver a faster improvement of sustainability performance, and also saves the embodied energy and embodied carbon emissions that already exist in existing buildings. Common activities for retrofitting office buildings are summarised by their potential improvement and interruption to building tenants.

Chapter 4. Sustainability assessment methods for buildings

This chapter reviews existing literature about assessment methods of sustainable buildings. To assess the environmental impacts of buildings, life cycle assessment (LCA) is the most popular method of assessing buildings' environmental impacts from a life cycle perspective. Moreover, different green building rating systems, such as the Leadership in Energy and Environmental Design (LEED) and Building Research Establishment Assessment Method (BREAM), can be used to rate the sustainability level of buildings. To assess buildings' economic impacts, life cycle costing (LCC) and cost benefit analysis (CBA) are two popular methods. To assess buildings' social impacts, social cost benefit analysis (SCBA) and social life cycle assessment (SLCA) are two common methods to quantify the social impacts of buildings. These are assessment methods for only one of the three sustainability dimensions.

Often the environmental and economic impact assessment methods are combined as a twodimensional assessment model to assess a building's sustainability performance. Life cycle sustainability assessment (LCSA) is an assessment method consisting of the life cycle assessments in all three sustainability dimensions – LCA, environmental life cycle costing (ELCC), and SLCA. Multi-criteria decision making is also discussed in this chapter due to its ability to integrate the three sustainability dimensions into sustainability performance assessment and generate one overall assessment result to assist in decision making of retrofitting strategies. This chapter discusses benefits and limitations of the assessment methods and justifies why multi-objective decision making, which is a branch of multi-criteria decision making, is selected in this study to develop a conceptual model for deciding retrofitting strategies.

Chapter 5. Research methodology

The chapter discusses the research methodology and research design to explain how this study is conducted. This research aims to develop a model for deciding retrofitting strategies for office buildings and uses a case study to quantify the model. This aim is achieved via a process from conceptualisation to operationalisation. First, a conceptual model is developed based on information and knowledge from the literature review. Then, a case study is conducted to verify the validity of the conceptual model and illustrate how to use the conceptual model to develop retrofitting strategies. There are two parts in the case study. The first part converts the conceptual model to an operating model by modifying retrofitting activities and assessment criteria to suit the local situation. To accomplish the conversion, a survey and three focus group discussions are conducted to collect opinions from local professionals and key stakeholders of retrofitting. The second part implements the operating model to develop retrofitting strategies for the case building. Based on detailed illustration, the process from conceptualisation to operationalisation can be regarded as a logically and methodologically correct framework for the work done and to be done in the future. By copying the process, the conceptual model can be adapted according to the specific situation of each retrofitting project and be used to develop retrofitting strategies.

Chapter 6. Development of a conceptual model for deciding retrofitting strategies for office buildings

This chapter develops a conceptual model for deciding retrofitting strategies for office buildings by using the triple-bottom line approach. The decision-making process in the conceptual model is designed based on multi-objective decision making, which can deal with multiple conflicting objectives. By following the steps in the conceptual model, the developed retrofitting strategies may meet all the identified retrofitting goals and maximise the improvement in the environmental, economic and social dimensions within project constraints.

Chapter 7. Case study - Data collection

To verify the validity and illustrate how to use the conceptual model to develop retrofitting strategies for an office building, a case study of retrofitting an office building in China is conducted. This chapter presents the first part of the case study – data collection to convert the conceptual model to an operating model to deal with locational variation. A two-stage data collection strategy is adopted to collect data from the broad to the specific. Stage one gathers broad opinions about suitable retrofitting strategies and assessment criteria for retrofitting in China. A survey, which can gather quantitative data in a relatively short time, is conducted with professionals and key stakeholders of retrofitting in northern and southern China. To consolidate the results of the survey and further modify the model to be suitable for the case building and other buildings in China, three focus group discussions are conducted with local professionals and key stakeholders. The participants of focus group discussions are also invited to determine weights for the three sustainability dimensions and the assessment criteria for each dimension using the Analytic Hierarchy Process (AHP) method.

Chapter 8. Case study – Quantifying the operating model

With finalised retrofitting activities and assessment criteria, this chapter describes how to use the operating model to develop retrofitting strategies for the case building. By following the steps in the conceptual model, five retrofitting strategies are developed for the case building. The building owners can choose the strategy that can best satisfy their demands as the final retrofitting strategy. The results of the case study are also compared with the Chinese national standard, Assessment Standard for Green Retrofitting of Existing Building (GB/T 51141-2015) (MOHURD 2015) to confirm the effectiveness of the developed model. This chapter discusses and analyses the outcome of the case study to deeply explore how the developed model can improve the sustainability performance of existing buildings.

Chapter 9. Summary and conclusion

This chapter summarises the research process and presents how the research questions are answered, how the research problems are addressed, and how the research aim is reached. It also discusses the general contribution and limitation of this research, and also offers recommendations for future research.
Chapter	Title	Content	Achieved research objective or solved research question
1	Introduction	Introduces research background, research question, research problem, research aim and objectives, and identifies research gap.	
2	Sustainable development of construction	Discusses necessity and limitation of sustainable construction.	Research objective (i), (iv)
3	Sustainable retrofitting of office buildings	Reviews previous literature about sustainable retrofitting regarding reasons for retrofitting, obstacles in conducting retrofitting, and common retrofitting activities for office buildings.	Research objective (i), (iii)
4	Sustainability assessment methods of buildings	Reviews existing methods of assessing sustainability of constructions and building in the environmental, economic and social dimensions, individually or jointly.	Research objective (ii), (iv) Research question (i), (iii)
5	Research methodology	Introduces the methodology and data collection methods used in the research and discusses the benefits and limits of each adopted method. Presents the research design on the research methodology and data collection methods to reach the identified research aim.	
6	Development of a conceptual model for deciding retrofitting strategies for office buildings	Uses the information and knowledge from the literature review to develop a conceptual model to decide retrofitting strategies for office buildings.	Research objective (v) Research question (ii), (iii)
7	Case study – Data collection	Conducts survey and focus group discussions to collect opinions about suitable retrofitting activities and assessment criteria for the local retrofitting of the case building.	Research objective (vi)
8	Case study – Quantifying the operating model	Implements the operating model to develop retrofitting strategies for the case building.	Research objective (vi)
9	Summary and conclusion	Summarises main content of the research, discusses the results of the research, and discusses the outcome of the research.	

Table	1.1.	Thesis	structure
raute	1.1.	1 110 515	Suuciuic

1.9. Summary

Improving the sustainable performance of existing buildings is a challenge for sustainable development, but it can also be an opportunity to make a difference. This research develops a model for deciding retrofitting strategies for office buildings by considering impacts on the three sustainability dimensions – environmental, economic and social dimensions. To build this model, a substantial literature review about sustainable development, sustainable retrofitting and sustainability assessment methods is undertaken. A case study is then conducted to verify the conceptual model and illustrate the detailed process of converting the conceptual model to an operating model and using the operating model to develop retrofitting strategies for the case building.

The outcome of this research is that the developed model can demonstrate the ability to assess multiple facets of sustainability and also to decide suitable retrofitting strategies for office buildings. This research is expected to promote deeper and broader study on building retrofitting by providing a new perspective of sustainability assessment, so that the overall sustainability level of buildings can be improved.

Chapter 2. Sustainable development of construction

2.1. Introduction

This study aims to develop a model for deciding retrofitting strategies for office buildings to improve their sustainability performance. To achieve the research aim, it is essential to investigate the background and tendency of sustainable development, especially in the construction and property sector. In addition, related studies about sustainable retrofitting such as the performance of existing office buildings, common retrofitting activities, assessment methods of sustainable development, and common decision-making frameworks should be reviewed to support the model development. Therefore, a literature review is conducted in Chapter 2 to Chapter 4 to illustrate related studies about sustainable retrofitting. First, existing studies about sustainable development and the triple-bottom line of construction are reviewed in this chapter. Then, the performance of existing office buildings and sustainable retrofitting application is investigated in Chapter 3. In Chapter 4, existing assessment methods and decision-making frameworks of sustainable development are reviewed. The literature review illustrates that it is possible to measure the level of complex changes in the level of sustainable development by adopting the specific view of sustainable development, suitable sustainability assessment methods, and a proper decision-making framework.

In this chapter, existing studies on the concept of sustainable development are reviewed first with analysis of its necessity and limitations. Then, studies about using the triple-bottom line approach to evaluate sustainability performance of construction are investigated. Some other significant aspects that should be considered in using triple-bottom line to achieve sustainable construction are also discussed.

2.2. Definition of sustainable development

The concepts of sustainability and sustainable development are becoming increasingly popular and are widely adopted to label the "green" traits of products, projects and even behaviours. The concept of sustainability is traceable to the forest industry, describing that the speed of harvesting should never be faster than the rate of growth of new trees (Wiersum 1995). The most well-known and widely used definition is given in the Brundtland Report "Our common future" by the World Commission on Environment and Development (WCED) in 1987. In the report, the concept of sustainable development is proposed and defined that humanity should make development sustainable such that it can meet the current generation's needs without jeopardising future generations' ability to meet their own needs (Brundtland 1987). Brundtland (1987) emphasised that the natural environment is a finite resource that places certain limits on development, and humans have the responsibility of preserving and protecting it from depletion for their descendants. Moreover, Brundtland (1987) pointed out that the essence of sustainable development is to realise a harmonious process of change in resource exploitation, investment direction, technology development orientation, and institutional change so that both current and future potential for meeting human needs and aspirations can be improved.

The WCED started a new era of sustainable development, and since then, the concept of sustainable development has been widely disseminated and implemented in various fields (Conte 2018; Sadollah, Nasir & Geem 2020). In past decades, multiple authors have discussed the concept of sustainable development, which now has over 200 definitions (Santos et al. 2019). Sadollah, Nasir and Geem (2020) provided a summary of the concept and definition of sustainable development as shown in Table 2.1. The definition of sustainable development is influenced by how people understand it and why they need it, such as the type of problems they need to solve. Thus, Sadollah, Nasir and Geem (2020) stated that the concept of sustainable development is prone to favouritism and is subjective.

Roots of sustainable development	Points of emphasis	Definition of sustainable development	Specific sustainable development emphasis
Ecological/carrying capacity	Maintenance of natural systems so that they can support human life and wellbeing	Carrying capacity	Optimum and maximum ability of Earth's systems to support human life and wellbeing
Resource/environment	Promoting economic growth only to the extent and in ways that do not cause deterioration of natural systems	Sustainable use of biological resources	Maximum sustainable yield from natural systems, such as forests and fisheries
Biosphere	Concern with the impacts of humans on the health of ecosystems and its ability to support human populations	Sustainable agriculture	Maintaining productivity of farming during and after disturbances such as floods and droughts
Critique of technology	Rejection of the notion that science and technology, by themselves, will protect and save the Earth	Sustainable energy	Renewable alternatives to fossil fuel reliance to produce heat energy
No growth-slow growth	Limit to the ability of the Earth to support the health and wellbeing of ever-growing populations	Sustainable society and economy	Maintaining human systems to support economic and human wellbeing
Ecodevelopment	Adapting business and economic development activities to realities of natural source and environmental limits	Sustainable economic and environmental development	Promoting economic growth only to the extent and in the ways that do not cause deterioration of natural systems

Table 2.1. A	summary of	concepts and	definitions	of sustainabl	le development
--------------	------------	--------------	-------------	---------------	----------------

Source: Adapted from Sadollah, Nasir & Geem 2020

There is a need to point out the difference between sustainability and sustainable development. In most studies, the terms are used interchangeably, but there is a subtle difference between them. Conte (2018) believed that the word "sustainability" represents a level of concept while "sustainable development" is more about the operational level of development. Similarly, Sadollah, Nasir and Geem (2020) explained that sustainability is the capacity for long-term development, and sustainable development is a dynamic process of achieving or considering the development. In this study, sustainable development is adopted since the study is about a process of realising sustainability (performance improvement) of existing buildings via the retrofitting strategies determined by using the model developed in this study (see Chapter 6). The definitions of sustainable society and economy, and sustainable economic and environmental development from Table 2.1 are combined to define sustainable retrofitting for

this study as a process of realising the reduction of cost of operating a building and increasing human wellbeing in ways that reduce deterioration of natural systems.

2.3. Significance of sustainable construction

The substantial consumption of energy and materials, the massive carbon emissions and the waste generation of the construction industry and the built environment make them the key areas in which to achieve sustainable development, thus contributing to the achievement of sustainable development for the whole of society (Amaral et al. 2020). Table 2.2 summarises energy and other resources consumed by buildings in different countries. Greenhouse gas emissions and waste generation from using and maintaining buildings also have harmful impacts on the natural environment. According to Table 2.2, about 30% to 50% of global energy consumption and 40% to 50% of greenhouse gas emissions are from the building sector. Around 30% of resource consumption, 40% of raw materials, and 12% to 25% of water consumption are also generated by the building sector, and it also generates about 25% to 40% of global waste.

Consumption/ emissions	Percentage of consumption/ emissions	Location	Source
Energy consumption	40%~42%	EU	Kolokotsa et al. 2009; Ardente et al. 2011; Cedllura et al. 2013; Asadi et al. 2014; Wang, Xia & Zhang 2014; Mauro et al. 2015; Kylili, Fokaides & Jimenez 2016; Mikulic, Bakaric & Slijepcevic 2016; Oregi, Hernandez & Hernandez 2017; Baumhof et al. 2018
	≈40%	US	Azhar, Brown & Farooqui 2009; Frey et al. 2012; Heo, Choudhary & Augenbroe 2012; Liu, Meng & Tam 2015
	39%	UK	Heo, Choudhary & Augenbroe 2012
	50%~60%	China	Peng, Wang & Zhang 2014
	30%~50%	Global	Lippiatt 1999; Miller & Buys 2008; Kolokotsa et al. 2009; Motawa & Carter 2013; Wang, Xia & Zhang 2014; Zuo & Zhao 2014; Chau, Leung & Ng 2015; Mauro et al. 2015; Dwaikat & Ali 2016; Radziejowska & Orlowski 2016; Oregi, Hernandez & Hernandez 2017; Pomponi & Moncaster 2017; Baumhof et al. 2018; Amaral et al. 2020
Resource consumption	33%	Global	Langston et al. 2008; Dwaikat & Ali 2016
Raw material	35%	EU	Cellura et al. 2013
consumption	40%	Global	Ardente et al. 2011; Eichholtz, Kok & Quiley 2013; Samandar 2015; Pomponi & Moncaster 2017
Water	13%	US	Frey et al. 2012
consumption	12%~25%	Global	Lippiatt 1999; Langston et al. 2008; Chau, Leung & Ng 2015; Samandar 2015; Dwaikat & Ali 2016
Waste	38%	Australia	Zuo & Zhao 2014
generation	25%~40%	Global	Langston et al. 2008; Chau, Leung & Ng 2015; Wong & Zhou 2015; Dwaikat & Ali 2016
GHG emissions	35%~40%	EU	Cellura et al. 2013; Asadi et al. 2014; Mikulic, Bakaric & Slijepcevic 2016
	40%~50%	US	Frey et al. 2012; Asadi et al. 2014; Menassa & Baer 2014
	41%	UK	Chidiac et al. 2011
	40%~50%	Global	Langston et al. 2008; Miller & Buys 2008; Azhar, Brown & Farooqui 2009; Woo, Wilsmann & Kang 2010; Ardente et al. 2011; Wilkinson 2012; Motawa et al. 2013; Chau, Leung & Ng 2015; Samandar 2015; Wong & Zhou 2015; Pomponi & Moncaster 2017

Table 2.2. Environmental pollution from the building sector

Regarding the massive negative impacts of construction and building related activities on the natural environment, researchers from different countries have focused on studying and analysing how to deliver successful sustainable construction and create a harmonious built environment for humans and protect the natural environment.

During the Final Session of the First International Conference of the International Council for Research and Innovation in Building and Construction (CIB) TG 16 on Sustainable Construction, in Florida in 1994, Kibert (1994) defined the objectives of sustainable construction as mitigating harmful impacts on the natural environment and improving indoor environmental quality. The definition of the objectives of sustainable construction is regarded as a milestone of applying the concept of sustainable development in the construction sector (Conte 2018). In addition, Kibert (1994) recognised criteria of sustainable construction that can be used to identify building materials, products and systems. He also established seven principles for sustainable construction: conservation, reuse, renewing/recycling, nature protection, non-toxics, economics, and quality. Integrating the established principles and different life stages of buildings, a conceptual model of sustainable construction was built as shown in Figure 2.1. It shows that sustainable construction has to consider various impacts on different life stages of buildings. From then, creating a sustainable built environment became a primary objective for the construction industry (Conte 2018).





Udomsap and Hallinger (2020) conducted a study to review research on sustainable construction from 1994 to 2018. They analysed 2877 sustainable construction related documents (published in English) from 98 countries by using a bibliometric method. The analysis showed that the interest in sustainable construction is a global phenomenon. Anglo-American-European countries dominate with an increasing interest from Asia (Udomsap & Hallinger 2020).

The first study about sustainable construction was in 1994, but over 80% of articles were published since 2010, which indicates that many areas of sustainable construction still remain to be exploited and studied (Udomsap & Hallinger 2020). By analysing different topics about sustainable construction through 2877 documents, Udomsap and Hallinger (2020) asserted that environmental and economic concerns are the main focuses and that there is a missing emphasis on social sustainability. In particular, there is a lack of standards for measuring and guiding the development of social sustainability in the construction sector. Udomsap and Hallinger (2020) forecast that the increasing research on sustainable construction would more than double in the next ten years. Their study emphasised the significance of sustainable construction, as by realising sustainable construction, it is possible to attain great progress and make a significant contribution to realising sustainable development in the whole of society (Udomsap & Hallinger 2020).

Indeed, buildings play an essential role in linking human daily life and the natural environment. By taking the perspective of a long-lasting coexistence of humans and nature, the improvement of the built environment can both mitigate unfavourable impacts on the natural environment, and also define a way to balance human life and nature (Conte 2018). Therefore, it is crucial to develop sustainable construction and improve the built environment to ensure a long future on the planet.

2.4. Sustainability dimensions

The various definitions of sustainable development indicate that different areas of our life are involved in sustainable development. Abu-Rayash and Dincer (2019) highlighted eight key areas covered by the concept of sustainable development as Figure 2.2 shows. These areas are interlaced in various ways. For instance, the economy field affects public policy while public policy influences the development and management of energy consumption. Their interplay makes it challenging to sort and analyse how human and social activities impact them. Therefore, they are commonly grouped into a manageable number of dimensions, and a series of relevant assessment criteria are identified for each dimension. For example, James (2015) studied urban sustainability and identified four dimensions of urban sustainability: economics, ecology, politics, and culture. Then, seven assessment criteria are identified under each dimension. Based on the evaluation of these assessment criteria, the level of urban sustainability can be measured.



Figure 2.2. The eight key areas covered by the concept of sustainable development

Source: Adapted from Abu-Rayash & Dincer 2019

Another four-dimensional model was implemented in studies by Dincer and Zamfirescu (2018) and Sadollah, Nasir and Geem (2020). Their studies were about sustainable development from the perspective of energy resources. Therefore, sustainable development is regarded as a synthesis of preservation of energy resources, environmental sustainability, economic sustainability, and social sustainability. Sadollah, Nasir and Geem (2020) explained that energy is a crucial factor in poverty reduction and the improvement of living standards. It should be integrated with the other three sustainability dimensions to pursue comprehensive sustainable development.

2.4.1. Three-dimensional model – triple-bottom line approach

Compared to the four-dimensional model, a three-dimensional model of sustainable development by integrating the environmental, economic and social dimensions is more widely recognised and applied (Purvis, Mao & Robinson 2019). Brundtland (1987) stated that economic and social development must be promoted simultaneously to eliminate world poverty

and release limitations by the state of technology and social organisation on the environment's ability. Combined with the emphasis on preserving natural resources, the Brundtland report was believed to be the early proposal of the three-dimensional model of sustainable development (Gimenez, Sierra & Rodon 2012; Klöpffer 2008). One way to integrate environmental, economic and social dimensions to achieve sustainable construction is explained by Conte (2018):

- starting point (environmental dimension): to alleviate negative environmental impact by saving energy, reducing resource consumption, and controlling emissions and waste generation from construction activities
- means (economic dimension): to invest in construction activities to realise sustainable construction and a sustainable built environment
- outcome (social dimension): to achieve long-lasting development for society.

The three-dimensional concept was later adopted in the assessment method of sustainability called triple-bottom line (TBL) by Elkington, which is a new approach to an accounting framework (Elkington 1997). In triple-bottom line, the three main concerns are referred to as people, planet and profit, or the three Ps, by Elkington (1997). Due to the well-known definition of sustainable development by WCED (1987), and following intense debate and discussion about sustainable development, the three Ps were later developed into environmental, economic and social dimensions (Gimenez, Sierra & Rodon 2012). The common assessment criteria under the three dimensions are listed in Table 2.3 based on the study by Savitz (2013).

	Economic	Environmental	Social
res	Sales, profits, return on investment	Pollutants emitted	Health and safety record
asu	Taxes paid	Carbon footprint	Community impacts
l me	Monetary flows	Recycling and reuse	Human rights; privacy
oica	Jobs created	Water and energy use	Product responsibility
Tyl	Supplier relations	Product impacts	Employee relations

Table 2.3. Typical measures in triple-bottom line

Source: Savitz 2013

As a popular measuring approach of sustainable development, triple-bottom line is also implemented to assess sustainable construction performance. Gou and Xie (2017) stated that green buildings are evolving through the push of two main concepts – regenerative design and

triple-bottom line. Regenerative design encourages improvements of positive impacts by design activities instead of reducing negative impacts, while triple-bottom line improves performance of green buildings by including different aspects of sustainable development in a process of assessment and decision making. Mathiyazhagan, Gnanavelbabu and Prabhuraj (2018) developed a sustainable assessment model based on the triple-bottom line approach, which can be used by construction companies in India to select suitable building materials by considering their impacts on environmental, economic and social dimensions. Jiang et al. (2019) conducted a study to analyse benefits and limits of modular prefabricated buildings in the construction stage using the triple-bottom line approach. The study outcomes showed that prefabrication can contribute to environmental improvement by saving 60% of steel, 56% of concrete, and 77% of formwork on site. However, it may cause some negative impacts on economic and social dimensions, including a relatively high initial cost compared to the conventional construction process, and limited labour availability.

In 2015, balanced development in environmental, economic and social dimensions was also advocated in the universal action by the United Nations (UN) – "Transforming our world: the 2030 Agenda for sustainable development" – to combat poverty, protect the planet, and make sure that all people can live in peace and prosperity by 2030 (Anderson et al. 2017; UN 2015). To conclude, the environment, economy and society are commonly recognised as three dimensions or pillars of sustainable development, which should be appropriately assessed and balanced when designing a new product or improving an existing product (Klöpffer 2008; Purvis, Mao & Robinson 2019). Therefore, this study also defines and assesses sustainable retrofitting in the environmental, economic and social dimensions.

2.4.2. Relationship among the three sustainability pillars

Based on the above discussion, it is generally accepted that sustainable development consists of three dimensions: environmental, economic and social. Therefore, it is important to understand the contents of each dimension, and also how the three dimensions are related. Based on the literature, the three sustainability dimensions are often expressed in one of three forms as Figure 2.3 illustrates.



Figure 2.3. Three dimensions of sustainable development

Source: Adapted from Purvis, Mao & Robinson 2019; Thatcher 2013; Yolles 2018

Since the three dimensions always refer to three pillars of sustainable development, they are straightforwardly represented as three pillars of a building, as shown in Figure 2.3 (a) (Purvis, Mao & Robinson 2019; Thatcher 2013). However, this representation is criticised for considering the three dimensions separately and equally important. As discussed above, different aspects of sustainable development interact in different ways, and their importance changes with different stakeholders (Thatcher 2013) which is why more people prefer to show the three dimensions using three circles with overlaps, as shown in Figure 2.3 (b) and (c).

The Venn diagram in Figure 2.3 (b) symbolises the three sustainability dimensions. There are three circles with intersections, and sustainability exists in the area where all three circles overlap (Yolles 2018). The three intersecting circles indicate that the importance level of these three sustainability dimensions may vary in different circumstances. Figure 2.2 (c) symbolises the three dimensions as three concentric circles, showing the interdependence of the three sustainability dimensions. The concentric circles illustrate the relation between the three sustainability dimensions as the social development depends entirely on the environment, and harmonious economic development can be gained only by realising social and environmental priorities (Joumard & Nicolas 2010; Yolles 2018). It emphasises the significance of the environmental dimension, where the damage to the natural environment cannot be compensated for by capital increase. These two representations indicate the interconnection and interdependency among these three sustainability dimensions. Essentially, it is the issue of how the trade-off works between the three dimensions, or more specifically, what is the level of capital that can substitute for natural resources (Biely, Maes & Van Passel 2018; Purvis, Mao & Robinson 2019).

In this study, the representation of three intersecting circles is adopted to illustrate different opinions of decision makers on the level of importance of the three sustainability dimensions. In Figure 2.3 (b), different sizes of overlapping area indicate that the trade-off, or level of substitution, between the three dimensions will change with the different opinions of stakeholders. It mainly depends on how important stakeholders or decision makers of a sustainability project consider each sustainability dimension to be. In this study, the level of importance of the three sustainability dimensions needs to be decided to indicate the emphasis of any retrofitting. In addition, the different level of substitution between the three dimensions can be expressed by different views of sustainable development, as discussed in the following section.

2.4.3. Weak, strong, and very strong view of sustainable development

The previous section discussed the interplay relationship between the three sustainability dimensions. Based on different levels of substitution between capital and natural resources, three views of sustainable development are proposed: weak sustainability, strong sustainability, and very strong sustainability (Ayres, Van den Berrgh & Gowdy 2001; Kuhlman & Farrington 2010).

Weak sustainability implies that although the loss of some natural resources is unavoidable, this can be offset by increased capital (Kuhlman & Farrington 2010). However, this view can result in extreme sensitivity to either natural disturbances or economic shocks (Ayres, Van den Berrgh & Gowdy 2001). An example given by Ayres, Van den Berrgh and Gowdy (2001) can be used to explain the reason. In 1900, people in the island nation of Nauru became very rich from phosphate mining. However, as a result of continuous mining for over 90 years, about 80% of the island had been destroyed by the 1990s. When the Asian financial crisis suddenly happened in 1997, it wiped out most of the nation's savings. The development of Nauru island followed the logic of weak sustainability. It also illustrates that it may be a one-way substitution between the natural resource and capital: once irreplaceable natural resources are transformed into manufactured or human capital, there is no way back to the previous situation (Ayres, Van den Berrgh & Gowdy 2001).

Standing in absolute opposition to the view of weak sustainability is very strong sustainability. It indicates that every natural environment component or subsystem, as well as every species and physical stock, must be protected (Ayres, Van den Berrgh & Gowdy 2001). However, very strong sustainability seems impossible in a culture that has created an industrial economy that relies on primary resources (Ayres, Van den Berrgh & Gowdy 2001).

Between weak sustainability and very strong sustainability, there is strong sustainability. In the view of strong sustainability, for those natural resources that are essential for our survival and the arrival of their tipping points cannot be prevented or slowed down by the current state of technological knowledge, a minimum amount of them must be preserved; for the environmental processes that are potentially reversible, they may fall under the criteria of weak sustainability (Ayres, Van den Berrgh & Gowdy 2001; Kuhlman & Farrington 2010). The view of strong sustainability can be used as a series of thresholds that cannot be crossed for realising sustainable development, and the impact assessment has to be constrained by these thresholds (Joumard & Nicolas 2010; Kuhlman & Farrington 2010). Actually, most significant variables in sustainability assessment are generated based on the thresholds of strong sustainability (Kuhlman & Farrington 2010).

To conclude, sustainable development is a matter of the degree of substituting natural resources for capital (Kuhlman & Farrington 2010). If the weak view is adopted, any natural resources can be substituted by capital. Oppositely, the loss of natural resources cannot be compensated by an increase in capital if the "very strong" view is adopted. By adopting the strong view of sustainable development, those natural resources that cannot be preserved or replaced by other resources based on the current state of technological knowledge should not be substituted by capital, but those that are potentially reversible may be substitutable by capital.

In this study, the strong sustainability view is adopted as the theoretical foundation for formulating the assessment process. First, the environmental, economic and social impacts of potential retrofitting activities are evaluated. Then, the importance of these three dimensions is determined to represent the trade-off or level of substitution between environmental protection and financial increase. By integrating the evaluation results of the three dimensions with associated weighting scores, the overall sustainability performance of potential retrofitting activities is evaluated. Details of the sustainability assessment are in Section 6.7 of Chapter 6.

2.5. Challenges of developing sustainable construction

The emergence of new terms like "smartness" or "resilience" questions whether the terms sustainability or sustainable development are still needed. Based on studies by the following scholars, sustainability is still needed. Increasing evidence shows that human activities are posing harmful impacts on climate change (Stocker et al. 2013). Moreover, the study by Motesharrei et al. (2016) suggested that the current carrying capacity of Earth cannot support the current rate of human development and population growth. Therefore, even though we have been immersed in the policies and practice of sustainable development for over 30 years, we still need to solve the problems that the Brundtland report posed (Conte 2018). In addition, there are several challenges that sustainable construction is facing and must overcome to achieve a wider application, as explained in the following sections.

2.5.1. High initial cost

First, the (perceived) high initial cost is one of the main obstacles to sustainable construction (Aghimien, Aigbavboa & Thwala 2019; Gunduz & Almuajebh 2020). Gunduz and Almuajebh (2020) identified critical success factors for sustainable construction by surveying professionals in the construction field worldwide to rank the importance level of 40 identified factors of successful sustainable construction. The result shows three essential factors that influence construction efficiency, one of which is finance. However, there is a general perception that the upfront investment in sustainable construction will be much more than in conventional construction. In the study by Rehm and Ade (2013), semi-structured interviews were conducted with 15 industry professionals across New Zealand. The collected responses indicated that all 15 participants believed that the construction cost of green buildings would be higher than for conventional buildings. Some people even assert that triple-bottom line is a zero-sum game because the cost premium of constructing green buildings (by considering all three sustainability dimensions) leads to negative income in the short term (Maltz, Bi & Bateman 2018). Consequently, investors and building owners avoid considering sustainable construction.

The initial cost of sustainable construction may be higher than traditional construction due to the relatively high cost of sustainable materials and new technologies (Cupido et al. 2010). In addition, the soft cost of green buildings, including consultants and the process of achieving a green building rating, also leads to a higher initial cost of green buildings (Cupido et al. 2010). However, this may not be true all the time. Based on the study by Zuo and Zhao (2014), green buildings in Australia rated as 4 stars may cost the same as conventional buildings, and ratings of 5 stars and 6 stars require construction cost premiums of 4% and 10% respectively. Dwaikat and Ali (2016) stated that green buildings are facilities that can cause less negative impacts on environment by consuming less natural resources¹. To investigate the cost premiums of green buildings, they conducted 20 studies in Australia, New Zealand, the UK and the USA to compare costs between green buildings and non-green buildings. Their study found that two of the 20 studies recorded a cost premium for green buildings of over 20% more than conventional buildings, three studies between 10% and 20% (including 20%), three studies between 5% and 10% (including 10%), and five studies between 0% and 5% (including 5%). Five of the 20 studies indicated that green buildings have similar costs as conventional buildings, and the other two studies indicated cost savings by constructing green buildings.

Indeed, a green building does not necessarily mean much higher initial costs than conventional buildings. By adopting an integrative design method to build passive green buildings, energy savings and indoor comfort can be realised with less or no mechanical intervention. As a result, costly mechanical building systems can be downsized or even removed (Hawken, Lovins & Lovins 2013). Moreover, the upfront cost required to construct green buildings can be compensated by the cost reductions during the operation and maintenance stages (Zuo & Zhao 2014). Therefore, the high initial cost is a challenge in sustainable construction, but it may be a challenge based on perception only – what people think it will be, not what it is.

To reduce concerns about the high initial cost of sustainable construction, a proper sustainable design and comprehensive cost evaluation should be conducted at the early stage of construction (Kovacic & Zoller 2015). The life cycle costing of buildings should also be integrated, which can predict the potential capital return in the long term (Conte 2018; Maltz, Bi & Bateman 2018). Incentive policy, legislation and support from government, locally and nationally, can also help improve the feasibility and applicability of sustainable construction

development

Note 1: The authors listed different definitions of green building by other scholars, and they summarise them as the one 32 that fits to their study.

2.5.2. A lack of attention to the social dimension

The second challenge is that not all the three dimensions are encapsulated in the assessment of performance of constructions as proposed under triple-bottom line (Goh et al. 2020). Previous studies have shown how environmental and economic impacts are assessed, but the social impact assessment is relatively vague and weak, regardless of the assessment methods or variables (Phillips et al. 2020).

For the environmental dimension, life cycle assessment (LCA) is the common method to assess the environmental impact of construction activities on different life stages of buildings (Dong & Ng 2016). The environmental impacts can include pollution of the natural environment based on different methods of life cycle inventory analysis (Kohler et al. 2010). In addition to LCA, many simulation tools are also available and can be used to estimate the environmental impacts of construction activities, especially for estimating energy consumption and carbon emissions. Building Information Modelling (BIM) is a popular simulation tool used to estimate energy use and predict performance for new buildings (Ilter & Ergen 2015; Woo, Wilsmann & Kang 2010).

For the economic dimension, in general, four different metrics can be calculated to measure whether the investment in construction projects is worthy (Cotter 2022): net present value (NPV), converting all cash flows during the life span of the building to current value based on the selected discount rate; equivalent annual cost (EAC), converting all cash flows to an equal amount per year over the life span of the building, or the annualised NPV; internal rate of return (IRR), the rate of discount at which the corresponding benefits and expenses are equal; and payback period, the number of time periods required to recover an investment's initial costs from the net cash flows generated at the selected interest rate. Different methods can be selected for different purposes. The details are discussed in Chapter 4.

For the social dimension, there is still no unified or commonly recognised method to assess the social impacts of construction activities. In fact, there is not even consensus on what social impacts should be covered for assessing sustainable construction (Santos et al. 2017). Most studies focus on the impact of construction materials for assessing environmental performance and related costs, jointly or individually. The social dimension is commonly only briefly

discussed, or even completely ignored. Goh et al. (2020) suspected the reason is that the social dimension is a human-centred aspect. For construction projects, stakeholders have different objectives and priorities. It is not easy to capture construction activities' impact on them, especially when the long-term perspective is adopted. However, to achieve sustainable construction, it is necessary to establish a balanced and optimal way to deliver sustainability in all the three sustainability dimensions (Klöpffer 2008; Purvis, Mao & Robinson 2019). To reach this goal, more effort is needed to investigate the details of the three pillars, both assessment methods and variables.

2.5.3. Difficulty of measuring trade-offs between different sustainability

dimensions

As discussed in Section 2.2, sustainable development is a multidisciplinary concept. Therefore, decision making on sustainable construction needs to integrate the analysis of trade-offs between environmental, economic and social dimensions (Epstein, Buhovac & Yuthas 2015). However, the three dimensions are closely intertwined, and how to assess trade-offs between them is a critical issue. The first challenge of measuring trade-offs between sustainability dimensions is that there is no standard reporting method for holistically measuring buildings' performance in the environmental, economic and social dimensions (Goh et al. 2020; Phillips et al. 2020). For example, to measure the performance of a green office building, in some cases only construction costs are measured, but in others, the profit growth due to improved productivity from the construction activities is also included.

The second challenge is from the different measurement units of the three dimensions, which makes it difficult to calculate a cumulative outcome (Maltz, Bi & Bateman 2018). As a result, it is hard to compare and make a decision among different options (Slaper & Hall 2011). For example, energy consumption by a construction activity is estimated in units of MJ, financial investment is calculated in monetary units like dollars, and the social impacts can be estimated using value scores which are unitless. Different measurement units make it difficult to declare one activity can bring more sustainability benefits than another. Therefore, it is challenging to make decisions on sustainable retrofitting strategies.

Based on past studies, two solutions are proposed to solve the problem. One is converting all measures into monetary values, so impacts by a construction activity in the three dimensions can be measured based on a unified scale of money (Slaper & Hall 2011). However, it is not easy to find an agreed price for endangered species and limited natural resources when many people and organisations think they are priceless. Moreover, as discussed in Section 2.4.3, converting environmental impacts to monetary value may result in irreversible environmental damage. Therefore, this solution is not considered in this study. The other solution is normalising measures into universal or unitless values (Krajne & Glavič 2005). In this way, the measurements for different dimensions are shown on the same scale, and a cumulative outcome can be reached for comparison and decision making. However, normalisation may lead to inaccuracies if the scale range is not properly determined. The typical range can be 0 to 1, 0 to 10, or 0 to 100. The range from 0 to 100 is more accurate than the other two if two decimals are required for the normalised results.

2.5.4. A lack of knowledge of sustainable construction

The last challenge in achieving sustainable construction is the requirement for knowledge, qualifications and skills about sustainable development by professionals, especially architects and engineers. They are the core group who make decisions and enforce the decisions into outcomes (Conte 2018). This challenge is more obvious in developing countries. Aghimien, Aigbavboa and Thwala (2019) stated that a lack of knowledge and understanding about sustainable development is the main obstacle to delivering sustainable construction. It is not only about professionals who cannot identify sustainable materials, make the right decisions, or adopt appropriate technologies, but also the public who lack the awareness of sustainable development. Consequently, the percentage of sustainable construction in developing countries is far less than in developed countries (Aghimien, Aigbavboa & Thwala 2019). By reviewing existing studies, the barriers from lack of knowledge to develop sustainable construction can be found in different locations around the world, as Table 2.4 summarises.

Location	Barriers
Cambodia	Lack of skilled professionals
	• Lack of training and education, leading to a lack of awareness of environmental concerns
Canada	• Lack of consideration of sustainability criteria in the evaluation of bids
	 Lack of knowledge of local conditions
Chile	Lack of knowledge on sustainable technologies
China	Lack of awareness of environmental protection
	 Lack of awareness of reducing construction waste, especially from designers
Finland	Lack of client understanding
	• Lack of knowledge and skills to conduct sustainable construction (procurement and
	tendering, timing, cooperation and networking, availability of methods and tools, and innovation)
Malaysia	Lack of awareness of sustainable buildings
	 Lack of training and education about sustainable construction
	Lack of professional capabilities/designers
UK	Lack of proven technology alternatives

Table 2.4. Studies about barriers to develop sustainable construction

Source: Adapted from Durdyev et al. 2018; Ghisellini et al. 2018

Two methods are proposed to eliminate this barrier: properly selecting the set of sustainability assessment indicators, and training stakeholders with sustainability knowledge (Atanda 2019; Eberhardt, Birgisdottir & Birkved 2019; Hossain et al. 2020). Sustainability indicators are used to assess sustainability performance of products and services and this is the area that researchers normally focus on (Stanitsas & Kirytopoulos 2021). Sustainable indicators embed physical and social science knowledge into decision making, which can help decision makers, even policy makers, to simplify and clarify information (Atanda 2019). Therefore, to achieve sustainable construction, sustainability indicators that can include primary building impacts and evaluate specific aspects of the socioeconomic context should be used (Vilnītis, Lapsa & Veinbergs 2019). In addition, the selection of sustainability indicators is crucial to achieve a balance between sustainability dimensions (Wu et al. 2018). Stanitsas and Kirytopoulos (2021) also supported this point by emphasising the importance of balancing sustainability dimensions under the indicators "umbrella" in order to successfully achieve sustainable construction.

Another method to eliminate this barrier is training stakeholders in construction with sustainability knowledge (Eberhardt, Birgisdottir & Birkved 2019; Hossain et al. 2020). Training is an important tool to help stakeholders understand the aim, indicators, frameworks, guidelines and policies of sustainable construction, and government plays a vital role in promoting and popularising training in sustainable construction (Eberhardt, Birgisdottir &

Birkved 2019). The study by Atanda (2019) highlighted the content that should be covered in education of sustainability, including materials selection, energy sources, water and waste management, pollution, biodiversity, knowledge of the physical environment, understanding of human impact on the environment, and capability to address environmental issues. In addition, training in sustainable construction is particularly meaningful to building users, who normally have limited knowledge about sustainability. By understanding the tangible and intangible benefits of sustainable construction, their willingness to act towards sustainability may increase.

To conclude the above discussion, although there has been a steady increase in studying sustainable construction in the past three decades, challenges are still encountered including the perceived high initial cost, a lack of attention to the social dimensions, difficulty of measuring trade-offs between different sustainability dimensions, and a lack of knowledge about sustainable construction among stakeholders. To overcome these challenges and achieve successful sustainable construction, the below points are significant and should be satisfied:

- Environmental, economic and social dimensions should be integrated and balanced in an optimal manner.
- A long-term perspective should be adopted for investigating impacts on the three sustainability dimensions.
- The assessment method and assessment indicators should be thoroughly studied and determined for each sustainability dimension.
- A universal unit should be adopted for comparison and decision making.

2.6. Summary

Reviewing past studies on sustainable development and construction showed that sustainable development is a significant concept for realising a continuous development process without damaging the benefits to future generations. It covers various aspects of our daily life. Considering the extensive resource consumption and pollution generation of the construction sector, it is a crucial field for realising sustainable development for the whole society. Combining existing definitions about sustainable society and economy, and sustainable economic and environmental development, sustainable retrofitting is defined in this study as a

process of reducing the cost of operating a building and increasing human wellbeing in ways that reduce deterioration of natural systems.

The environmental, economic and social dimensions are commonly recognised as three dimensions of sustainability. The popular sustainability assessment tool – triple-bottom line – also includes these three dimensions to assess sustainability performance of construction and buildings. Regarding different levels of substitution between capital and natural resources, three views are proposed: weak view, strong view, and very strong view. The weak view implies that the loss of natural resources can be offset by increased capital. In the view of strong sustainability, for those natural resources that are essential for our survival and where the arrival of their tipping points cannot be prevented or slowed down by the current state of technological knowledge, a minimum amount of them must be preserved; for the environmental processes that are potentially reversible, the loss of them may be offset by increased capital. In the very strong view, every natural environment component or subsystem, as well as every species and physical stock, must be protected. In this study, the strong view of sustainability is adopted to design assessment of sustainability performance of retrofitting activities. The level of importance of environmental, economic and social dimensions is determined to represent the trade-off or level of substitution between environmental protection and economic increase.

This chapter identified four challenges of achieving sustainable construction: perceived high initial costs, a lack of attention on the social dimensions, difficulty of measuring trade-offs between the three sustainability dimensions, and a lack of knowledge of sustainability. To overcome these challenges and realise sustainable construction, four suggestions are given based on existing studies. First, the environmental, economic and social dimensions should be integrated and balanced in an optimal manner. Second, a long-term perspective should be adopted for investigating impacts on the three sustainability dimensions. Third, the assessment method and assessment indicators should be thoroughly studied and determined for each sustainability dimension. Fourth, a universal unit should be adopted for comparison and decision making.

The next chapter presents a review of the performance of existing office buildings and sustainable retrofitting and discusses why sustainable retrofitting is an effective solution for improving existing office buildings' sustainability performance.

Chapter 3. Sustainable retrofitting of office buildings

3.1. Introduction

The previous chapter discussed the significance and necessity of sustainable construction and implementation of triple-bottom line to assess the environmental, economic and social sustainability of construction. Continuing the topic, this chapter discusses the necessity of improving the sustainability of existing buildings, especially office buildings, due to their extensive energy consumption and carbon emissions, and massive environment pollution (Che et al. 2019). Retrofitting is recognised as an effective remedy for the poor performance of existing buildings, and can be implemented to reach sustainability goals (Krstić-Furundžić, Vujošević & Petrovski 2019). To justify this statement, this chapter first investigates the current environmental, economic and social performance of existing buildings based on information derived from the literature (Section 3.2). Then, the reasons for retrofitting are analysed by comparing demolition and construction of new buildings, with the discussion of benefits and barriers of retrofitting (Section 3.3). Lastly, common retrofitting activities for office buildings are introduced, highlighting the potential improvement from them (Section 3.4 and 3.5).

3.2. Current performance of existing office buildings

The performance of buildings declines over time of use, not only because of aging building materials and components, but also because stricter building regulations over time increase demands on buildings, such as the demand for better indoor environmental quality. The phenomenon is more obvious for office buildings because their performance largely relies on the operation of mechanical and electrical facilities and systems. With the large number of existing office buildings around the world, their poor performance has caused extensive negative environmental, economic and social impacts. The following sections discuss existing buildings' performance in these three pillars in greater detail.

3.2.1. Environmental performance

In most countries about 20% to 40% of total energy consumption is from existing buildings (Hong et al. 2019). The massive energy use by buildings causes about 8.6 Gt CO₂ emissions

each year (Munarim & Ghisi 2016). According to 2021 Global Status Report for Buildings and Constructions (United Nations Environment Programme 2021), the building sector is responsible for about 47% of the global annual carbon emissions. More than half of the 47% is from building operation. Due to the increasing affordability and advancements in air conditioning technologies, the demand for thermal comfort is also increasing, leading to more energy consumption by buildings expected in the future (Luo et al. 2018). Energy use in buildings is significantly higher in cities with high-rise commercial buildings (Che et al. 2019). Che et al. (2019) found that buildings account for 64% of Hong Kong's total final energy use, and commercial buildings account for 43% of the total consumption. Australia, as a representative developed country, has been practising sustainability in the built environment for decades, but about one quarter of the total greenhouse gas emissions in the country originate from building-related activities, and about 12% of total emissions are from office buildings (Drosou et al. 2018; Wilkinson 2014). Similarly, about 40% of final energy consumption in the United States and the EU is attributed to the buildings sector (Toosi et al. 2020; Wagiman et al. 2020). According to Li et al. (2019), about 22% of the energy used in the United States is consumed by commercial buildings. In Europe, the floor area of non-residential buildings is up to about 25% of the total floor area, and 26% of the space of non-residential buildings is office buildings (Gimeno-Frontera et al. 2018). However, non-residential buildings have 40% more energy consumption than residential buildings in Europe (la Cruz-Lovera et al. 2017; Toosi et al. 2020).

The situation is worse in developing countries due to more construction activities and relaxed energy policies (la Cruz-Lovera et al. 2017). As the largest developing country, China is currently experiencing rapid urbanisation, leading to a dramatic increase in energy consumption and carbon emissions (Zheng, Yu & Wang 2019). The annual added construction area in China represents up to half of the world's building construction (Ding & Ying 2019). In China, about 28% to 34% of national carbon emissions are from the building industry, which is approximately equal to the total carbon emissions in the Middle East (He et al. 2020). According to World Data Atlas (Andrew & Peters 2021; Friedlingstein et al. 2021), in 2020, China's CO₂ emissions were 8.2 tonnes per capita, almost twice as high as the world's per capita emissions in the same year.

Definitions of building types vary by national context. In western countries public buildings mean government buildings. In China, public buildings are designed for people for a variety of public activities, including commercial buildings, transportation buildings, tourism buildings, office buildings, hotels, and others (Wei & He 2017). The energy consumption by public buildings in China is 5 to 15 times that of residential buildings (Xu, Chan & Qian 2012). This statement is supported by Guo et al. (2022), stating that the operation energy of existing buildings in China accounts for 22% of the total domestic energy consumption, and public buildings are the largest contributor.

Indeed, commercial buildings, primarily office buildings, are some of the largest energy consumers and carbon emitters (Krstić-Furundžić, Vujošević & Petrovski 2019; Niemelä et al. 2017). According to IEA (2021), in 2020 about 27% of global energy consumption in buildings was from non-residential buildings with about 37% of energy-related CO₂ emissions. There was a slight decline in CO₂ emissions from the building sector in 2020 (from 9.6 Gt in 2019 to 9 Gt in 2020), which was mostly caused by lower activity in the services sector due to the global COVID-19 pandemic (IEA 2021). As social activity resumes, consumption and emissions are expected to gradually increase back to the pre-pandemic level (IEA 2021). Therefore, it is still urgent to reduce energy consumption and carbon emissions in existing office buildings.

Water is a very valuable resource, and due to the growth of global population and change in precipitation patterns associated with climate change, water demands are dramatically increasing (Lani et al. 2018). The construction and property sector is responsible for about 12% to 25% of water consumption (Dwaikat & Ali 2016). The majority of water used in public facilities is for flushing toilets (up to 50%), and with taps and showers together accounting for up to 80% of all water use (Bertone et al. 2018). To reduce water consumption in buildings, solutions include either reducing water consumption by installing water saving fixtures, or using alternative water resources such as treated or recycled grey water, or rain water (Sousa, Silva & Meireles 2019). The building sector is also responsible for about 30% of resource consumption, 40% of raw materials, and 25% to 40% of waste generation globally (see Table 2.2 in Chapter 2).

In addition to impacts on the natural environment, the indoor environment of existing office buildings is another essential aspect for achieving a sustainable built environment (Niemelä et al. 2017). Office buildings are designed to provide a workspace for people who typically stay in the provided work environment for at least 8 hours every day (Che et al. 2019; Cheong et al. 2020). Considering this long-hour occupancy, thermal comfort, visual condition and acoustic comfort are crucial for both long-term workplace safety and the physical comfort of occupants (Cheong et al. 2020).

Work productivity, in addition to safety and comfort, is another vital aspect affected by indoor environmental quality (IEQ). A nationwide survey conducted in the UK with professionals in different sectors indicated that sound indoor environmental quality contributes to up to 20% improvement in productivity of the building occupants, which is equivalent to £135 billion per year (Horr et al. 2016). Correspondingly, poor indoor environmental quality can reduce working performance. Among the three aspects of indoor environmental quality, thermal comfort has the greatest impact on work productivity. For example, 21 °C to 25 °C is observed to be an ideal indoor temperature range for office buildings. Every 1 °C above 25 °C up to 30 °C can cause a 2% decrease in productivity (Kaushik et al. 2020).

Acoustic comfort and visual comfort are two other crucial aspects impacting work performance in office buildings. Acoustic performance has an equivalent effect on an employee as the thermal performance. Horr et al. (2016) showed that a 2.6 dB increase in noise has the same impact on productivity as a 1 °C increase in indoor temperature. Visual comfort of office occupancies mainly relies on indoor lighting performance. Several studies have shown that office lighting has a considerable impact on work performance (Kim, Wang & McCunn 2019; Ma, Lee & Cha 2022; Wagiman et al. 2020). Moreover, prolonged exposure to a harsh visual environment in office buildings can lead to both physical and psychological issues, such as fatigue, headaches, back pain, annoyance and stress (Ma, Lee & Cha 2022). Due to these impacts on occupants, poor indoor environmental quality of office buildings is one of the main incentives for organisations to conduct retrofitting projects (Newsham et al. 2013). Indoor environmental quality is recognised as a vital assessment criterion for measuring environmental performance of existing office buildings (Shrubsole et al. 2019; Yang et al. 2020). In summary, existing office buildings are consuming massive amounts of energy, emitting large quantities of carbon dioxide, and causing unfavourable impacts on the natural environment. Moreover, indoor environmental quality of office buildings has great impacts on occupants' physical and psychological health as well as work productivity. Considering the large number and long life span of existing office buildings, improving environmental performance becomes urgent for alleviating the impact on global warming and enhancing occupants' comfort level.

3.2.2. Economic performance

The general economic performance of existing buildings is suffering mainly because of high operation and maintenance costs (Oregi, Hernandez & Hernandez 2017). Based on the international standard ISO 15685-5 (ISO 2017), these two types of cost occur during the operation stage of buildings. Past studies have estimated that operation and maintenance costs account for approximately 75% to 80% of the total costs in a building's whole life cycle if the service life is 50 years (Hauashdh, Jailani & Rahman 2022). The operation cost can also be called an environmental cost because it includes fuel costs for heating, cooling, power, lighting, water and sewerage costs, and related environmental taxes (ISO 2017). Therefore, the economic dimension of an existing property is highly related to its environmental performance, and energy price is the most relevant factor of existing buildings' economic performance (Cetiner & Edis 2014; Mikulić, Bakarić & Slijepčević 2016; Santos et al. 2019).

The HVAC system is responsible for a substantial portion of energy consumption in commercial buildings to provide indoor thermal comfort and good indoor air quality (Che et al. 2019). Based on existing studies, the HVAC system consumes up to almost half of energy use in commercial buildings and dominates peak electricity demand (Che et al. 2019; Department of Climate Change, Energy, The Environment and Water 2022; Kim 2017). Following the HVAC system, the lighting system is the second largest energy consumer in commercial buildings, responsible for about 20% of the total energy use (Jin et al. 2021). Due to the large amount of energy consumption, the operation cost from these service systems also accounts for a sizable amount of overall building costs (Department of Climate Change, Energy, The Environment and Water 2022). Moreover, due to the growing recognition of the importance of improving indoor environmental quality of office buildings for health, a good

work environment, and higher productivity, expenditures have been rapidly increasing (Kibert 2016). In the US, the annual cost of improving indoor environmental quality for office buildings is estimated to be hundreds of billions of dollars (Kats 2003; Kibert 2016).

For maintenance cost, eight kinds of costs are included (ISO 2017): maintenance management; adaptation or refurbishment of the building; minor repair and replacement cost; major systems or components replacement cost; cleaning; ground maintenance; redecoration; and tax on maintenance goods and services.

The quality and performance of building components and systems tend to diminish over time if no intervention like refurbishment or retrofit is applied. Moreover, building maintenance practices are becoming more and more difficult due to the environmentally friendly requirements and regulations and more complex building systems (Hauashdh, Jailani & Rahman 2022). Therefore, whether it is to attain acceptable building functions or to overcome maintenance challenges by contemporary requirements of sustainable buildings, maintenance costs are increasing and represent a certain portion of the total cost during a building's operation stage. Based on the data collected by Consulting Engineer (2022), for most buildings, about 80% to 90% of maintenance costs are attributable to 30% to 40% of individual asset items. Therefore, considerable savings can be attained by improving the efficiency of these items.

In addition to expenses on operation and maintenance, the income of properties also relies on the price, quality and competition of the property in the market (Wilkinson & Remoy 2017). There is no doubt that an existing building becomes less marketable if its performance is poor, especially compared to newer buildings. Based on the above discussion, it can be concluded that with the expected higher energy price and fierce competition in the market for existing buildings, the investment opportunities for energy-efficient retrofitting will increase and it will become one of the driving forces for owners to retrofit existing buildings (Pombo, Rivela & Neila 2016).

3.2.3. Social performance

As discussed in Chapter 2, compared to the environmental and economic dimensions, the social dimension does not gain equal attention due to a vague definition and a lack of assessment methods (Phillips et al. 2020). Without a clear definition of social sustainability and what is covered in the social dimension, it is difficult to assess the social performance of a building as good or poor. Therefore, instead of discussing how existing buildings perform socially, different opinions about the social sustainability of buildings are summarised and discussed.

Tweed and Sutherland (2007) pointed out that the protection of cultural heritage is indispensable for achieving the social sustainability of buildings. The ability to preserve the cultural heritage embodied in existing buildings should be evaluated to alleviate or avoid the damage caused by construction activities. Valdes-Vasquez and Klotz (2012) stated that the social dimension reflects different aspects of the stakeholders of a project, including the mediation among employees, local communities, clients and the supply chain, to meet the demands of current and future populations and communities. Deuble and de Dear (2012) believed that cultural and behavioural factors, as part of the social context, are crucial for green building developments. Similar, Mateus and Bragança (2011) and Zuo and Zhao (2014) believed that the education level and awareness level of sustainable development of stakeholders, especially occupants, are important components of social sustainability. In the study by Samandar (2015), social sustainability is considered as a broad concept covering a variety of aspects including processes that can improve social health and wellbeing.

Even though different aspects are emphasised by different scholars, most aspects are related to people, particularly stakeholders. This point is also agreed by other scholars. Watson et al. (2016) stated that the social context can be expressed as building user group dynamics, which is an integration of institutional norms, culture and management. Wilkinson and Remoy (2017) and Parida et al. (2021) identified social aspects for office buildings, which are work-related flow, comfort, wellbeing and job satisfaction of occupants. Social sustainability essentially involves various social values that are influenced by different stakeholders (Liu & Qian 2019). In terms of a building project, it is supposed to meet the diverse requirements of multiple stakeholders involved in the whole project process including construction teams, suppliers, end users and local communities (Liu & Qian 2019).

It is normally difficult to quantify human-related content due to its "soft" characteristics. However, even though the advantages of social sustainability are intangible sometimes the benefits can be returned in many ways (Samandar 2015). For example, a study conducted by Onat, Kucukvar and Tatari (2014) assessed buildings' performance in the UK using the triple-bottom line approach. In their study, three major social impacts were identified for commercial buildings which are the construction phase, electricity consumption, and commuting. Onat, Kucukvar and Tatari (2014) explained that the income of commercial buildings is heavily influenced by the construction phase. More than half of the tax categories of commercial buildings are about electricity consumption. Construction (43%) and commuting activities (34%) are the major sources of work-related injuries in commercial buildings. A similar study by Li, Ding and Runeson (2018) also confirmed that the social dimension covers a wide range of aspects, and that some financial benefits and environmental benefits can be achieved if the social performance of buildings is improved. Therefore, social sustainability is indispensable to achieve the overall improvement of sustainable buildings.

Based on the above discussion, construction activities generate various social impacts on individuals, communities, and even the whole of society. The wide coverage and the intangible characteristics of the social context make the social dimension challenging in pursuing overall sustainable buildings (Dendena & Corsi 2015). As an indirect way to preserve the cultural and societal assets that have been embodied in existing buildings, building retrofitting should not be carried out without social impact assessment (Jagarajan et al. 2017). However, researchers have not reached complete agreement on what social content should be included in the context of sustainable buildings for different analysis. In this thesis study, the impacts of retrofitting construction on stakeholders, including building owners, tenants, people who visit the building, people who live or work in neighbouring buildings, and people who pass by the buildings (or the retrofitting construction) are considered. The selected social impacts are also checked against the impacts included in the environmental and economic dimensions to prevent double or triple counting.

3.3. Sustainable retrofitting

3.3.1. Definition of retrofitting

According to the Cambridge Dictionary (Walter 2005), retrofit means "to provide a machine with a part, or a place with equipment, that it did not originally have when it was built". When it is used in the construction and property sector, retrofit refers to modification of existing buildings to make their facilities easy to operate, more efficient and with less impact on the environment (Hong et al. 2019). This may involve activities such as adjustment, reuse or upgrade to existing building envelopes and mechanical systems to meet new conditions or requirements (Bruce et al. 2015; Hong et al. 2019; Wilkinson 2014).

Apart from retrofit, there are other terms that are interchangeable for expressing a similar meaning but in different scales from repairing building fabric to refurbishing the whole building, including renovation, refurbishment and conversion (Wilkinson 2012; Zhang et al. 2021). Among these four terms, conversion is the easiest to distinguish because the function of existing buildings is only changed when building conversion is conducted (Remøy & Wilkinson 2012). In other words, a new use is fitted in the existing building via conversion, but not by the other three terms. In the conversion process, the building structure including loadbearing members would be affected, while the major structure would not be changed in retrofit and renovation projects, even though the intervention may encompass the whole building (Giebeler et al. 2009).

Renovation is defined as intervention activities that do not extend the existing building structure, nor change anything with new substitution (Giebeler et al. 2009). Therefore, the value and function of the renovated building will not change, but its quality may be improved to some degree. According to Brown and Teernstra (2008), renovation is part of the maintenance process with large investments to enhance building quality and maintain its basic function.

Similar to renovation, refurbishment is also a necessary intervention for maintaining existing buildings. According to the international standard ISO 15685-5 (ISO 2017), refurbishment cost is one component of maintenance cost. However, in contrast to renovation to keep the basic function of buildings, refurbishment is always adopted to meet new standards caused by

increasing demands or new technical regulation (Giebeler et al. 2009). It is divided into three categories according to the degree of change: partial refurbishment, "normal" refurbishment, and total refurbishment (Giebeler et al. 2009). Partial refurbishment refers to changes involving only one component or one part of existing buildings. "Normal" refurbishment makes additions or changes to the entire building or at least one part of it. A total refurbishment project involves replacing the current infrastructure or upgrading all existing components of the building, and often involves demolition activities.

Instead of being regarded as a maintenance measure, retrofit is more like a new stage in the life cycle of existing buildings (see Figure 3.1) that can modernise outdated buildings to comply with current energy efficiency rules, building standards, and requirements for indoor comfort (Munarim & Ghisi 2016; Wilkinson 2012). The latest technologies are normally adopted in a retrofit project to prolong service life and improve the performance of existing buildings (Menassa 2011; Shaikh et al. 2017). Retrofit is regarded as an effective way to improve the sustainability level of existing buildings not only by improving buildings' energy efficiency performance, but also by largely reusing existing components and structures with minor additions of materials and energy use (Jagarajan et al. 2017; Latham 2016).



Figure 3.1. Extended life cycle of buildings by retrofitting

Source: Adapted from Munarim & Ghisi 2016

As discussed in Chapter 1, this study is designed to investigate how to improve the sustainability performance of existing office buildings, including how to upgrade building performance where no structural element would be affected. The expected outcome is that existing office buildings' sustainability performance can be effectively improved, embodied energy in existing structures can be preserved, and the service life of existing buildings can be reset. Therefore, the term "retrofit" is more specific and appropriate for use in this study.

3.3.2. Reasons for retrofitting

Atkinson (1988) introduced a "sinking stack" theory to explain the relationship between new buildings and existing buildings as Figure 3.2 shows. The new buildings are added at the top annually representing current building conditions, and old buildings with poor quality are at the bottom of the stack. With the passage of time, all buildings age, and resources and energy are needed to maintain their quality and performance. From the perspective of sustainable development, the existing buildings with very poor performance, at the bottom of the stack in Figure 3.2, should be removed from use. The quantity of new buildings should be minimised

to reduce the energy and resources required in construction. Most resources and effort should be allocated to improve the performance of the existing stock in the middle of the stack in Figure 3.2.





Buildings are a long-lasting product that require regular maintenance and renewal, but still cannot avoid becoming obsolete or redundant due to changing demands and regulations. Eventually, the obsolescent buildings are demolished and new buildings are built complying with current standards and regulation, but the obsolescence stage can be delayed by retrofitting (Ongpenga et al. 2020; Solanki, Rastogi & Paul 2022). The reason for preferring to retrofit is that, compared to demolition and new construction, retrofitting can provide a quicker delivery of sustainability in time and save energy (Thomsen & Van der Flier 2011).

First, retrofitting is a more feasible strategy than demolition because of the huge number of existing buildings around the world. Existing buildings may represent approximately 87% of buildings that we will use until 2050 (Wilkinson & Remoy 2017). In the UK, almost 85% of current buildings will be occupied until 2050 (Dowson et al. 2012). About 66.3% of buildings in the US and 75% in Europe were built before 1990 (Lee, Shepley & Choi 2019). Moreover, over 25% of existing stocks in Europe are older than 70 years and do not reach the requirement of energy efficiency in the new energy codes (Invidiata, Lavagna & Ghisi 2018; Vilches, Garcia-Martinez & Sanchez-Montañes 2017). Regarding the large quantity of existing
buildings, the rate of new buildings added to total stock is only 1% to 2% annually (Ding & Ying 2019; Wilkinson 2012). Therefore, compared to demolishing existing buildings and building new ones, retrofitting seems a more feasible strategy to improve the sustainability performance of the whole building sector.

Second, retrofitting can improve performance more quickly than demolition new buildings (Wilkinson & Remoy 2017). As a product with a long service life, the negative impacts of buildings also last for a long time if no effective intervention is adopted. As discussed in Section 3.2, existing buildings are blamed for posing environmental, economic and social obstacles to overall sustainable development (Lee et al. 2020). With the large quantity, long service life, and massive negative impacts of existing buildings, compared to demolition, retrofitting can provide a faster process than demolition and then building a new building since the existing structure can be preserved, and existing infrastructure can be largely reused (Munarim & Ghisi 2016; Power 2010).

Last but not least, there is the promising potential of retrofitting to save energy and reduce carbon emissions, both embodied and operational. The operation stage accounts for about 70% to 90% of the total energy consumption in the whole life span of a building (Menassa & Baer 2014; Toosi et al. 2020; Yuan, Nian & Su 2019). Previous studies have shown that sustainable retrofitting can effectively improve the energy efficiency of existing buildings (Che et al. 2019; Hong et al. 2019; Luo, Lu & Ge 2021; Munarim & Ghisi 2016; Yin 2011). Savings of about 30% to 40% of energy use in existing buildings can be achieved by retrofitting (Hong et al. 2019). In commercial buildings, retrofitting can assist in achieving about 40% to 74% of energy saving (Yin 2011). Retrofitting existing buildings also has huge potential to reduce carbon emissions, estimated to result in a drop of up to 80% globally in 2050 compared to emissions in 2005 (Luo, Lu & Ge 2021).

With the big potential of saving operation energy and carbon emissions, the proportion of the impact from embodied energy and embodied carbon emissions is rising (Salehian, Ismail & Ariffin 2020; Toosi et al. 2020). For now, about 10% to 30% of the total energy consumption of a building is embodied energy (Ingrao et al. 2018; Toosi et al. 2020). It is estimated that the embodied carbon emissions in new buildings constructed between 2020 and 2050 will be equal

to their operation carbon emissions (SDSN & FEEM 2019). Compared to demolition and new buildings, retrofitting can preserve the embodied energy and embodied carbon emissions in existing buildings by prolonging building use (Munarim & Ghisi 2016). If demolition is conducted, the embodied energy and embodied carbon will be wasted, and additional energy and emissions will be produced by new construction (Merlet et al. 2021).

In summary, given the low rate of new buildings being added to the building stock, the focus of saving energy and reducing carbon emissions should be on existing buildings. There are two strategies for the purpose: retrofitting existing buildings, or demolishing and replacing them with new buildings. Based on existing studies, retrofitting is often more desirable than demolishing buildings. First, the number of existing buildings is huge, and compared to demolishing them and building new ones, retrofitting is a more feasible strategy. Second, retrofitting can improve the sustainable performance of existing buildings more quickly than demolishing and re-building since the existing structure and infrastructure can be largely reused. Third, retrofitting can save operation energy consumption and carbon emissions, but also preserves embodied energy and embodied carbon emissions in the existing buildings, which is wasted if they are demolished. For these reasons, it can be concluded that rather than demolishing and re-building, retrofitting is more desired and effective to improve existing buildings' sustainability performance.

3.3.3. Benefits and barriers to retrofitting

In addition to the above-mentioned benefits, sustainable retrofitting also brings other tangible and intangible benefits. For the environment, retrofitting projects (without demolition) helps reduce landfill waste (Wilkinson & Remoy 2017). Retrofitting is also an effective strategy to reduce reliance on fossil fuel and contributes to the transition to a decarbonised energy system (Bleyl et al. 2019). From the economic perspective, retrofitting can contribute to higher rents and real estate values, lower running costs, higher work productivity, and lower vacancy rates. Eventually, the marketability of existing buildings can be improved (Bleyl et al. 2019; Bruce et al. 2015; Wilkinson 2014; Xu, Chan & Qian 2012). Kok, Miller and Morris (2012) conducted 14 case studies in different cities in the US showing that if the cost of retrofitting is USD 0.93/m² to USD 1.86/m², 8.79 kWh/m² to 14.65 kWh/m² energy can be saved annually, which can be converted to financial benefits of USD 0.14/m² to USD 0.23/m². From a social

perspective, sustainable retrofitting can bring better indoor environmental quality, which is good for tenants' physical and psychological health (Lee et al. 2020). For office buildings, sustainable retrofitting is able to effectively improve work-related flow, reduce the absence rate, and improve job satisfaction, which helps achieve better organisation outcomes (Wilkinson & Remoy 2017).

Even though people are well aware of the variety of benefits from retrofitting, the annual rate of sustainable retrofitting is quite low, at about 0.4% to 1.2% of the total building stock in Europe (Tavakolan et al. 2022; Zhang et al. 2021). Lee, Shepley and Choi (2019) explained that the low rate is because of the high initial cost, insufficient incentives, lack of related information, lack of reliable advice, uncertain outcomes, and split benefits received by stakeholders, in which building tenants gain the benefits of retrofitting paid for by building owners. Indeed, sustainable retrofitting faces a number of challenges. To attain wide application of sustainable retrofitting, these challenges must be addressed.

First, even though the potential of saving energy and reducing carbon emissions by retrofitting is recognised, existing research aimed at improving the efficiency of buildings is still concentrated mainly on new buildings (Ding & Ying 2019). There is an absence of incentives through regulations and policies to improve existing buildings' energy efficiency and environmental performance (Munarim & Ghisi 2016). Munarim and Ghisi (2016) gave an example. In England, upgrades to existing buildings are subject to a tax of 17.5% (including Value Added Tax) while new buildings are exempt from Value Added Tax, which is assessed on expenditure or consumption. Additionally, most certification schemes for green buildings give priority to new buildings instead of existing buildings (Munarim & Ghisi 2016). For instance, in the LEED scheme, only 2 out of 144 points can be received for reusing at least 75% of existing structures.

Second, as discussed in Chapter 2, sustainable retrofitting also faces a financial barrier (Juliardi et al. 2019). Drosou et al. (2018) stated that even though technologies, tools and policies about sustainable retrofitting are already in place, only the confirmation of value added on existing properties can convince building owners to invest in sustainable retrofitting.

Third, a lack of knowledge is another challenge facing sustainable retrofitting (Bertone et al. 2016; Liu et al. 2020). Building owners are the essential promoter of sustainable retrofitting, but they normally have limited knowledge and poor awareness of sustainable retrofitting (Bertone et al. 2016). Moreover, retrofitting is a complicated process involving a wide range of technologies (Liu et al. 2020). Even if information about energy efficiency and water efficiency is available, the information is too complicated to analyse and choose (Bertone et al. 2016).

Lastly, the benefits split unevenly across stakeholders also lead to a low rate of sustainable retrofitting (Lee, Shepley & Choi 2019). Building owners pay for sustainable retrofitting, but tenants gain from the generated savings. Building owners would like to keep the capital costs of the building as low as possible (without giving much thought to energy and water saving), while tenants would like to maximise energy and water efficiency to save operation costs. As a consequence, sustainable retrofitting is often not considered (Bertone et al. 2016).

In conclusion, sustainable retrofitting can bring environmental, economic and social benefits for existing buildings, which helps improve the sustainability performance of the whole building sector. However, the rate of retrofitting existing buildings across the world is still low. Sustainable retrofitting faces various barriers. More research on the implementation of sustainable retrofitting is needed to maximise benefits and eliminate barriers.

3.4. Common retrofitting activities for office buildings

Retrofitting activities for office buildings refer to installing or upgrading building envelopes (external walls, windows and external shadings), energy-related systems (HVAC, lighting, lifts, energy supply system, etc.), water-related systems (water supply system, sanitary fixtures, etc.), and other electrical appliances by adopting the latest technologies (Bruce et al. 2015; Hong et al. 2019; Wu, Wang & Xia 2016; Xu, Chan & Qian 2012). Currently, building retrofitting is conducted mainly for energy saving and carbon emissions reduction (Che et al. 2019). However, there is no doubt that considerable water consumption in existing buildings can be reduced if suitable retrofitting activities are implemented (Bertone et al. 2018). By reviewing existing studies, commonly used retrofitting activities are mainly about upgrading and installing new components on building fabrics and building service systems, which are

summarised in Table 3.1. The potential improvement from these retrofitting activities is discussed in the following sections.

Parts retrofitted	Retrofitting activities	Potential improvement	Reference
External walls and roofs	A1. Install/Upgrade insulation of building envelopes	 Reduce energy demand Reduce carbon emissions Increase indoor thermal comfort 	Bruce et al. 2015; Hong et al. 2015; Krarti & Deneuville 2015; MOHURD 2015; Braulio-Gonzalo & Bovea 2017; Drosou et al. 2018; Illankoon, Tam & Le 2018; Lee et al. 2019; Zheng et al. 2019; Australian Government 2020; Chen et al. 2020; Kumar et al. 2020
Roof	A2. Adopt extensive green roof	 Reduce energy demand (mitigate the heat island effect) Reduce carbon emissions Reduce air pollution Increase indoor comfort level 	Bianchini & Hewage 2012; Peri et al. 2012; Besir & Cuce 2018; Shafique, Kim & Rafiq 2018; Aboelata 2021
Windows	A3. Replace existing windows with energy efficient windows	Reduce energy demandReduce carbon emissionsIncrease indoor thermal comfort	Krarti & Deneuville 2015; Cuce 2018; Drosou et al. 2018; Zheng et al. 2019; Evangelisti et al. 2020; Fulton et al. 2020; Haule et al. 2020; Dabbagh & Krarti 2021; Simko & Moore 2021
	A4. Install sun shading devices	 Reduce energy demand Reduce carbon emissions Increase indoor thermal comfort Improve indoor visual comfort 	
Management and control system	A5. Install building management control system (BMCS)	Reduce energy consumptionReduce carbon emissions	Yildiz, Bilbao & Sproul 2017; Drosou et al. 2018; Chen et al. 2020
Lifts	A6. Upgrade lifts to more energy efficient ones	 Reduce energy consumption Reduce carbon emissions 	De Almeida et al. 2012; Bogach & Wang 2013; Carrillo et al. 2013; Al-Kodmany 2015; Zheng et al. 2019; Ali et al. 2021
HVAC system	 A7. Upgrade parts of existing HVAC system A8. Replace existing HVAC system with more energy efficient one 	 Reduce energy consumption Reduce carbon emissions Improve indoor air quality Improve indoor thermal comfort 	Bogach & Wang 2013; Bruce et al. 2015; Hong et al. 2015; Krarti & Deneuville 2015; Nguyen 2017; Che et al. 2019; Ding & Ying 2019; Far & Far 2019; Hong et al. 2019; Lee et al. 2019; Zheng et al. 2019; Energygovau 2020; Lu et al. 2021; Simpeh et al. 2021
Lighting system	A9. Install motion sensors for lighting system A10. Replace fluorescent bulbs with T8, T5 or LED A11. Install daylight dimming control system	 Reduce energy consumption Reduce carbon emissions Improve indoor visual comfort (greater luminescence and better colour rendering) 	Stansbury & Mittelsdorf 2001; Dubois & Blomsterberg 2011; Lecamwasam, Wilson & Chokolich 2012; Bogach & Wang 2013; Bruce et al. 2015; Krarti & Deneuville 2015; BEEX 2017; Drosou et al. 2018; Riyanto et al. 2018; Beccali et al. 2019; Han et al. 2019; Hong et al. 2019; Zheng et al. 2019; Haule, Chaiwiwatworakul & Chirarattananon 2020; Wagiman et al. 2020
Energy generation equipment	A12. Install PV panels	 Reduce traditional energy (electricity) demand Reduce carbon emissions 	Akhimien et al. 2017; Nguyen 2017; Drosou et al. 2018; Belussi et al. 2019; Hong et al. 2019; de Cunha & Aguiar 2020; Wang et al. 2020; Chahidi et al. 2021

Table 3.1. Common retrofitting activities for office buildings

Water system	A13. Install water control system	Reduce water consumption	Friedler & Alfiya 2010; Liu & Ping 2012; Cook, Sharma & Gurung,
	A14. Apply water-saving		2014; Hong et al. 2015; Bertone et al. 2016; Nguyen 2017; Bertone
	appliances and fixtures (taps,		et al. 2018; Balachandran, Mahanta & Samuel 2020; Metallidou,
	toilet flushing)		Psannnis & Egyptiadou 2020
	A15. Install water treatment system		
	and reuse recycled water		
	(storm, black and grey water)		

3.4.1. Install and/or upgrade insulation of building envelopes (A1)

Effective insulation of building envelopes can help downsize the required coverage area or reduce the working load of the HVAC system. Therefore, appropriate thermal insulation is crucial for saving energy consumption and providing a comfortable indoor environment (Illankoon, Tam & Le 2018). In warm climate regions, due to improved insulation performance, a HVAC system may not even be needed in winter, contributing to energy saving by reducing the heating loads. The application of upgrading or installing insulation of building envelopes can save up to 40% of the energy consumption of an office building (Chen, Hammad et al. 2020).

In Australia, thermal resistance, expressed as R-value, is used to rate the insulation effectiveness. The higher the R-value, the greater the insulation effectiveness (Illankoon, Tam & Le 2018). The R-value of an external wall is the summation of all the R-values of the material constituting the external wall. After retrofitting, the R-value of the external wall should be greater than the limitation regulated by the National Construction Code (NCC) (ABCB 2019). In Green Star, Australia's national green building rating tool, up to 5 out of a total 20 points are allocated for energy saving by improving insulation performance, which indicates the significance of building insulation (Illankoon, Tam & Le 2018).

In China, instead of R-value, U-value is used to measure the insulation effectiveness. In contrast to the R-value, a higher U-value represents poorer insulation performance. The limitation of the U-value is different based on different climate zones and building categories. In China, there are five climate zones: severe cold region, cold region, hot-summer cold-winter region, hot-summer warm-winter region, and temperate region. Public buildings (non-residential buildings) in China are divided into two categories, A and B. According to the national design standard for energy efficiency of public buildings (MOHURD 2015), category A includes single buildings with an area more than 300 m² and building groups with each building area less than 300 m². Category B includes single buildings with an area less than 300 m². The U-value of the building's envelope after retrofitting should be smaller than the limits for these two categories provided in the national design standard (MOHURD 2015). Table 3.2 shows the largest U-values for public buildings in category B.

	U-value (W/(m ² •K)											
	Severe cold region	Cold region	Hot-summer cold-winter region	Hot-summer warm-winter region								
Roof	≤0.35	≤0.55	≤0.70	≤0.90								
External walls (including opaque curtain walls)	≪0.45	≤0.60	≤1.0	≤1.5								
Windows	≤2.2	≤2.5	≤3.0	≪4.0								

	Table 3.2. Li	mit U-valı	e for bui	ldings in	category B	in Chi	na
--	---------------	------------	-----------	-----------	------------	--------	----

Source: MOHURD 2015

In a field study in Guangzhou China, a hot-summer warm-winter region, Song et al. (2017) measured an office building to indicate how much energy can be saved by different retrofitting activities on the building envelope. They created 27 scenarios to measure the different level of improvement on energy saving. By increasing thickness of insulation material to 4 cm and 10 cm, 0.49% and 0.83% of energy consumption can be saved respectively. These energy saving ratios can be applied for evaluating energy saving by the same retrofitting activity on buildings in this climate zone. However, it is important to note that the effect of energy saving by increasing the thickness of insulation materials can only be realised in a certain range (Braulio-Gonzalo & Bovea 2017). If the thickness of insulation materials is too small, the heat resistance ability will be poor. The energy saving ability from the increased thickness of insulation materials cannot compensate for the increased embodied energy of the insulation materials if it is too thick (Braulio-Gonzalo & Bovea 2017).

Reviewing existing studies (Braulio-Gonzalo & Bovea 2017; Chen, Hammad et al. 2020; Kumar et al. 2020) shows that cellulose, fibreglass, rock wool and polyurethane are four commonly used insulation materials for external walls and roofs. A case study by Chen et al. (2020) to evaluate the performance of these four insulation materials by implementing them on a house in Australia showed that cellulose is the optimal insulation material if embodied energy is the only objective considered, but fibreglass can contribute to the lowest energy requirement compared to the other three. Polyurethane has the best performance in insulation effectiveness, however it also has the biggest embodied energy of the four materials.

Existing buildings have illustrated the tremendous contribution to energy conservation by improving the insulation performance of building envelopes (Hong et al. 2019). With the great capacity of energy saving, the cost of improving insulation can be considered less. For example,

if expanded polystyrene (EPS), a common insulation material, is used, the cost is about AUD 342 per metre square of insulation area (Braulio-Gonzalo & Bovea 2017). In Australia, the maintenance cost of building insulation ranges from 13% to 29% of the total life cycle cost (Illankoon, Tam & Le 2018).

3.4.2. Install green roof (A2)

A green roof is highly recommended for improving buildings' energy efficiency (Hong et al. 2019). Two kinds of green roof are available: intensive roofs and extensive roofs (Shafique, Kim & Rafiq 2018). The intensive roof is more like a roof garden with a thick layer of soil, leading to more weight and high cost. If an intensive roof is constructed on an existing building, the structure of the building has to be strengthened. Compared to an intensive roof, an extensive roof is more applicable for existing buildings. Only a thin layer of soil is needed, and minimum maintenance is required (Shafique, Kim & Rafiq 2018). Therefore, a green roof in this thesis study refers to an extensive roof only. The common material for constructing an extensive roof is polymer, which is light and cheap. There are six layers from top to bottom: vegetation, growing medium, water retention, drainage, root barrier and roof assembly (Shafique, Kim & Rafiq 2018). The study by Bianchini and Hewage (2012) calculated the air pollutants from the recycled and non-recycled polymer for both intensive and extensive roofs. Tables 3.3 and 3.4 illustrate the calculation data for extensive roofs only. The expected operation life of green roofs is 40 to 55 years (Bianchini & Hewage 2012). Based on these provided data and the area of the green roof, the amount of materials and air pollution can be calculated.

Table 3.3. Data for constructing a typical extensive roof

Layers	Materials	Density (g/cm ³)	Thickness (cm)
Root barrier	Low density polyethylene (LDPE)	0.92	0.05
Drainage	Semi-crystalline polypropylene (PP)	0.95	1.50
Water retention	Polymeric fibers	0.95	1.00
		0.95	1:00

Source: Bianchini & Hewage 2012

			We	ight (kg)			
Substance	Unit	Non-r	ecycled	Recycled			
		LDPE	PP	LDPE	PP		
NO ₂	Kg	3.8x10 ⁻³	3.0x10 ⁻³	-2.22x10 ⁻³	6.75x10 ⁻²⁶		
SO ₂	Kg	5.3x10 ⁻³	3.79x10 ⁻³	5.03x10 ⁻³	0		
O3	Kg	4.16x10 ⁻⁹	2.88x10 ⁻⁴	4.16x10 ⁻⁹	6.75x10 ⁻²⁶		
PM10	kg	4.75x10 ⁻³	4.06x10 ⁻⁴	4.75x10 ⁻³	6.75x10 ⁻²⁶		

Table 3.4. Amount	of substances re	eleased to the ai	r per 1 kg of	polymer for	green roof
					0

Note: LDPE is low density polyethylene; PP is semi-crystalline polypropylene Source: Bianchini & Hewage 2012

Green roofs have been regarded as an effective retrofitting activity to reduce buildings' cooling energy. The study by Aboelata (2021) showed that green roofs can reduce buildings' cooling energy by 5% to 15% in France, Singapore, Shanghai and Iran. Specifically, extensive roofs can mitigate the cooling energy of buildings by 1.3% to 13.3% (Aboelata 2021). In general, green roofs can reduce 80% of heat penetration into building roofs in summer, which contributes to 2.2% to 16.7% reduction of energy consumption compared to traditional roofs in summer (Besir & Cuce 2018). Green roofs can also effectively reduce a building's carbon emissions with an annual carbon capture in a range of 0.375–30.12 kg carbon/m² (roof area) (Besir & Cuce 2018). Green roofs can also improve indoor air quality and acoustic comfort, since the indoor environmental quality is largely affected by the proximity to outdoor sources (Barmparesos et al. 2018; Aboelata 2021). Green roofs can protect roof structures from extreme outdoor temperatures and large temperature fluctuations (Barmparesos et al. 2018). Moreover, green roofs also work as acoustic insulation to mitigate diffracting sound waves over roofs and reduce sound transmission through the roof system (Van Renterghem 2018).

3.4.3. Upgrade windows (A3 & A4)

Highly insulated windows not only save energy consumption by reducing heating or cooling requirements, but also contribute to a steady indoor comfort and better wellbeing and health for occupants (Simko & Moore 2021). Two values are commonly used to measure the insulation performance of windows: U-value and solar heat gain coefficient (SHGC). The definition of U-value is in Section 3.4.1. SHGC is a fraction (from zero to one) that represents the proportion of incident solar radiation passing through a window (Dabbagh & Krarti 2021). Windows with high SHGC allow more solar radiation to pass through. Therefore, a high SHGC is desired in a heating dominated climate such as northern China for the "free" heating offered by solar radiation. Conversely, a low SHGC is preferred in a cooling dominated climate such

as Sydney in Australia for keeping the indoor temperature cool (Cuce 2018; Simko & Moore 2021).

Generally, two strategies can be adopted individually or jointly: replacing existing windows (glazing with/without frames) with energy efficient ones, and installing sun shading devices.

Replacing existing windows with energy efficient windows depends on types of glazing and materials of frames (Simko & Moore 2021). Double glazing is safer, more airtight, and soundproof than single glazing (Hong et al. 2019). Filling the gap between window panes with argon or krypton gas can lower thermal conductivity by 34% compared to air in the gap (Berardi 2018). A low emittance (low-e) coating that can suppress radiative thermal heat is normally installed to gain a low SHGC (Simko & Moore 2021). Simko and Moore (2021) conducted a case study using a house in Australia to analyse energy saving and cost efficiency performance of energy-efficient windows in Australia. Based on their study, the thermal conductivity and cost of different combinations of glazing types and frame materials are summarised in Table 3.5. Apart from double glazing, installing triple glazing or multiple glazing windows, or adding secondary glazing on existing windows are also potential activities for better energy efficiency, and thermal and acoustic insulation performance compared to single glazed windows (Bulut et al. 2022). However, the price of these activities varies in different countries.

		Double glazing	5	Double glazing with low-e coating					
Frame option	U-value (W/m ² •K)	SHGC	Cost (AUD/m ²)	U-value (W/m ² •K)	SHGC	Cost (AUD/m ²)			
Timber	2.6	0.5	609	1.7	0.17	703			
Aluminium	3	0.64	508	2.1	0.36	585			
Thermally broken aluminium	2.8	0.64	529	1.8	0.36	608			
uPVC	2.3	0.41	356	1.7	0.24	410			

Table 3.5. U-value, SHGC and cost of different options of glazing types and window frames in Australia

Source: Simko & Moore 2021

Installing sun shading devices to reduce heat gain from solar radiation is also an effective way to save energy use. This retrofitting activity is more recommended in tropical areas where solar radiation is abundant. Sun shading devices include external vertical shading, horizontal shading

and internal blinds (Haule, Chaiwiwatworakul & Chirarattananon 2020). Appropriate application of sun shading devices can reduce incoming heat fluxes by about 39% which needs to be compensated by an air conditioning system (Evangelisti et al. 2020). The exact energy saving by sun shading devices can only be estimated by considering the particular situation of the building such as location, climate, shape coefficient of the building, window/wall ratio (WWR), etc.

Valladares-Rendón, Schmid and Lo (2017) investigated the potential energy saving by using sun shading devices in Singapore, South Korea and Taiwan. They found that for a rectangular building in Taiwan, about 8.92% energy saving can be achieved if a horizontal shading is applied. In South Korea, a south-facing building with a horizontal shading is able to lower the cooling loads during May to September by 14.81 kWh/m² (or 19.7% energy demand reduction). If a fixed tilted overhang with 60° adjustable slats is installed on the south-facing windows, about 66% energy saving can be achieved (Valladares-Rendón, Schmid & Lo 2017). In Singapore, the yearly space cooling loads can be lowered by 21.2% in a building by installing the overhang with 30° incline downward on the west-facing windows (Valladares-Rendón, Schmid & Lo 2017). Sun, Cui and Jiang (2018) conducted a case study in the cold zone of China that found about 5.32% potential energy saving in a building if external shading is applied.

In addition to energy saving, sun shading devices are used for a variety of other purposes. They can maintain acceptable visual and thermal comfort conditions, protect against heat and glare on sunny days, and reduce cooling loads and lighting requirement (Stazi, Naspi & D'Orazio 2017).

3.4.4. Install building management control system (A5)

Installing a building management control system (BMCS) can control and monitor the use of electricity of related building systems like the HVAC and lighting system. It can automatically turn on and turn off these building systems according to the schedule of tenants and external weather parameters like ambient humidity, temperature and solar radiation (Yildiz, Bilbao & Sproul 2017). The basic BMCS strategy is to schedule the operation of the HVAC system and automatically turn it off during non-working hours, during holidays, and when the required

indoor temperature is the same as the outdoor temperature, and fresh air only is provided. A BMCS monitors use of electricity by different consumers in real time, which can alert building managers to breakdowns and mistakes during the operation of these building systems to avoid electricity waste. The ranges of potential energy saving through the BMCS of the HVAC system and lighting system are 10% to 28% and 43% to 71% respectively (Chen, Zhang et al. 2020).

3.4.5. Upgrade lifts to more energy efficient ones (A6)

Lifts account for 1% to 15% of overall energy consumption by an office building, which is the lowest compared to other equipment because the consumption depends on the actual appliance (Ali et al. 2021). During peak time, lifts may consume about 40% of the energy use of a building (Al-Kodmany 2015). Advanced technology can be applied to both the hardware and software of lifts to achieve more energy saving. Energy efficient hardware of lifts includes alternating current (AC) motors and direct current (DC) motors, geared and gearless motors, machine-room-less technology (MRL), regenerative drives, elevator ropes, twin systems, double deck lifts, and LED lighting. Energy efficient software involves destination dispatching system, people flow solution, and standby solutions. These technologies do not only bring energy saving, but are also space saving and have more efficient traffic flow. New lifts with these advanced technologies can save about 30% of energy compared to conventional lifts (Al-Kodmany 2015; Carrillo et al. 2013).

3.4.6. Upgrade HVAC system (A7 & A8)

The HVAC is the biggest or second biggest energy consumer in a building, accounting for 30% to 80% of total energy use depending on HVAC options and climate zones (Simpeh et al. 2021). The main reason is believed to be the heating and cooling loss due to the poor insulation performance of building envelopes (Far & Far 2019). In Australia, about 40% of total energy used by buildings for meeting heating and cooling requirements is due to the poor thermal performance of buildings (DEWHA 2008). As for HVAC itself, related equipment and technology are also mature and have been widely applied for improving energy efficiency. The study by Simpeh et al. (2021) discussed measures for enhancing the energy efficiency of HVAC systems at a low cost, such as re-commissioning the system, to a major approach such as installing a smart management and control system which has a longer payback period.

Accordingly, the energy consumption by HVAC can be effectively reduced by two ways: reducing energy demand via improved insulation performance of building envelopes; and improving the energy efficiency of HVAC itself (Australian government 2022).

The strategy for improving buildings' insulation performance has been discussed in Sections 3.4.1 to 3.4.3. It is important to point out that energy saving from the HVAC system can only reach an optimum with the pre-condition that the building is well-insulated (Simpeh et al. 2021). Even though both improvements can benefit environmental saving, the efficiency of the HVAC system has a limited impact on the optimum insulation solution of building envelopes (Landuyt et al. 2021). In addition, the cost of a building's insulation is generally less than the cost of upgrading the HVAC system. Therefore, improving building insulation should be prioritised over improving the energy efficiency of the HVAC system.

To improve the energy efficiency of the HVAC system, different types of upgrade can be implemented: replacing part of the existing HVAC system with a more efficient one; and replacing the whole HVAC system with a more efficient one.

A case study of HVAC retrofitting was conducted by the Australian Government (2010) on a commercial building in Canberra, Australia. The adopted retrofitting activities for upgrading HVAC include:

- rezoning the HVAC system to improve occupant comfort and reduce conflict between the operation of cooling and heating systems
- converting the existing constant volume air distribution system to a semi variable air volume (VAV) system modulated at branch ducts, allowing variable rates of air flow depending on air conditioning requirement
- replacing an old reciprocating R22 chiller with a modern high-efficiency centrifugal machine with magnetic bearings and adiabatic cooling pads
- installing a modern building management system (BMS) to effectively control and monitor the new HVAC system.

The outcome showed that the retrofitting contributed to a 70% reduction in annual greenhouse gas emissions (about 786 tonnes CO_{2-eq}), AUD 120,000 saving on annual energy cost, and improvement in occupant comfort.

Che et al. (2019) conducted a similar study in Hong Kong, which illustrated that about 50% of energy use reduction can be achieved by the below retrofitting activities:

- adopting a sensor-based building management system for better ventilation and energy performance through auto-adjustment of the amount of air flow and cooling water based on sensed indoor CO₂ concentration and temperature
- adding dehumidification coils to remove moisture from the outdoor air to provide acceptable indoor thermal comfort in a hot and humid climate, saving energy from extra cooling and reheating processes
- updating a filtration system from an aluminium filter to a two-stage filtration system with aluminium filter and pleated filter to reduce the ingress of outdoor particles.

Similarly, a study by Lu et al. (2021) indicated that about 15% energy saving can be achieved by replacing the water-cooled chiller of the existing HVAC system with a more efficient one. In addition to energy saving and carbon emissions reduction, the HVAC system is also crucial to both indoor air quality and thermal comfort. In particular, the design and operation of the HVAC system greatly impacts the levels of indoor temperature and humidity (Che et al. 2019).

3.4.7. Upgrade lighting system (A9–A11)

About 26% of a building's energy use is by the lighting system, making it the second largest electricity consumer (Haule, Chaiwiwatworakul & Chirarattananon 2020). There are three main strategies to reduce its electricity demand: retrofitting existing luminaires with energy efficient luminaries, such as LED; reducing the illumination level; and implementing control systems for better working efficiency (Dubois & Blomsterberg 2011; Wagiman et al. 2020). By appropriate retrofit, the energy consumption of a lighting system can be reduced by as much as 75%, with about half from upgraded fixtures, and half from the installation of controls (BEEX 2017).

By reviewing studies about lighting retrofitting, the below three activities are commonly used to achieve energy saving and indoor visual comfort:

- Installing motion sensors which can automatically turn on lights when motion is detected and turn off lights when people leave the area. With the installation of motion sensors, the electricity consumption is anticipated to be 30% to 40% lower than without sensors installed (Riyanto et al. 2018).
- Changing the lighting system to T5 or LED lighting system can provide great system savings on energy consumption (Lecamwasam, Wilson & Chokolich 2012). A study by Han et al. (2019) investigated energy saving by upgrading the lighting system of office buildings in South Korea and found that about half of the energy saving can be achieved if upgrading fluorescent lamps to LED.
- Installing a daylight dimming control system can automatically dim indoor lighting if day lighting is enough. By adopting an appropriate dimming strategy, a range of 31% to 60% of energy saving can be achieved (Beccali et al. 2019).

3.4.8. Install solar/PV-assisted units (A12)

Photovoltaic (PV) technology is the most widely used renewable energy system in the building sector due to the ability to provide on-site electricity (Belussi et al. 2019). Abundant solar energy provided globally also makes solar energy a feasible renewable energy. The sun provides the entire land surface with energy of about 5×10^{24} J each year (da Cunha & de Aguiar 2020). This amount is approximately 10,000 times greater than the real amount consumed globally each year (da Cunha & de Aguiar 2020). Renewable energy can help reduce reliance on energy supply based on fossil fuels and reduce associated carbon emissions. Installing PV panels can generate power, and also reduce annual cooling loads by about 38% (Wang et al. 2020). The limit of adopting PV technology for existing buildings is the availability of roof space to install PV panels (Belussi et al. 2019). The installation of PV panels to generate energy can reduce electricity demand by an office building by about 26% (Nguyen 2017).

Wang et al. (2020) investigated potential energy saving by three types of PV systems in 13 cities in China. When taking into account the double effect of shading and power generation the daily overall energy-saving efficiency was 63.35% for the horizontally-mounted PV roof, 62.73% for the tilted PV roof, and 59.54% for the firmly-attached PV roof. A case study by Chahidi et al. (2021) identified that in the Mediterranean climate about 16% of energy demand can be reduced in the cooling period by adopting a PV system. In conclusion, installing PV panels can reduce energy demand by existing buildings, but the effectiveness of power generation depends on the local climate and geographic conditions.

3.4.9. Upgrade water system (A13–A15)

Office buildings are a major consumer of urban water, and about 50% to 90% of water use in office buildings is for toilet flushing and cooling tower blowdown (Cook, Sharma & Gurung 2014). There are several activities available for improving water efficiency:

- Installing water control sensors can minimise water use from water taps, toilet flushing and urinals. The study by Nguyen (2017) illustrated that using water control sensors can achieve 4–5 litres water per flush of toilets and urinals, and a flow rate of 4–4.5 L/minute for tapware. This strategy can contribute to water saving of 20%–34% (Metallidou, Psannis & Egyptiadou 2020).
- Replacing existing water fixtures with more water efficient ones, such as a water system with a high-pressure toilet water tank, sub-water meter, water control sensors, leak detection monitors, and waterless urinals can reduce water use. Using water saving fixtures can achieve water conservation of up to 40% compared to traditional fittings (Balachandran, Mahanta & Samuel 2020).
- Installing a water treatment system to reuse stormwater, and/or recycle grey or black water can help reduce water use. Up to 50% of water consumption in office buildings can be treated on site and reused for toilet flushing and/or landscape irrigation (Friedler & Alfiya 2010; Nguyen 2017).

3.5. Interruption to building tenants

There is no doubt that sustainable retrofitting can bring environmental, economic and social benefits if an appropriate retrofitting strategy is designed and implemented. However, the retrofitting construction may cause interruption to tenants, and building owners may be afraid that tenants may terminate the lease because of it. In fact, the earlier the understanding of disruption in terms of type and extent, the better the disruption can be accepted or managed (Chaves et al. 2016). Chaves et al. (2016) categorised disruption to users into four types:

- disruption of utilities: gas, electricity, and/or water supply is disrupted
- disruption of traffic: internal traffic flow, and/or access to the building is interrupted
- disruption of physical space: physical comfort is interrupted, and/or work space is occupied by retrofitting construction
- disruption of internal environment: noise, dust, and/or debris from retrofitting construction.

These disruptions impact existing tenants to different extents. Tzortzopoulos et al. (2019) classified these disruptions into three levels: high, medium and low. High-level disruption refers to disruption inside buildings or interruption of normal work activities or building service provision. Medium-level disruption is caused by retrofitting activities that have a long construction duration, may lead to limited access to the building or service, or cause excessive unfavourable impact on the indoor environment. Low-level disruption is normally caused by retrofitting activities executed on the outside of buildings. Based on this classification of interruption, the retrofitting activities listed in Table 3.1 are recategorised in Table 3.6.

Disruption level	Retrofitting activities	Reason
Low-level disruption	A2. Adopt extensive green roof	Occurs outside of buildings
_	A4. Install sun shading devices	
	A12. Install PV panels	
Medium-level disruption	A1. Install/upgrade insulation of	Occurs outside of buildings
_	building envelopes	Causes noise and waste
	A5. Install BMCS	Occurs inside building, but duration of
		limited access to building service is
		short
	A6. Upgrade lifts to more energy	Occurs inside the building, but
	efficient ones	construction duration is relatively
	A9. Install motion sensors for lighting	short and can be done during off-work
	system	hours
	A13. Install water control sensors	
High-level disruption	A3. Replace existing windows with	Occurs inside building, and disrupts
	energy efficient windows	normal daily work activities
	A7. Upgrade parts of existing HVAC	Disrupts use of HVAC
	system	
	A8. Replace existing HVAC with more	Disrupts use of HVAC
	energy efficient one	
	A10. Replace fluorescent lamps with	Disrupts use of lighting
	T5 or LED	
	A11. Install daylight dimming control	Disrupts use of lighting
	system	
	A14. Replace existing water fixtures	Disrupts water use
	with more water efficient ones	
	A15. Install water treatment system	Disrupts water use

T 11 2 (D' /'	1 1 0	· · · · ·	
lable 16	Disruption	levels of	refromiting	activities
1 4010 5101	Distaption		recronicenne	

Office buildings vary in terms of location, materials, construction type, energy type, age, size, and occupancy characteristics. Given the different situations of existing buildings and different levels of upgrade, the improvements from and the negative impacts of these retrofitting activities are different. Therefore, they should be selected by considering the specific condition and situation of the target building. The estimation of the environmental, economic and social impacts of retrofitting activities can help decide retrofitting strategies effectively.

3.6. Summary

This chapter first discussed the current performance of office buildings in the environmental, economic and social dimensions. Existing office buildings are responsible for large energy consumption, carbon emissions, and high operation and maintenance costs. On the social dimension, aging office buildings may lead to poor indoor comfort and less job satisfaction for building occupants. Considering the large quantity of existing buildings, their long service life, and the identified massive negative impacts, retrofitting is recognised as a better remedy than demolition and new construction. Sustainable retrofitting can quickly improve existing

buildings' environmental, economic and social performance, and can also retain the embodied energy and embodied carbon emissions in existing buildings and avoid more of them being created by new construction. Based on previous studies, 15 common retrofitting activities for office buildings were identified and the potential contribution of each retrofitting activity was discussed. In addition to the contribution, the interruption to existing tenants by implementing these retrofitting activities was also discussed. Then, based on different levels of interruption, the identified retrofitting activities were categorised, which may help select suitable retrofitting activities.

The identified retrofitting activities can be used as a checklist. When the decision model developed in this study is implemented in practice, the retrofitting team can select suitable activities from it. However, the optimal retrofitting strategy can only be generated based on the estimation of the performance of the proposed retrofitting activities. Therefore, the next chapter reviews existing sustainable assessment methods and decision-making methods. By identifying the benefits and limits of each method, suitable ones are adopted to develop a conceptual model for retrofitting, as illustrated in Chapter 6.

Chapter 4. Sustainability assessment methods for buildings

4.1. Introduction

The last chapter discussed the current sustainability performance of existing office buildings and concluded that the poor performance of existing office buildings causes massive environmental pollution, operation costs and associated social impacts. To alleviate these negative impacts, sustainable retrofitting or demolition and new construction are two possible solutions. The existing studies show that retrofitting existing buildings can provide faster delivery of sustainable development compared to demolition and new construction. To assess the effectiveness of retrofit, this chapter discusses existing sustainability assessment methods for buildings. First, the assessment methods for evaluating buildings' environmental, economic and social impacts are introduced. Then, the assessment models incorporating two or more sustainable dimensions are discussed regarding their assessment framework and limitations.

4.2. Environmental impact assessment

Environmental assessment methods are important tools to assess and monitor buildings' environmental performance, and also link a building's environmental context to the decisionmaking framework of design strategies (Carvalho, Bragança & Mateus 2021; Sartori et al. 2021). Life cycle assessment (LCA) and green building rating systems (GBRSs) are two main approaches that can holistically evaluate the environmental performance of buildings (Mahmoud, Zayed & Fahmy 2019; Mattoni et al. 2018; Sartori et al. 2021). The assessment framework and associated benefits and limitations of these two approaches are discussed.

4.2.1. Life cycle assessment

Life cycle assessment (LCA) is popular as the only standardised method of environmental impact assessment (Dong & Ng 2016). It is widely used to quantify the environmental impact of a product or service from a life cycle perspective (Llatas, Soust-Verdaguer & Passer 2020). The international standards ISO 14040 (ISO 2006a) and 14044 (ISO 2006b) are the foundation

for performing LCA. These two standards govern four phases of LCA: the goal and scope definition, life cycle inventory (LCI) analysis, life cycle impact assessment (LCIA), and interpretation of results. Each phase is discussed in the following sections.

4.2.1.1. Assessment framework: goal and scope definition

The first phase of LCA is to define the goal and scope of a study. This phase affects other aspects of an LCA study, including the selection of methods and adoption of details for LCI, the method of impact assessment interpretation, and reporting format (Desideri & Asdrubali 2018). Based on ISO 14040 (ISO 2006a) and 14044 (ISO 2006b), the following items should be identified to define the scope of an LCA study:

(1) Functional unit (FU): FU defines the quantification of the identified function and subfunctions of the studied system. It serves as the basis for the quantification of all inputs and outputs. In addition, FU allows assessment results of different products or services to be compared based on an equivalent functional performance (Souza et al. 2021). Different FU options for LCA studies of buildings are illustrated in Table 4.1. Among all the provided options, a square metre (m²) of floor area is the most commonly used FU for building LCA (Saade, Guest & Amor 2020). In this study, the whole building (represented as gross floor area) is adopted as the functional unit.

Dimension	FU	Description						
Space	m ²	Net floor area						
		Gross internal area						
		Gross floor area						
		Air conditioned area or unconditioned area						
Time	year(s)	Each year during lifetime						
		Years during lifetime						
Service	Occupancy							
Space per time	m ² /year							
Space per service	m ² /occupancy							
Space per time per service	m ² /year/occupan	ICV						

Table 4.1. Functional unit options for LCA studies of buildings

Source: Adapted from Saade, Guest & Amor 2020; Souza et al. 2021

The space dimension considers the whole building, a section of it, or a specific amount of space, expressed as area (m^2) or volume (m^3) . The time dimension refers to a building's life span. When addressing particular quality levels, the service dimension

should be defined. If people use a building or items stored in a building, the FU per occupancy can be adopted. In cases of product manufacturing or service provision, per product output can be adopted as the FU (Souza et al. 2021).

- (2) Reference study period (RSP): RSP is the time frame used to investigate the time-dependent characteristics of the object being evaluated (Desideri & Asdrubali 2018). For LCA studies of buildings, it refers to the use phase of a building, during which maintenance, repair and replacement activities will take place. Due to nontechnical factors like occupants' behaviour, local climate, surrounding environment and maintenance schedules, it is difficult to define an exact remaining service life for buildings. Therefore, a common time frame is used in research studies as a reference study period to assess buildings' performance. The common RSP used by research practice and international certification schemes, is 50 years or 60 years depending on different depreciation principles for construction investment (Rasmussen et al. 2020). However, a 50-year study period is typically used as a default RSP in LCA (Desideri & Asdrubali 2018, Rasmussen et al. 2020).
- (3) System boundary: The boundary determines the unit process that is considered for the LCA study and indicates which stages and what processes during each stage are included (Birgisdottir & Rasmussen 2016). According to EN 15978 (CEN 2011), Figure 4.1 illustrates modular information for different stages of the building assessment contained in LCA. Different settings of system boundaries for LCA studies are also illustrated in EN 15978 (CEN 2011).

Based on different study purposes, different stages of buildings can be selected for formulating system boundaries. Common system boundaries include cradle to gate, where only the product stage of a product is considered (A1 to A3 in Figure 4.1); gate to grave, where the construction stage, use stage, and end-of-life stage are considered (A4 to C4 in Figure 4.1); cradle to grave, where the product stage, construction stage, use stage, and end-of-life stage are considered (A1 to C4 in Figure 4.1); and cradle to cradle, where all life stages are considered (A1 to D in Figure 4.1). Cradle to grave is the most widely used system boundary for LCA of buildings (Anand & Amor 2017).

Pr /	oduc \1-3	t	С	Cons A	tructic \4-5	on			L B	Jse 1-7				Eı	nd of C1-4	life 4			Beyond the system boundary - D
Raw material supply	Transport	Manufacturing		Transport	Construction – installation process		Use	Maintenance, repair,	replacement,	refurbishment	Operational energy	Operational water use	De-construction /	demolition	Transport	Waste processing	Disposal		Reuse, recovery, recycling potential
← Cradl ←	e to	Gate	¥											Cr	Gate adle	to Gr to Gi	ave rave	XX	
<u> </u>																			Cradle to Cradle

Figure 4.1. Modular information for different stages of the building assessment in LCA

Source: Adapted from Sartori et al. 2021

Following goal and scope definition, LCI analysis is conducted to create an inventory of flows that can be based on quantifying the inputs and outputs of the studied system in FU (Desideri & Asdrubali 2018). LCI of buildings is recognised as very complicated due to the different materials and processes involved and the dynamic nature of operating a building (Anand & Amor 2017). Figure 4.2 illustrates the LCI process applied to a building system, and the LCI phase is used for the data collection and modelling of the system (Hauschild, Rosenbaum & Olsen 2018).



Figure 4.2. LCI process applied to a building system

Source: Adapted from Hauschild, Rosenbaum & Olsen 2018

Three LCI methods are identified: the Process Analysis, the Input–Output (I-O) Analysis, and the Hybrid Analysis (Vilches, Garcia-Martinez & Sanchez-Montañes 2017). Process Analysis is a bottom-up approach that only includes processes within the boundaries of the studied product system. However, this method does not take into account the environmental impacts of inputs and outputs located outside of the system boundaries. As a result, omission of processes may be caused due to the "left out" impacts in the upstreaming stages (Lenzen 2008; Vilches, Garcia-Martinez & Sanchez-Montañes 2017). This method is suitable to compare different options of the same product or service, since the omission of processes will impact different options in the same way (Lenzen 2008).

The I-O Analysis is conducted based on national data, which makes it an appropriate method for national research (Vilches, Garcia-Martinez & Sanchez-Montañes 2017). However, if it is not certain the assessed product is representative, I-O Analysis may not be a suitable LCI method (Vilches, Garcia-Martinez & Sanchez-Montañes 2017).

The last method is Hybrid Analysis, combining Process Analysis and I-O Analysis (Vilches, Garcia-Martinez & Sanchez-Montañes 2017). In this method, the LCI is conducted based on Process Analysis and includes I-O data (Majeau-Bettez, Strømman & Hertwich 2011). There

is a distinction between process-based Hybrid Analysis and I-O-based Hybrid Analysis in terms of the tier in which the I-O data is included (Treloar 1997; Vilches, Garcia-Martinez & Sanchez-Montañes 2017). Normally, the I-O-based Hybrid Analysis yields more adverse environmental impacts compared to the process-based Hybrid Analysis (Praseeda, Reddy & Mani 2015).

In addition to national data, environmental product declarations (EPD) or related databases can be used as sources of the secondary inventory data (Anand & Amor 2017). Numerous EPD certification programs are available in markets, including Eco Platform, International EPD, Bau-EPD, Inies, Global EPD, EcoLeaf, Milieu Relevante Product Informative (MRPE), Institut Bauen und Umwelt (IBU), Environdec, EPDnorge, NHO, dapC, PEP Eco PASSPORT, Korea Eco-Labeland, etc. (Desideri & Asdrubali 2018). As for databases, different LCI databases are available (shown in Table 4.2) and can be adopted to create LCI.

Region	Database	Description		
Global	Ecoinvent v3.1	International LCI database with most of the industrial,		
		construction and transport processes, and systems		
	GaBi LCA database	International LCI database with most of the industrial,		
		construction and transport processes, and systems		
	GEMIS 4.5	Free database that includes energy and transport processes,		
		materials, processes, recycling, and waste treatment		
	Agri-footprint	Comprehensive LCI database of feed, food and biomass, with		
		around 3500 products and processes		
	LC-inventories	Over 1000 process datasets, which are corrections, updates or		
		extensions of Ecoinvent v2.2 database, created by ESU-		
		Services and other authors		
	NEEDS	Database designed for long-term environmental assessment,		
		with around 800 processes of future energy supply systems,		
		future material supply, and future transport services		
Europe	Reference life cycle	LCI data from front-running EU-level business associations and		
	database 3.1	other sources for key materials, energy carriers, transport and		
		waste management		
	ELCD	Database of the Joint Research Centre (JRC) of the European		
		Commission with more than 300 datasets on energy, material		
		production, disposal, and transport		
Denmark,	IO-database	Input-output database based on the Danish national economic		
Europe		and environmental accounting statistics for 1999		
	LCA Food	Danish database containing more than 600 datasets on basic		
		food products and related processes from agriculture,		
		aquaculture, fishery, industry, wholesale, and supermarket,		
	~ .	including waste treatment processes		
France,	Diogen	Environmental impacts of the NF P 01–010 standards for		
Europe	Due De c	Mana then 8000 detects on ensure material and heating		
Germany,	ProBas	More than 8000 datasets on energy, material production,		
Europe		focuses on mesosages within Commonly		
	Öltahav	Cormon database for construction materials and huilding		
	Okobau	contribute and the second line of the Environment		
		Nature Conservation Building and Nuclear Safety (BMUB)		
Luvembourg	Leitfaden	Public database of materials, components, and construction		
Europe	Lettraden	products		
Snain	ITec	Economic information of the components with more		
Europe	1100	environmental data of each constructive element		
Sweden	National I CA database	More than 500 well-documented I CI data sets in SPINE format		
Europe		for a wide range of industrial processes and household goods		
Larope		and services		
US	LCA Commons	More than 18,000 datasets for US agriculture production and		
0.2		agriculturally derived products		
	NREL	US-American database with around 300 datasets related to the		
		production of materials, components, or assembly in the US		
	Life-cycle inventory database	Provides individual gate-to-gate, cradle-to-gate, and cradle-to-		
		grave accounting of the energy and material flows into and out		
		of the environment that are associated with producing a		
		material, component or assembly in the US		
	Minnesota building	Database of construction materials with rating about issues such		
	database	as environmental, cost, health, sourcing, end of use or lifecycle		
		thinking (based on Athena and BEES databases)		
Canada	Athena database v.4	Comprehensive, comparable LCI databases for building		
		materials and products		

Source: Adopted from Desideri & Asdrubali 2018; Hauschild, Rosenbaum & Olsen 2018

After the LCI analysis, the next phase is life cycle impact assessment (LCIA). Characterisation factors of environmental impacts can be derived in two common ways: at midpoint level and at endpoint level (Huijbregts et al. 2017). At the midpoint level, 11 environmental problems are identified. From midpoint to endpoint where three areas of protection are identified – human health, ecosystem quality and resource scarcity, the damage pathway of environmental impacts are sorted as Figure 4.3 shows (Huijbregts et al. 2017). Accordingly, the environmental interventions can be assessed by linking LCI results via the impact categories (midpoint) to the damage categories (endpoint) shown in Figure 4.3 (Desideri & Asdrubali 2018).



Figure 4.3. The framework of environmental LCIA

Source: Adopted from Desideri & Asdrubali 2018

According to ISO 14040 (ISO 2006a), LCIA contains five elements, and the first three are mandatory:

- (1) Selection: refers to selecting impact categories that should be part of goal and scope definition.
- (2) Classification: refers to classifying inventory flows by assigning them to impact categories based on how much they can affect the chosen indicator.
- (3) Characterisation: refers to using environmental impact assessment models to quantify the potential impact of the allocated elementary flows on the indicator of the category.

- (4) Normalisation: is conducted to calculate the relevant magnitude of each characterised category indicator.
- (5) Weighting: is a process of using numerical factors to express how severely each impact category is relevant to the other categories. With weights assigned, all the weighted impact scores can be aggregated into one overall environmental impact score for the studied system. It is especially useful if the LCA result is integrated with other condensed information like LCC to support decision making.

A variety of LCA tools are available for assessing the environmental impacts of buildings, which are listed in Table 4.3. Some of them are favoured because the data from design tools can be imported, and some are selected due to the ability to integrate multiple impact assessments.

	Included indicators		
LCA tools	Cost	Environmental impact	Greenhouse gases
Gabi	\checkmark		
SimaPro			
Umberto NXT LCA software			
OpenLCA		\checkmark	
TEAM TM 5.2		\checkmark	
EIO-LCA (Economic Input-Output Life Cycle Assessment)		\checkmark	
Boustead Model		\checkmark	
Athena (Impact Estimator for Buildings)		\checkmark	
LEGEP-Life cycle Assessment		\checkmark	
Envest 2		\checkmark	
ECOSOFT		\checkmark	
BeCost		\checkmark	
BEES (Building for Environmental and Economic	al	2	
Sustainability)	v	V	N
EQUER			
EcoEffect			
ECO-BAT 4.0			

Table 4.3. Common building LCA tools

Source: Anand & Amor 2017

The last phase is interpretation of the result, a systematic process to identify, quantify, check, and evaluate information from the LCI and LCIA results. Finally, based on the results, a conclusion can be drawn, and the limits and recommendations should be given to the intended audience (Desideri & Asdrubali 2018; Hauschild, Rosenbaum & Olsen 2018).

4.2.1.2. Benefits and limitations

As the first and only internationally standardised environmental impact assessment method (Klöpffer 2003), LCA is being more widely used to assess potential environmental impacts of products, services and associated resource use (Birgisdottir & Rasmussen 2016). Three specific features of LCA are identified below (Finkbeiner et al. 2010; Hunkeler, Lichtenvort & Rebitzer 2008; Klöpffer 2003):

- (1) The life cycle perspective: accounts for the impact from extracting and processing resources to distributing, transporting, consuming, and recycling and/or disposing of the product. The whole process has to be assessed for all relevant materials and energy flows.
- (2) Setting measures in a functional unit: benefits of the system(s) are measured by quantifying all mass and energy flows, resource and land use, and any potential impact associated with these "interventions" in a functional unit.
- (3) Comparative method: it is essentially a comparative method for decision making between alternatives. Also, it is used to compare the improvements of one system to the status quo.

LCA is also employed in the building sector, acting as a vital component of the evaluation of buildings' environmental sustainability (Birgisdottir & Rasmussen 2016; Lei et al. 2021). In addition to assessing environmental impacts of building-related activities, LCA can also be applied to support decision making for better interventions or material selection toward environmental sustainability of buildings, and to compare environmental performance of two buildings with the same function (Desideri & Asdrubali 2018). However, some limitations prevent the broader use of LCA in building practice. Remarkably, four limitations are highlighted: uncertainty of LCI data input, uncertainty from a long life span of buildings, difficulty in providing a holistic assessment, and temporal issues (Desideri & Asdrubali 2018; Favi et al. 2018; Meex et al. 2018).

First, the environmental impact is assessed in LCA via inventory analysis based on a data process of input and output; thus, the accurate assessment heavily relies on the accuracy of LCI data input (Favi et al. 2018). Different from other products, each building is unique with a specific design and material use, which makes it impossible to standardise the assessment

process. Even though the standards EN 15804 on EPDs and EN 15978 on the assessment of the environmental performance of buildings have been established for years, different interpretations of and additions to these standards are still made by various countries (Meex et al. 2018). Moreover, depending on different study scopes, input data may include information about the type and consumption of energy and materials from resource excavation, manufacturing and transportation to life stages of construction, operation and demolition. Even though numerous EPDs and LCI databases are available, environmental data for all materials is not available (Lei et al. 2021; Meex et al. 2018). Therefore, a lack of accurate LCI databases is reported as one of the major challenges of LCA studies (Desideri & Asdrubali 2018; Hunkeler & Rebitzer 2005; Llatas, Soust-Verdaguer & Passer 2020; Toosi et al. 2020).

Second, there are certain limits of LCA to forecast the future (Favi et al. 2018). Buildings have a long life span that can reach over a century, which increases uncertainty, particularly in the use stage due to refurbishment, repair, replacement, and occupants' behaviour (Meex et al. 2018).

Third, LCA simulation tools cannot provide a holistic environmental assessment. In most existing LCA software tools (Table 4.3), the focus is on the assessment of the embodied impact of materials, leading to the neglect of the estimation of operation energy demand (Meex et al. 2018). Moreover, these tools cannot consider occupants' comfort, wellbeing and health, thus failing to provide a holistic assessment approach and a complete LCA study (Desideri & Asdrubali 2018). In addition, there is a lack of studies applying LCA to building refurbishment (Meex et al. 2018; Vilches, Garcia-Martinez & Sanchez-Montañes 2017).

Lastly, the temporal issue in LCA is still a controversial topic (Lueddeckens, Saling & Guenther 2020). In LCA, the time value of environmental impacts is not considered, leading to most LCA studies making the assumption that environmental impacts occur with absolute certainty and constant magnitude during the study period (Zhang 2017). However, as discussed above, a building is a product with a long life span during which environmental impacts vary with the conduct of maintenance, repair and replacement activities, as well as changing use behaviours of occupants. Therefore, there is a need to distinguish short-term and long-term environmental impacts (Zhang 2017). However, this demand raises an issue of whether

environmental impacts should be monetised or discounted. The proponents of monetising environmental impacts believe that environmental impact can be monetised as long as the following factors are taken into account when discounting environmental impacts: changes in damage magnitude; productivity of capital; pure time preference; and uncertainty (Hellweg, Hofstetter & Hungerbuhler 2003).

Nevertheless, monetising environmental impacts does not comply with the principle of intergenerational equity in which future generations deserve the same quantity and quality of natural resources and treatment as the current generation with a zero-discounting rate (Hartwig 2020). Moreover, in the discussion about the relationship among three sustainability pillars, it is neither desirable nor appropriate to put a price on natural resources, especially for intangible items. Therefore, most studies do not recommend economic discounting of environmental impacts. Thus, this thesis study uses the conventional environmental impact assessment method in which impact discount is not considered.

4.2.2. Green building rating system

Green building rating systems (GBRSs) are widely used to evaluate and verify sustainability of buildings (Shan & Hwang 2018). A GBRS usually consists of a set of explicit performance requirements that buildings must satisfy to be certified. In addition, it will also provide recommendations that may guide the project team in designing and constructing buildings to meet those performance requirements (Shan & Hwang 2018; Varma & Palaniappan 2019). GBRSs are widely used to assess building performance and have received close attention from construction authorities, businesses and academics around the world in recent years.

4.2.2.1. Green building rating schemes

Green building certification labels buildings as sustainable, high performance and energy efficient (Berawi et al. 2019). Since the end of the 20th century, hundreds of green building rating schemes have been introduced to assess, evaluate and categorise buildings at different levels (Cordero, Melgar & Márquez 2019; Varma & Palaniappan 2019). Chew and Das (2008) identified four generations of green building assessment methods: first generation refers to nominal type pass or fail certification systems; second generation refers to simple additive systems, in which assessment results are the addition of rating scores earned in different

assessment categories; third generation refers to weighted additive systems, in which criteria are ranked by experts, and weightings are then assigned by analysing the data using a variety of methods; and fourth generation refers to tools which are operated based on cutting-edge concepts like the energy efficiency of buildings or life cycle impact and cost. Most GBRSs are second or third generation (Varma & Palaniappan 2019). GBRSs normally classify the impacts into different categories and assign a weight to each. The assessment is based on a system that assigns points to determine credits regarding their effect on the severity of environmental loads. Then, a final score is rated as a weighted score, and the degree of certification can be obtained by exceeding the required point thresholds (He et al. 2018).

The Leadership in Energy and Environmental Design (LEED, the USA), Building Research Establishment Environmental Assessment Method (BREEAM, the UK), and Green Star (Australia) are three GBRSs that are widely customised for use in different countries. Assessment Standard of Green Building (ASGB, China) is the national green building assessment standard in China, which is the world's largest construction market. Table 4.4 summarises basic information about these four GBRSs which are discussed in detail in the following sections.

	BREEAM ¹	LEED ²	Green Star ³	ASGB ⁴
Country	UK	US	Australia	China
Certification body	Building Research Establishment Group	US Green Building Council	Green Building Council of Australia	China's Ministry of Housing and Urban- Rural Development
First version	1990	1998	2002	2006
Latest update	2016	2019	2017	2019
Flexibility	83 countries	167 countries	3 countries	1 country
Impact categories and associated weights	 Management (11%) Health and wellbeing (19%) Energy (20%) Transport (6%) Water (7%) Materials (13%) Waste (6%) Land use and ecology (8%) Pollution (10%) 	 Location and transportation (17%) Sustainable sites (10%) Water efficiency (11%) Energy and atmosphere (33%) Material and resources (13%) Indoor environmental quality (16%) 	 Management (14%) Indoor environmental quality (17%) Energy (22%) Transport (10%) Water (12%) Materials (14%) Land use & ecology (6%) Emissions (5%) 	 Safety and durability (17%) Health and comfort (17%) Life convenience (17%) Resources saving (32%) Environmental liveability (17%)
Maximum rating score	100 points (including extra 10 points for innovation)	110 points (including extra 10 points for innovation)	100 points (including extra 10 points for innovation)	110 points (including extra 10 points for promotion and innovation)
Certification levels	 Outstanding (≥ 85 points) Excellent (70-84 points) Very good (55-69 points) Good (45-54 points) Pass (30-44 points) Unclassified (≤ 30 points) 	 LEED platinum (≥ 80 points) LEED gold (60-79 points) LEED silver (50-59 points) LEED certified (40-49 points) 	 Six stars (> 75 points) Five stars (60–75 points) Four stars (45–59 points) One to three stars (10–44 points) 	 Three stars (≥ 85 points) Two stars (70-84 points) One star (60-69 points)

Table 4.4. Summary of four green building rating schemes

Source: Adapted from He et al. 2018; Suzer 2019; Liu & Leng 2021; Sartori et al. 2021 Note:

1. The weighting for BREEAM is based on BREEAM International New construction for non-residential buildings

2. The weightings for LEED is based on LEED V4.1 for BD+C New Construction

3. The weighting for Green Star is based on Green Star – Design & As built V1.3

4. The weighting for ASGB is based on ASGB - Design

BREEAM

BREEAM is the oldest protocol of GBRSs in the world (Cordero, Melgar & Márquez 2019; Mattoni et al. 2018). It was initiated for the construction stage of new structures in the UK. It now includes the whole building life cycle from the design stage to in-use and retrofitting and is available for global application (Mattoni et al. 2018; Varma & Palaniappan 2019). BREEAM consists of five systems: BREEAM–Community, BREEAM–New construction, BREEAM–In use, BREEAM–Refurbishment and fit-out, and BREEAM–CEEQUAL. Each system has a different scoring and weighting system (Cao 2022).

LEED

LEED is one of the most well-known GBRSs for evaluating sustainability for different life stages, including design, construction, maintenance and operation (Berawi et al. 2019; Mattoni et al. 2018). LEED consists of eight rating tools: LEED–Building Design and Construction; LEED–Operations and Maintenance; LEED–Interior Design and Construction; LEED–Homes; LEED–Neighbourhood Development; LEED–Cities and Communities; LEED–Recertification; and LEED Zero. It is regarded as the most influential and widespread international GBRS, which has been adapted into other national versions (Mattoni et al. 2018). Moreover, according to Sartori et al. (2021), LEED is the most cited of all green building rating schemes in the relevant studies. LEED is more like a design-guide scheme that provides the project team with a series of design strategies and measures (He et al. 2018). However, LEED is criticised as an energy-oriented environmental assessment tool since it gives the highest weight to the assessment criterion of energy and atmosphere. The overemphasis on energy efficiency may lead to other environmental impacts being overlooked (He et al. 2018).

Green Star

Green Star is a voluntary rating system for buildings and communities initially in Australia (He et al. 2018). It is also a popular GBRS, which has been customised with national versions in New Zealand and South Africa (Mattoni et al. 2018). Four Green Star rating tools are available: Green Star–Communities, Green Star–Design & as built; Green Star–Interiors, and Green Star–Performance (Green Building Council of Australia 2020). In contrast to LEED as a design-guide scheme, Green Star is a performance-based rating scheme that primarily uses quantitative data to anticipate environmental performance (He et al. 2018). Even though the two assessment criteria of energy and indoor environmental quality are assigned most weight in Green Star, environmental issues are treated in a relatively balanced way compared to LEED. It is noted that Green Star explicitly requires the process management assessment over the building life span (He et al. 2018).

Assessment Standard of Green Building (ASGB)

ASGB is the only national environmental rating system in China, the world's largest construction market (He et al. 2018). ASGB is a single manual that is used to assess all civil buildings in China (Liu & Leng 2021). There are two assessment stages in ASGB. The first
stage is the pre-assessment, which is the evaluation conducted after the construction design is finished. The second stage is the building assessment, which evaluates sustainability performance after the building is constructed (Liu & Leng 2021). Like Green Star, ASGB is also a performance-based rating scheme, which effectively encourages "appropriate" design innovation toward sustainability (He et al. 2018). Different from the previous version (established in 2016) focusing on resource saving, the latest update in 2019 emphasises the concept of being people-oriented by adding three new assessment categories: safety and durability, health and comfort, and life convenience (Cao 2022). It is noted that, unlike the other three discussed GBRSs, CO₂ reduction is only considered as a bonus in ASGB, and it is not a mandatory item. Additionally, no specific measures have been suggested to reduce CO₂ emissions in ASGB (Liu & Leng 2021).

4.2.2.2. Benefits and limitations

GBRSs have become increasingly significant in the development of green buildings due to their ability to assist buildings owners in baselining (i.e., setting a baseline measurement for future performance calibration), benchmarking (i.e., establishing a basis for comparing to competitors), decision making (i.e., developing a basis for selecting among alternatives), and documentation (i.e., documenting evidence to comply with sustainable regulations) (Shan & Hwang 2018). GBRSs are often used as a benchmark of quality, which aids in effectively communicating the goal and objectives of sustainable development with project stakeholders (Shan & Hwang 2018; Varma & Palaniappan 2019). In addition, Varma and Palaniappan (2019) stated that GBRSs can facilitate resource conservation and alleviate environmental impact while meeting users' needs. However, there are some limits to existing GBRSs, which have been discussed in relevant studies.

First, in the globalised world, the sustainability level of buildings should be able to be compared among different countries (Mattoni et al. 2018). Many studies have been conducted to analyse and compare various green building rating schemes. The finding illustrates that each rating scheme is established based on the local climatic and geographic condition, which differs from place to place. The variations have a significant impact on the selection of impact categories, scoring method, credit allocation, and especially, the weighting system for each rating scheme. Consequently, different final scores may be achieved using different schemes to evaluate the same building (Karaca et al. 2020; Mattoni et al. 2018).

Second, many GBRSs claim that they are able to address the sustainable development goal holistically; nevertheless, most of them cannot even include all the three sustainability dimensions (Varma & Palaniappan 2019). According to Table 4.4, most impact categories in these four rating schemes focus on environmental impacts (Cordero, Melgar & Márquez 2019; Newsham et al. 2013). Therefore, it is stated that current GBRSs are still environmentaloriented, even energy-oriented tools (He et al. 2018; Lazar & Chithra 2020; Varma & Palaniappan 2019). LEED-Neighbourhood and BREEAM-Communities are two rating tools considering social and economic aspects. However, they are still criticised for insufficient analysis of social and economic impacts, and ambiguities in the weighting, scoring and rating system (Deng, Peng & Tang 2019). Without full coverage of the three pillars of sustainability, the trade-offs between energy and other sustainability issues cannot be carefully considered, leading to poor performance in other aspects, even though the building is certified as a green building (He et al. 2018). Therefore, the research gap is identified that a GBRS needs a framework that can implement the triple-bottom line concept of conserving the environment, improving the health, wellbeing and safety of building occupants, and being economically rational (Francis & Thomas 2022; Varma & Palaniappan 2019).

The last limit is that most existing GBRSs are designed for new buildings, and there is still a lack of knowledge and tools for sustainability design for existing buildings (Andersen, Jensen & Ryberg 2021). As discussed in Chapter 3, existing buildings are responsible for many negative environmental impacts, but the GBRSs designed for them are limited (Munarim & Ghisi 2016). For example, LEED–Operations and maintenance and Green Star–Performance are two common rating schemes for existing buildings. However, they are still criticised for focusing on the environmental dimension by assigning the largest weight to the assessment category "Energy" (Karaca et al. 2020; Solla, Ismail & Yunus 2016). Therefore, there is a need for GBRSs to evaluate existing buildings' sustainability performance holistically.

4.3. Economic impact assessment

Life cycle cost (LCC) and cost benefit analysis (CBA) are two commonly used tools to assess the economic impacts of a product or service. The discounted cash flow needs to be calculated in both methods. However, it is for evaluating costs that may occur during the product's life span in life cycle cost, while evaluating the product's profitability in cost benefit analysis (Hoogmartens et al. 2014; Larsen et al. 2022). The benefits and limitations of each tool are discussed below.

4.3.1. Life cycle costing

The conventional economic impact assessments mainly focus on calculating the investment or the initial cost of a project, which is the primary concern of building owners or investors. However, the financial benefits returned in other life stages cannot be considered. Without potential financial return in the future being confirmed, building owners or investors may avoid adapting a project with a long life span due to the high initial cost. Therefore, the assessment method from the life cycle perspective should be used to holistically assess the economic performance of a building project. Life cycle costing (LCC) is a well-known economic appraisal that can estimate the costs of maintaining a facility over a period of analysis (Dwaikat & Ali 2018). The following sections discuss the details of life cycle costing and associated uncertain factors.

4.3.1.1. Background

Life cycle costing (LCC) has been developed and used for decades to assess a project's economic validity and attractiveness (Dong & Ng 2016). The initiation of LCC can be traced back to the 1960s when the US Department of Defence first carried it out to purchase high-cost military equipment (Fauzi et al. 2019; Guinee et al. 2011). It developed out of the need to thoroughly understand the financial flows over a product's life cycle. It allows decision-makers to consider not only the initial cost but also the costs of operation, maintenance and end-of-life treatment (Fauzi et al. 2019).

Kubba (2010, p. 325) stated that "LCC is a technique of combining both capital and operating costs to determine the net economic effect of an investment, and to evaluate the economic

performance of additional investments that may be required for green building". LCC is conducted to make more informed decisions about how well a building will function economically through its useful life cycle (Marzouk, Azab & Metawie 2018). It is primarily helpful to evaluate economic profitability and capital investments, and the LCC analysis from both a system and end-user perspective could be widely adopted in the building sector (Larsen et al. 2022). LCC takes into account all costs related to a product's life cycle that are directly borne by any actor in the product's life cycle, such as the supplier, producer and user (Fauzi et al. 2019; Hunkeler & Rebitzer 2005). In addition, it is also possible to include externalities anticipated to be internalised in the decision-relevant future (Hunkeler, Lichtenvort & Rebitzer 2008; Visentin et al. 2020). Generally, all costs over a lifetime of a building can be considered in LCC, which makes it a cradle-to-grave assessment method (Larsen et al. 2022). Based on the international standard ISO 15686-5: 2017 (ISO 2017), the major cost categories in LCC include design and construction cost, operation cost, maintenance cost, and end-of-life cost.

When conducting LCC in the construction and property sector, uncertainty is mainly from two factors – selection of discount rate and prediction of the building's life span, which are discussed in the following sections. Different decision rules under LCC are introduced and a suitable one for this study is selected.

4.3.1.2. Discount rate

In LCC calculation, a discount rate is adopted to represent the time value of money (Jafari, Valentin & Russell 2014; Younis, Ebead & Judd 2018). The selection of the discount rate has a significant impact on LCC results. Sterner (2000, p. 388) stated that "choosing a discount rate which is too high will bias decisions in favour of short-term low capital cost options, while a discount rate which is too low will give an undue bias to future cost savings". According to the International Standard ISO 15686-5 (ISO 2017), a discount rate, either real or nominal, should be included in the calculation to evaluate discounted impacts on investment. In addition, the discount rate may also be used to calculate the intermediate cash flows that are reinvested at the opportunity cost in the net present value (NPV) method (Arjunan & Kannapiran 2017). Based on the re-investment assumption, the intermediate cash flows can also be reinvested at the calculated internal rate of return in the internal rate of return (IRR) method (Arjunan & Kannapiran 2017). However, if the re-investment rate in IRR is not equal to the one adopted in

the NPV method, conflicts can be caused (Arjunan & Kannapiran 2017; Keef & Roush 2001; Lohmann 1988).

The nominal discount rate is estimated considering inflation effects, while a real discount rate represents the true power of earning money without consideration of the inflation rate (Park 2016; Younis, Ebead & Judd 2018). The Fisher Equation (Equation 4.1) can be used to represent the relation between the three rates (nominal, real, and inflation rate) (Fisher 1907).

$$(1+n) = (1+r)(1+f)$$
 Equation 4.1.

Where,

n – nominal interest rate expressed as decimal

r – real interest rate expressed as decimal

f – inflation rate expressed as decimal

When LCC is adopted for comparing macro investment alternatives, the real discount rate is preferable to the nominal discount rate (Miraj, Berawi & Utami 2021) for three reasons. First, because it does not incorporate the effects of inflation, the real discount rate is more stable than the nominal discount rate. This is also supported by Eldomiaty et al. (2020) by stating that money illusion may be caused by using nominal discount rates to discount real cash flows, resulting in behavioural issues that cause inflation-induced valuation errors. Moreover, it is challenging to predict the inflation rate for a product with a long life span. Second, the calculation is very complicated and time-consuming if inflation is included in the calculation (Miraj, Berawi & Utami 2021). It means that every cash payment or income has been subjected to inflation and should be adjusted accordingly. This includes net uniform series, where every cash flow needs to be adjusted yearly. Third, inflation affects all types of products roughly the same way most of the time. Therefore, there is not much difference between the two methods in the desirability of any alternatives (Miraj, Berawi & Utami 2021).

It is noted that the discount rate may vary over the building's life span. However, it is usually assumed to be constant in LCC studies due to the uncertainty and complexity of predicting the value of money in the future (Jafari, Valentin & Russell 2014; Tavakolan et al. 2022). Therefore, a current real discount rate based on existing studies is adopted in this study to discount the

economic impacts of each proposed retrofitting activity, and it is assumed to be constant during the study period.

4.3.1.3. Building service life

Building service life significantly impacts deciding the study period for a LCC study (Miraj, Berawi & Utami 2021). Jafari and Valentin (2015) examined the impacts of different lengths of remaining service life of retrofitted buildings on LCC results and concluded that retrofitting strategies can be more cost-effective if the remaining service life of the building is extended.

To estimate the service life of a building, two factors have to be considered – the physical lifetime of the building structure and the functional performance of the building system and facilities (Newton, Hampson & Drogemuller 2009). The building structure could remain for hundreds of years if proper maintenance is applied. Therefore, functional performance is the dominant factor determining the service life of a building. As one of the benefits from retrofitting (see Section 3.3.3 in Chapter 3), a building's service life can be prolonged to some extent. However, how long it can be extended is not known since the service life of the main building systems and facilities such as the HVAC system, lighting system, water system, building envelope, and doors and windows are different, and are affected by many factors such as quality of materials and components, users' behaviour, use frequency, and replacement frequency. For these reasons Goulouti et al. (2020) stated that uncertainty is always embedded in service life calculation.

Table 4.5 illustrates the life expectancy of typical building materials and building systems in Australia, which varies from 15 to 50 years. In LCC calculation, the building materials and components are replaced with new ones at the end of the life expectancy. Correspondingly, the recurrent costs occur at the time. Therefore, the building's remaining service life determines how many times the replacement of building materials and components is needed, and thus the total estimated replacement cost as well as the operation cost and maintenance cost. However, the periodic replacement of building materials and components may contribute to extending the building's service life, making it difficult to reach a consensus on a precise length for the life cycle study (Dwaikat & Ali 2018; Miraj, Berawi & Utami 2021).

Building materials/service systems	Life expectancy (years)	Building materials/service systems	Life expectancy (years)
Stucco	50	External sun shading panels	25-35
Stone veneer	50	Internal window blinds	10
Concrete poured-in-place	Building lifetime	HVAC	30
Brick, block, and stone	50	Lighting fittings	15
Precast concrete panel	30	Motion sensors for lighting system	15
Brownstone	40	LED lights	30
External thermal insulation composite system	20	Daylight dimming control system	30
EPS insulation	35-50	PV panels	30
XPS insulation	Building lifetime	Copper (pipe)	75
Cellulose fibre insulation	50	PVC (pipe)	50
Foam glass insulation	Building lifetime	Sewer pipes	75
Stone wool insulation	Building lifetime	Urinals	20
VIP (insulation)	40	Faucets (sink)	20
PUR (insulation)	50	Faucets (shower)	20
Aluminium siding	40	Flush valves (toilet)	20
Opaque modular cladding	30	Flush valves (urinal)	20
Energy efficient windows	20	Toilet fixtures	20
Glazed cladding/Curtain walling	35	Water control sensors	15
Communications installations and controls	15	Water fixtures	50
Lift	30	Water recycle system	30

Table 4.5. The life expectancy of building materials and building systems

Source: Alam et al. 2017; Australian Cost Management Manual, Volume 3, p. 52; EATS 2015; Kono et al. 2016; Kubba 2010; Penny 2015; RICS 2018; Tavares, Silva & de Brito 2020

Based on existing retrofitting studies, a 50-year study period is usually assumed in life cycle studies. Jafari, Valentin and Russell (2014) used a 50-year study period to analyse the life cycle cost of retrofitting for sustainable housing. Rodrigues and Freire (2017) also decided to use 50 years as a study period for analysing the environmental impacts on a building envelope by retrofitting. Similarly, Piccardo, Dodoo and Gustavsson (2020) assumed 50 years as the remaining service life of the case building to identify the life cycle carbon balance by

improving the insulation performance of the building's envelope. Therefore, if the remaining service life of a retrofitted building cannot be estimated, a 50-year remaining lifetime can be regarded as an acceptable study period for analysing the sustainable performance of retrofitting.

4.3.1.4. Decision rules

Several decision rules are implemented in LCC to test the economic validity of competitive alternatives. Common ones include net present value (NPV), equivalent annual cost (EAC), internal rate of return (IRR), payback period (PBP), and benefit/cost ratio (BCR).

(a) Net present value

Net present value (NPV) is the most frequently used LCC calculation method for assessing an asset's economic viability over its life cycle (Marzouk, Azab & Metawie 2018; Toosi et al. 2020). It is the difference between the discounted revenues and expenses at the expected discount rate (Cotter 2022). It can be calculated based on Equation 4.2 (ISO 2017). NPV is usually calculated to compare different building components with different initial costs and maintenance costs for deciding on investments. The positive result indicates the investment is profitable, and the higher NPV calculated, the more desirable the project. Therefore, the alternative with the highest NPV should be selected to achieve maximised profit (Cotter 2022).

$$NPV = \sum_{n=1}^{p} \frac{c_n}{(1+d)^n}$$
 Equation 4.2.

Where,

NPV – net present value C_n – the cost in year, n d – the expected real discount rate per annum n – the number of years between the base date and the occurrence of the cost p – the period of analysis

(b) Equivalent annual cost or income

The equivalent annual cost (EAC) can also be used to assess the life cycle economic viability of assets if the residual service life of different existing buildings is uncertain (Rodrigues & Freire 2017). To calculate EAC, different costs over the building's remaining service life can be converted to an equivalent annual amount at the expected discount rate (Rodrigues & Freire 2017). The calculation can be achieved based on Equation 4.3 (ISO 2017). Positive value

represents income, and negative value represents cost. The alternative with the lowest annual cost or highest annual income should be selected (Cotter 2022).

$$EAC = \frac{Cd}{(1+d)^{n}-1}$$
 Equation 4.3.

Where,

EAC – equivalent annual cost or income

C – the cost in year n

d – the expected real discount rate per annum

n – the number of years between the base date and the occurrence of the cost

EAC is a more suitable LCC technique to calculate a product's life cycle cost when the natural replacement cycle is not an exact multiple of the study period (ISO 2017). Moreover, it is especially meaningful for building owners or investors, since EAC can illustrate whether this project creates enough cash flow to pay back the loan and produce enough after-tax cash flow to reinvest in other future products (Cotter 2022).

(c) Internal rate of return (IRR)

The internal rate of return (IRR) is the compound rate of interest that makes costs equal to benefits when cash flow is reinvested at a given interest rate (ISO 2017). It can be calculated based on Equation 4.4. The alternative that can maximise IRR should be selected (Cotter 2022).

$$NPV(cost, r) = NPV(benefit, r)$$
 Equation 4.4.

Where,

NPV – net present value

r – discount rate that makes the NPV equal to zero

IRR is usually used to rank investment sizes of alternatives and associated patterns of cash flow over time, ultimately to decide whether they are acceptable (Cotter 2022; ISO 2017). However, IRR may violate the re-investment assumption and can also have several different outcomes depending on changes of signs of cash flow, such as the extra cost caused by maintenance or refurbishment (Arjunan & Kannapiran 2017; Lohmann 1988).

(d) (Adjusted) Payback period

Payback period (PBP) is the time it takes to cover investment costs (ISO 2017). A simple payback period uses non-discounted values for future money, while an adjusted (discounted) payback period uses the present value (ISO 2017). The alternative with the smallest (adjusted) payback period is desirable (Cotter 2022). The major problem with this decision rule is it ignores the benefits and costs that occur after the end of the payback period (ISO 2017).

(e) Benefit/cost ratio

In contrast to the above-discussed decision rules for measuring profits by an investment, benefit/cost ratio (BCR) is used to measure the profitability per dollar invested of an investment. It can be calculated based on Equation 4.5 (Cotter 2022). A BCR greater than one means the project is profitable, and the bigger the BCR, the more desirable the project.

$$BCR = \frac{\sum_{n=0}^{n} \frac{B_n}{(1+d)^n}}{\sum_{n=0}^{n} \frac{C_n}{(1+d)^n}}$$
Equation 4.5.

Where,

BCR – benefit/cost ratio B_n – project benefits during period n C_n – project costs during period n n – number of periods d – discount rate

BCR is regarded as convenient for investors to select the most profitable alternative by ranking the BCR of all potential projects. Therefore, it is commonly used to assess competitive projects' validity and relative merit (Tung 1992).

This study develops a model for deciding retrofitting strategies in which different retrofitting activities are contained. These retrofitting activities are selected to improve the performance of associated building components and systems. According to Table 4.5, typical building materials and building system life expectancy varies from 15 to 50 years. It is difficult to ensure the life expectancy of all upgraded building materials and building systems is an exact multiple of the study period – estimated or assumed remaining service life. Moreover, compared to

construction for new buildings, reinvestment and management issues are more associated with retrofitting projects. Therefore, EAC is selected in this study to calculate the life cycle cost of adopted retrofitting activities.

4.3.1.5. Limitations

Even though LCC has been used to assess the economic performance of buildings for decades, there is a certain limitation remaining to be solved – how to treat externalities in LCC (Hunkeler & Rebitzer 2005; Wulf et al. 2019; Zamagni, Pesonen & Swarr 2013). First, it is a major concern to define the scope of LCC. From the perspective of life cycle sustainability, economic, environmental and social conditions evolve over time, and double counting may occur if no clear interface is defined between the three pillars (Hunkeler & Rebitzer 2005). Second, if externalities are included in LCC, the issue of whether environmental and social values should be monetised arises (Wulf et al. 2019). As discussed in Section 4.2.1.2, there is little agreement on attaching monetary value to environmental impact. Moreover, it is also a controversial issue to monetise social value. Therefore, instead of monetising environmental and social values that can be regarded as externalities in LCC, a comprehensive assessment should be conducted by incorporating environmental and social dimensions to track trade-offs between the three pillars from a long-term perspective (Wulf et al. 2019; Zamagni, Pesonen & Swarr 2013).

4.3.2. Cost benefit analysis

Cost benefit analysis (CBA) is a common economic assessment method to assess the attractiveness of projects, from a whole of society perspective (Hoogmartens et al. 2014). It is widely used in decision making by ranking alternatives based on their cost and different benefits (Hoogmartens et al. 2014). The costs and benefits assessed in CBA may not be simply summed up, so conversion fractions are used to convert them to one metric – money (Jeswani et al. 2010).

By monetising costs and benefits, CBA can represent assessment results as a single criterion (money), which enables intuitionistic comparison of different alternatives (Jeswani et al. 2010). However, monetisation is criticised as a simplistic way, which may lead to limitations in applying CBA. First, it is challenging to get consensus on an exact price of damages (Hoogmartens et al. 2014). Second, benefits and costs are monetised in CBA by deterministic values, but the uncertainty of these values is often unknown (Dong et al. 2018). Lastly, as an economic analysis tool, CBA is conducted from a whole of society view by considering external factors. Therefore, it is not suitable to be taken to assess economic performance of a project by only considering effects on the stakeholders (Hoogmartens et al. 2014).

4.4. Social impact assessment

As discussed in Section 3.2.3 of Chapter 3, the social dimension has not received the same attention as environmental and economic dimensions (UNEP/SETAC 2011). It can be seen that environmental and/or economic dimensions are considered in most well-known assessment schemes, but the social dimension is rarely involved (Fatourehchi & Zarghami 2020). There is not an international standard similar to the environmental and economic dimensions which can standardise assessment criteria and assessment methods that is widely accepted. Therefore, the assessment methods for social impact are still limited. Reviewing related studies, two methods are commonly used to assess the social impacts of buildings: the social cost benefit analysis (SCBA) and social life cycle assessment (S-LCA). Apart from these two, a social assessment standard EN16309:2014 was initiated by the European Union (BSI 2014). However, like most other social assessment methods, it mainly relies on qualitative criteria and checklists, leading to less accurate assessments.

4.4.1. Social cost benefit analysis

Social cost benefit analysis (SCBA) is an alternative form of the original CBA, but with socialeconomic criteria incorporated in the assessment (de Nooij 2011; Dodgson et al. 2009). The investment decision based on pure economic cost analysis may also result in impacts on society such as climate change, traffic congestion, etc. Therefore, SCBA as a socio-economic evaluation method is adopted to support decisions, especially those made by government (Hauck et al. 2016).

Similar to the original CBA, all relevant costs and benefits are also evaluated in SCBA in monetary terms (Dodgson et al. 2009). However, the difference is that the estimated costs and benefits in SCBA can quantify societal differences with and without a project. Moreover, in addition to evaluating impacts on decision makers, externalities are also taken into account in

SCBA (de Nooij 2011). Based on the study by de Nooij (2011), SCBA ought to include investment costs, changes in losses, project risk analysis, potential synergies and interconnections across projects. It can also consider socio-economic factors like the exchange of ancillary services, the benefits of a more integrated market, security of energy supply, and optimisation of energy generation. Due to its ability to evaluate projects with multiple aspects, SCBA has been used to support decision making about infrastructure design, such as bridges, building materials, and road safety improvement (Hauck et al. 2016).

A benefit of SCBA is that all effects are formulated precisely by allocating a price on each assessed impact (de Nooij 2011). However, the monetary-equivalent measure in SCBA also receives many critiques (Dodgson et al. 2009; Fleurbaey & Abi-Rafeh 2020). Some specialists in welfare economics even object to this approach because no weights are introduced in SCBA (Fleurbaey & Abi-Rafeh 2020). They argue that it is difficult to reach a consensus on attaching an exact price to welfare variation (referring to the willingness to pay or willingness to accept) (Fleurbaey & Abi-Rafeh 2020). Indeed, social values tend to change in different situations. For rational estimates, the same values should be used across the board because society needs to ration investments of scarce resources. However, it does not mean those values should be used in other situations, since the willingness to pay or accept changes with different situations.

4.4.2. Social life cycle assessment

Social life cycle assessment (S-LCA) is a method for assessing positive and negative impacts throughout the product's life cycle by evaluating its social and socio-economic elements (UNEP/SETAC 2009). Although assessing social impacts by a life cycle approach has been studied since the 1990s, little progress has been made after its initiation, leading to it falling behind the other two life cycle methods – LCA and LCC (Hunkeler & Rebitzer 2005; Klöpffer 2003). The review study by Huertas-Valdivia et al. (2020) analysed publications about S-LCA from 2003 to 2018. The result showed that about 66% of all 187 relevant articles were published, primarily by European authors, after 2009, when the first international S-LCA assessment guideline – UNEP/SETAC Guidelines for Social Life Cycle Assessment of Products – was launched. Additionally, only three articles out of the total analysed articles were about building and construction. The study by Fan et al. (2018) also indicated the first time S-LCA was applied in the area of civil engineering was in 2013. Therefore, even though rapid

growth in S-LCA study has occurred in recent years, it is still at an early developing stage (Dong & Ng 2016; Fan et al. 2018; Huertas-Valdivia et al. 2020).

It is undeniable that there is an urgent need for S-LCA. The social life cycle performance should be assessed and optimised in the early stage of a construction project to support decision making (Liu & Qian 2019). Most importantly, there is a need for a life cycle approach for assessing social impacts along with environmental life cycle assessment (ELCA) and environmental life cycle costing (ELCC) to complete a comprehensive sustainability assessment (Benoît et al. 2010; Fan et al. 2018). ELCA and ELCC, and the differences between LCA and ELCA, and between LCC and ELCC, are discussed in Section 4.5.2.

4.4.2.1. The assessment framework

The S-LCA guideline initiated in 2009 explains how the ELCA approach could be adapted to assess the social and socio-economic impacts of products (UNEP/SETAC 2020). The methodological framework of S-LCA is similar to the assessment framework of LCA, consisting of goal and scope, inventory life cycle analysis, impact assessment, and interpretation (UNEP/SETAC 2009). The updated version in 2020 gives a more detailed introduction to each phase, shown in Figures 4.4 and 4.5.



Figure 4.4. S-LCA decision tree

Source: UNEP/SETAC 2020

Figure 4.	.5. Assessment	framework	of S-LCA	by reference	scale approach
					sector approact

Phase 1. Goal and scope	 Identify assessment goal Define functional unit Define reference flow Define product system Define system boundary Identify activity variable Identify cut-off criteria Identify cut-off criteria Identify assessment method and impact strategies
Phase 2. Life cycle inventory	 Identify the data to be prioritised for collection Collect data for hotpots assessment if this is part of the Goal and Scope Collect data for the selected/relevant stakeholders and subcategories Collect complementary data for the impact assessment Collect site specific (primary) and generic (secondary) data for unit process and activity variables Collect data for scoring and/or weighting
Phase 3. Impact assessment	 Establish reference scales for impact assessment Data collection Assess data against the reference scale Apply an impact assessment method to group by subcategory or impact category and aggregate results over the value chain using an activity variable Finally weight results Present the results
Phase 4. Interpretation	 Completeness check Consistency check Sensitivity and data quality check Materiality assessment Conclusions, limitations, and recommendations

Source: UNEP/SETAC 2020

The first phase is similar to LCA in that the assessment goal and scope should be identified first. As Figure 4.4 shows, whether the assessment will focus on a product or an organisation needs to be clarified. Following that, the study scope should be defined, including system boundary, functional unit, product system, cut-off criteria, etc. (as Figure 4.5 shows). In S-LCA, two main impact assessment approaches are highlighted, and need be clarified in this phase (UNEP/SETAC 2020):

- (1) Reference Scale Approach (Type I): the assessment aim is to describe a product system with an emphasis on social performance and social risk.
- (2) Impact Pathway Approach (Type II): the assessment aim is to predict the effects of the product system with an emphasis on identifying potential social impacts.

By applying the Reference Scale Approach, the performance reference points (PRPs) that are specific referent points of expected activity need to be identified. They are used as a reference to assess the social performance of these activities of organisations in the product system. Therefore, the assessment mainly relies on data, information or judgment. It is normally conducted to assess the immediate effect of evaluated activities and the magnitude and significance of potential further social impacts (UNEP/SETAC 2020).

The Impact Pathway Approach is more in line with E-LCA, in which impacts are assessed based on a "characterisation" process of analysing causal or correlation/regression-based directional relationships between the product system/organisations' activities and the subsequent potential social impacts. Therefore, the assessment is normally used to identify and track the effects of possible activities on longer-term ramifications along an impact pathway (UNEP/SETAC 2020).

In this study, social impact assessment is adopted mainly for assessing the magnitude and significance of the immediate social impacts (in the construction stage) and potential social impacts further down the line (in the operation stage) by retrofitting activities. The cause-and-effect relationship between activities and the associated social consequence is not the focus of this study. Therefore, the Reference Scale Approach (Type I) is more suitable and is applied in this study.

Following the goal and scope definition, the next phase is inventory analysis, where a data collection strategy needs to be developed and decided, in particular, whether to use an S-LCA database or other data resources, and whether to collect site-specific data (UNEP/SETAC 2020).

The third phase is to conduct impact assessment. Since the Reference Scale Approach is adopted in this study, Phase 3 in Figure 4.5 lists the steps of implementing the Reference Scale Approach. The first step is establishing the performance reference points. The second step is collecting required data. The third step is assessing the data against the reference scale. The fourth step is applying a specific assessment method to group impacts based on impact (sub)categories, in which the assessment results can be aggregated. In the final step, the weighted results can be obtained (UNEP/SETAC 2020).

The last phase is to interpret the assessment results built according to ISO 14044 (ISO 2006b). First, the whole assessment process should be reviewed. Then, checks for completeness, consistency, sensitivity and data quality, and materiality assessment need be conducted. Based on the review and check results, the assessment process can be interpreted by summarising results, summarising limitations, and providing recommendations (UNEP/SETAC 2020).

4.4.2.2. Limitations

Two publications by UNEP/SETAC (2020) provide a conceptual framework of social life cycle impact assessment, which guide practitioners and researchers to assess positive and negative social impacts of products or organisations. However, two main limitations of S-LCA remain to be solved.

First, there is no consensus on the social life cycle impact assessment method regulated in S-LCA (Fan et al. 2018). Researchers have to design the assessment method by selecting the specific scoring system, weighting methods, and even stakeholders. Therefore, different assessment results may be achieved in the same study by different researchers.

The second limitation of S-LCA is the lack of consensus on assessment indicators (Huertas-Valdivia et al. 2020). Assessment indicators play a crucial role in S-LCA for providing shortand long-term information, which can help organisations better comprehend their current situation and development tendencies (Kühnen & Hahn 2017). However, the S-LCA assessment framework by UNEP/SETAC does not provide social indicators that can be directly attributed to products or processes. As a result, researchers must use indicators from a different reference level (organisations, regions), and the product relation must then be made based on appropriate methodological assumptions (Finkbeiner et al. 2010).

Regarding these two limitations of S-LCA, even though the Reference Scale Approach is used in this study, the rating scale, assessment method, weighting method, and assessment indicators have to be customised according to the specific situation of this study. It must be noted that, in contrast to SCBA where social impacts are quantified by discounting monetised values, value scores decided based on judgment of social impacts are used as proxy variables in S-LCA. However, how to time-discount impacts judged by subjective information is still an unsolved problem. Moreover, if converting judgment into a monetary scale similar to SCBA, the same critique of monetising social value would apply. Therefore, in this study, social impacts are not discounted, and it is assumed that the social impacts would be constant in the remaining service life of the building.

4.5. Multi-dimensional sustainability model

Assessment methods of sustainable development in a single dimension are available. However, sustainable development requires integration and balanced environmental, economic and social development (as defined in Section 2.4 of Chapter 2). For a long time, researchers have made efforts to amalgamate the environmental, economic and social dimensions to support decision making of sustainable designs. As discussed in Section 4.4, the social dimension is the least developed pillar among the three pillars. Therefore, the two-pillar model combining the environmental and economic dimensions is commonly implemented to assess sustainability performance and support decision making. This section discusses the two-pillar model and three-pillar model of sustainability assessment and multi-criteria decision making (MCDM).

4.5.1. Integrate two pillars into sustainability assessment

Since the concept of sustainable development is implemented in the construction and property sector, the concept of "cost-optimality" is regarded as the drive for designing new buildings and retrofitting existing buildings (Ascione et al. 2017). Energy efficiency and costs are the main indicators used for comparison during the multi-stage optimisation process due to the common application in other studies (Ascione et al. 2017).

Many studies have integrated LCA and LCC to evaluate buildings' performance (Fouche & Crawford 2017). Based on the study by Miah, Koh and Stone (2017), there are six types of LCA–LCC integration, which can be categorised into three groups based on integration approaches: conducting LCA and LCC on the same product or services, but showing assessment results separately; using LCA and LCC as two criteria to evaluate decision-making

options, and aggregating LCA and LCC assessments based on the mechanism of decision making; and expanding existing LCC to include LCA.

The first integration approach is to conduct LCA and LCC on the same product or service but show the assessment results of LCA and LCC separately. Therefore, a portfolio of results is represented as tables, bar graphs, etc. in this integration category (Miah, Koh & Stone 2017). This integration method was adopted in the study by Pombo, Rivela and Neila (2016), which compared the sustainability performance of common renovation strategies. In their study, the LCA results and LCC results of selected renovation strategies were presented in two separate tables, and according to these results, they were ranked for environmental performance and economic performance respectively.

The second integration category regards LCA and LCC as two assessment criteria in a decisionmaking process. The LCA and LCC results can be aggregated based on the mechanism of the decision-making method, such as optimisation in multi-criteria decision making (MADM) (Miah, Koh & Stone 2017). In 1994, a project named "the Building for Environmental and Economic Sustainability" (BEES) was initiated by the US National Institute of Standards and Technology (NIST) for selecting environmentally and economically balanced building products (Alamu et al. 2021; Lippiatt 1998; Xu et al. 2022). In BEES, LCA and LCC work as the assessment baseline, and the LCA and LCC results are aggregated as one based on the decision-making mechanism of multi-attribute decision analysis to select the "most" sustainable product (NIST 2020).

The last integration category is expanding the existing LCC to include LCA (Miah, Koh & Stone 2017; Zuo et al. 2017). One method is to give a monetary value to external environmental impacts (Zuo et al. 2017). The study by Kneifel (2010) monetises the CO_2 emissions of a building and includes the value in LCC to evaluate the building's economic performance by considering the external environmental impact (CO_2 emissions). Environmental life cycle costing (ELCC) is another example to include environmental impacts in LCC (Miah, Koh & Stone 2017; Zuo et al. 2017). Compared to traditional LCC, the assessment aim, perspective and scope in ELCC are much closer to those in LCA. Therefore, it is fully compatible and

always conducted with LCA (Miah, Koh & Stone 2017; Zuo et al. 2017). Moreover, ELCC is one pillar in life cycle sustainability assessment (LCSA), which is discussed in Section 4.5.2.

Similarly, CBA can also be integrated with LCA. It can be realised by conducting LCA and CBA in parallel to evaluate economic and environmental performance of a product or service (Dong et al. 2018). Alternatively, the CBA can be expanded to include LCA by monetising external environmental impacts – environmental cost benefit analysis (ECBA) (Dong et al. 2018; Hoogmartens et al. 2014). The study by Manzo and Salling (2016) assessed the environmental and economic impacts of vehicles by including monetised air pollution into CBA. The result showed that the indirect environmental effects account for a significant portion of the project's estimated expenditure.

4.5.2. Life cycle sustainability assessment

With the wide recognition of the concept of sustainability development, the demand to integrate social and economic dimensions of sustainability into LCA has become increasingly strong. As discussed above, many assessment frameworks can incorporate environmental and economic dimensions, but the social dimension is absent. In 2003, the social dimension (S-LCA) was first integrated with LCA and LCC by Klöpffer (2003). Then, the formation and integration of all the three pillars was proposed and discussed by Hunkeler and Rebitzer (2005) as life cycle sustainability assessment (LCSA).

LCSA is a three-pillar model built on the principle of triple-bottom line (Kloepffer 2008; Llatas, Soust-Verdaguer & Passer 2020; Moslehi & Reddy 2019). With the three pillars included, LCSA provides a complete view of the positive and negative impacts along the product's life cycle. In addition, LCSA also supports decision making by prioritising resources and investing them where there are more opportunities for positive impacts and fewer negative ones (Ciroth et al. 2011).

In LCSA, environmental impact is assessed using environmental LCA (ELCA), the economic dimension is assessed using environmental life cycle costing (ELCC), and social life cycle assessment (S-LCA) is used to assess social impacts. These three techniques have the same

methodology based on the standard ISO 14040 (ISO 2006a), and assessment results for each method are used to represent the performance of life cycle sustainability, as Equation 4.6 shows (Llatas, Soust-Verdaguer & Passer 2020; Soust-Verdaguer et al. 2022).

$$LCSA = ELCA + ELCC + S-LCA$$
 Equation 4.6.

Where,

LCSA – life cycle sustainability assessment ELCA – environmental life cycle assessment ELCC – environmental life cycle costing S-LCA – social life cycle assessment

As discussed in Section 4.2.1, LCA is the only environmental impact assessment method with a standardised assessment framework (see Section 4.2.1). It is also used to assess environmental impacts throughout a product's life cycle in LCSA. ELCC is adopted to evaluate the performance in the economic dimension. The concept of ELCC was initiated by Hunkeler, Lichtenvort and Rebitzer (2008). The UNEP report about LCSA identifies three types of LCC with different boundaries of analysis: conventional LCC, where the post-production stages are overlooked; ELCC, which is a mirror of LCA and considers all costs associated with the life cycle of a product that are directly covered by any one or more of the actors (stakeholders) in the product's life cycle; and social LCC where all external and private costs and benefits are monetised (Ciroth et al. 2011; Fauzi et al. 2019). It has generally been accepted that ELCC should be used in LCSA because it complies with the concept of LCA and is consistent with the environmental dimension for avoiding double-counting (Finkbeiner et al. 2010). S-LCA is the third pillar in LCSA to assess a product's or organisation's social impacts (see Section 4.4.2).

With the three pillars integrated, LCSA is believed to be a comprehensive method that can draw a whole picture of sustainability assessment, and it is becoming increasingly significant (Toosi et al. 2020). However, certain limitations of LCSA are identified and urgently need to be solved. First, compared to the other two techniques, S-LCA is still weak and uncommon, leading to the social pillar being left out of consideration in many studies (Ostermeyer, Wallbaum & Reuter 2013; Toosi et al. 2020; Vilches, Garcia-Martinez & Sanchez-Montañes 2017; Zamagni, Pesonen & Swarr 2013). Second, LCSA models each pillar of sustainability independently, and each pillar's assessment results are combined in a final step of decision

analysis. Even though this assessment mechanism can emphasise the sustainability view for each pillar, it hinders a comprehensive knowledge of the system due to the neglect of considering trade-offs between the three sustainability pillars (Zamagni, Pesonen & Swarr 2013). As a result, ELCA, ELCC and S-LCA do not facilitate the decision-making process unless they are amalgamated into a decision-making support system (Toosi et al. 2020).

4.5.3. Muti-criteria decision making

Decision making is a complex and challenging task in most decision-making processes where multiple, even conflicting, objectives need to be considered. Moreover, uncertainty from different sources, such as subjective opinions from decision makers and changing demands of stakeholders, adds complexity (Mohammad 2021). This is why multi-criteria decision making (MCDM) is preferred over one-dimensional decision making, which can only consider single criterion (Ekel, Pedrycz & Pereira Jr 2020; Geneletti 2019). Moreover, MCDM can deal with subjective components of the decision-making process, which are always ignored by the one-dimensional model (Munier, Hontoria & Jiménez-Sáez 2019).

Munier, Hontoria and Jiménez-Sáez (2019, p. 5) defined MCDM as "a process of selecting one of the several courses of action, alternatives or options, which must simultaneously satisfy many different conflicting and even contradictory criteria". The essence of MCDM is to explore balance or trade-offs between the benefits and drawbacks of different alternatives by illustrating their performance across all criteria (Geneletti 2019). After all, compared to finding the best solution which is almost impossible in reality, reaching a consensus or compromise amongst all the involved parties is more preferred (Munier, Hontoria & Jiménez-Sáez 2019). Furthermore, with the capability to deal with group decisions, MCDM can identify similarities or potential areas of conflicts amongst stakeholders. Eventually, it can contribute to a more comprehensive understanding of the values held by others (Kiker et al. 2005).

Based on formulated decision problems, MCDM is categorised into two types: multi-objective decision making (MODM) and multi-attribute decision making (MADM). Table 4.6 summarises their characteristics and application. The typical steps in MODM and MADM are illustrated in Figure 4.6 (a) and (b), respectively.

		MODM	MADM
Similarity	nilarity • Multiple criteria: each problem has multiple criteria		
		• Conflict criteria: multiple criteria conflict with each other	
		• Incommensurable units: criteria may be measured in different units	
Difference	Type of	A design model for designing the	A selecting model for selecting the
	model	optimal solution	best from a pool of predetermined
			alternatives
	Decision	Decisions are taken via several	Decisions are made based on
	types	objectives in a continuous decision	preferences with discrete decision
		space	spaces
	Alternatives	• No predetermined alternatives are	 Predetermined alternatives are
		needed	needed
		• Infinite alternatives can be dealt with	• Limited, finite alternatives can be
			dealt with
	Mechanism	functions which describe chiestives	A systematic method with which a
	of decision	and constraints of the multi-shipstive	discrete performance rating mechanism
	making	decision machines	the norfermance of these
		decision problems	predetermined alternatives
	Fields of	Mainly in the field of operational	Different fields, but mainly for solving
	application	research and management science	management and evaluation decision-
	application	research and management science	making problems
Benefits		Promotes clear thinking	Copes with conflicting_even
Denemis		 Provides comprehension and insight 	commensurable objectives
		• Explains decision rationale	• Is more relevant and significant for
		• Enables communication and	dealing with real-world decision
		understanding among multiple	problems
		stakeholders	r · · · · · · · · · · ·
		Deals with conflicting, even	
		commensurable objectives	
Limits		Decision making is constrained by inform	mation availability and domain
		knowledge	
		Since infinite alternatives can be	The decision results largely depend on
		considered, it is a constant challenge	the quality of predetermined items
		for decision makers to select the best	
		materials and constructions to satisfy	
		complex design problems.	
Sources		Asmone & Chew 2018; Alinezhad &	Asmone & Chew 2018; Alinezhad &
		Khalili 2019; Brownley 2013; Ekel,	Khalili 2019; Ekel, Pedrycz & Pereira
		Pedrycz & Pereira Jr 2020; Jones &	Jr 2020; Yu, Fei & Li 2018;
		Tamiz 2010; Zavadskas, Turskis &	Zavadskas, Turskis & Kildiene 2014;
		Kildiene 2014	Zolghadr-Asli et al. 2021

Table 4.6. Characteristics of MODM and MADM



Figure 4.6. Basic steps in MODM and MADM

(a) MODM (Source: Brownley 2013)

(b) MADM (Source: Zolghadr-Asli et al. 2021)

Decision making in sustainable projects is complex and sometimes even difficult to control for the inherent trade-offs between the three sustainability dimensions. Clearly understanding and effectively managing the trade-offs is key to achieving comprehensive sustainability (Kiker et al. 2005). Kiker et al. (2005) further stated that sustainable decisions always involve multiple facets with different priorities and objectives. A sustainable decision is typically based on support from diverse areas of expertise with consideration of different perspectives that represent the requirements of different stakeholders (Ekel, Pedrycz & Pereira Jr 2020; Kiker et al. 2005).

By applying MCDM to retrofitting projects to decide on sustainable retrofitting strategies, two approaches are available: maximum performance, where the retrofitting strategy with the best performance is selected by assessing only one indicator or one dimension of sustainability thus weighting is not necessary; and optimal performance, where the retrofitting strategy with the highest overall performance across a number of indicators or dimensions of sustainability is selected. Thus, weighting needs to be considered (Ostermeyer, Wallbaum & Reuter 2013).

As previously stated, sustainable development requires integration and balanced environmental, economic and social development. In addition, for retrofitting projects, the retrofitting options for the site might be measured against each other instead of reaching hard targets, which allows achieving optimal solutions within project constraints instead of reaching a certain goal (Ostermeyer, Wallbaum & Reuter 2013), such as achieving a certain level of environmental performance improvement regardless of cost. Therefore, the optimal solution (or saying as good as possible) is more desired for retrofitting projects (Ostermeyer, Wallbaum & Reuter 2013). Due to the nature of retrofitting projects, the possibilities for the site should be compared to one another rather than to strict goals. The optimisation model is more suitable for generating retrofitting strategies since it can provide the best overall and balanced performance regarding the three sustainability pillars. Therefore, compared to MADM, MODM is a more suitable decision-making methodology for this study for the ability to generate an optimal solution within project constraints. In addition, it enables dealing with conflicting, even non-commensurable issues by creating a vector based on the assessment criteria relating to the decision, objective functions, and problem constraints (Asmone & Chew 2018).

4.6. Summary

This chapter discussed common assessment methods in each sustainability dimension regarding the assessment framework and their benefits and limitations. LCA and green building rating systems are two common methods to assess environmental impact. For the economic dimension, LCC and CBA are widely used to evaluate income and costs throughout a product's life cycle. SCBA and S-LCA are two methods that can be adopted to assess social impact. Compared to assessment considering only one sustainability dimension, methods that can include two or more dimensions are more able to achieve comprehensive sustainability. The

two-pillar model refers to assessment frameworks integrating environmental and economic dimensions to assess sustainability performance and support decision making.

Based on the requirement of sustainable development, the environmental, economic and social dimensions should be integrated in a balanced way. LCSA is an assessment framework that can include the three dimensions to support decision making. In LCSA, ELCA is used to assess environmental impact; ELCC is used to assess life cycle costs; and S-LCA is used to assess social impact. The assessment results of these three assessment techniques are then aggregated by integrating them in a decision-making process. Therefore, even though the three sustainability pillars are covered in LCSA, their mutual relations are neglected in this method. To solve this problem, MCDM is suggested to solve complicated problems with multiple, even conflicting objectives. More importantly, MCDM is an effective method to analyse and balance trade-offs between different assessment criteria. MODM and MADM are two categories under MCDM. MODM can generate optimal solutions within project constraints, while MADM is usually used to select the "best" solution from a pool of predetermined alternatives. Due to the nature of retrofitting projects, the possibilities for the site should be compared to one another rather than to strict goals. The optimisation model is more suitable for generating retrofitting strategies since it can provide the best overall and balanced performance regarding the three sustainability pillars. Therefore, MODM is adopted in this study to build the conceptual model for deciding on retrofitting strategies.

In summary, Chapters 2 to 4 present the literature review about sustainability, retrofitting, and assessment tools of sustainability. Based on the review, a conceptual model for deciding retrofitting strategies for office buildings is developed. The details of the conceptual model are presented in Chapter 6. The next chapter illustrates the research methodology on how the model is developed and how this study is conducted.

Chapter 5. Research methodology

5.1. Introduction

In the previous chapters, existing studies about sustainable development, sustainable retrofitting, and sustainability assessment methods were reviewed. Based on the literature review, it can be concluded that existing buildings are imposing massive negative impacts on the natural environment. Retrofitting is recognised as an effective strategy to alleviate those negative impacts. To attain comprehensive sustainable development, the economic and social dimensions should be integrated with the environmental dimension, and these three dimensions need to be balanced. Therefore, a complete assessment of sustainability needs to be conducted, in which the life cycle impacts in the three dimensions are integrated to evaluate the performance of potential retrofitting activities. Based on the literature review, this study aims to develop a model for deciding retrofitting strategies for office buildings.

Before presenting the details about how the conceptual model is developed, this chapter first gives an overview of how the research is conducted – the research methodology. Common types of epistemology, theoretical perspectives, methodology, and data collection methods are introduced. Based on discussions of benefits, limits and applicability, suitable ones are selected for this study.

In the beginning of the research, a research proposition is stated based on the generally accepted theory that humans contribute to climate change and the corollary that we can reduce the pressure on the climate by changing the way we interact with the environment. The proposition investigated here is that it is possible to improve sustainability, which could potentially be sufficient to create room for economic growth and improved social wellbeing, by retrofitting existing buildings; and that this can be done by using the triple-bottom line approach. A process from conceptualisation to operationalisation is conducted to test the proposition, and a positivist data collection strategy is employed to realise this process. First, a conceptual model, a fuzzy expression of the research proposition, is developed. This conceptual model can then be converted into an operating model by specifying retrofitting activities and assessment criteria. A case study is conducted to demonstrate how this can be done. In this way, the conceptual model is verified, and the process of using the conceptual model to develop

retrofitting strategies can also be illustrated. As a result, the research proposition can be tested, indicating whether the research is valid.

5.2. The research process

Crotty (1998) described the research process as "scaffolded learning", which can establish structured frameworks for the long-term purpose of researching or learning. The research process consists of four components: epistemology, theoretical perspective, methodology and research methods (Figure 5.1). Combining the explanations from Crotty (1998) and Scotland (2012), the meaning of these four components can be described as follows: epistemology is considered a set of philosophical assumptions about the nature of knowledge. The theoretical perspective is the philosophical stance behind the adopted methodology, which provides theoretical support for the process and grounding its logic and criteria. The methodology is the strategy or plan to rationalise the held research assumptions and link the research actions to the desired outcomes. Research methods refer to a series of techniques or procedures used to collect and analyse the research process. The relationship among these four elements is illustrated in Figure 5.1 and Table 5.1.



Figure 5.1. Basic elements of research process

Source: Crotty 1998

Epistemology	Theoretical perspective	Methodology	Method
 Objectivism Constructionism Subjectivism 	 Positivism (and postpositivism) Non-positivism Interpretivism Symbolic interactionism Phenomenology Hermeneutics Naturalistic inquiry Interpretive communities Critical theory/Critical race theory Postmodernism Feminism 	 Experimental research Survey research Ethnography Phenomenological research Grounded theory Heuristic inquiry Action research Correlation analysis Narrative enquiry Cause-comparative study Discourse analysis Etc. 	 Sampling Measurement and scaling Questionnaire Observation Participant Non-participant Interview Focus group discussion Case study Content analysis Documents Audio-visual materials Statistical analysis Etc.

Table 5.1. Relationship among the four elements of the research process

Source: Crotty 1998

5.2.1. Epistemology and ontology

Epistemology is about how people understand things and what people can consider acceptable knowledge in a discipline (Walliman 2016). It is inherent in the theoretical perspective, and the methodology researchers have applied. Maynard and June (1994, p. 10) further explained, "Epistemology is concerned with providing a philosophical grounding for deciding what kinds of knowledge are possible and how we can ensure that they are both adequate and legitimate".

Often merging with epistemology, ontology is the study of being. If epistemology deals with how we know what we know, ontology is concerned with "what is" (Crotty 1998). Therefore, ontology is described as the theory of social entities, which is about what exists to be investigated (Walliman 2016). In the research literature, ontology and epistemology are generally discussed together to inform the theoretical perspectives. When discussing the construction of meaningful reality, we cannot avoid talking about the construction of meaning. Nowadays, the term "ontology" is seldom mentioned unless the matter of "being" (radical ontology) needs to be strictly dealt with (Crotty 1998).

There are three main stances to epistemology: objectivism, subjectivism and constructionism (Crotty 1998; Gray 2014).

5.2.1.1. Objectivism

Objectivism takes meanings, and therefore meaningful reality, as an existence that is independent of the operation of any consciousness. In the view of objectivism, the objective truth can always be identified if the researcher goes about it in the right way, regardless of who the researcher is (Crotty 1998; Walliman 2016).

5.2.1.2. Subjectivism

Subjectivism still believes in an external and measurable social reality but emphasises the mediation by people's perceptions which gives meaning to the reality (Curtis & Curtis 2011). Crotty (1998) explains that there is an object (reality) in the view of subjectivism, but it contributes nothing to the generation of meaning.

5.2.1.3. Constructionism

Constructionism refuses to take social reality as an independent phenomenon. Instead, it believes that meaning cannot be discovered but only constructed. It states that meaning can be built in many ways based on different modes of consciousness, and it comes out of an interaction between object and subject to which it is ascribed (Crotty 1998; Curtis & Curtis 2011; Walliman 2016).

The three stances of epistemology help researchers express how they understand what they know or how people look at the world. In this study, the stance of objectivism is taken to reach the research aim: to develop a model for deciding retrofitting strategies to improve the sustainability of existing buildings. Objectivism is chosen because the reality that the sustainable performance of existing buildings can be enhanced by carrying out suitable retrofitting strategies is not affected by who the researcher is or who the people involved in the process are. Moreover, developing suitable retrofitting strategies using the developed model is based on numerical data and calculation, which can be considered objective information.

5.2.2. Theoretical perspectives

The theoretical perspective is concerned with the philosophical attitude embedded in the chosen methodology. To ensure which theoretical perspective should be picked for research, it needs to be clear how the theoretical perspective provides a context for the whole research process and how it works to ground its logic and criteria. Inevitably, the researcher needs to clarify the philosophical assumptions that are stated and brought to the chosen methodology (Crotty 1998). According to the study by Burrell and Morgan (1979), there are four assumptions for social research: (1) ontological assumption, about the nature of reality; (2) epistemological assumption, about the nature of the specific form of knowledge; (3) agency assumption, about the nature of human beings; and (4) methodological assumption, about the process of accessing the knowledge. These assumptions reflect the specific stance or worldview when choosing the particular research methodology and methods. Two main types of theoretical perspectives are highlighted: positivism and non-positivism (Creswell & Poth 2018; Crotty 1998; Gray 2014; Walliman 2016). It is noteworthy here, at the level of theoretical perspectives, the expression of positivism and non-positivism is not a matter of quantitative against qualitative research methods. The quantitative methods can still be adopted in nonpositivism research, and vice versa (Crotty 1998).

5.2.2.1. Positivism and Post-positivism

Positivists are those who take a scientific approach to research (Creswell & Poth 2018). It is an objective approach with the elements of being logical and reductionistic and emphasising an empirical process of data collection. Eventually, the relationship between causes and effects can be established (Creswell & Poth 2018; Walliman 2016). It is a process of accumulating facts about the world, and the generated causes and effects are regarded as a scientific theory (Gray 2014). Positivism has the greatest confidence in science since positivists believe that scientific knowledge is certain and accurate. Scientific knowledge is sharply opposed to the information from opinions, feelings, beliefs and assumptions gained by non-scientific approaches (Crotty 1998). For this reason, positivism ignores issues like power, subjectivity, and cultural relativism etc., because the reliability and validity of their data are hardly tested judging by a positivistic measure (Curtis & Curtis 2011). The theoretical perspective of positivism can be adopted in various kinds of studies, such as grounded theory, phenomenology

and other analysis strategies (Creswell & Poth 2018). In general, positivism holds three arguments (Gray 2014):

- What can be verified by our sensations or calculation constitutes the reality.
- Scientific observation (as opposed to philosophical speculation) is the primary inquiry method.
- Both natural and human sciences share common logical and methodological principles that deal with facts and not with values.

However, it is noted that there is a difference between what positivists are supposed to do and what actually has been done because many so-called "facts" that serve as the components of scientific theories are not directly observed at all (Crotty 1998). As a result, uncertainty may be caused by holding a positivism theoretical perspective. For this reason, instead of passively noting theories found in nature, post-positivism claims to actively construct scientific knowledge (Creswell & Poth 2018; Crotty 1998).

5.2.2.2. Non-positivism – Interpretivism

There are two main anti-positivist stances: interpretivism and pragmatism. Interpretivism is a major anti-positivist stance which recognises the significance of subjective meanings in social reality (Walliman 2016). It is a process of looking for the culturally traced and historically situated interpretations of the world (Crotty 1998; Gray 2014). Gray (2014) further elaborates on this by saying that, in the view of interpretivism, the direct relationship between ourselves (subjects) and the world (object) does not exist, and the world can only be interpreted through classification schemas existing only in our minds. There are four common interpretive approaches: symbolic interactionism, phenomenology, hermeneutics, and naturalistic inquiry. The epistemology can be closely linked to constructionism (Gray 2014). If the process, however, is recognised to be influenced by researchers, and the aim of the research is to understand the deep structure of the phenomenon or to understand the phenomenon via the meaning that the insiders assign to it, interpretivism is more appropriate than subjectivism (Burrell & Morgan 1979; Crotty 1998). There are a few additional non-positivism stances, including critical theory/critical race theory, postmodernism, and feminism (Crotty 1998; Gray 2014). All these stances can be categorised in the field of interpretivism because this research

attempts to interpret communities' stances, which is why Creswell and Poth (2018) named this type of interpretivism interpretive communities.

5.2.2.3. Non-positivism – Pragmatism

Pragmatism holds a kind of worldview that focuses on the research outcomes and solutions to problems rather than antecedent conditions (Creswell & Poth 2018). Therefore, there is no particular type of method assigned. Methods are suitable and effective as long as their application "works". The direction for the basic ideas can be summarised as (Cherryholmes 1992; Murphy 1990):

- Pragmatism is not committed to any system of philosophy and reality.
- Individual researchers have the freedom to choose any research procedures, methods and techniques as they best meet their requirements and purposes.
- Pragmatists do not see the world in a purely quantitative or qualitative way. Both quantitative and qualitative methods can be used to collect and analyse data.
- Pragmatists agree that research occurs from time to time in social, historical, political and other contexts.
- Pragmatists look for "what" and "how" to research based on the pursued research outcomes.

In practice, multiple data collection methods (quantitative and qualitative) are employed to best answer research questions. Again, pragmatism emphasises the importance of conducting research to best address the research problems (Creswell & Poth 2018).

In this study, the theoretical perspective of positivism is held to develop the conceptual model because the whole process of model development is objective. It deals with the fact that conducting suitable retrofitting strategies can improve the sustainable performance of existing office buildings. It will not change with the different opinions of different people.

5.2.3. Research methodology

Research methodology refers to tasks, strategies and criteria that shape the scientific inquiry, including all aspects of a research process (Gerring 2011). As discussed above, there are

different theoretical perspectives; thus, various methodologies are needed to deal with different kinds of problems and seek different types of answers. Quantitative and qualitative methodologies are the two major categories implemented in social science (Cresswell 2012; Fellows & Liu 2015; Taylor, Bogdan & DeVault 2015). Quantitative methodologies put considerable trust in numerical data; in contrast, qualitative methodologies attempt to use words and observations to describe reality (Amaratunga et al. 2002). Even though there is a clear difference between these two methodologies, qualitative and quantitative methods are not mutually exclusive, and both can be used jointly to address research problems (Hughes 2012; Taylor, Bogdan & DeVault 2015).

5.2.3.1. Quantitative methodology

A quantitative methodology is regarded as an objective research approach. It is used to study the current circumstances of people and events with concern for amounts and frequencies (Thomas 2003). Quantitative studies usually collect data as numbers that can be used to build a systematically organised set of materials for either testing or generating hypotheses (Olsen 2012). This study discusses four types of quantitative methods: survey, correlation analysis, causal-comparative study, and experiment (Creswell 2003; Mertler 2018; Thomas 2003). These four methods can generate numerical results that allow researchers to draw a precise difference between members of a group and between groups as units (Thomas 2003). The advantages and limitations of these four methods are listed in Table 5.2.

Type of method	Description	Advantages	Limitations
Correlation analysis	Description Survey methods involve gathering information about the current status of some target variables within a particular collectivity and then summarising the findings. Correlation studies are designed to answer the general question: What happens to one variable when another variable changes? Descriptions of the relationship between variables can range in precision from very general verbal observations to highly specific	 Advantages Surveys are most helpful in revealing the current status of a target variable within a particular entity. The accuracy of description is enhanced if the status of variables is cast in numerical form than if the results are reported using imprecise verbal expressions such as many, a few, or significantly more. Using statistical techniques to calculate the degree of relationship between phenomena can provide more precise information than estimates of relationships cast in phrases such as "not much of a connection among". 	It fails to show the unique way that the target variable fits into the pattern of the individual units within the collectivity.
Causal- comparative study	Causal- comparative study is for exploring the reasons behind existing differences between two or more groups.	It is an effective alternative to experimental designs, particularly when the independent/grouping variables cannot or should not be manipulated.	Because the cause under investigation has already occurred, the researcher has no control over it, which is incredibly limiting when researchers are trying to conclude cause-and-effect relationships.
Experimental research	An experiment consists of treating objects in a defined way and then evaluating the outcome to determine how the treatment influenced the objects and why the treatment had such an effect.	Experiments provide information about the apparent causes of changes in characteristics.	The requirements for designing and conducting true experimental studies are extremely stringent and, in some cases, prohibitive. Researchers must also go to great lengths to ensure that their designs, data and conclusions are not subject to a variety of threats to validity.

Table 5.2. Summary of advantages and limitations of four quantitative methods

Source: Adapted from Creswell 2003; Mertler 2018

Quantitative research methodology is an effective way to gain information about selected features of either a population's members or the whole population. The numerical data collected is applicable for comparing and summarising trends of the regular phenomenon, which can be used to generate theory or formula. However, they are not effective for drawing the patterning of characteristics of the lives of individuals, groups or institutions.

5.2.3.2. Qualitative methodology

Qualitative research is usually described as a large umbrella covering a broad range of techniques and philosophies (Hennink, Hutter & Bailey 2010). It is normally used to examine people's experiences in detail or to generate a theoretical understanding of social phenomena (Hennink, Hutter & Bailey 2010; Kanazawa 2017). The most distinctive feature of qualitative research is that it enables researchers to identify phenomena using the understanding, experience and knowledge gained from their study (Hennink, Hutter & Bailey 2010). Common research methods of this methodology are interviews, observation, life histories, focus group discussions, visual methods and content analysis. Table 5.3 illustrates the characteristics of qualitative research methods. According to Table 5.3, qualitative methodology is used to identify the ontological position that researchers established for interpreting their study and the way they understand the social world. Thus, the quality of research heavily depends on researchers' professionalism (Mason 2018).

Qualitative research	Qualitative researchers
 takes place in the natural world 	• view the social world holistically
uses multiple methods	 systematically reflect on who they are
focuses on context	 are sensitive to personal biography
• is emergent rather than tightly prefigured	use complex reasoning
 is fundamentally interpretive 	conduct systematic inquiry

Table 5.3. Characteristics of qualitative research and researchers

Source: Rossman & Rallis 2011

There are five common interpretive methods of qualitative methodology: case study, ethnography, grounded theory, phenomenology, and narrative enquiry. The content, advantages and limitations of each are summarised in Table 5.4.
Type of method	Description	Advantages	Limitations	
Case study	A case study	It permits a researcher to	The generalisations or principles	
	typically consists of a	reveal the way a	drawn from one case or multiple	
	description of an	multiplicity of factors has	cases can be applied to other	
	entity and its action	interacted to produce the	cases only at considerable risk of	
	and offers	unique character of the	error.	
	explanations of why	entity that is the subject of		
	the entity acts as it	the research.		
	does.			
Ethnography	Ethnography is a	It can serve several	Researchers cannot expect	
	special kind of case	purposes, including	ethnography to portray the	
	study in which the	revealing characteristics	"objective truth" about a group or	
	researcher	shared among members of	organisation.	
	participates in the	a group – characteristics	-	
	activities of the	that render the group's	Conclusions drawn from the	
	people, organisation,	culture distinctive, also	ethnographic study of one group	
	or event being	exposing the internal	can be applied to other groups	
	investigated over a	operations of a group or	only at considerable risk because	
	period of time. It is	organisation.	of the unique conditions that may	
	the chief method used		determine the pattern of life in	
	by cultural		each setting.	
	anthropologists.			
Grounded theory	Grounded theory is a	It provides systematic	Researchers have to understand it	
	research method that	procedures for shaping and	is difficult for researchers,	
	employs a systematic	handling rich qualitative	especially novices, to build the	
	set of procedures to	materials, and the rigorous	trustworthiness of their ability to	
	develop an	procedures enable	identify when data collection	
	"inductively derived"	researchers to check, refine	should cease. Otherwise, misuse	
	grounded theory	and develop their ideas and	or abuse might occur.	
	about a particular	intuitions about the data.		
	phenomenon.			
Phenomenological	Phenomenology	It is a very appropriate	It is challenging to determine	
research	seeks to understand	methodology to facilitate a	sample size. A small sample size	
	and make explicit the	meaningful understanding	may lead to less reliability, but a	
	subjective	of the lived experience. In	large sample size may cause a	
	interpretation of the	other words, it is the most	common misunderstanding that	
	essence of human	appropriate method to	the results should be statistically	
	experience.	study human activity.	reliable.	
			Gathering data and data analysis	
			may be time-consuming and	
			laborious	
Narrative	Narrative research is	It has the potential to	Narrative research is not an	
research	stories about	demonstrate both the	effective device for revealing	
	influential incidents	uniqueness of individuals'	how characteristics are	
	in a person's own	lives and the similarity	distributed throughout a	
	life.	among lives that are lived	population.	
		under different	I - I	
		circumstances.		

Table 5.4. Summary of advantages and limitations of qualitative methods

Source: Charmaz & Belgrave 2007; Lai & To 2015; Lester 1999; Mason 2017; Sander 1982; Thomas 2003

A qualitative methodology can flexibly gather various types of information to analyse some issues without concern for quantities, but the research process cannot be easily copied from

one case to another. In addition, the research results may become vulnerable due to the dependence on the quality of researchers.

This research aims to develop a model for deciding retrofitting strategies for office buildings. The model needs to be applied based on the evaluation of potential retrofitting activities' performance. Therefore, what process and criteria of assessment will be applied need to be clarified. It is also necessary to decide the theory foundation as the base to establish the decision model and how to verify the developed model to make sure it can work in practice. To cover the variety of data and study the issue in both broad and deep dimensions, the "within" mixed method is adopted in this study to determine assessment criteria and verify the developed model. Three data collection methods are chosen in this study: questionnaire survey, focus group discussion, and case study. The following sections explain why and how these methods are employed.

5.2.4. Data collection methods

The data needed for studying social phenomena is categorised into two types: primary and secondary data (Walliman 2016). Primary data is collected through observing, recording and measuring the activities and opinions of real people, inspecting objects, or experiencing events if the research target is an object or phenomena (Creswell & Creswell 2018; Walliman 2016). Secondary data is that gathered from purely literary works, such as newspapers and journals, and from drawings and photographs, fiction, databases, etc. It is used to understand the background of the study. For most research, this includes the resources explored for the literature review.

According to the traits of collected data, quantitative method and qualitative method are two main types of data collection methods. However, there is a trend to combine these two methods for broader and more reliable data collection, referred to as mixed method or multi-method according to different forms of combination.

5.2.4.1. Quantitative methods and qualitative methods

Quantitative methods and qualitative methods are two main types of data collection processes. Their benefits and limitations are summarised in Table 5.5 and Table 5.6.

Data collection type	Content	Advantages	limitations	
Questionnaire	The word questionnaire is typically used in a very general sense to mean any form (printed or digital) of questions that participants in a survey are asked to answer.	 They enable a researcher to collect a large quantity of data in a relatively short period of time. The researcher does not need to be present at the time the information is provided, and data can be collected from people in distant places if the questionnaires are sent by regular mail or over the internet. 	 Low reliability and validity of the research due to possible low return rate if the researcher is not present to supervise the participants as they complete the questionnaire. Questionnaires rarely allow participants to receive clarification of confusing items, nor do questionnaires offer a convenient way for respondents to elaborate their answers and explain conditions that affect their opinions. 	
Inventories	Inventory means a document on which participants in a research project are asked to report their attitudes or preferences.	The strengths and weaknesses of inventories are the same as those of questionnaires.		
Measurement	Measurement links the theoretical framework of a survey-based study to its data-gathering instrument. Four different levels of measurement are usually distinguished: nominal, ordinal, interval, and ratio.	A higher level of measurement is not necessarily better than a low level of measurement. Standpoints refer to differences of perspective that give interesting angles on the same events or situation. Therefore, the benefits and limitations of measurement cannot be asserted without discussing the context.		

Table 5.5. Summary of advantages and limitations of quantitative data collection methods

Source: Bhattacherjee 2012; Creswell & Creswell 2018; Olsen 2012; Thomas 2003

Collection type	Options within type	Advantages	Limitations
Observations	 Complete participant: researcher conceals role Observer as participants: role of the researcher is known Participant as an observer: observation role secondary to participant role Complete observer: researcher observes without participating 	 Researcher has firsthand experience with participants Researcher can record information as it is revealed Unusual aspects can be noticed during observation Useful in exploring topics that may be uncomfortable for participants to discuss 	 Researcher may be seen as intrusive Researcher cannot report "private" information that may be observed Researcher may not have good attending and observing skills Researcher may find it difficult to gain rapport with certain participants (e.g., children)
Interviews (structured, semi- structured & unstructured interviews)	 Face-to-face: one on one, in-person interview Telephone: researcher interviews by phone Video: researcher interviews by video such as Zoom Group: researcher interviews participants in a group 	 Useful when participants cannot be observed directly Participants can provide historical information Allows researcher "control" over the line of questioning Interviewers can clarify any issues raised by the respondent or ask probing or follow-up questions. 	 Provides indirect information filtered through the views of interviewees Provides information in a designated place rather than the natural field Researcher's presence may bias responses Not all people are equally articulate and perceptive Time-consuming and resource-intensive
Focus group (more structured & less structured)	 Face-to-face: a small group of participants (typically 6–10) at one location Video: a group of participants (6–10) have online meetings such as Zoom 	 Useful to build a holistic understanding of the problem situation based on participants' comments and experiences Suited for exploratory research 	 Internal validity cannot be established due to a lack of controls The findings may not be generalised to other settings because of the small sample size
Documents	 Public documents such as minutes of meetings, and newspapers Private documents such as journals, diaries, and letters Email discussions 	 Can obtain the language and words of participants Can be accessed at a time convenient to the researcher – an unobtrusive source of information Represents data that is thoughtful, as participants have given attention to compiling it Saves researcher the time and expense of transcribing 	 May be protected information unavailable to the public or private access. Requires the researcher to search for the information in hard-to-find places Materials may be incomplete Documents may not be authentic or accurate
Audio-visual materials	 Photographs Videotapes Art objects Computer software Film 	 May be unobtrusive method Allows participants to directly share their "reality" May be creative as it captures attention visually 	 May be difficult to interpret May not be accessible publicly or privately Presence of an observer (e.g., photographer) may be disruptive and affect responses

Table 5.6. Summary of advantages and limitations of qualitative data collection methods

Source: Bhattacherjee 2012; Creswell & Creswell 2018; Thomas 2003

5.2.4.2. Mixed method and multi-method strategies

Some researchers have strongly argued that qualitative and quantitative research should be regarded as complementary rather than opposites. The key differences between qualitative and quantitative research are listed in Table 5.7. Amaratunga et al. (2002) declared that the probability of failing to explore all components needed in a study is high if using only a single data collection method. Mason (2018) supported this point by stating that most researchers would apply a mixed method at some point to explore different aspects of the investigation. A combined qualitative and quantitative method for studying the same research problem was defined by Denzin (1978) as triangulation.

	Qualitative research	Quantitative research	
Objective	To gain a detailed understanding of underlying reasons, beliefs, motivations	To quantify data and extrapolate results to a broader population	
Purpose	To understand why? How? What is the process? What are the influences or contexts?	To measure, count and quantify a problem. How much? How often? What proportion? Relationships in data.	
Epistemological positions	Constructivist	Objectivist	
Relationship between researcher and subject	Close/insider	Distant/outsider	
Research focus	Meanings	Facts	
The nature of data	Data based on textual materials	Data based on numerical materials	
Study population	A small number of participants or interviewees are selected purposively (non-randomly)	A large sample size of representative cases	
	Referred to as participants or interviewees	subjects	
Data collection methods	In-depth interviews, observation, group discussions, etc.	Population surveys, questionnaires, exit interviews, etc.	
Analysis	Analysis is interpretive	Analysis is statistical	
Outcomes	To develop an initial understanding, to identify and explain behaviour, beliefs or actions	To identify prevalence, averages and patterns in data. To generalise to a broader population.	

Table 5.7. Key differences between qualitative and quantitative research

Source: Adapted from Gray 2014; Hennink, Hutter & Bailey 2010

With the development of both qualitative and quantitative research in social and human science, the application of data collection in both forms is expanding, and the desirability of mixed methods has been increasing in recent decades (Creswell 2003). Two types of triangulation (defined on page 128) are introduced: between-method and within-method (Singleton Jr & Straits 2018). "Between-method", also called "cross-method", is the most popular. Researchers use multiple methods to study the same dimension of one phenomenon; then, comparative data is yielded. The benefit is the certainty and reliability of the research results is increased if all results reach the same conclusion (Hennink 2014). The "within-method" uses multiple techniques to collect and interpret data. For example, the researcher can undertake a survey (quantitative method) to develop research using multiple scales or indices and then conduct interviews (qualitative method) to interpret the cause-and-effect behind the emergent theory.

Another term, "multi-method research", is frequently mentioned with mixed methods. It investigates a social phenomenon using the combination of empirical research methods instead of single-shot studies to generate more reliable and accurate research results (Wood et al. 1999). Like mixed method, multi-method research is also at the level of method choice (Brewer & Hunter 2006). The coordination and comparison of different data collection methods (mainly the different observation ways) enrich the collective efforts to research findings. Eventually, it can make a firmer empirical base.

To conclude, quantitative and qualitative methods may be involved in the research process at different times and interact in different ways (Walliman 2016). Nevertheless, the mixed method emphasises the combination of quantitative and qualitative data collection methods in different forms: "between-method" triangulation assists the improvement of external validity while "within-method" triangulation allows cross-checking for internal reliability and conformance. Multi-method also allows the combination of either several qualitative or several quantitative methods.

This research adopts mixed method data collection methods. A questionnaire survey and focus group discussions are conducted in sequence to identify applicable retrofitting activities and generate suitable assessment criteria so that the conceptual model can be converted to an operating model. Then, a case study is conducted to quantify the operating model.

5.3. Research design

5.3.1. Research inquiry approaches

The strategies of inquiry are different types of inquiries within the selected quantitative, qualitative, mixed method or multi-method approaches that shape the direction and procedures of a research study (Creswell & Creswell 2018). In addition, the research design is deeply influenced by the research problem. After all, the purpose of undertaking research is to solve the research problem. All research elements should answer research questions, complete research objectives, and eventually solve the research problem (Creswell & Creswell 2018).

To plan research, the researcher needs to specify the need to employ quantitative or qualitative methodology, which implies the researcher's worldview and applied philosophical assumptions. Some typical paradigms are introduced to illustrate why and how to apply a particular research inquiry approach (Creswell & Creswell 2018):

- The positivist worldview is adopted in quantitative research. Normally, the experimental design is applied, and the attitude of post-test measures is embedded in the research process.
- Qualitative research may hold the constructivist view to observe participants' behaviours, so an ethnographic design is needed. In addition, the stance of interpretive communities can also be found in qualitative research to interpret through individual interviews how participants have experienced oppression. A narrative design may be needed.
- For mixed method research, the pragmatic worldview is held, and both quantitative and qualitative data are sequentially collected for the best understanding and solving of the research problem.

In addition to the epistemology and theoretical perspective, suitable data collection strategies should be designed. In general, data collection strategies can be categorised into two types based on different reasoning approaches: positivist method and interpretive method (Bhattacherjee 2012). Positivist methods employ deductive reasoning to research, beginning with a theory about how things work and deriving testable hypotheses from it. Then, empirical data is collected to test the theoretical hypotheses (Bhattacherjee 2012; Walliman 2016). In contrast, interpretive methods employ inductive reasoning, beginning with data collection and attempting to generate a conclusion about the phenomenon of interest from the collected data

(Bhattacherjee 2012; Walliman 2016). In theory, positivist methods are used to test theory, while interpretive methods are used to build theory, but mostly there is some overlap between the two in practice to achieve a comprehensive data collection.

Considering what has been discussed above, the research design process for this study is illustrated in Figure 5.2. The research question, sub-questions, research aim and research objectives were identified in Chapter 1 and also consolidated in Figure 5.3. This study takes an objectivist stance of epistemology (referring to the discussion in Section 5.2.1). This study holds the theoretical perspective of positivism, even though both quantitative and qualitative data collection methods are employed (referring to the discussion in Section 5.2.2). A numerical calculation can provide the environmental and economic performance of retrofitting activities. As for social performance, a value score is used to express the performance level of retrofitting activities. Therefore, all of them are based on numerical data, and the process can be considered logical and reductionistic.

	888				
Research question	How can we develop retrofitting strategies that can effectively improve a building's environmental, economic, and social performance in a balanced manner?				
Research sub-questions	How can all the three pillars of sustainability be considered for retrofitting?How can the resulting outcomes of retrofitting activities be used to generate suitable retrofitting strategies?How can the performance in the environmental, economic, and social dimensions be improved in a balanced manner?				
Research aim	Develop a model for deciding retrofitting strategies for office buildings to improve their sustainability performance.				
Research objectives	Develop a conceptual model that can be used to develop retrofitting strategies based on the evaluation results of potential retrofitting activities.				
Epistemology	Objectivism				
Theoretical perspective	Positivism				
Theory	Humans contribute to climate change.				
Research proposition	It is possible to improve sustainability while having economic growth and improved social wellbeing by retrofitting existing buildings. This can be done by using the triple-bottom line approach.				
Strategy of inquiry	Conceptualisation Deductive reasoning Operationalisation Deductive reasoning Deductive reasoning				
	Develop a conceptual model Convert the conceptual model to an operating model Quantify the operating model				
	▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲ ▲				
Methodology	Quantitative methodology within mixed method				
Data collection and analysis methods	Document (literature review) Survey Case study Focus group discussions Case study				

Figure 5.2. Process of research design

Chapter 3 discussed that existing office buildings are responsible for massive negative impacts on the natural environment. In addition, regarding the theory that humans contribute to climate change, the research proposition is that it is possible to improve sustainability while potentially allowing economic growth and improved social wellbeing by retrofitting existing buildings using the triple-bottom line approach. In research, the theoretical proposition consists of relationships between abstract constructs, and careful measurement of these constructs is necessary to test the theoretical proposition (Bhattacherjee 2012).

In general, constructs in research can be measured by the processes of conceptualisation and operationalisation (Bhattacherjee 2012). Conceptualisation refers to the mental process by which vague and abstract constructs (concepts) and their constituent parts are defined in concrete and precise terms. Once a theoretical construct, or conceptual model, is defined, the process of operationalisation should be conducted to measure it. Therefore, operationalisation refers to the process of quantification that develops indicators or items to measure the constructs (Bhattacherjee 2012).

This study adopts a positivist data collection strategy (deductive reasoning approach) to test the research proposition. First, a conceptual model for deciding retrofitting strategies using the triple-bottom line approach is developed based on literature review (conceptualisation). Then, the conceptual model is converted to an operational model by specifying retrofitting activities and assessment criteria (operationalisation). The operating model is then quantified via a case study (operationalisation). As a result, the conceptual model considers all the various strategies for retrofitting, and the operating model is a quantified set for a specific location or region. The whole process can test whether the research proposition is correct or not. Then the process from converting the conceptual model to an operating model to quantifying the operating model using a case study can be recognised as a logically and methodologically correct framework for the work done and to be done in the future. By copying the process, the conceptual model can be adapted according to the specific situation of each retrofitting project and used to develop retrofitting strategies.

5.3.2. Research process

Based on the discussed research design, the research process is illustrated in Figure 5.3. The research aim, objectives and questions were identified in Chapter 1. Existing literature relevant to sustainable retrofitting and assessment methods of sustainability was reviewed to understand the development of sustainable retrofitting and identify general assessment criteria in environmental, economic and social dimensions (see Chapters 2 to 4). A conceptual model for deciding retrofitting strategies for office buildings is also developed based on the literature review (see Chapter 6). To verify and demonstrate the conceptual model, it needs to be converted to an operating model first (see Chapter 7). Then, the operating model is applied to the case building to develop the retrofitting strategies that can improve sustainability in environmental, economic and social dimensions (see Chapter 8).





5.3.2.1. Converting the conceptual model to an operating model

A conceptual model is designed to be generic, which means it can be adapted in any country, region or place. Therefore, it has to be modified for local situations before use. To illustrate that the conceptual model can be converted to be operational in reality, China is selected as the place where the model is verified and demonstrated. China is the most populous developing country in the world and has a large number of buildings built in the past 30 years, now due for retrofit. In addition, China has regions with varying climate conditions, making it a good opportunity to test the flexibility of the conceptual model.

An operating model is defined as a bridge between strategies and operational activities (Bateman 2017). It works as a blueprint for a building that can capture the design team's critical decisions that others can conveniently refer to during the transformation (Campbell & Gutierrez 2021). However, the operating model is more dynamic than a blueprint for regular changes (de Vries et al. 2011). The complexity and uncertainty make it challenging to interpret an operating model (de Vries et al. 2011).

Different frameworks are used to define elements that compose an operating model. The framework developed by de Vries et al. (2011) uses people, process and technology to structure an operating model. A new operating model from McKinsey has three elements: structure, processes and people (Álvarez et al. 2022). Campbell and Gutierrez (2021) developed an operating model covering six elements: process, organisation, location, information, suppliers and management. Meanwhile, Campbell and Gutierrez (2021) concluded three steps for transforming strategies into implementation, which are the essence of operating models:

- Step 1. Design changes to current organisation
- Step 2. Transform organisation guided by the design
- Step 3. Lead the transformed organisation

In this study, the operating model is not created from scratch but converted from the conceptual model. Most elements, such as process, people and general information, have been defined in the conceptual model. Therefore, the conversion process is mainly for designing changes to suit the local situation such as the local market, specific structure, activities, climate, and culture. The changes in the model are retrofitting activities and assessment criteria, which

should be modified according to the specific local situation mentioned above. Therefore, these two elements are determined during the conversion process.

To identify suitable retrofitting activities and assessment criteria for the case building, opinions from key stakeholders need to be collected, such as building owners and tenants, and professionals in the construction and property sector, such as engineers, architects, project managers and facility managers. As discussed in Section 5.2.4, different data collection methods are available to gather people's opinions, and suitable one(s) can be selected according to the specific situation of the conducted retrofitting project. In this study, an office building in Hangzhou, China is used to conduct the case study (See Section 7.2 in Chapter 7). Considering the large size of territory and different climate zones in China, the within mixed method is adopted to collect data from the broad to the specific. An explanatory sequential design or two-phase design, as Figure 5.4 shows, is conducted (Creswell & Creswell 2018).





Source: Creswell & Creswell 2018

The quantitative method is used in the first phase. In this study, a questionnaire survey is conducted in northern and southern China to broadly identify applicable retrofitting activities and assessment criteria for retrofitting projects in China (See Section 7.4.1 in Chapter 7). The quantitative results from the first phase typically inform the types of participants in the second phase (Creswell & Creswell 2018). The data from the second phase helps provide more depth and insight into the quantitative results (Creswell & Creswell 2018). Therefore, in this study, the same type of respondents to the survey are also invited as participants in the second phase – the focus group discussions (See Section 7.4.2 in Chapter 7). Focus group discussions can explain outcomes of the questionnaire survey. Moreover, consensus on suitable retrofitting activities and assessment criteria for the case building can also be reached via group discussions. In this way, the conceptual model can be converted into an operating model, which is ready to be applied in the case study.

5.3.2.2. Quantifying the operating model

The last step in demonstrating validity of the conceptual model is to conduct a case study (See Chapter 8). Orum, Feagin and Sjoberg (1991, p. 2) defined a case study as "an in-depth, multifaceted investigation, using qualitative research methods, of a single social phenomenon". Case study method enables the researcher to take a complicated and broad topic, or phenomenon, and condense it into a manageable research question (s). By collecting data from qualitative and/or quantitative datasets, a deeper understanding of the topic or phenomenon can be acquired than using only one type of data (Heale & Twycross 2018). Case study method can be used to both test propositions (positivist method) and build a theory (interpretive method) (Bhattacherjee 2012). In this study, a case study is conducted to test the research proposition. The expected result of the case study is that a preferable retrofitting strategy (within project constraints) can be developed for the case building by considering the three sustainability pillars.

5.3.2.3. Language and ethics

Since the case study is conducted in China, the survey and focus group discussions were conducted in Chinese to communicate with respondents and participants effectively and accurately. The questionnaire form was created in English first, then translated into Chinese. Both English and Chinese versions were sent to professionals who can speak both languages to check whether the translation is accurate. The Chinese questionnaire form was only distributed after the professionals confirmed that the translation did not distort the meaning. For the focus group discussions, recordings were transcribed and analysed in Chinese. Then the analysis results were translated into English to reduce the risk of invalid analysis results due to inaccurate translation. In addition, it can also reduce the workload of translation review as only the analysis results need to be translated.

Ethical issues have to be considered when conducting research. Gajjar (2013) introduced five reasons why ethical consideration is essential for academic research. First, the development of norms is essential for pursuing knowledge and truth, which is the aim of doing research. Second, many different people with different cultures, religions and backgrounds are involved in research. The value generated by ethical standards, such as trust and respect, can help build the bridge of cooperation and coordination. Third, ethical standards can ensure researchers abide by the policy of public responsibility. Fourth, public trust in research can also be built by ethical norms. Finally, many ethical considerations of research can also promote other social values, like health and safety, human rights, and social welfare. According to the Australian Code for the Responsible Conduct of Research (2007, p. 1.3), the below ethical concerns of research should be considered:

- honesty and integrity
- respect for human research participants, animals and the environment
- good stewardship of public resources used to conduct research
- appropriate acknowledgment of the role of others in research
- responsible communication of research results.

Ethical consideration is an indispensable part of this research because different people are involved when conducting the questionnaire survey, focus groups and case study. In this study, respondents of the survey and participants of focus group discussions are kept anonymous. In the analysis, their identity is represented as symbols without allegorical meanings, such as A and B. The collected questionnaire forms and recordings of focus group discussions are encrypted and stored in a computer to which only the researcher and the supervisors have access. Furthermore, all the participants were informed before participating in the research that their identity would not be revealed in any form, and they could quit at any time without giving a reason. All these ethical issues were considered in the application for ethical approval, which

was submitted to the UTS Human Research Ethics Committee. The study received ethics approval (No. UTS HREC ETH18-2810), and data collection was conducted by following the requirements in the ethics approval.

5.4. Summary

This study aims to develop a model for deciding retrofitting strategies for office buildings to answer the research question: how can we develop retrofitting strategies that can effectively improve a building's environmental, economic and social performance in a balanced manner? To achieve the research aim and answer the research question, the epistemological stance of objectivism is taken because the fact that the sustainability performance of existing buildings can be improved by using suitable retrofitting strategies will not be affected by who the researcher is or who is involved in the process. For the theoretical perspective, the stance of positivism is held because developing the model is an objective process. It addresses that retrofitting strategies can improve the sustainability performance of existing office buildings. It will not change with the different opinions of different people.

To justify the validity of the conceptual model, the research is designed as a process of giving a research proposition and testing the proposition via a measured process of conceptualisation and operationalisation. Based on the theory that humans contribute to climate change, the research proposition is that it is possible to improve sustainability while having potential for economic growth and increased social wellbeing by retrofitting existing buildings; and it can be done by using the triple-bottom line approach. Then, the proposition is conceptualised as a conceptual model that can be used to decide retrofitting strategies for office buildings. The next step is to convert the conceptual model to an operating model (operationalisation) by specifying retrofitting activities and assessment criteria to suit the local situation of the case building. A case study is conducted to quantify the operating model. As a result, the validity of the conceptual model can be justified, which reflects that the research proposition is correct.

With a clear understanding of how this research is designed, the next chapter describes the process of developing the conceptual model. The data from the questionnaire survey and focus group discussions is analysed in Chapter 7, which illustrates the process of converting the

conceptual model to the operating model for the case building. Chapter 8 conducts a case study demonstrating how an optimal retrofitting strategy can be developed using the operating model.

Chapter 6. Development of a conceptual model for developing retrofitting strategies for office buildings

6.1. Introduction

Chapters 2 to 4 reviewed existing studies about sustainable development, sustainable retrofitting, and assessment methods of sustainable construction. Based on the literature review, it can be stated that existing buildings are responsible for massive negative impacts on the natural environment, and building retrofitting is regarded as an effective remedy for the poor performance of existing buildings. It is also the last chance for existing buildings to achieve sustainability. Regarding this situation and requirements of sustainable construction, a research proposition is proposed: it is possible to improve sustainability while having economic growth and social wellbeing by retrofitting existing buildings using the triple-bottom line approach.

To test this research proposition, a conceptual model is developed in this chapter, which can consider all three sustainability dimensions when applying it to develop retrofitting strategies for office buildings. First, the general steps of MODM and the requirement of retrofitting projects are reviewed and adapted to develop the conceptual model. Then, details about each step of the conceptual model are elaborated.

6.2. Model development

Based on the research proposition, the conceptual model should be able to develop a retrofitting strategy that addresses and balances environmental, economic and social concerns complying with the principles of the triple-bottom line approach. However, the three pillars of sustainability often compete with one another. For example, the improved performance of an existing building by a retrofitting activity, such as improving indoor comfort performance by installing a powerful HVAC system, usually means high initial cost and/or unfavourable social impacts from the installation. Apart from the three sustainability pillars, the developed retrofitting strategy should also be able to satisfy project constraints. Regarding multiple conflicting issues that need to be dealt with, as discussed in Section 4.5.3 of Chapter 4, MODM is a suitable decision-making methodology for this study. MODM enables dealing with

conflicting, even non-commensurable issues by creating a vector based on the assessment criteria relating to the decision, objective functions, and problem constraints (Asmone & Chew 2018).

As discussed in Chapter 3, retrofitting is more complex than new construction. A study by Ma et al. (2012) revealed the general process of building retrofitting by considering the ability to reduce energy consumption and improve cost efficiency of existing buildings by retrofitting. The whole retrofit process is divided into three stages: pre-retrofit stage, during retrofit stage, and post-retrofit stage. The pre-retrofit stage is mainly about selecting and amalgamating activities as retrofitting plans shown in Figure 6.1. The during retrofit stage is about implementing and commissioning the retrofitting strategy. The post-retrofit stage is for measuring and verifying results of the determined retrofitting strategy.



Figure 6.1. Retrofit process at pre-retrofit stage

Source: Adapted from Ma et al. 2012

Based on Figure 6.1, the process at the pre-retrofit stage mainly consists of two parts: identifying problems of the existing building to decide whether retrofitting or demolishing is a feasible solution, and designing a retrofitting plan if it is decided the building is to be retrofitted. Therefore, the part labelled identifying problems in Figure 6.1 is simplified into two steps in the conceptual model: set retrofitting goals and objectives, and identify problems of the retrofitted building.

The developed retrofitting plan in Figure 6.1 is to reduce the energy consumption of the existing building with consideration of cost efficiency. Therefore, energy efficiency is the main

objective. However, the conceptual model developed in this study considers all the environmental, economic and social dimensions of sustainable development. Different retrofitting activities may be proposed to meet different retrofitting objectives, such as improving energy efficiency, improving water use efficiency, improving indoor environmental quality, decreasing operation cost, and/or improving accessibility to building facilities and services. As the general MODM model indicated in Figure 4.6 (a) in Chapter 4, the performance of an individual retrofitting activity is assessed based on identified assessment criteria. However, whether an activity can be selected for developing a retrofitting strategy is not purely determined by the evaluation results but also how well the amalgamation of different retrofitting activities can satisfy project constraints. Therefore, compared to the general MODM model – develop retrofitting strategies that can meet retrofitting objectives and satisfy all project constraints at the same time.

By considering elements in MODM and the required steps for generating retrofitting strategies, a conceptual model for deciding retrofitting strategies for office buildings is developed, as Figure 6.2 shows. The conceptual model represents a relatively linear process of decision making but with several feedback loops. In fact, the methodology of MODM is an iterative process that allows decision makers to add new information and refine decisions (Brownley 2013; Geneletti 2019). Each step contained in the conceptual model is discussed in great detail in the following sections.



Figure 6.2. The conceptual model of deciding retrofitting strategies for office buildings

6.3. Set retrofitting goals and objectives

The first step of the process is to set retrofitting goals and objectives. Retrofitting goals are thresholds regarding different criteria for measuring whether alternatives achieve or not, while objectives indicate the desired direction of achievement (Ekel, Pedrycz & Pereira Jr 2020). For example, if a retrofitting objective is set to improve energy efficiency, the associated

retrofitting goal can be to achieve a 30% energy saving. Setting retrofitting goals and objectives requires the retrofitting team to gather different perspectives and opinions on the decision situation, such as the requirements and needs of clients (building owners or investors) via consultation, and the demands of users, such as tenants and service staff, via interviews or surveys.

In addition, project constraints such as the financial, political and external also need to be identified at the early stage of the decision-making process (Nijkamp, Rietveld & Voogd 1990). Financial constraints are related to factors that restrict the amount or quality of project development, such as the budget of projects. Political constraints refer to regulations and policies that affect the direction of development. External constraints should also be considered, which are influenced by the development impacts on the community and the natural environment, reflecting the environmental and social requirements of sustainability. Investigation of project constraints at the early stage of a project is crucial to identify precise alternatives contributing to minimising the gap between the generated solution and the desired outcome (Nijkamp, Rietveld & Voogd 1990).

With the above information in mind, a retrofitting team can identify objectives and structure an initial, general frame of evaluation, including priority ranking of objectives, assessment criteria, and associated assessment methods to quantify performance of retrofitting activities. Identifying objectives is one of the most challenging decision-making processes since it needs constructive thinking and creation based on raw information. Success in this step significantly contributes to optimising alternative solutions (Brownley 2013).

6.4. Identify problems of the proposed building

The second step involves identifying the problems of the proposed building. Usually, a walkthrough is undertaken first for a preliminary and rough check of where the problems are. Then, a building audit is conducted to diagnose the problems of the building and provide decision makers with a whole picture of the situation of the proposed building. The retrofitting scope can also be defined based on the audit result. There is no doubt that a problem can have impacts on different sustainability dimensions. For example, a lack of building management control system may lead to high energy consumption and high operation cost (environment and economic dimensions), and tenants do not have a building where the indoor temperature, humidity, and dimming light level cannot automatically adjust with the change of outside climate (social dimension) Another example is that the existing HVAC system is outdated, and its components are not easily available. Therefore, it may be difficult and expensive to get replacements if the components break, leading to high maintenance cost and poor maintainability. Based on the literature review, Table 6.1 summarises the primary problems of existing office buildings on environmental, economic and social dimensions.

	1 0	e
Environmental problems	Economic problems	Social problems
 High energy consumption 	 High operation cost 	 Some building services are not
 Massive greenhouse gas 	 High maintenance cost 	provided, such as a lack of
emissions	 High vacancy rate due to poor 	building management control
• Use of materials with a short life	performance	system which can automatically
span		adjust indoor temperature,
 High water consumption 		humidity, dimming level with
 Excessive waste generation 		changes of outside climate
 Poor indoor environmental 		 Lack of building facilities for
quality		people with additional needs,
		including disabled people,
		pregnant women, the elderly,
		etc.
		 Safety and security issues
		 Difficulty in cleaning and
		maintaining the building

Table 6.1. Common problems with existing office buildings

Source: Ascione et al. 2017; Cellura et al. 2013; Dolezal & Spitzbart-Glasl 2015; Dwaikat & Ali 2016; Kylili et al. 2016; Liang et al. 2018; Oregi et al. 2017; Wong & Zhou 2015; Wu et al. 2014

By understanding the problems the existing building has, building owners or investors may get a better understanding of why the building performs as it does. Meanwhile, based on audit results, they may modify retrofitting goals and objectives set in the first step by adding objectives or emphasising the expectation of improvement on a specific aspect. Then the decision-making process will restart from the beginning like the loop shown between the first two steps in Figure 6.2.

6.5. Propose retrofitting activities

Regarding each identified problem from the last step, relevant retrofitting activities that may solve those problems can be proposed based on the expertise of the retrofitting team in the problem area. In addition, identifying retrofitting activities also significantly depends on the cost and availability of information (Ekel, Pedrycz & Pereira Jr 2020). Table 3.1 in Chapter 3 lists 15 common retrofitting activities for office buildings based on the literature review. Based on building components where retrofitting activities are implemented, these common retrofitting activities can be categorised into two groups, as Figure 6.3 shows. Figure 6.3 can be used as a checklist for selecting suitable retrofitting activities for proposed projects. When using this conceptual model on a real case, the retrofitting team can substitute, expand, delete or modify these retrofitting activities to suit the specific situation of the case.



Figure 6.3. Common retrofitting activities for office buildings

How and when to stop identifying retrofitting activities is a crucial issue. Based on the study by Ekel, Pedrycz and Pereira Jr (2020), the number of retrofitting activities is defined in terms of identified objectives and decision problems. In this conceptual model, to make sure the developed retrofitting strategy is the optimal solution regarding existing project constraints, all the potential activities should be identified first. If the set of suitable alternatives for solving one problem is too large, a screening process may be adopted to reduce the number, making them viable and manageable. The screening process prioritises proposed activities based on key elements of the project, such as cost, availability of local materials, construction technologies, etc. In other words, the screening process identifies the scope of impact assessment, which controls the cost and time of the assessment (Dougherty & Hall 1995).

6.6. Develop assessment criteria

This step is related to developing assessment criteria for evaluating the performance of proposed retrofitting activities. Criteria are desired attributes of solutions to decision problems, and alternatives are measured against criteria to see how well they solve the decision problems (Jones & Tamiz 2010). Correspondingly, assessment criteria for green buildings reflect aspects of benefits that a green building should have or produce.

Assessment criteria can be identified via four different processes according to different traits of projects. The first method is to identify assessment criteria based on retrofitting goals and objectives when they are the only concerns of the project. The second one is to identify assessment criteria through the retrofitting team that has a deep understanding of the situation of the project. Then, they identify assessment criteria based on their expertise and with the consideration of the needs of clients, requirements of different stakeholders, and project constraints. The third method is to identify assessment criteria purely based on a literature review. Table 6.2 summarises environmental, economic and social assessment criteria that are suitable for assessing the performance of existing buildings based on the literature review. Similar to Figure 6.3, Table 6.2 can also be modified according to the specific situation of the retrofitting project, which makes the conceptual model operational. The last method is to have public participants joining in the process so that the external influence of the project can also

be identified, such as the impact of construction on people randomly visiting the building, passing by the building, and living or working in the neighbouring buildings.

	Life stage		References	
	Retrofitting	E1. Energy	E1-1. Embodied energy of materials	Luzkendorf &
	stage	efficiency	E1-2. Energy consumption by retrofitting	Lorenz 2005;
			activities	ISO 2006;
		E2. Carbon	E2-1. Embodied carbon emissions of materials	Kylili,
		emissions	E2-2. Carbon emissions of retrofitting activities	Fokaides &
		E3. Material use	E3-1. Use reusable/recyclable materials and	Jimenez 2016;
			components	Mahmoud,
		E4. Waste	E4-1. Waste generation from retrofitting activities	Zayed &
		generation		Fahmy 2019
al	Operation	E1. Energy	E1-3. Embodied energy from materials for	
ent	stage	efficiency	maintenance, repair, and replacement	
m			E1-4. Energy consumption of operation activities	
LOI		E2. Carbon	E2-3. Embodied carbon emissions from materials	
nvi		emissions	for maintenance, repair, and replacement	
E			E2-4. Carbon emissions of operation activities	
		E3. Material use	E3-2. Use reusable/recyclable materials and	
			components	
		E4. Waste	E4-2. Waste generation from maintenance, repair,	
		generation	and replacement	
		E5. Indoor	E5-1. Indoor thermal comfort	
		environmental	E5-2. Indoor visual comfort	
		quality	E5-3. Indoor acoustic comfort	
		E6. Water	E6-1. Water consumption of operation activities	
		efficiency		
	Retrofitting EC1. Initial (retrofitting) cost		fitting) cost	Wang, Xia &
nic	stage		Zhang 2014;	
101	Operation	EC2. Operation co	ost	Kylili,
C01	stage	EC3. Maintenance	e cost	Fokaides &
E		EC4. Repair cost		Jimenez 2016;
		EC5. Replacemen	t cost	180/2017
	Retrofitting	S1. Impact on	S1-1. Noise impact on the neighbourhood	BSI 2014;
	stage	neighbourhood	S1-2. Emission impact on the neighbourhood	Kylılı,
			(e.g., particulate matters, dust from retrofitting	Fokaides &
			construction)	Jimenez 2016;
		S2. Safety	S2-1. Site safety during retrofitting	Orlowski &
		S3. Impact on	S3-1. Damage to cultural heritage	Radžiejowska
ial		cultural heritage		2017; Sentes et al
Soc	Operation	S4. Accessibility	S4-1. Accessibility to building facilities for people	2017
•1	stage		with special needs	Zorghami &
			S4-2. Accessibility to building services (such as	Fatourehchi
			automatically adjusted indoor temperature,	2020
		C5 Maintenan	numerity, and dimming level by BIVICS)	2020
		55. Maintenance	55-1. Impact on maintenance and maintainability	
		and	irom newly added building fabrics or building	
1		maintainability	systems	

Table 6.2. Assessment criteria of retrofitting

The process of developing assessment criteria can also remind the retrofitting team that there may be other problems they neglected in Step 2 (Section 6.4). If any, they should be added, and the decision-making process should restart from Step 2 (Section 6.4).

6.7. Evaluate performance of retrofitting activities

The fifth step is to evaluate the performance of retrofitting activities. It is achieved by analysing the performance of proposed retrofitting activities by measuring their ability to satisfy each assessment criterion. To aggregate evaluation results as one, normalisation is employed to convert evaluation into unitless values. In this step, the level of importance of assessment criteria is considered by integrating a weighting system. In this way, a weighted assessment score can be attained for each retrofitting activity, which can be used to develop retrofitting strategies in the next step (Section 6.8). These three sub-steps are explained as follows.

6.7.1. Evaluate performance and convert to numerical data

Both quantitative and qualitative issues are included in the identified assessment criteria from the last step. The techniques for measuring quantitative criteria are well-developed and have been applied for decades. The difficulty in this step is quantifying subjective assessment criteria. The evaluation methods for assessing objective and subjective criteria are discussed below.

6.7.1.1. Evaluation of objective criteria

Most of the criteria under the environmental and economic dimensions are objective criteria. As discussed in Chapter 4, LCA is one of the most widely used methods to quantify the environmental impacts of construction projects. It is adopted in this model to evaluate the environmental assessment criteria from Table 6.2, including embodied energy, carbon emissions, water efficiency, use of reusable/recyclable materials, and waste generation. If data or information is not available to undertake an LCA study, simulation tools can be adopted to simulate the environmental performance of retrofitting activities, such as indoor environmental quality.

When retrofitting using the LCA methodology, the system boundary, study period and functional unit need to be defined first based on EN 15978 (CEN 2011). Figure 4.1 in Chapter 4 illustrated the modular information for different stages of the building assessment contained in LCA. Since this step is to assess the difference by retrofitting activities, which mainly occurs in the retrofitting stage and operation stage, the cradle-to-use system boundary is taken, including modules A1 to A5 at the product and construction process stages, and modules B2 to B7 at the use stage.

Based on EN 15978 (CEN 2011), the energy and water consumption in the operation stage is calculated. However, in this study, comparing the amount of consumption, the saving on energy and water use is more desired, representing the ability of potential retrofitting activities to improve the building's energy efficiency and water efficiency. Accordingly, the calculations required in the conceptual model include:

Modules A1 to A3	Includes energy use by extracting, transporting and					
	manufacturing raw materials into materials or products at product					
	stage.					
Modules A4 and A5	Includes energy use by transporting to site and installing the					
	manufactured materials or products on site.					
Modules B2 to B5	Includes the energy saving by retrofitting activities from					
	maintaining, repairing and replacing the upgraded building					
	systems and/or components during the operation stage.					
Modules B6 and B7	Includes energy saving and water saving from operating the					
	newly installed building components or systems during the					
	operation stage.					

The whole building is defined as the functional unit for this study since the improvement by implementing the developed retrofitting strategy is evaluated for the whole building. The residual service life of retrofitted buildings also influences evaluation results. As discussed in Chapter 4, if residual service life can be estimated, modules B2 to B7 can be assessed based on it. If it cannot be estimated, a service life of 50 years can be regarded as an acceptable study period (Desideri & Asdrubali 2018). With the system boundary and functional unit defined, environmental impacts can be evaluated using the assessment criteria previously developed.

(1) Energy use and carbon emissions

At the retrofitting stage, embodied energy (EE) and embodied carbon (EC) emissions of building materials, and energy consumption and carbon emissions at retrofitting construction are evaluated. Two methods are available to calculate them. First, LCA software, such as Gabi, Simapro, Umberto and openLCA, can be used to evaluate embodied energy and embodied carbon emissions of materials from retrofitting activities (Silva et al. 2019). However, most software packages include only one database and assessment model (Kalverkamp & Karbe 2019; Pauer, Wohner & Tacker 2020). Even when some software, like GaBi and SimaPro, has more than one database to choose from, it cannot guarantee that an updated database for every country is available. Therefore, local variations may be an issue in using simulation software.

Another method is using embodied energy and materials' carbon coefficients to calculate the consumption and emissions. It can be realised by multiplying them with the quantity of installed materials. The coefficients vary with advancement of technology and different locations. In the long term, technology may have impacts on determining the carbon emission coefficient, but in the short term, such impacts will be at the margin. Due to different fuel sources and methods of extracting raw materials, manufacturing and transportation, the coefficients are different in different locations. Therefore, the latest updated local data about embodied energy and embodied carbon emissions should be adopted when implementing the model in a real retrofitting case. However, it is often challenging to obtain local data of embodied energy and materials' carbon coefficients. The Inventory of Carbon and Energy (ICE) developed by the Department of Mechanics of University of Bath (UK) provides data about embodied energy and materials' carbon coefficients (Hammond & Jones 2011). It is often used as a data resource to calculate embodied energy and carbon of building materials, such as in the study by Rodrigues et al. (2018) and the study by Zeitz, Griffin and Dusicka (2019). However, as discussed above, relying on one data resource cannot deal with local variations. Therefore, the local data adopted from existing studies and literature may contribute to more accurate evaluation results. A database, such as ICE, can be used as a supplement if no local data is available. Indeed, there is a compromise between what is theoretically correct and what is reasonable accuracy in practice. The accuracy of evaluation will be affected by the available data. It is important to get as much data as possible that can reflect the local situation.

The energy consumed and carbon emitted on the construction site are evaluated based on the quantity, type, power and working duration of equipment and machines used on the retrofitting site. The data can be obtained by consulting with specialists who are familiar with retrofitting construction. Alternatively, the data from existing studies in which the same retrofitting activities are used to retrofit a building of similar scale can be adopted.

In the operation stage, recurrent embodied energy and carbon emissions from replacement and operation energy consumption and carbon emissions are taken into account. Recurrent embodied energy and embodied carbon emissions are the aggregation of embodied energy and embodied carbon emissions of the installed building components during the residual lifetime of the building. Energy consumption by operating building service systems can be estimated using simulation tools or calculating the usage based on the quantity, power and operation hours of electricity-related building systems. The amount of carbon emissions can be attained based on the conversion relation with the type of energy supplied for the building. For example, the national database in the US, the Emission and Generation Resource Integrated Database (eGRID), can be used as a data resource to estimate the environmental characteristics of most electric power generated in the US (EPA 2022). The study by Xu, Schwarz and Yang (2020) summarised the carbon emission coefficients for coal, oil, natural gas and hydro-electricity by four institutes, as Table 6.3 shows. Again, the coefficients vary with advancement of technology and different locations. The latest updated local data should be adopted.

	Carbon emission coefficients				
Data resource	Coal (tonne/tonne)	Oil (tonne/tonne)	Natural gas (tonne/million cubic metre)	Hydro & nuclear	
Institute of Energy Economics, Japan	0.756	0.586	0.449	0	
US DEO/US EIA	0.702	0.478	0.389	0	
Chinese State Scientific & Technological Commission: Climate Change Program	0.726	0.583	0.409	0	
Chinese Energy Research Institute	0.748	0.583	0.444	0	

Table 6.3. Carbon emission coefficients

Source: Xu, Schwarz & Yang 2020

(2) Water use

The saved water use in the operation stage is calculated in this study. In contrast to estimating energy consumption, simulation tools are rare for estimating water use by buildings (Bertone et al. 2016). The common way to estimate water use by buildings is based on statistical data (Bertone et al. 2016). However, data collection is also a challenge to estimate water saving by retrofitting (Bertone et al. 2016; Bertone et al. 2018). The retrofitting team needs to conduct a statistical survey to collect data about how much water can be saved by implementing potential retrofitting activities. Alternatively, the necessary data, such as the flow rate of taps, use frequency and time of each use, can be collected through existing publicly available sources.

(3) Waste generation

In the retrofitting stage, waste is mainly from removing obsolescent building materials and components. However, the materials are removed due to their poor performance, which means they will be removed anyway for regular maintenance even if retrofitting is not conducted. Therefore, this amount of waste generation will not be reduced by retrofitting, so only construction waste from implementing retrofitting activities is counted in this study.

During the operation stage, the waste generated from operating, maintaining, repairing and replacing the upgraded building components or systems needs to be calculated. The weight of replaced building materials is relatively easy to estimate. For example, the weight of insulation materials of external walls can be estimated based on area of external walls (excluding area of windows), and density and designed thickness of required insulation materials. As for the replaced components of building service systems, they may be reusable or recyclable, such as the motors from replaced lifts, which will not be landfilled directly. Therefore, the waste generation by replacing them may not be accurately estimated in practice. Considering the small amount of waste generation by operating, maintaining and replacing the building service system, if the waste cannot be accurately estimated, it will be neglected in this study. However, post-assessment can be conducted to obtain real-time data about waste generation to calibrate the estimation results and can also be used as a reference for future study. Due to the time constraint, the post-assessment was not realised within the period of the thesis study. It may be conducted as a follow-up study to this research.

(4) Material use

The amount of reusable or recyclable materials is estimated for the retrofitting and operation stage. The higher the percentage of reusable or recyclable materials used in the retrofitting stage, the better the estimation result for material use. In the operation stage, the same type and quantities of materials are used for replacement when it is at the end of the service life of the building materials and/or systems. Therefore, the percentage of reusable or recyclable materials in the operation stage would be consistent as it is in the retrofitting stage.

(5) Initial cost, operation cost, maintenance cost, repair cost and replacement cost

LCC is used to assess the economic performance of potential retrofitting activities. In the conceptual model, the initial cost of materials and building systems in the retrofitting stage is counted. In the operation stage, the costs for the use of electricity and water for operating different electricity-related and water-related systems, for maintaining the building such as cleaning and facility management, and for repairing or replacing broken and obsolescent materials and components are included. In practice, the retrofitting team needs to conduct a statistical survey to obtain the initial cost, maintenance cost, repair cost and replacement cost for potential retrofitting activities. However, it is an ideal data collection method only when time and budget are not issues. Alternatively, the data collected through existing publicly available sources, such as research studies, reports and websites, can be adopted to approximately evaluate the economic performance of potential retrofitting activities.

As discussed in Section 4.3.1.4 of Chapter 4, EAC is calculated in this conceptual model to evaluate the economic performance of potential retrofitting activities. With the determined study period in the LCA study and collected costs by potential retrofitting activities, the EAC of them can be calculated based on Equation 4.3.

6.7.1.2. Evaluation of subjective criteria

The subjective criteria identified in this conceptual model are indoor environmental quality in the environmental dimension and all assessment criteria in the social dimension. Indoor environmental quality is a special criterion, which can be assessed using objective methods to quantify sensations (temperature, humidity, CO₂ concentration) or objective methods to measure how people feel or are satisfied with indoor environmental quality. Section 6.7.1.3 elaborates on how to assess indoor environmental quality using objective and subjective methods.

Assessment criteria in the social dimension are used to measure retrofitting activities' social impacts on people who use the building, or live or work in surrounding communities, and even the whole of society. As Table 6.2 shows, the impacts may be negative impacts on the retrofitting stage, including impacts on the health and comfort of people living or working in neighbouring buildings (S1-1 and S1-2 in Table 6.2), impacts on safety issues on the construction site (S2-1) and damage to cultural heritage (S3-1). Moreover, retrofitting activities may increase or decrease tenants' satisfaction with the use of the retrofitted building by increasing or decreasing accessibility to building facilities and services (S4-1 and S4-2) and imposing impacts on building maintenance (S5-1). As discussed in Section 4.4 of Chapter 4, SCBA and S-LCA are two common methods to assess social impacts.

SCBA is normally used to evaluate what the impact of an investment is on society, so it is often used by government to support decision making. However, the monetary-equivalent measure in SCBA is much critiqued since there is no agreement on attaching a price to welfare variation. S-LCA is another popular method to assess social impacts, which is compatible with the assessment framework in LCA. Since the environmental impacts are assessed by using LCA in the conceptual model, S-LCA is a more suitable method to assess social impacts caused by retrofitting activities compared to SCBA.

In S-LCA, a value score is used to express levels of measures on social impacts of retrofitting activities. Four common types of rating scales that use value scores to express measurements at different levels are summarised below (Bhattacherjee 2012; Dodgson et al. 2009; Hobbs & Meier 2000):

- Nominal scale: a nominal scale, also called a categorical scale, reflects the attributes of different categorical data like the gender of participants, religious affiliation, etc. A binary scale is a special nominal scale, which only allows two options like yes or no.
- Ordinal scale: is used to measure order-oriented data, such as ranking of students by their grades. The Likert scale is a prevalent ordinal scale that uses simply worded
statements to which respondents can express their level of agreement or disagreement on a three- four-, five-, six-, or seven-point scale.

- Interval scale: an internal scale where not only the order of different options but also the magnitude of difference between each level of options is calculable.
- Ratio scale: a scale with all the characteristics of nominal, ordinal and interval scales also has a "true zero" point, where the value zero denotes the lack or non-availability of the underlying construct.

All the social criteria from Table 6.2 are about the satisfaction level of stakeholders who use the building, visit the building, randomly pass by the building, or who work or live in neighbouring buildings. A rating scale acting as a proxy variable is commonly used to express different levels of social impacts (Ornaghi & Van Beveren 2011).

A study by Menadue (2014) aimed to determine whether green office buildings outperform conventional ones by conducting a survey to understand the different occupants' satisfaction levels with building design and facilities of both green and non-green office buildings. A rating scale from 1 to 7 was adopted to express different levels of tenants' satisfaction, where 7 represents "satisfactory" or "good". Similarly, D'Oca et al. (2017) conducted a study to explore how social-psychological and demographic factors impact tenants' behaviour and the intention of sharing the control system of buildings. A survey was also developed and sent to the targeted group – users of teaching buildings in universities across four countries, with a five-point Likert scale provided for respondents to use. In this way, the impacts of social-psychological and demographic factors can be measured.

A five-point Likert scale is also used in the conceptual model to estimate social impacts, as Table 6.4 shows. According to the developed social assessment criteria, the social impacts on the retrofitting stage are negative. Therefore, the less impact, the better the performance. For the social impact on the operation stage, the impacts from retrofitting activities are compared with the current situation before retrofitting is conducted. If it is same as the situation before retrofitting, a score of two should be given, representing the same performance level as the current situation. If the performance is better than the current situation, scores of three to five can be given based on the improved extent. If the performance is worse than the current situation, a score of one should be given.

Impact range on retrofitting stage	Value score	Performance range in operation stage
Minimum impact	5	Excellent performance
Minor impact	4	Very good performance
Average impact	3	Good performance
Significant impact	2	Same as current performance
Very significant impact	1	Unsatisfactory performance

Table 6.4. Rating scale for subjective issues

Source: Adapted from Vagias 2006

The final score for each assessment criterion is an average score of the value scores given by the retrofitting team, which can be achieved based on Equation 6.1.

$$S_{ij} = \frac{\sum S_{ij}^n}{n}$$
 Equation 6.1.

Where,

- S_{ij} social assessment score of retrofitting activity *i* regarding assessment criterion *j*
- S_{ij}^{n} the rating score of retrofitting activity *i* regarding assessment criterion *j* given by the nth assessor
- n the number of assessors in the retrofitting team

6.7.1.3. Evaluation of indoor environmental quality

Indoor environmental quality is a complicated issue involving various influential factors, such as the building's insulation performance, tenants' use behaviours, tenants' age and gender etc., and different instruments may need to be used to assess different aspects of indoor environmental quality (Horr et al. 2016). Normally, an assessment of indoor environmental quality is conducted for two purposes. It can be conducted at the stage of designing and planning to identify improvement areas for which potential retrofitting activities can be proposed, which is called pre-assessment. Pre-assessment can also be conducted to evaluate effect by any changes on the indoor environmental condition – evaluating improvement of potential retrofitting activities. If it is conducted at the operation stage, the assessment is to measure the improvement by implementing retrofitting activities, or the level of satisfaction of occupants, which is called post-assessment. In this study, the conceptual model is developed to decide retrofitting strategies; thus, only pre-assessment is needed.

As discussed in Section 3.2.1 of Chapter 3, indoor environmental quality consists of three main components: indoor thermal quality, visual quality, and acoustic quality. The components can be expressed and assessed in either quantitative or qualitative ways (Cheong et al. 2020; Horr et al. 2016). For example, the thermal sensation is an objective item that can be quantified by evaluating indoor temperature, relative indoor humidity and CO₂ concentration. However, thermal comfort is a subjective issue about how occupants feel or are satisfied with indoor thermal condition (Horr et al. 2016; Kaushik et al. 2020). Similarly, either the sensation or comfort regarding the indoor visual quality and indoor acoustic quality can be evaluated.

(1) Quantitative assessment method

By implementing a quantitative method to evaluate indoor environmental quality, different simulation tools can be adopted to estimate the deviation of the current state from the required performance or improvement by any intervention (retrofitting activity). Based on the study by Wang and Zhai (2016) and Cheong et al. (2020), computation fluid dynamics (CFD) software has been extensively used to estimate indoor thermal quality. Commonly used simulation tools for estimating indoor visual quality include IESVE, SuperLite, Micro Lumen, Radiance, Lightscape, Daylight Visualizer Velux, Ecotect, and PKPM-daylight, and available simulation software tools for estimating indoor acoustic quality include Odeon, Epidaure, Soundplan, Raynoise, Ramsete, EASE, and Cadna. By using simulation-aided design methodology to estimate indoor environmental quality, three stages are commonly conducted: Step 1 – Model the indoor space in a virtual environment using modelling software, such as Sketchup and Autodesk CAD; Step 2 - Input the virtual model and indoor environmental quality-related factors in the simulation software, such as local climate, building orientation, HVAC arrangement, and lighting conditions; and Step 3 - Simulate indoor environmental quality performance based on the running results of the implemented simulation software (Cheong et al. 2020).

Different simulation tools can be used to assess specific parameters against the required performance. For example, the optimum temperature range for comfort in office buildings is from 21 °C to 25 °C based on the study by Kaushik et al. (2020). The natural sound level of a typical air conditioned office is between 45 dB and 70 dB (Horr et al. 2016). The recommended illuminance level of office buildings is 500 lux based on the Illuminating Engineering Society of North America (IESNA) standard (IESNA 2011; Park et al. 2021). The indoor

environmental quality of office buildings is affected by various factors, which include but are not limited to work position, hours spent at work, clothing, physical activities and seating, posture and mental state, accessibility to indoor environmental quality controls, local climate and geography, and human factors such as age, gender and metabolism (Chen et al. 2021; Kaushik et al. 2020). Therefore, the benchmarks of indoor environmental quality differ by the location and work type of office buildings. The related local requirement or standard should be adopted as an assessment benchmark when applying the conceptual model to a specific retrofitting project.

(2) Qualitative assessment method

Since various factors affect indoor environmental quality, it is more suitable to express indoor environmental quality as a cumulative response of occupants toward the state of the indoor environment (Kaushik et al. 2020). Therefore, qualitative assessment methods are usually used to assess the level of comfort or satisfaction of occupants with indoor environmental conditions. A field survey can be used to collect opinions of occupants on the current state or changes in the physical environment via interviews and/or questionnaires (Horr et al. 2016). Interviews are intended for use in a detailed study with a small sample size, while a field survey can be used to obtain the perception of multiple occupants about indoor environment conditions through questionnaire surveys (Horr et al. 2016).

By collecting subjective opinions on indoor environmental quality, rating scales may be adopted to express different levels of satisfaction. For example, thermal comfort can be measured through the metric, the predicted mean vote (PMV), via a field survey (Cheong et al. 2020; Horr et al. 2016). The PMV predicts the mean response of building occupants based on the thermal sensation scale regulated by the American Society of Heating, Refrigerating and Air-Conditioning Engineers (ASHRAE) Standard 55 (ASHRAE 2009; Che et al. 2019; Cheong et al. 2020; Horr et al. 2016). In ASHRAE, a seven-point scale ranging from -3 to +3, equating to cold and hot relative to the comfort optimum, is used to measure occupants' perception of indoor thermal conditions (Cheong et al. 2020; Horr et al. 2016). The Likert scale is also commonly used to measure the level of satisfaction of occupants for indoor environmental quality (Likert 1974). More discussion about rating scales can be found in Section 6.7.1.2.

There are also some survey instruments applicable for office buildings to conduct field surveys for measuring indoor comfort (Horr et al. 2016):

- BOSTI (Buffalo Organisation for Social and Technological Innovation)
- BUS (Building Use Studies Occupant Survey)
- HOPE (Health Optimization Protocol for Energy-efficient Buildings)
- REF (Ratings of Environmental Features)
- CWRE (Checklist of Work Related Experiences)
- AMA WorkWare (Alexi Marmot Associates)
- DQI (Design Quality Indicator)
- BASE (Building Assessment Survey and Evaluation)
- ProKlima
- ASHRAE RP-884
- CBE Survey (Centre for the Built Environment)
- OPN (Office Productivity Network)

(3) Combining quantitative and qualitative assessment methods

Quantitative and qualitative assessment methods can be conducted jointly to measure the indoor visual and acoustic environment of existing office buildings. For example, three simulation methods are available to simulate indoor visual condition: field lighting simulation, rendering-based lighting simulation, and immersive virtual simulation (IVS) (Ma, Lee & Cha 2022). Field lighting simulation is a traditional simulation method by changing the lighting conditions in a real office or mock-up rooms. Rendering-based lighting simulation is to investigate users' reaction to lighting by displaying rendered representations of the lighting designs of 3D computer models. IVS offers an immersive experience of one-to-one scale lighting designs of 3D computer models through visualisation devices such as head-mounted display and cave automatic virtual environment (Ma, Lee & Cha 2022). After experiencing the indoor visual conditions simulated by these three methods, occupants need to complete a field survey or interview to measure the impact on their visual perception and task performance (Cheong et al. 2020; Ma, Lee & Cha 2022). Similar simulation methods integrated with opinion collection methods, such as survey and interview, can also be used to assess indoor acoustic performance.

Based on the discussion above, multiple factors affect indoor environment quality, and both quantitative and qualitative methods are available for assessing indoor environment condition or estimating improvement by potential retrofitting activities. As a quantitative method, different simulation tools can be used to measure the state of indoor environment quality or improvement by potential retrofitting activities. However, it may be expensive and time-consuming to build virtual models.

In addition, there is no point measurement for indoor comfort, but a comfort zone for each factor (Horr et al. 2016). Either too high or too low measurement is not acceptable. However, even though the estimation results from using simulation tools fall in the comfort zone (benchmark), it cannot ensure that the occupants are satisfied with the condition of indoor environmental quality, since it is difficult to include all influential factors in the simulation process. Consequently, it will be less meaningful to include evaluation results of the improvement on indoor environmental quality by potential retrofitting activities to develop retrofitting strategies. Therefore, if time and budget are not problems, simulation methods may be conducted to estimate the improvement by potential retrofitting activities, but it would be better if the evaluation results from a qualitative assessment method can be combined to increase assessment accuracy. Otherwise, only a pre-assessment of indoor environmental quality to identify improvement areas for which potential retrofitting activities can be proposed is conducted in this study.

6.7.2. Weight assessment criteria

In most cases, assessment criteria in a decision-making process have different levels of importance representing a weighting. It is the value assigned to criteria which reflects preferences of decision makers and/or stakeholders among possible outcomes (Geneletti 2019). Different weighting generation methods can provide necessary frameworks for decision makers to compute weights. Delphi and Analytic Hierarchy Process (AHP) are two of the most prevalent methods for developing weights, especially in the context of group decision-making (Liu 2014).

The Delphi method engages experts to collect and distil judgments via an iterative process. The process stops only when the purpose of the study is reached, like theoretical saturation is

reached, or group consensus is attained. The iterative process can refine and improve the reliability of the outcome (Skulmoski, Hartman & Krahn 2007). Zolghadr-Asli et al. (2021) stated that Delphi can help decision makers select the most relevant assessment criteria, and also assign proper weights to them. However, time and cost may hinder the applicability of this method (Liu, Ding & Samali 2013; Yu et al. 2015).

The other method, the AHP method, was developed by Saaty (1980), and mainly focuses on solving MCDM problems (Yu 2002). Decision makers need to give a rating score by pairwise comparison, which is used to calculate the weighting score (Steele et al. 2009). The rating scale adopted for the AHP is a nine-point scale, describing the relative importance level for pairwise comparisons. A matrix of pairwise comparison can be structured based on given rating scores, and weightings of assessment criteria are normalised values of the eigenvector that is related to the maximum eigenvalue for this matrix. Saaty (1980) explained that this determination process in AHP is the best way to reduce the impact of inconsistencies in the ratios.

AHP is recognised as a popular mathematically based trade-off technique. The formulation embodied can be used to construct and express trade-offs; eventually, it can assist in identifying the optimal solution (Kassab 2013). Moreover, the traits of the relatively low requirement on computation and the reliance on subjective assessment make AHP the most commonly used method for generating weights for evaluation criteria in MCDM (Zolghadr-Asli et al. 2021).

In this study, the conceptual model is developed based on MODM, which is one of the branches in MCDM. It is intended to consider trade-offs between the environmental, economic and social dimensions when using it to develop retrofitting strategies. Based on the above discussion, AHP is more suitable to be adopted in this study to estimate weights compared to Delphi. It is worthy to mention that there is a generalisation of AHP, Analytic Network Process (ANP), always discussed with AHP (Görener 2012). Both of them can be used to generate cardinal rankings of the alternatives, but ANP is usually adopted when the decision problems cannot be structured with a uni-directional hierarchical structure as they are in AHP (Görener 2012; Ossadnik, Schinke & Kaspar 2016). Since assessment criteria under each sustainability dimension have been identified as shown in Table 6.2 (see Section 6.6), they can be easily

structured as a uni-hierarchical structure. Therefore, AHP is more suitable than ANP to be used in this study.

Two levels of weights need to be determined in this study. The first level is the three dimensions of sustainability: environmental, economic and social. The second level is assessment criteria under environmental and social dimensions. For assessment criteria under the economic dimension, based on the above discussion, the LCC method is used to calculate the EAC of potential retrofitting activities. In this method, the level of importance of these costs is regarded as equal. Therefore, no additional weighting is needed.

The AHP method should be conducted by a retrofitting team that has expertise in construction and property management. Meanwhile, they are playing a role in communicating with different stakeholders in the decision-making process in order to understand the needs of different stakeholders for building retrofitting. Ideally, different stakeholders or representatives of them can be included so that the determined weights can truly reflect the actual preference of different stakeholders. However, conducting an AHP study with too many people can be timeconsuming and challenging.

6.7.3. Normalise measurements for criteria aggregation

Based on Section 6.7.1, assessment criteria are quantified into different units, which makes it difficult to have one final score to express overall performance. To solve the problem, normalisation is used to convert assessment results into a preference scale which is a unitless expression of the level of desirability of proposed retrofitting activities (Geneletti 2019).

6.7.3.1. Normalise measurements

The four most popular normalisation procedures (Equations 6.2–6.6) are introduced based on the study by Munier, Hontoria and Jiménez-Sáez (2019):

(1) Sum of performance values in a row:

$$a_{ij}^* = \frac{a_{ij}}{\sum_{1}^{n} a_{ij}}$$
 Equation 6.2.

Where,

 a_{ij} – original performance value of option *i* under criterion *j* a_{ij}^* – the normalised value of option *i* under criterion *j n* – number of normalised values

(2) Largest value in a row:

$$a_{ij}^* = \frac{a_{ij}}{\max_{a_{ij}}}$$
 Equation 6.3.

Where,

 $max_{a_{ii}}$ – the largest performance value

(3) Euclidean formula:

$$a_{ij}^* = \frac{a_{ij}}{\sqrt{\sum_{1}^{n} (a_{ij})^2}}$$
 Equation 6.4.

It computes the formula in the denominator and divides each performance value by it.

(4) Maximum/minimum ratio:

$$a_{ij}^* = \frac{a_{ij} - min_{a_{ij}}}{max_{a_{ij}} - min_{a_{ij}}}$$
 if assessed performance is positive impact Equation 6.5.

$$a_{ij}^* = \frac{\max_{a_{ij}} - a_{ij}}{\max_{a_{ij}} - \min_{a_{ij}}}$$
 if assessed performance is negative impact Equation 6.6.

Where,

$$min_{a_{ij}}$$
 – the smallest performance value

Theoretically, the normalisation results by different methods should be the same (Munier, Hontoria & Jiménez-Sáez 2019). However, the last method (maximum/minimum ratio) is regarded as advantageous in MCDM since it incorporates subtracting members, which allows discrimination, while the first three produce a concentration of values. By favouring discrimination, the normalised results based on the maximum/minimum ratio can assist decision makers in distinguishing the optimal solution. In addition, the maximum/minimum ratio can ensure that the bigger normalised value can always represent better performance. In the assessment process of the conceptual model, both positive and negative impacts of retrofitting activities are evaluated. For positive impacts, a bigger evaluation score means better

performance of the measured retrofitting activity. In contrast, a smaller evaluation score represents a better performance of the measured retrofitting activity if it causes negative impacts. With different evaluation directions, it is difficult to combine different assessment scores into one score to represent the overall performance. The normalisation method, maximum/minimum ratio, converts the negative direction to the positive by measuring the absolute distance between the evaluation and the target. In this way, a bigger normalised value is always desired to represent better performance. Due to the asserted benefits, the normalisation method, the maximum/minimum ratio (Equations 6.5 and 6.6), is adopted in the model to normalise evaluation values into a unitless scale.

6.7.3.2. Aggregate normalised measurements

The last task of this step is to aggregate normalised scores as one to express overall performance. The criteria aggregation is realised based on a rule that can combine assessment outcome and associated weights (Geneletti 2019). One of the most widely used rules for criteria aggregation is the weighted linear combination (WLC) (Steele et al. 2009). Moreover, this aggregation procedure permits "trade-offs" between criteria, in which poor performance regarding some requirements can be compensated by achieving better performance on other criteria (Geneletti 2019).

Trade-off is defined as the increased amount of one criterion while that of the value decreases regarding the other criterion in a particular solution (activity a_i in Equation 6.7) is replaced by another (a_j in Equation 6.7). Trade-off can be expressed as a ratio as in Equation 6.7 (Nowak 2010):

$$T_{ji}^{XY} = \frac{X_j - X_i}{Y_i - Y_j}$$
 Equation 6.7.

Where,

- T_{ji}^{XY} trade-off between a pair of criteria X and Y when activity a_i is replaced by activity a_i
- $X_j X_i$ value increase in criterion X when activity a_i is replaced by activity a_j
- $Y_i Y_j$ value decrease in criterion Y when activity a_i is replaced by the activity a_j

Nowak (2010) further explained Equation 6.7 by assuming a decision-making scenario that when analysing retrofitting activity a_i , it is realised that the evaluation for criterion X should be improved while the evaluation for criterion Y can be decreased. Therefore, there is no doubt that another retrofitting activity a_{j^*} is preferable if the evaluation $X_{j^*} > X_i$, and $Y_{j^*} \ge Y_i$. However, if the activity a_{j^*} does not exist, which is common in MCDM, Equation 6.7 can be used to look for activity a_j that can maximise the trade-off between criterion X and Y. The trade-off represented in Equation 6.7 is a point-to-point trade-off. In this study, the three pillars of sustainability – environmental, economic and social – are considered. Instead of the tradeoff at a point, a range of acceptable trade-offs between the three pillars is more desired. The derivative of Equation 6.7 by considering trade-offs between the three sustainability dimensions is given in Appendix A.

In the weighted linear combination, the normalised criterion scores are multiplied by their weights, which then adds the results across all criteria to generate an overall score for each retrofitting activity (Geneletti 2019). It means that although a retrofitting activity is evaluated as poor performance regarding one or several assessment criteria, it is still possible to combine it with other retrofitting activities to form the optimal retrofitting strategy, which has the highest overall evaluation score, or maximised trade-off (at Step 6 in Section 6.8). In general words, the optimal retrofitting strategy is an amalgamation of retrofitting activities that can best use their strengths to complement others' weaknesses.

Based on the above discussion, weighted linear combination is adopted in this conceptual model to attain overall evaluation scores of proposed retrofitting activities. Based on Equation 6.8, the evaluation score of a retrofitting activity regarding one of the three sustainability dimensions can be attained by aggregating scores across all assessment criteria in this sustainability dimension.

$$P_j = \sum_{i=1}^m a_{ij}^r \times w_j^r + \sum_{i=1}^n a_{ij}^o \times w_j^o \qquad \text{Equation 6.8.}$$

Where,

- P_j the evaluation score of retrofitting activity *j* for one of the three sustainability dimensions
- a_{ij}^r the normalised score of retrofitting activity *j* regarding criterion *i* at the retrofitting stage

- a_{ij}^{o} the normalised score of retrofitting activity *j* regarding criterion *i* at the operation stage
- w_i^r the weighting score for criterion *i* at the retrofitting stage
- w_i^o the weighting score for criterion *i* at the operation stage
- m the number of assessment criteria included at the retrofitting stage
- n the number of assessment criteria included at the operation stage

By now, the contribution the proposed retrofitting activities can make to meeting retrofitting goals has been evaluated. It is time to consider whether there are other activities that may have better performance. If yes, they should be added. Therefore, the process goes back to the third step (Section 6.5) – propose retrofitting alternatives. In this way, the process can be refined more effectively compared to reacting to the feedback when the whole process is completed.

6.8. Develop retrofitting strategies

This step is to develop retrofitting strategies based on performance evaluation results of retrofitting activities. Rey (2004, p. 267) defined a retrofitting strategy as "a set of interventions, dictated by a coherent architectural attitude and technically optimised, in particular through a full coordination of the interventions on the sheathing surfaces and the technical installations". Interventions refer to retrofitting activities in this study. Therefore, retrofitting strategy in this study can be defined as an amalgamation of retrofitting activities that can improve the performance of building envelopes and building service systems to achieve overall performance improvement of the existing building.

In this conceptual model, a binary linear mathematical model is established to determine which of the proposed retrofit activities should be adopted in order to achieve the highest overall evaluation scores while satisfying all project constraints. Before outlining the mathematical model, the concept of ceteris paribus is introduced. It forms the basis on which the performance of retrofitting strategies is evaluated.

6.8.1. Ceteris paribus

Ceteris paribus is a widely used assumption for analysing a series of impacts of a complex problem. It has been used as an important economic clause for a long time, meaning that the effect of an event is analysed in isolation and assuming that other factors remain constant (Kremling, Bruns & Schildmann 2019). In this study, the concept of ceteris paribus is used to evaluate the performance of retrofitting strategies, which the assumptions below are based on:

- (1) The effect of one single retrofitting activity is measured for each analysis.
- (2) The effect of each retrofitting activity is analysed assuming that no other retrofitting activities have been implemented prior to this one.
- (3) The effect of a retrofitting strategy is the aggregated effect of included retrofitting activities.

According to the three assumptions, the performance of a retrofitting strategy can be evaluated by aggregating the performance of contained retrofitting activities. However, there is an exception as discussed in Section 3.4.6 of Chapter 3. The performance of HVAC and the insulation of building envelopes are closely related, and improving building insulation should be prioritised over improving the energy efficiency of the HVAC system. Therefore, if both of them are proposed as retrofitting activities, the effect of improving HVAC should be superimposed on the effect of improving the insulation of building envelopes.

In summary, the concept of ceteris paribus is adopted not only to simplify evaluation but also to help make it clear how much benefit the retrofitting activity can bring, and how adverse an impact the activity can cause. Then, different activities can largely borrow others' strong points and offset weaknesses to develop a retrofitting strategy, reflecting "trade-offs" in the weighted linear combination discussed in Section 6.7.3.

6.8.2. A binary linear mathematical model for developing retrofitting strategies

Multiple objective programming is often represented as a mathematical model that coordinates different criteria to select the best element from a set of alternatives (Alinezhad & Khalili 2019). Three common methods of multiple objective programming are multiple linear objective programming (MLOP), multiple nonlinear objective programming (MNLOP), and goal programming (GP). MLOP should be adopted to achieve an optimal solution if all objective functions can be optimised. If some constraints or objective functions are not linear, which means not all objective functions can be optimised, MNLOP can be used to generate solutions

that can "best" satisfy most of the objectives. As for GP, preferred solutions are developed instead of the optimal solution by minimising unwanted deviation variables (Romero 2001).

As discussed in Section 4.5.3 of Chapter 4, an optimal solution is desired for a retrofitting project, which means the developed retrofitting strategy can have the maximised overall evaluation score regarding the three sustainability dimensions within project constraints. In addition, the performance of retrofitting strategies is evaluated in a linear form, as Equation 6.9 shows. Therefore, MLOP is suitable to be adopted in the conceptual model to develop retrofitting strategies. However, there is a chance that no solution can be generated by using MLOP. In this case, other retrofitting activities that may better meet retrofitting goals should be proposed. Otherwise, some compromise has to be made to get the optimal solution, such as increasing the project budget.

To establish a linear mathematical model, three elements should be clarified (Vanderbei 2020):

• **Decision variables** are physical quantities that decision makers have control over. They are usually denoted as:

$$x_i, j = 1, 2, ..., n$$

• **Objective functions** are mathematical functions of the decision variables that turn a solution into a numerical value. In linear programming, they are always used to maximise or minimise linear functions, which can be represented as:

 $\operatorname{Max}/\operatorname{Min} Z = c_1 x_1 + c_2 x_2 + \dots + c_n x_n$

• **Constraints** are a set of functional equalities and/or inequalities that describe physical, economic, technological, legal, ethical, or other constraints with numerical values that can be assigned to decision variables. They can be represented as linear combinations of decision variables:

$$a_1x_1 + a_2x_2 + \dots + a_nx_n \begin{cases} \geq \\ \leq \\ \leq \end{cases} b$$

In this conceptual model, binary linear programming, a special branch of MLOP, is used to develop retrofitting strategies. In binary linear programming, each decision variable can only have a value of 0 or 1. It could refer to the rejection or acceptance of a choice, the turning off or on of switches, a no or yes response, or a variety of other scenarios (Chinneck 2004). In the conceptual model, 0 represents that the proposed retrofitting activity is not selected, and 1 represents that the retrofitting activity is selected and will be amalgamated with other selected activities as a retrofitting strategy.

In MLOP, two common approaches are used to deal with conflicts between different objectives (Benayoun et al. 1971). The first approach is to define a hierarchy for objectives that represents an order of achievement according to the project goals. It regulates that the second objective can only be reached with the precondition that the first one has been reached, and so on. The second approach uses "utility" in which each objective function is assigned a weight *w*, then a unique objective function $Z = \sum_j w_j c_j$ can be maximised or minimised.

In this conceptual model, the "utility" method is adopted, since the direction of projections can be determined by integrating weights in MLOP (Joro, Korhonen & Wallenius 1998). By determining weights, decision makers can directly or indirectly influence decision outcomes that reflect their preference structure. In addition, integrating weights in the evaluation also complies with the form of a weighted linear combination, representing trade-offs between different objectives (see Section 6.7.3).

Based on the above discussion, the problem of developing an optimal retrofitting strategy for getting the best environmental, economic and social performance within project constraints can be put in the form:

- The objective function has the form maximise $Z = \sum_{j=1}^{n} c_j x_j$ which indicates the expectation of better sustainability performance.
- All of the x_j where j = 1, 2, ..., n are binary values (can only have the value 0 or 1).
- Weight *w* for each sustainability dimension is combined into the objective function to represent a "trade-off" between the three sustainability dimensions (see discussion in Section 6.7.3).

If a specific solution region can be defined to satisfy a retrofitting goal, such as a retrofitting goal to achieve at least 30% energy saving, it should be formulated as a constraint function in the "subject to" bracket as: ∑_{j=1} a_jx_j ≥ 30% (where a_j represents the energy saving ability of each proposed retrofitting activity). In this way, it can ensure that the generated retrofitting strategy can reach this retrofitting goal.

By combining the above elements, the binary linear mathematical model for developing retrofitting strategies can be built as Equation 6.9.

 $\max Z = W_{en} \times \sum_{j=1}^{n} P_{j}^{en} x_{j} + W_{ec} \times \sum_{j=1}^{n} P_{j}^{ec} x_{j} + W_{s} \times \sum_{j=1}^{n} P_{j}^{s} x_{j}$ Equation 6.9. $s.t. \begin{cases} \sum_{j=1}^{n} a_{1j} x_{j} \ge (=, \le) b_{1} \\ \sum_{j=1}^{n} a_{2j} x_{j} \ge (=, \le) b_{2} \\ \vdots \\ \sum_{j=1}^{n} a_{mj} x_{j} \ge (=, \le) b_{m} \\ x_{j} = 0, 1 (j = 1, 2, ..., n) \end{cases}$

Where,

- Z the overall performance of a retrofitting strategy
- W_{en} the weighting score of the environmental dimension
- W_{ec} the weighting score of the economic dimension
- W_s the weighting score of the social dimension
- P_j^{en} the evaluation score of retrofitting activity *j* in the environmental dimension, which can be calculated based on Equation 6.8
- P_j^{ec} the evaluation score of retrofitting activity *j* in the economic dimension, which can be calculated based on Equation 6.8
- P_j^s the evaluation score of retrofitting activity *j* in the social dimension, which can be calculated based on Equation 6.8
- $\sum_{i=1} a_{mj} x_i \ge (=, \le) b_m$ the constraint function for subjecting to project constraint b_m
- n the number of proposed retrofitting activities
- m the number of project constraints
- $x_j = 0,1$ whether retrofitting activity x_j is selected. When $x_j = 0$, it is not selected. When $x_j = 1$, it is selected.

Based on Equation 6.9, the retrofitting activities that can be aggregated as a retrofitting strategy with the maximised overall evaluation score while satisfying project constraints can be identified (denoted as 1). In this way, the optimal retrofitting strategy within project constraints can be developed.

However, with multiple assessment dimensions, criteria and objectives involved, the calculation of Equation 6.9 is very complex and time-consuming. To solve this problem, computer programming, such as Excel and MATLAB, can be adopted to calculate the mathematical model.

6.9. Conduct sensitivity analysis

There is no doubt that most decisions are affected by uncertainties, which may be caused by measurement and conceptual errors, limited knowledge about process, simplification and data scarcity, different values and opinions, and the future (Geneletti 2019). Sensitivity analysis can be conducted to test how, and how much, those uncertainties affect the results (Saltelli et al. 2019). Sensitivity analysis can help develop better monitoring strategies and experiment designs by, for instance, highlighting the significance and amount of data to be collected (Douglas-Smith et al. 2020). In addition, by identifying parameters that may be "insensitive" or "inactive", sensitivity analysis can also help limit the parameter space to those affecting model outcomes only marginally or not at all (Douglas-Smith et al. 2020). In particular, in the context of building performance evaluation, sensitivity analysis can be used to identify significant and negligible parameters in a building energy model to indicate the possibility of simplifying, optimising, calibrating or correcting the model (Pang et al. 2020).

Many sensitivity analysis methods are available and can be selected according to the traits of the given problems and available computational resources (Pang et al. 2020). The common sensitivity analysis methods and their pros and cons are listed in Table 6.5. Based on the specific situation of the implemented model and the characteristics of different sensitivity analysis methods, a suitable one can be selected referring to Table 6.5 and Figure 6.4.

Method	Pros	Cons
Local	• Easy to understand and implement	• Not suitable when the model is nonlinear or
	Computationally efficient	nonmonotonic
		• Parameters cannot be varied simultaneously
		Does not support uncertainty analysis
Morris	Computationally efficient and robust	• Not the most robust method
	 Produces results comparable to 	• Not able to quantify the individual interaction
	computationally expensive methods like the	between two parameters
	Sobol method and the FAST method	Does not support self-verification
PEAR,	 Easy to understand and implement 	• Not robust
SRC	Computationally efficient	• Only suitable for linear and monotonic
		models
PCC	Easy to understand and implement	• Not robust
	Computationally efficient	• Only suitable for linear and monotonic
	Parameter interaction is eliminated (PCC)	models
SPEA,	• Easy to understand and implement	• Not robust
SRRC	Computationally efficient	• Only applied to monotonic models
	Suitable for nonmonotonic models	
PRCC	• Easy to understand and implement	• Not robust
	Computationally efficient	• Only applied to monotonic models
	Suitable for nonmonotonic models	
	Parameter interaction is eliminated	
Sobol	Most robust	Lower computational efficiency compared
	• The results often serve as a benchmark for the	with the GAST/e-FAST method
	testing of other sensitivity analysis methods	
	Provide high-order interactions	
FAST	• Model-independent	• Computationally complex for a large number
	• The variance of the outputs can be	of inputs
	apportioned to the inputs	Cannot address high-order interactions (e-
	• Computationally efficient compared with	FAST can solve this problem)
	Sobol	• The result is not as robust as that of Sobol
Metamodel	• A potentially computationally intensive	• The calculation to generate a metamodel can
	model can be simplified to a mathematical	be resource intensive
	model	• The valid domain and applicability of the
	• The coefficients of the model may indicate	model are highly dependent on the training
	sensitivities directly	data
RSA	Computationally efficient	• Cannot quantify the interactions among input
-		parameters
Lasso	• Computationally efficient	Cannot generate robust results
	• Can be used as a screening technique	

Table 6.5. Pros and cons of common sensitivity analysis methods

Source: Pang et al. 2020



Figure 6.4. Decision diagram to guide the selection of sensitivity analysis methods

Source: de Rocquigny, Devictor & Tarantola 2008; Pang et al. 2020

Based on the discussion in Section 6.8, the established mathematical model in this study is a linear model. Moreover, by using the normalisation method, maximum/minimum ratio, the direction of assessment is the same: the bigger the estimation score, the better the performance of potential retrofitting activities. Therefore, the model is monotonic. According to Figure 6.4, the sensitivity analysis method, local differential or regression/correlation, may be selected for a linear, monotonic model, but local differential can bring a more efficient analysis process. Therefore, local differential is adopted in this study for the sensitivity analysis.

The local sensitivity analysis method often computes the derivative of the model at a given point or the average derivative of several points in the parameter space to assess the significance of an input parameter (Douglas-Smith et al. 2020; Pang et al. 2020). By using the local sensitivity analysis method, only one parameter is changed at a time while other input parameters are fixed; thus, it is also known as One-At-a-Time (OAT) (Pang et al. 2020; Saltelli et al. 2019). There are various methods for determining how much a parameter should be perturbed, but the most frequently used one is to employ a proportional increment (Razavi & Gupta 2015). For example, a parameter might be increased or decreased by 10% of its nominal value up to and including a specified bound. Ding (2005) developed a multi-criteria approach for measuring the performance of sustainability, and three different discount rates, 5%, 10%, and 15%, were used for calculating the NPV of three operations as a sensitivity analysis of the study. The result showed that with a discount rate of 5%, all three options are acceptable, while all of them are unacceptable with a discount rate of 15%. Only one option is acceptable with the 10% discount rate.

As discussed in Section 4.3 of Chapter 4, uncertainty in this study is mainly from the selection of discount rate, study period and coefficient. Therefore, sensitivity analysis can be conducted by replacing the values with other possibilities to test the stability of the results. Based on the results of the sensitivity analysis, if any component should be refined, the process should be repeated from the modified step. Otherwise, the process can proceed to the next step.

6.10. Choose the "best" solution

By following the previous seven steps, an optimal retrofitting strategy for best improving environmental, economic and social performance within the current project constraints can be developed. However, decision makers may prefer more options from which the one that "best" satisfies their demands can be selected. It can be realised by relaxing or tightening one or several project constraints so that the optimal retrofitting strategies regarding the changed project constraints can be obtained accordingly. For example, the retrofitting goal is to have at least a 30% energy saving. Following the above seven steps, an optimal retrofitting strategy is developed, which can save 32% of energy. However, by relaxing the project constraint, for example, increasing the budget by 5%, the retrofitting strategy developed with the relaxed project constraint can have a 50% energy saving. Then, the building owners or investors can decide if they would like to pay 5% more to get an additional 18% in energy saving.

With several retrofitting strategies developed, the retrofitting team is responsible for explaining to building owners or investors how the strategies meet retrofitting goals and objectives with different project constraints. They should also provide suggestions on the "best" solution based on their expertise and experience. With the given suggestions and uncertainty involved, the final decision makers, usually building owners or investors, can simply select the one that can primarily meet their requirements as the "best" solution.

6.11. Summary

This chapter developed a conceptual model for deciding retrofitting strategies by considering essential elements in MODM and the required steps for developing retrofitting strategies. There are eight steps in the conceptual model, which can be followed to develop an optimal retrofitting strategy that can maximise improvement on environmental, economic and social dimensions within project constraints.

Following the illustration of the conceptual model, each step included is elaborated. The first step is to set retrofitting goals and objectives based on communication with different stakeholders and consideration of project constraints. Then, the retrofitting team needs to conduct a building audit to identify the problems the building has. The third step is to propose retrofitting activities that may solve the identified problems.

The fourth step is to develop assessment criteria for measuring the achievement of retrofitting goals and considering the three pillars of sustainability. With developed assessment criteria, the retrofitting team can evaluate the ability of proposed retrofitting activities to reach retrofitting goals. Moreover, the environmental, economic and social impacts of these activities can also be measured. In this step, the measurement of retrofitting activities' performance is normalised to unitless so that they can be aggregated to one measure to show overall performance. Meanwhile, a weighting system by AHP is combined to represent different importance levels among assessment criteria and the three sustainability dimensions.

With the evaluation results of the performance of potential retrofitting activities, retrofitting strategies can be developed based on an MLOP mathematical model established in Step 6. The developed strategy has the highest overall score, indicating the maximised improvement in environmental, economic and social dimensions within project constraints. The next step is to conduct a sensitivity analysis to test how, and how much, uncertainty affects the results. Based on the results of the sensitivity analysis, if no components need to be refined, the decision process can go to the final step – the "best" solution is chosen by building owners or investors.

Since the developed model is a conceptual model, which is intended to be general, the framework of the model can be adapted for any retrofitting project. Therefore, the conceptual model needs to be converted to an operating model before applying it to a real case. The next chapter explains how the conceptual model can be converted to an operating model, and the process is illustrated in a case study.

Chapter 7. Case study – Data collection

7.1. Introduction

The last chapter developed a conceptual model for deciding retrofitting strategies for office buildings. A conceptual model is general and can be adapted to suit the specific situation of any retrofitting project. Therefore, the conceptual model needs to be converted to an operating model before applying it to an actual retrofitting project. To validate the model for developing retrofitting strategies, a case study, which is retrofitting an office building in Hangzhou city, Zhejiang province, China, is conducted. The case study has two parts. The first part collects data about suitable retrofitting activities and assessment criteria in China, which can be used to convert the conceptual model to an operating model for the Chinese retrofitting market. The second part quantifies the operating model by applying it to develop retrofitting strategies for the case building, illustrated in the next chapter.

This chapter focuses on the first part of the case study. First, the rationale behind the case study is discussed (Section 7.2). The background information about the case building is presented (Section 7.3). Then, a two-stage process of data collection for the case study is demonstrated (Section 7.4 and Section 7.5). The first stage is to collect broad information about sustainable retrofitting in China by a survey with professionals and stakeholders of retrofitting. In stage two, focus group discussions are conducted with same types of participants of the survey to consolidate the result of the survey and further modify retrofitting activities and assessment criteria to be suitable for use in the case study and other retrofitting projects in China. Focus group discussions are also conducted to estimate weights of the three sustainability dimensions and assessment criteria.

7.2. Rationale behind the case study

A case study was selected to demonstrate the process of using the conceptual model to develop retrofitting strategies for an actual office building. When using a case study approach to describe steps undertaken, it allows the researcher to break down a complex and broad topic, or phenomenon, into manageable research questions (Heale & Twycross 2018). For example, by conducting the case study, the process from modifying retrofitting activities and assessment

criteria to deal with location variations, to developing retrofitting strategies for the case building by taking steps in the conceptual model (Figure 6.2 in Chapter 6), can be illustrated, the methods to assess environmental, economic and social performance of retrofitting activities can be described, and the way to develop retrofitting strategies by balancing the three sustainability dimensions can be explained.

A single case study or a multiple case study can be conducted to reach an illustrative or verifiable conclusion (Gustafsson 2017). A multiple case study is usually conducted to understand the differences and similarities between different cases (Gustafsson 2017; Heale & Twycross 2018), while a single case study is an ideal option if the researcher only wants to look at one aspect (for instance, one person from a particular group) or one group (for example, a group of people) (Yin 2009). In this study, a single case study was conducted to illustrate the process of using the conceptual model to develop retrofitting strategies for an office building. There is no doubt that conducting more case studies can increase the validity of the developed conceptual model. The validity in this study means that by using the developed model, retrofitting strategies can be developed for an existing office building, and the potential improvement from the developed retrofitting strategies can be estimated, and this can be done with a single case study. The case study is also conducted to illustrate how to use the conceptual model to develop retrofitting strategies for an existing office building by considering the environmental, economic and social dimensions. It is a complex process, so it is important to provide a clear and detailed presentation of the process, such as collecting both quantitative and qualitative data, assessing performance of retrofitting activities in the three sustainability dimensions, and establishing the mathematical model and running it in a program for developing retrofitting strategies. Studying only one case allows the researcher to deliver a more careful study, contributing to a deeper understanding of the subject (Gustafsson 2017) improving the environmental, economic and social performance of existing office buildings by retrofitting.

7.3. Background information for the case study

The case building is in Hangzhou city, Zhejiang province, China. Based on Zhejiang Statistical Yearbook (ZPBS & SOTNBS 2021), the average temperature in Hangzhou is 28 °C in summer and 8 °C in winter in 2021. The average relative humidity ranges from 76% to 81%. To

maintain a comfortable indoor environment, buildings in Hangzhou city often have high operation costs, energy consumption, and carbon emissions.

Zhejiang province is ranked fourth in GDP in the country, and has a relatively fast development and acceptance of green buildings (Ge et al. 2018). However, based on the national green building standard, Assessment Standard for Green Building (ASGB) (BG/T 50378-2014), only 276 buildings were granted green building labels in Zhejiang province in 2016. The national standard regulates green buildings at two stages: the design label at the building construction stage, and the operation label when the building is in use (MOHURD 2014). Of the 276 green buildings, only 20 have been given operation labels. Moreover, only 79 of the labelled projects can reach the requirements of operation management due to insufficient supervision by building owners and facility managers (Shen, Zhao & Ge 2020).

The case building is an office building located in the centre of Hangzhou city. The building was built in 1985 and is a relatively old office building in Hangzhou city. The building's total floor area (the sum of the horizontal projected areas of each floor of the building) is about 19,683 m², and it has a land area of 2,386 m². The building has 13 storeys, and no basement. In China, public buildings are designed for people for a variety of public activities, and an office building is one type of public building (Wei & He 2017). Characteristics of the case building are summarised in Table 7.1.

Age	37 years					
Total floor area	19,683 m ²					
No. of storeys	13					
Height	42.7 m					
Category of public buildings ¹	Level II					
Shape factor ²	0.22					
Structure type	Frame-shear wall structure					
Glazing type	Single glazing					
Area ratio of window to wall	East: 0.19					
	West: 0.26					
	South: 0.29					
	North: 0.20					
Average annual energy consumption	1,836,262 kWh					
Average annual energy consumption per floor area	93.29 kWh/m ²					
Average annual CO ₂ emissions	1,830,753 kg CO ₂					
Average annual CO ₂ emissions per floor area	93.01 kg CO ₂ /m ²					
Average annual water consumption	23,970 kL					
Average water consumption per floor area	1.22 kL/m^2					
Note:						
1. Public buildings in Zhejiang Province can be grouped into three categories based on the local						
standard, Design Standard for Energy Efficiency of Public Buildings (UAD, ZIAD & CMA 2007):						
• Category I: A public building with a building area greater than or equal to 20,000 m ² , or with a full						
air conditioning system fully equipped.						

• Category II: A public building with a building area of less than 20,000 m² and without or partially equipped with air conditioning systems.

• Category III: A public building where the building is out of use during the summer and winter months of the year when the cooling and heating loads are at their peak, and where no air conditioning is provided.

2. Shape factor refers to the ratio of the exterior area of a building in contact with the outdoor atmosphere to the volume it encloses.

Hangzhou is in the hot-summer-cool-winter climate zone where air conditioners or heaters are the heating supply. A reverse cycle air conditioning system was installed five years ago for cooling in summer and heating in winter, and the HVAC system can cover at least two-thirds of the area of the building, with the remaining one-third naturally ventilated. Before that, fans and heaters were used to adjust indoor temperature. Usually, in this climate zone, the cooling period (summer) is from June to September and the heating period (winter) is from December to February. The indoor environment is good enough during spring and autumn, and no intervention (HVAC or fans) is needed. Since the building was built, some upgrades have been undertaken from time to time when the building owner realised there was a problem or due to demand by tenants. Upgrades include the newly installed reverse cycle air conditioning system and changing some steel window frames to PVC frames. No planned and comprehensive retrofitting has ever been conducted on the building.

7.4. Data collection strategy for the case study

The case study is conducted to demonstrate how to use the conceptual model to develop retrofitting strategies for an office building. The information in the conceptual model is intended to be general, which may not be applicable to the case building. To deal with locational variations, the conceptual model needs to be converted to an operating model by modifying retrofitting activities and assessment criteria to suit the local situation. As discussed in Section 5.3.2.1 of Chapter 5, many data collection methods are available to gather people's opinions, such as surveys, interviews and focus groups. In this case study, an explanatory sequential strategy is adopted to collect data in two stages from the broad to the specific.

In stage one, a survey was conducted to collect opinions about retrofitting activities and assessment criteria of sustainable retrofitting from professionals and key stakeholders of retrofitting in northern and southern China. The survey can help gather opinions about sustainable retrofitting in China in a relatively short time. However, the collected information tends to be broad and general. In stage two, more detailed data is collected to consolidate the results of the survey and further modify the retrofitting activities and assessment criteria to be suitable for local use. In this case study, three focus group discussions were conducted with local professionals. Focus group discussion is a suitable method when the purpose of data collection is to build a holistic understanding of the problem situation based on participants' comments and experiences (Bhattacherjee 2012). Hearing others speak often prompts responses or ideas that participants have not considered previously, which provides chances for an in-depth examination of complex issues (Bhattacherjee 2012), such as how to balance the improvement in the three sustainability dimensions, how important different stakeholders consider each sustainability dimension to be and why, etc. However, the findings may not be generalised to other settings because of the small sample size (Bhattacherjee 2012). Therefore, if using the conceptual model in other locations, focus group discussion or other forms of data collection need to be conducted with local professionals to make sure the model is modified to comply with the local situation. The details of each adopted method in this case study are discussed in the following sections.

7.4.1. Stage 1 – Questionnaire survey

In the first stage of data collection for the case study, a questionnaire survey was conducted to collect general opinions on suitable retrofitting activities and assessment criteria for retrofitting office buildings in China. Opinions about the development of sustainable buildings and the trend of retrofitting were also collected.

7.4.1.1. Development of the questionnaire survey

The questionnaire was designed in four parts with 17 questions based on the literature review. Both closed and open-ended questions were created to gather respondents' attitudes and opinions about sustainable retrofitting and their working and life experiences about retrofitting. The four parts of the questionnaire are explained below.

Part I: General information

This part had seven questions on some basic information about respondents like gender, age and experience of retrofitting. Different roles in retrofitting projects have different requirements, and respondents' experience and background can influence their answers. Therefore, respondents' background information can be used to analyse possible reasons for the provided answers. Other questions included how many retrofitting projects respondents have ever undertaken, and what kinds of buildings they have ever retrofitted to know whether they have experience in retrofitting office buildings.

Part II: Understanding about sustainable retrofitting

This part had four questions to capture the general understanding of sustainable retrofitting. Based on respondents' retrofitting-related experience, they were asked to give their opinions on "what" are common retrofitting activities and assessment tools that they ever used, "who" is responsible for improving sustainability, and "what" are drivers of retrofitting of buildings.

Part III: Level of importance of different aspects of sustainability

This part had three questions where respondents needed to give rating scores on different aspects of sustainability and the identified assessment criteria under environmental and social dimensions. Apart from the identified criteria in the survey, respondents were also able to provide other criteria that they think should be included in this study by asking about "others".

Part IV: Further discussion about improvement of sustainable retrofitting

This part had three open-ended questions for further discussion on "what" can be improved by sustainable retrofitting and "how" to improve the sustainable performance of existing buildings. Even though the survey always allowed respondents to give their own opinions by asking "other" after most of the questions, respondents could still share more opinions in this part to help analyse the provided data and explain the results. In addition, the last question asked what other actions can be taken to improve the performance of existing buildings which can be used to estimate future trends of retrofitting.

7.4.1.2. Pilot study

Before the main survey, a pilot study took place in December 2018 for about two weeks with three Australians and two Chinese professionals whose study fields are relevant to the built environment. The purpose of the pilot study was to verify whether the content was clear and easy to understand, whether respondents could answer questions in a proper way, and how long it would take to complete the survey. Based on their feedback in the pilot study, the questionnaire form was revised and refined until it was ready to send out for the main survey.

Respondents' feedback showed that it took about 10 to 15 minutes to finish the survey. Generally, the questions were clear and easy to understand, but there were still some vague expressions, including Question 3 (see Appendix B-2), it should be "property management industry" instead of "industry of building construction"; and the question "what do you expect to be improved by sustainable retrofitting" should be asked of professionals, building owners and tenants separately to identify gaps between provided services and requirements. Therefore, this question was split into two questions, Question 15 for building owners and tenants and Question 16 for professionals. After revising the survey based on these professionals' comments, the survey was finalised (see Appendix B-2).

The survey was designed to be conducted with professionals and key stakeholders of retrofitting in China, so the finalised questionnaire form was translated into Chinese. The translated questionnaire form was sent to the same two Chinese professionals who participated in the pilot study to check whether the translation distorted any questions. Their feedback showed that the meaning of the translated questionnaire form was consistent with the English version. The English version is in Appendix B-2 and the Chinese version is in Appendix B-3.

7.4.1.3. Sampling method and participant recruitment

Two sampling methods can be used to recruit respondents to a study: probability sampling and non-probability sampling methods (Sedgwick 2013). In the probability sampling method, each object has an equal chance to be selected as a sample, and researchers' opinions do not influence who is selected in this method. If the sample size is big enough, it is likely to be representative of the whole population so that findings generated based on the analysis of the sample can be considered representative of the whole population (Naderifar, Goli & Ghaljaie 2017). The common probability sampling techniques are simple random, stratified random, random cluster, and systematic sampling. They are often used to collect data in quantitative research (Naderifar, Goli & Ghaljaie 2017).

Non-probability sampling methods are usually used in qualitative research (Naderifar, Goli & Ghaljaie 2017). In contrast to probability sampling, not everyone has an equal chance of being selected as the sample in these sampling methods. Usually, samples are selected because they are available to researchers. In this sampling method, whether the selected samples can represent the whole population is unclear. As a result, the sampling rate in error cannot be calculated (Naderifar, Goli & Ghaljaie 2017). Common non-probability sampling methods include convenience, purposeful and quota sampling.

One method of convenience sampling is called snowball sampling as the sample size gets bigger over this sampling method, just like rolling a snowball (Cohen & Arieli 2011). Snowball sampling is a respondent-driven sampling method, also called chain-referred or link-tracing sampling (von der Fehr, Sølberg & Bruun 2018). By using snowball sampling, researchers typically begin with a small number of initial contacts who meet the research criteria and are invited to participate in the study. The respondents are then asked to recommend other contacts

who meet the research criteria and may be willing respondents as well, who in turn recommend other potential respondents, and so on (Parker, Scott & Geddes 2019). It allows researchers to reach a population that is difficult to sample when using probability methods (Naderifar, Goli & Ghaljaie 2017).

Snowball sampling is believed to be simple and cost-efficient with little planning and fewer resources needed (Dudovskiy 2016). This sampling method also allows researchers to communicate better with respondents because subsequent respondents are acquaintances of the first participant who is linked with researchers (Naderifar, Goli & Ghaljaie 2017). The sampling process often ends when a research saturation point is reached, such as until no more significant information can be offered from further sampling (Geddes, Parker & Scott 2018). However, there are some disadvantages to using this sampling method. Geddes, Parker and Scott (2018) stated that researchers have little control over the snowball sampling method since the subsequent selections mainly rely on previous contacts. Therefore, the selection of initial contacts is very important for guaranteeing the sample quality. Moreover, there may be sampling bias generated by snowball sampling. Referrals by previous respondents are normally to people they know well and they may be from the same group and share similar opinions.

There are three patterns of snowball sampling (Dudovskiy 2016):

- Linear snowball sampling: The sampling process starts with one subject, and the subject provides only one referral. This pattern is continued until the sample group is fully formed.
- Exponential non-discriminative snowball sampling: Multiple referrals are provided by the first subject recruited. This pattern is continued until sufficient primary data from the identified samples is collected.
- Exponential discriminative snowball sampling: Subjects give multiple referrals, but only one new subject can be recruited each time. The determination of a new subject is guided by the aim and objectives of the study.

This case study adopted the exponential non-discriminative snowball sampling method to quickly collect opinions about suitable retrofitting activities and assessment criteria in northern and southern China. China is divided into the north and the south by the Qinling Mountains–

Huaihe River Line, which is also a dividing line between warm temperate and subtropical climate zones in China (Xu et al. 2021). Therefore, the north and the south in China have different natural conditions and sociocultural customs. For example, due to historical energy scarcity and relevant heating standards, central heating is only available in the north for varying lengths of time. In the south, no district heating substations and long-distance heating pipelines have ever been provided (Yan et al. 2019). As a result of economic development over the past decades people who do not have access to central heating have begun to use air conditioners or heaters. Li et al. (2018) found that, apart from office buildings built before the 1980s that have different cooling and heating facilities such as fans and central heating supply, office buildings built after the 1980s are air conditioned across the whole country. It means that office buildings are similar in their heating and cooling source. In the north, HVAC is mainly used as a heating source for office buildings, while it is mainly used to cool down the indoor temperature in summer for office buildings in the south (Wang et al. 2020). Therefore, even though China is a big country with various climate conditions, the office buildings across the country have a similar way of operating and it is a good opportunity to test the flexibility of the conceptual model.

To understand the opinions of different roles in retrofitting projects, six groups of stakeholders were included in the survey: architects, engineers, project managers, facility managers, building owners, and tenants. To avoid the bias mentioned above, instead of one initial subject recruited, 24 initial respondents who were suggested to be active in the industry by scholars from two universities in China¹, and they were recruited in northern China (12 respondents) and southern China (12 respondents).

7.4.1.4. Survey process

The survey was conducted from January to March 2019 via the online survey tool SurveyStar. The survey link to the final survey was sent to the 24 initial respondents via WeChat, a commonly used communication tool in China. The respondents were invited to complete the questionnaire form and send the link to other people they thought were eligible for this study. A reminder was sent to them once a month for three months to make sure they had received the survey link and to remind them to send the link to other potential respondents. The information sheet and consent form were attached to th1e questionnaire form (the details are in Appendix B-1). Any referrals could also read and determine for themselves whether they were eligible for this study.

The online survey can set whether a question is compulsory so that only completed surveys can be successfully submitted. Therefore, it is not possible to receive a form with missing answers. The only reason for invalid submission is that the respondent does not fit in any of the six roles identified in the questionnaire. By screening the role of respondents, 13 of the collected responses were not valid. Therefore, the final sample size of this study was 128.

Since the snowball sampling method was used, it is not known whether the sample size was representative. To consolidate the outcome of the questionnaire survey and further specify the model to be operational for the case building and other office buildings in China, three focus group discussions were organised with professionals in Hangzhou, as discussed in Section 7.3.2.

7.4.1.5. Survey analysis

a) Background of respondents

The background characteristics of respondents, including gender, age range, location and experience about building retrofitting, were investigated first since respondents' backgrounds can help explain the survey results. Table 7.2 shows, among all respondents, 66 were male and 62 were female. The role group of engineers was the largest group of respondents, representing 21.1% of all respondents. The number of respondents in each of the other five role groups was similar, at around 15% in each group. Generally, the distribution of role groups and genders can be considered balanced.

Roles	Male	Female	Sub-total	% of total
Architects	9	10	19	14.8
Engineers	13	14	27	21.1
Facility managers	13	7	20	15.6
Project managers	17	5	22	17.2
Owners	5	14	19	14.8
Tenants	9	12	21	16.5
Total	66	62	128	100

Table 7.2. Respondents in different roles by gender (128 responses)

The survey provided five age groups to be chosen by respondents. Table 7.3 shows most respondents were in the age group 26 to 35 years old (51 or 39.8%) or the age group 46 to 55 (38 or 29.7%). Very few respondents (6 or 4.7%) were over 55 years old. A similar age distribution can be found in each role group.

	Number of respondents						Sub	0/
	Architects	Engineers	Facility	Project	Owners	Tenants	Sub- total	% of total
			managers	managers				
Age								
≤ 25	3	3	2	2	3	4	17	13.3
26–35	9	10	9	12	5	6	51	39.8
36–45	4	1	2	4	3	2	16	12.5
46–55	3	12	6	2	6	9	38	29.7
≥ 56	0	1	1	2	2	0	6	4.7
Total	19	27	20	22	19	21	128	100
Region								
North	9	15	9	13	14	12	72	56.2
South	10	12	11	9	5	9	56	43.8
Total	19	27	20	22	19	21	128	100

Table 7.3. Characteristics of respondents (128 responses)

Table 7.3 also shows the regions respondents were from. Over half (56.2%) of respondents were from northern China, and the rest (43.8%) were from southern China. By different roles, 14 of the 19 owners were from the north, and only 5 were from the south, but in other role groups, the location distribution was approximately even.

To understand respondents' experience with building retrofitting, three questions were asked: how long they have been working in the field of property management, how often they conduct retrofitting projects, and how many retrofitting projects they have conducted in the past three years. Table 7.4 shows that 53.9% of the respondents have worked in the field for 5 years or less, 15.6% have worked in the field for 6 to 10 years, and the rest (30.5%) of them have worked in the field for over 10 years. It can be stated that respondents of the survey can represent the group of professionals with working experience in the property management industry.

	Number of respondents						Sub	0/	
	Architects	Engineers	Facility	Project	Owners	Tenants	SUD- total	% 01 total	
			managers	managers			totai	totai	
Working years in the field of property management									
≤ 5	11	9	12	10	11	16	69	53.9	
6–10	2	7	2	6	2	1	20	15.6	
11-15	2	3	3	2	0	0	10	7.8	
16–20	3	4	3	1	2	0	13	10.2	
≥ 21	1	4	0	3	4	4	16	12.5	
Total	19	27	20	22	19	21	128	100	
Frequency of	of conducting	retrofitting	projects						
Always	1	1	1	0	1	0	4	3.1	
Very often	1	3	4	3	0	1	12	9.4	
Sometimes	11	11	7	13	7	5	54	42.2	
Rarely	4	7	6	4	8	9	38	29.7	
Never	2	5	2	2	3	6	20	15.6	
Total	19	27	20	22	19	21	128	100	
Numbers of	retrofitting p	orojects cond	ucted in the	past 3 years					
0	2	7	5	5	7	8	34	26.6	
1–3	10	18	11	14	11	12	76	59.4	
4–7	5	2	1	2	0	0	10	7.8	
8-11	1	0	3	0	0	0	4	3	
12–14	1	0	0	0	0	1	2	1.6	
≥ 15	0	0	0	1	1	0	2	1.6	
Total	19	27	20	22	19	21	128	100	

Table 7.4. Related experience of respondents (128 responses)

Based on Table 7.4, most respondents were sometimes (42.2%) or rarely (29.7%) involved in a retrofitting project. In the past three years, 59.4% of respondents had conducted 1 to 3 retrofitting projects, and 26.6% had not undertaken a retrofitting project. Only 14% of respondents were involved in more than three retrofitting projects in the past three years. The low frequency of conducting retrofitting projects reflected that, in China, retrofitting has not been commonly considered a solution for improving the sustainable performance of existing buildings. New construction is still attracting more attention than existing buildings in China. Based on Huo et al. (2019) and Liu et al. (2019), from 2000 to 2015, about 28.4 billion m² of new construction was added in China, and China is still witnessing a dramatic increase in new construction. Therefore, it is urgent to improve the performance of existing buildings in China to maintain an average of sound performance for the whole building sector.

The type of buildings the respondents had ever retrofitted included residential buildings, office buildings, industrial buildings, shopping malls, school buildings and hospitals, as Table 7.5 shows. Residential buildings were the most common type of building being retrofitted (63.3%),

followed by office buildings (34.4%). Hospital buildings were seldom retrofitted by respondents (3.1%). A similar situation can be found in each role group.

	% of the role group						
Building type	Architects	Engineers	Facility managers	Project managers	Owners	Tenants	total
Residential	68.4	51.9	70	72.7	63.2	57.1	63.3
Office	31.6	33.3	25	54.5	31.6	18.6	34.4
Industry	21.1	14.8	5	18.2	0	9.5	11.7
Shopping mall	21.1	11.1	20	22.7	5.3	9.5	14.8
School building	15.8	18.5	0	9.1	21.1	14.3	13.3
Hospital	0	7.4	0	9.1	0	0	3.1

Table 7.5. Type of buildings ever retrofitted by respondents (128 responses, more than one choice possible)

b) Development of sustainable retrofitting in China

Four questions were asked to understand the development of sustainable retrofitting in China. First, respondents were asked to choose retrofitting activities that they had ever used from provided options. More than one retrofitting activity could be chosen. Apart from the provided options, if there were others they had ever used, they could also add them at "others". The frequency of chosen retrofitting activities is illustrated in Figure 7.1. The three most undertaken retrofitting activities were A6 – Upgrade lifts to more energy efficient ones (56.3%), A1 – Install/upgrade insulation of building envelopes (51.6%) and A9 – Install PV panel (43.8%). The three least common activities were A4 – Install sun shading devices (14.8%), A5 – Install BMCS (13.3%) and A10 – Install water control sensors (11.7%).


Figure 7.1. Frequency of commonly undertaken retrofitting activities by respondents (128 respondents, more than one activity possible)

A1. Install/upgrade insulation of building envelopes

A2. Adopt extensive green roof

A3. Change windows to energy efficient windows

A4. Install sun shading devices

A5. Install BMCS

A6. Upgrade lifts to more energy efficient ones

A7. Upgrade HVAC system
A8. Upgrade lighting system
A9. Install PV panel
A10. Install water control sensors
A11. Replace existing water fixtures with water efficient ones
A12. Install a water treatment system

As discussed in the beginning of Section 7.3.1.3, the geographical features, natural conditions and people's living behaviour in northern and southern China are different. Therefore, the Chisquare test was used to test whether regions affect the choice of common retrofitting activities, with a 95% confidence level. If the *p*-value (asymptotic significance) is less than 0.05, the region of respondents is associated with the type of retrofitting activities; otherwise, they are not associated. The results are summarised in Table 7.6, and the SPSS results are in Appendix B-4. Five of the total activities are associated with the region of respondents: A1 – Install/upgrade insulation of building envelopes, A4 – Install sun shading devices, A7 – Upgrade HVAC system, A8 – Upgrade lighting system, and A12 – Install water treatment system.

Retrofitting activities		Chi-Square test	Result
A1. Install/upgrade insulation of building envelopes	p<0.05	$\chi(1) = 10.012, p=0.002$	Associated
A2. Adopt extensive green roof	p>0.05	$\chi(2) = 1.329, p=0.514$	Not associated
A3. Change windows to energy efficient windows	p>0.05	$\chi(1) = 1.405, p=0.236$	Not associated
A4. Install sun shading devices	p<0.05	$\chi(1) = 11.232, p=0.001$	Associated
A5. Install BMCS	p>0.05	$\chi(1) = 1.810, p=0.179$	Not associated
A6. Upgrade lifts to more energy efficient ones	p>0.05	$\chi(1) = 1.580, p=0.209$	Not associated
A7. Upgrade HVAC system	p<0.05	$\chi(1) = 11.997, p=0.001$	Associated
A8. Upgrade lighting system to improve energy efficiency	p<0.05	$\chi(1) = 5.003$, p=0.025	Associated
A9. Install PV panel	p>0.05	$\chi(1) = 2.612, p=0.106$	Not associated
A10. Install water control system	p>0.05	$\chi(1) = 0.059, p=0.859$	Not associated
A11. Replace existing water fixtures with more water efficient ones	p>0.05	$\chi(1) = 0.431, p=0.512$	Not associated
A12. Install a water retreatment system	P<0.05	$\chi(1) = 4.195, p=0.041$	Associated

 Table 7.6. Chi-Square test between commonly used retrofitting activities and region (128 responses)

As discussed in Section 7.4.1.3, northern China has a district heating supply for its cold and dry winter, and this supply is not provided in the southern region. Therefore, HVAC is important for people living in the south to adjust the indoor temperature, while the insulation performance of building envelopes is important for buildings in the north to reduce heat loss in winter and gain in summer. Therefore, more respondents from northern China have undertaken the retrofitting activity of upgrading the insulation of building envelopes.

The survey showed that about 50% of respondents from the south have ever undertaken the retrofitting activity A8 – Upgrade lighting system, but only 31% of those from the north have used it. With sufficient natural light in southern China, sun shading devices are essential if no window film adheres to glazing. Therefore only 6% of northern respondents had undertaken activity A4 – Add sun shading devices, which is much lower than the southern respondents. Due to the greater precipitation in southern China, the retrofitting activity A12– Install water treatment system was more likely to have been undertaken by respondents from southern China than those from northern China.

Based on the above discussion, each of the retrofitting activities offered had been used at one time or another by some respondents. However, due to location variations, some of them were more likely to be used in the north, such as improving insulation of building envelopes; and some of them were more likely to be used in the south, such as upgrading HVAC, upgrading the lighting system, installing sun shading devices, and installing a water treatment system. The operating model is intended to be suitable for use across China. Therefore, all these retrofitting activities should be included, and suitable ones can be selected when using the operating model to develop retrofitting strategies for the case building.

To understand what sustainability assessment tools are commonly used in China, the survey provided options for respondents, including popular tools used both widely in China (ASGB) and across the world (Green Star and NABERS in Australia, LEED in the US, and BREEAM in the UK). The frequency of selections by respondents is illustrated in Figure 7.2. It shows that 33% of respondents have used ASGB to assess the sustainable performance of existing buildings, mainly because it works as a design guide established by the Chinese government. However, the focus of the ASGB assessment tool is on new buildings instead of existing buildings. Moreover, it is a self-assessment system, in which assessment results may be less accurate and reliable.





Other assessment tools were seldom used by respondents as less than 10% of respondents had ever used any of them. Moreover, over half of respondents (55%) indicated that they have never

used any assessment tools to assess the performance of existing buildings. It can be considered that using assessment tools is not a major measure to improve the sustainability performance of existing buildings in China.

For the question on who should be responsible for improving sustainability, four options were provided in the survey, and more than one option could be chosen. The result in Table 7.7 shows that respondents believed the government should mainly be responsible for improving sustainability. In total, 68.8% of respondents believed that government should take responsibility by establishing relevant regulations. 65.6% thought that government and industry should improve sustainability jointly, and 58.6% thought that government should take responsibility by using financial incentives. In contrast, only 25.8% of respondents said individual owners and facility managers should take responsibility.

Table 7.7. The parties who should take responsibility for improving sustainability (128respondents, more than one choice possible)

	% in each role group						
Responsibility	Architects	Engineers	Facility managers	Project managers	Owners	Tenants	76 01 total
Government through regulation	73.7	63	90	63.6	63.2	61.9	68.8
Government through financial incentives	47.4	59.3	55	68.2	47.4	71.4	58.6
Government and industry jointly	78.9	78.9	75	59.1	63.2	66.7	65.6
Individual owners and managers	36.8	42.1	25	27.3	21.1	14.3	25.8

Based on Table 7.7, an interesting phenomenon is observed. In most role groups, the options of government through regulation, and government and industry jointly, received the most support for taking responsibility for improving sustainability. However, the option that government should take responsibility through financial incentives received the most support from project managers and tenants. Moreover, even though most role groups did not consider that individual owners and managers are responsible for improving sustainability, 42.1% of engineers still believed owners and managers should take responsibility for promoting sustainability. It indicates that the understanding and demands of sustainable buildings vary across stakeholders.

The last question in this part was about the drivers of sustainable retrofitting. Respondents could choose more than one from the provided options. The responses by different groups of roles are illustrated in Table 7.8. More than half of engineers (56%), facility managers (75%), project managers (59.1%), owners (52.6%) and tenants (66.7%) believed that tenants' requirements were the driver, while 75% of facility managers, 86.4% of project managers, 52.6% of architects, and 59.1% of project managers believed that market requests were the driver. Overall, the main drivers of sustainable retrofitting were believed to be tenants' requirements and market requests, indicating that sustainable retrofitting would be more desired and promoted if tenants and the professionals in the market could better understand the benefits of sustainable retrofitting.

 Table 7.8. Drivers of sustainable retrofitting by role group (128 respondents, more than one choice possible)

	% in each role group						
Drive	Architects	Engineers	Facility managers	Project managers	Owners	Tenants	76 of total
Tenants' requirement	36.8	55.6	75	59.1	52.6	66.7	57.5
Government	52.6	48.1	45	45.5	42.1	33.3	44.5
Economic payback	47.4	37	55	59.1	26.3	28.6	42.2
Images/Brand	36.8	37	30	36.4	21.1	33.3	32.8
Market request	52.6	48.1	75	86.4	47.4	42.9	58.6
Support and promotion from related organisations	26.3	22.2	15	4.5	5.3	23.8	16.4

c) Importance level of different aspects of sustainability

As discussed in Chapter 2, sustainable development is evaluated in a three-dimensional model of environmental, economic and social dimensions. To understand the importance of these dimensions, a five-point Likert scale was provided for respondents to express their judgment, where 1 is not important, 2 is slightly important, 3 is moderately important, 4 is important, and 5 is very important. The rating scores given by respondents are shown in Table 7.9.

A spects of	Lavalaf	Number of respondents						Sub	0/ of
sustainability importance		Architects	Engineers	Facility managers	Project managers	Owners	Tenants	total	total
	Not or Slightly	0	1	2	0	0	0	3	2.3
Environmental	Moderately	2	4	0	0	1	3	10	7.8
Environmental	Important or Very	17	22	18	22	18	18	115	89.9
	Total	19	27	20	22	19	21	128	100
	Not or Slightly	1	2	2	1	4	2	12	9.4
Economia	Moderately	1	9	2	2	3	4	21	16.3
Economic	Important or Very	17	16	16	19	12	15	95	74.3
	Total	19	27	20	22	19	21	128	100
	Not or Slightly	1	6	2	0	2	1	12	9.4
Social	Moderately	7	9	6	8	3	3	36	28.1
Social	Important or Very	11	12	12	14	14	17	80	62.5
	Total	19	27	20	22	19	21	128	100

Table 7.9. Importance ratings of different aspects of sustainability by role group (128 respondents)

About 90% of respondents agreed that environmental impact is important or very important as an aspect of sustainability, about 74.3% of respondents agreed that economic is an important or very important dimension of sustainability, and 62.5% of respondents agreed that social is important or very important. For each role group, a similar situation can be observed: most of them believed these aspects are important or very important. Among these three dimensions, social received the least support, consistent with the gap identified in the literature review that the social dimension does not get as much attention as the environmental and economic dimensions get.

To explore whether the given rating scores are affected by respondents' age, Table 7.10 illustrates the distribution of rating scores by age group. As Table 7.9 showed most respondents believed that the three sustainability dimensions are important or very important, Table 7.10 summarises results by age group.

Sustainability	Level of	Nur	nber of resp	ondents in	each age gi	oup	Tatal
dimensions	importance	≤25	26–35	36–45	46-55	≥56	Totai
Environmental	Not or Slightly	1	1	0	1	0	3
	Moderately	3	1	1	5	0	10
	Important or Very	13	49	15	32	6	115
	Total	17	51	16	38	6	128
Economic	Not or Slightly	1	3	0	7	1	12
	Moderately	3	4	3	9	2	21
	Important or Very	13	44	13	22	3	95
	Total	17	51	16	38	6	128
Social	Not or Slightly	1	5	1	3	2	12
	Moderately	6	13	4	12	1	36
	Important or Very	10	33	11	23	3	80
	Total	17	51	16	38	6	128

Table 7.10. Importance rating scores on the three sustainability dimensions by age group (128 respondents)

On the environmental dimension, 13 respondents believed it is not, slightly, or moderately important, with 6 of them (46.2%) in the age group 46 to 55 years old. Of the 33 respondents believing the economic dimension is not, slightly, or moderately important, about 57.6% of them are over 46 years old. On the social dimension, 48 respondents believed it is not, slightly, or moderately important, with about 37.5% of them over 46 years old, and another 37.5% in the age group 26 to 35 years old. In general, more responses for not, slightly, or moderately important of the three sustainability dimensions were from respondents of older ages, with younger people more interested in sustainability.

To test whether the identified assessment criteria under the environmental dimension are applicable in China, respondents were asked to give rating scores for them. As Table 7.11 shows, most of them believed they are important or very important.

Environmental	Level of	Number of respondents						Sub-	% of
variables	imnortance	Architects	Engineers	Facility	Project	Owners	Tenants	total	total
variables	importance			managers	managers			totai	totai
Reduce energy	Not or Slightly	0	1	2	0	0	0	3	2.3
consumption	Moderately	3	1	1	3	1	1	10	7.8
	Important or Very	16	25	17	19	18	20	115	89.9
	Total	19	27	20	22	19	21	128	100
Reduce carbon	Not or Slightly	0	2	0	1	0	0	3	2.4
emissions	Moderately	1	2	3	2	3	1	12	9.4
	Important or Very	18	23	17	19	16	20	113	88.2
	Total	19	27	20	22	19	21	128	100
Reduce water	Not or Slightly	0	1	0	0	1	0	2	1.6
consumption	Moderately	1	2	0	3	2	3	11	8.6
	Important or Very	18	14	20	19	16	18	115	89.8
	Total	19	27	20	22	19	21	128	100
Use reusable or	Not or Slightly	1	2	0	1	0	2	6	4.7
recyclable	Moderately	6	6	4	6	6	5	33	25.7
materials or	Important or Very	12	19	16	15	13	14	89	69.6
components	Total	19	27	20	22	21	21	128	100
Reduce waste	Not or Slightly	0	2	1	1	0	2	6	4.7
generation	Moderately	3	4	2	5	0	2	16	12.5
	Important or Very	16	21	17	16	19	17	106	82.8
	Total	19	27	20	22	19	21	128	100
Improve indoor	Not or Slightly	0	0	1	0	0	0	1	0.8
environmental	Moderately	1	1	1	2	2	4	11	8.6
quality	Important or Very	18	26	18	20	17	17	116	90.6
	Total	19	27	19	22	19	21	128	100

 Table 7.11. Importance ratings of different criteria under environmental dimension by role group (128 respondents)

As Table 7.10 shows that there are about 34.4% of all the respondents who have ever worked on projects involving office buildings. Since these identified assessment criteria are for office buildings, it is necessary to test whether the choice of importance level of these environmental assessment criteria are impacted by whether the respondents have experience of working on office buildings. The Chi-square test with a 95% confidence level was conducted to check whether the association exists. If the *p*-value (asymptotic significance) is less than 0.05, the experience of working on office buildings is associated with the decision on importance level of environmental assessment criteria; otherwise, they are not associated. The SPSS results are in Appendix B-5, and the results indicate that all the *p*-value are bigger than 0.05, which means that the decided importance level of the environmental assessment criteria is not associated with respondents' experience of working on office buildings.

Similarly, respondents also gave rating scores for identified social assessment criteria. The same as the environmental dimension, respondents in different role groups held the same opinion. Most believed that all the identified assessment criteria are important or very

important, as Table 7.13 shows. Except for the listed assessment criteria, one social assessment criterion was also mentioned, impact on surrounding traffic and pedestrians. Whether the criteria are applicable, and how important they are, is discussed in the focus group discussion in Section 7.4.2.

	Lovalof		N	umber of r	esponden	ts		Sh	0/ of
Social variables	importance	Architects	Engineers	Facility managers	Project managers	Owners	Tenants	total	total
	Not or Slightly	0	0	0	0	0	0	0	0
Noise impact on	Moderately	5	4	2	3	1	2	17	13.3
neighbourhood	Important or Very	14	23	18	19	18	19	111	86.7
	Total	19	27	20	22	19	21	128	100
Б	Not or Slightly	0	0	0	0	0	0	0	0
Emission impact	Moderately	2	4	2	3	2	2	15	11.7
on	Important or Very	17	23	18	19	17	19	113	88.3
neighbournood	Total	19	27	20	22	19	21	128	100
Impact from	Not or Slightly	0	0	0	0	0	0	0	0
glare or	Moderately	3	2	2	2	1	3	13	10.2
overshadowing	Important or Very	16	25	18	20	18	18	115	89.8
neighbourhood	Total	19	27	20	22	19	21	128	100
	Not or Slightly	1	1	0	0	0	0	2	1.6
Safety and	Moderately	2	1	1	2	0	0	6	4.7
security	Important or Very	16	25	19	20	19	21	120	93.7
	Total	19	27	20	22	19	21	128	100
	Not or Slightly	0	1	0	0	0	1	2	1.6
Impact on	Moderately	2	3	4	2	1	2	14	10.9
cultural heritage	Important or Very	17	23	16	20	18	18	112	87.5
	Total	19	27	20	22	19	21	128	100
Accessibility to	Not or Slightly	0	3	0	1	0	0	4	3.2
building	Moderately	3	3	0	3	2	4	15	11.7
facilities for	Important or Very	16	21	20	18	17	17	109	85.1
people with special needs	Total	19	27	20	22	19	21	128	100
A 11.11.4	Not or Slightly	0	2	0	1	0	1	4	3.1
Accessibility to	Moderately	3	6	2	1	4	5	21	16.4
building	Important or Very	16	19	18	20	15	15	103	80.5
services	Total	19	27	20	22	19	21	128	100
Impact on	Not or Slightly	0	1	0	0	0	0	1	0.8
maintenance and	Moderately	3	3	3	6	3	4	22	17.1
maintainability	Important or Very	16	23	17	16	16	17	105	82.1
from newly added building fabrics or building systems	Total	19	27	20	22	19	21	128	100

Table 7.13. Importance ratings of different criteria under social dimension by role group (128 respondents)

Same as the environmental assessment criteria, the Chi-square test with a 95% confidence level was also conducted here to check whether there is an association between the decided importance level of the social assessment criteria and the respondents' experience of working

on office buildings. The SPSS results are in Appendix B-6, and the results show that all the *p*-value are bigger than 0.05, which means that the association does not exist.

In summary, the identified aspects of sustainable development and assessment criteria under environmental and social dimensions were generally believed to be important and should be included in this study. The other assessment criteria mentioned by respondents are included in focus group discussions to determine their necessity and importance.

d) Further discussion about improvement of sustainable retrofitting

Three open-ended questions in the last part of the survey further discussed the methods to improve the development of sustainable retrofitting, but they were not mandatory. Therefore, not all the respondents answered these questions, but some valuable responses are discussed.

The first question was for owners or tenants of office buildings who were asked about what improvement they expect from sustainable retrofitting. Two of the owners expressed that they expected improvement in several aspects of the building, including comfort level, the beauty of the interior and exterior of the building, and safety level, but the most important is that the cost of retrofitting is low. They explained that cost could be reduced by using materials and labour at a low price and in other ways like reusing recycled materials and components. The other owner expressed that the most expected improvement by retrofitting is upgraded building facilities by replacing aging facilities and systems such as pipes and circuit systems.

For tenants, a safe and healthy working environment was the most expected improvement. Even though responses were limited, the different demands of owners and tenants for retrofitting can still be summarised that both owners and tenants expect improvement of buildings' performance from retrofitting. Owners would like improved performance with lower cost, and most tenants do not care about cost much.

The same question was asked to professionals who provided their opinions based on their professional experience. They expected five aspects to be improved by sustainable retrofitting:

(1) All professionals believed sustainable retrofitting could be realised by appropriate

planning and wise determination of a retrofitting strategy, which can ensure building space is used more reasonably and effectively and meets different tenants' needs.

- (2) Selection of building materials is essential for achieving high quality of retrofitting. Materials with good quality should be encouraged to reduce the frequency of replacement. The use of reusable and recyclable materials can also help alleviate adverse impacts on the environment and ecology.
- (3) Architects, engineers and facility managers emphasised that different stages of existing buildings should be considered for determining retrofitting strategies. In other words, the long-term perspective should be taken to achieve sustainable retrofitting.
- (4) Project managers believed that advanced technology and equipment should be applied for retrofitting, such as smart control systems, so existing buildings can become "smarter" to manage and control.
- (5) The public's awareness of sustainability should be improved. If people understand the benefits of sustainable retrofitting that can be obtained by different stakeholders, at different stages, and in various forms, more people would support retrofitting existing buildings.

The last question was on what other actions, except for retrofitting, should be taken to improve the performance of existing buildings. Fifteen responses were received from six groups of respondents. Summarising these responses, two major actions are suggested: providing education for tenants and owners about how to properly use and manage existing buildings, and establishing relevant standards for regulating the retrofitting process. Since owners and tenants do not have sufficient knowledge about property management, most do not know how to use building facilities properly or do not pay attention to energy and resource-saving when they use the building. For example, if motion sensors are not installed in an office building, the lights would not be turned off automatically after people leave the office. In this case, if tenants were more aware of energy efficiency, they would manually turn off the lights so that energy is not wasted. Another example is that much energy can be saved if tenants understand to close windows and doors when the air conditioner is on. In addition, the relevant standards play a significant role in regulating the performance level of buildings. If the benchmarking is reasonable and regulation is strict, improvement by retrofitting can be largely achieved.

7.4.1.6. Outcomes of the survey

Based on the literature review and analysis of the questionnaire survey, all provided retrofitting activities are applicable for retrofitting office buildings in China (as shown in Figure 6.3 in Chapter 6). The assessment criteria under the three dimensions applicable in the context of sustainable retrofitting in China are listed in Table 7.13.

r	1		
Life stages	Environmental	Economic	Social
Life stuges	assessment criteria	assessment criteria	assessment criteria
Retrofitting	 Energy consumption 	 Initial cost 	 Noise impacts on neighbourhood
stage	 Carbon emissions 		• Emission impacts on neighbourhood
	• Use of		 Safety of retrofitting construction
	recyclable/reusable		 Impact on cultural heritage
	materials		 Impact on surrounding traffic and
	Waste generation		pedestrians*
Operation	 Energy consumption 	 Operation cost 	• Impacts from glare or overshadowing
stage	 Carbon emissions 	 Replacement cost 	neighbourhood
	• Use of	 Maintenance cost 	• Safety and security of users (e.g.,
	recyclable/reusable		security entrance, CCTV, etc.)
	materials		• Accessibility to building facilities for
	 Waste generation 		people with special needs
	 Indoor environmental 		 Accessibility to building services
	quality		 Impact on maintenance and
	 Water consumption 		maintainability from newly added
			building fabrics or building systems

Table 7.13. Assessment criteria from questionnaire survey

Note. * the item is added via survey and is further determined by focus group discussions

7.4.2. Stage 2 – Focus groups

Since the information collected by the survey is broad, and the response rate and sample size cannot be calculated when using snowball sampling, it was necessary to conduct another data collection method to consolidate and explain the survey results. As discussed at the beginning of Section 7.4, focus group discussion is a suitable method, which can consolidate the survey results, and also modify the assessment criteria further to be suitable for adapting the conceptual model to use for the case building and other buildings in China. The focus group discussion is a good chance to understand the interplay of different aspects of sustainability among various stakeholders. As well as to modify the conceptual model, the focus group discussions are also conducted to estimate weights for the three sustainability dimensions and assessment criteria.

7.4.2.1. Participants in the focus group discussions

To consolidate the results of the survey and further specify the model to suit the local situation of the case building, the same role groups as the questionnaire survey were invited for focus group discussions. One of the researcher's supervisors has connections with local professionals and key stakeholders of retrofitting in China. Twenty invitations were sent, and 14 replies were received expressing interest in participating in the focus group discussion. Since no replies from owners of office buildings were received, no building owners participated in the focus group discussion. Due to time clashes, three focus group discussions were organised at different places and times for participants' convenience. Each focus group discussion was held for one to two hours, depending on the number of participants. The participants in each focus group are summarised in Table 7.14.

Category	Role	Number of participants
Professionals from academia	Architect	3
	Engineer	2
	Facility manager	1
	Project manager	1
Professionals from industry	Architect	1
	Engineer	2
	Project manager	1
Others	Owner	0
	Tenant	3
Total		14

Table 7.14. Categories of participants in the focus group discussion

7.4.2.2. Focus group discussion

The details of the three focus group discussions are listed in Table 7.15. The participant information sheet and consent form were sent to each participant prior to the discussion (see Appendix C-1 and C-2). It took about 15 minutes for them to read, sign and return the completed consent form before the commence of the discussion. Four main questions were asked to help participants express their opinions and guide discussions within the scope of the study.

	Date	Location	Period	Moderator	Notetaker	Participants
1	7 May 2019	Zhejiang University City College	2 hours	External supervisor	Researcher	7 professionals from academia
2	10 May 2019	An architectural design company in Hangzhou	1.5 hours	Researcher	Researcher	4 professionals from the industry
3	15 May 2019	A cafe in Hangzhou	1 hour	Researcher	Researcher	3 office building tenants

Table 7.15. Details about focus group discussions

- Question 1: What is your understanding of the environmental, economic and social dimensions of sustainability?
- Question 2: What are the challenges of retrofitting?
- Question 3: What retrofitting activities are commonly applied in Hangzhou?
- Question 4: Are the identified environmental and social assessment criteria relevant to retrofitting projects in Hangzhou, and how important are they?

7.4.2.3. Analysis of focus group discussion

The discussion is categorised into four parts. First, issues concerned with the three sustainability pillars are discussed. Then, obstacles faced by local sustainable retrofitting mentioned throughout the discussion are also summarised in Table 7.16. Based on the understanding of local sustainable retrofitting, participants were asked to give rating scores on the provided retrofitting activities and assessment criteria to justify whether they are suitable for the case study in Hangzhou. If there was something that they believe should be added, modified or deleted, they needed to give a reason for these changes.

Discussion		Professionals from academia	Professionals from industry	Tenants
Concerns on three pillars	Environmenta issues Economic issues	 Indoor environmental quality Waste management Carbon emissions are not emphasised in China Acoustic comfort is always neglected Impacts from retrofitting construction are always neglected Poor thermal performance is the main reason for retrofitting This dimension is only considered for meeting relevant regulations The initial cost is the primary concern of owners The payback period is important for assessing the economic performance of retrofitting LCC should be employed to assess the economic 	 This dimension is not emphasised at the stage of planning and construction Energy consumption is only considered for achieving a green building label The environmental impact will be considered if the retrofitting activities can bring financial benefits Owners do not think this dimension is important The budget of retrofitting determines the scale and quality of retrofitting and frequency of retrofitting Owners consider initial costs more 	 Indoor visual comfort Indoor air quality Greenery level The most expected improvement by retrofitting is the performance of HVAC Structure defect is the main reason for retrofitting This dimension is not the primary consideration of tenants Tenants also pay for retrofitting as retrofitting is regarded as part of maintenance
	Social issues	 performance of retrofitting from a long-term perspective Split benefit between owners and tenants is an obstacle to retrofitting Inappropriate use by tenants Retrofitting frequency Lack of consideration for people with additional needs Impact on culture 	Important than other costs Retrofitting frequency	 Accessibility to building facility and building service Indoor environment for good wellbeing and higher work efficiency Tenants' use behaviour
Obstacles		 Voices from tenants are hardly heard A limited database about energy consumption is available in China Different work formats for retrofitting from western countries 	 Information about retrofitting construction is limited in China Lack of mature assessment standards for retrofitting in China Limited support from the government, especially financial support 	Tenants' opinions should be considered for retrofitting

Table 7.16. Summary of focus group discussions

For environmental issues, indoor environmental quality, including indoor thermal comfort, air quality, acoustic comfort and visual comfort, is the primary concern. However, acoustic performance is always neglected at the early stage of planning and construction. Professionals from academia also mentioned that waste management is essential to environmental sustainability. They also stated that energy consumption and carbon emissions are rarely a concern for retrofitting in China unless for meeting relevant regulations or achieving a green building label. Moreover, the focus on reducing negative impacts on the environment is mainly at the operation stage. Impacts from retrofitting construction are seldom considered. In summary, improving indoor environmental quality is believed to be the main content under the environmental dimension. Reducing negative impacts on the environment is only considered under the have-to situation, which is normally not considered by building owners and tenants.

For the economic dimension, the importance of initial cost is emphasised, especially for owners who pay for retrofitting. Whether the initial cost of retrofitting is acceptable determines the scale and quality of the retrofitting. Another reason for owners avoiding retrofitting is the split benefit. Tenants receive the benefits of high initial costs instead of the owners who pay for it. The split benefit between owners and tenants is regarded as a significant obstacle for retrofitting.

Compared to the environmental and economic dimensions, the social dimension receives less attention. Some existing social issues were discussed, including lack of knowledge about sustainability leading to inappropriate use behaviour of tenants, and lack of consideration for people with additional needs such as the elderly, people with disability, and pregnant people. Tenants' wellbeing and work efficiency impacted by the indoor environment should also be considered. The impact of the frequency of retrofitting should also be included under the social dimension. Determining a proper time and frequency of retrofitting is important for reducing interruption to tenants.

In addition to issues in the three sustainability dimensions, participants also discussed obstacles faced by local sustainable retrofitting. First, they stated that tenants' requirements are the main driver of promoting sustainable retrofitting. However, through the process of determining retrofitting strategies, tenants are rarely involved. A new mechanism of deciding retrofitting

strategies that allows tenants' voices to be heard is required. Second, in China, retrofitting is conducted mainly to improve thermal comfort. It is rare in China to consider all the environmental, economic and social dimensions for retrofitting. Besides, a mature assessment standard for sustainable retrofitting in China does not exist yet. China is still at the early stage of developing sustainable retrofitting.

Based on the above discussion, respondents were asked to modify the retrofitting activities and assessment criteria to be suitable for local retrofitting projects. First, tables containing retrofitting activities and assessment criteria based on the survey results were sent out. If respondents believed something needed to be modified, added or deleted, they needed to explain why these changes were required. The related discussion is summarised in Table 7.17.

Discussion	Professionals from academia	Professionals from industry	Tenants
Retrofitting activities	• Adding or upgrading acoustic insulation for the walls that subdivide the tenancy areas	• No other changes are needed	• No other changes are needed
Assessment criteria	 The problem of glare or overshadowing neighbourhood is rarely considered when conducting retrofitting projects The adaptability for different users should be considered 	 Improved adaptability by retrofitting is desired since other tenants may require a different function 	 Concern about relocating tenants during retrofitting

Table 7.17. Modification of retrofitting activities and assessment criteria

In the three focus group discussions, professionals from academia proposed one new retrofitting activity of adding or upgrading acoustic insulation for the walls that subdivide the tenancy areas. They believed that it is a crucial activity for retrofitting office buildings, which can improve indoor acoustic comfort and offer more private space for tenants.

On assessment criteria, professionals suggested removing the assessment criterion of impact from glare and overshadowing the neighbourhood from the social dimension. They explained that the problem could not easily be solved by retrofitting and they could not recall any related practice based on their professional experience. Two assessment criteria are added for the social dimension. The first one is room flexibility. Based on the newly added retrofitting activity mentioned above, professionals from both academia and industry believed that the improvement of room flexibility needs to be added to the operation stage of the social dimension. If the adaptability of the building can be improved by retrofitting, different types of tenants can be attracted, such as office space catering for both architectural firms and IT companies. This makes the office building easy to rent out and may attract higher rents. The second criterion is added to the retrofitting stage in the social dimension by tenants – impacts on relocating current tenants. Tenants stated that whether tenants could be relocated during retrofitting can sometimes determine if retrofitting is acceptable or not.

Once no more changes were proposed, participants were asked to assign rating scores to determine the effectiveness of retrofitting activities and the importance of assessment criteria for applying them to local retrofitting projects. A 10-point scale was adopted, where a score of 1 represents the least effective or important, while a score of 10 represents the most effective or important. The average rating scores for retrofitting activities and assessment criteria are shown in Tables 7.18 and 7.19, respectively.

Retrofitting activities	Average rating score (max 10)
A1. Install/upgrade insulation of building envelopes	9.09
A2. Adopt extensive green roof	7.64
A3. Replace existing windows (glazing and/or frames) with energy efficient ones	8
A4. Install sun shading devices	7.45
A5. Install building management control system (BMCS)	7
A6. Upgrade lifts to more energy efficient ones	7.23
A7. Replace components or the whole HVAC system with a more efficient one	8.18
A8. Upgrade lighting system (including install motor sensors for lighting system, replace bulbs with T8, T5 or LED, and/or install daylight dimming control system)	7.90
A9. Install PV panels	8
A10. Install water control sensors	7.85
A11. Apply water-saving appliances and equipment (taps, toilet flushing equipment)	7.73
A12. Install water treatment system and reuse recycled water (storm, black and greywater)	7.55
A13. Add or upgrade acoustic insulation for the walls subdividing tenancy areas	7.91

Table 7.18. Rating score on the effectiveness of retrofitting activities

		Assessment criteria	Average rating score (max 10)
u	tt še	Energy consumption	8.21
Si01	ofi	Carbon emissions	7.5
len	tetr 1g s	Use of recyclable/reusable materials	7.36
lin	н. К	Waste generation	8.64
al e		Energy consumption	8.29
ent	uc	Carbon emissions	7.79
un	atio	Use of recyclable/reusable materials	8.29
iro	per sta	Waste generation	7.86
Nu	Ō,	Indoor environmental quality	9.36
ш		Water consumption	8.14
		Noise impact on neighbourhood	8
	ng	Emission impact on neighbourhood	8
	fitti ge	Safety of retrofitting construction	8.93
uo	tro] sta	Impact on cultural heritage	7.64
isni	Re	Impact on surrounding traffic and pedestrians	7.79
me		Impact on relocating tenants	7.93
l di		Accessibility to building facilities for people with special needs	8.43
cia	n	Accessibility to building services	8.71
So	oeratic stage	Impact on maintenance and maintainability from newly added building fabrics and building systems	8.43
	Ō	Safety and security of users	8.71
		Room flexibility for different demands	8.14

Table 7.19. Rating score on importance of assessment criteria

Based on Tables 7.18 and 7.19, the lowest rating score assigned to a retrofitting activity is 7, and the lowest score assigned to an assessment criteria is 7.36. It can be concluded that all provided retrofitting activities and assessment criteria are suitable to apply to local retrofitting projects. According to the specific situation of the case study, the necessary retrofitting activities and assessment criteria can be selected from Tables 7.18 and 7.19, respectively.

7.4.2.4. Outcomes of focus group discussions

Based on the focus group discussions, retrofitting activities and assessment criteria suitable for local retrofitting projects are finalised in Tables 7.18 and 7.19. The operating model for the case study has now been converted from the conceptual model. It can be used to develop retrofitting strategies for the case building based on steps illustrated in Chapter 6. At the end of the focus group discussions, participants were given a survey questionnaire to determine the weights of the three sustainability dimensions and assessment criteria under them using the AHP method. The details for determining weights are described in the following section.

7.5. Estimate weights by AHP method

According to the conceptual model developed in Chapter 6, the AHP method is adopted to determine weights. Figure 7.3 shows the steps to determine the level of importance or priority of assessment criteria using the AHP method (Saaty 2008).





Source: Saaty 2008

The developed assessment criteria should be structured in a hierarchical form. According to the discussion in Chapter 6, two levels of importance should be determined: the three sustainability pillars and the assessment criteria under them. These two levels are structured in a decision hierarchy as Figure 7.4 illustrates. Since both retrofitting and operation stages are considered, assessment criteria are categorised into these two stages correspondingly. The pairwise comparison, therefore, is done for each stage separately. In Figure 7.4, assessment criteria under the economic dimension are not shown because their level of importance is regarded as equal in the LCC method (see Section 6.7.2 in Chapter 6).



Figure 7.4. The hierarchy structure of assessment criteria

After structuring the decision in a hierarchy form, a fundamental scale, as Table 7.20 shows, is provided for participants of the focus group discussions to make the pairwise comparison.

Intensity of Importance	Definition	Explanation
1	Equal importance	Two criteria contribute equally to the objective
2	Weak or slight	
3	Moderate importance	Experience and judgment slightly favour one over another
4	Moderate plus	
5	Strong importance	Experience and judgment strongly favour one over another
6	Strong plus	
7	Very strong or demonstrated	A criterion is favoured very strongly over another: its dominance demonstrated in practice
8	Very, very strong	
9	Extreme importance	The evidence favouring one over another is of the highest possible order of affirmation
Reciprocals of above	If criterion <i>i</i> has one of the above non-zero numbers assigned to it when compared with criterion <i>j</i> , then <i>j</i> has the reciprocal value when compared with <i>i</i>	A reasonable assumption
1.1 – 1.9	If the criteria are very close	It may be challenging to assign the best value, but when compared with other contrasting criteria, the size of the small numbers would not be too noticeable, yet they can still indicate the relative importance of the criteria.

Table 7.20. Fundamental scores for the AHP method

Source: Saaty 2008

The last step in Figure 7.3 is to obtain relative weights, which is realised by collecting and calculating the geometric mean of individual judgment. The steps of this process are described in the following sections.

7.5.1. Design a questionnaire form to collect individual judgment

In this study, the participants of focus group discussions were also invited to complete a questionnaire form, designed to collect individual judgment on the level of importance between assessment criteria and the three sustainability dimensions. Assessment criteria under different life stages are structured in a pairwise form, as Table 7.21 shows. Each participant in the focus group discussion needed to fill in the table using the value score provided in Table 7.20. Due to the reciprocal relation between a pair of assessment criteria compared, only the triangle in the upper right hand in Table 7.21 needs to be defined. The lower left-hand triangle can be calculated based on the reciprocal relation, $c_{ij} = 1/c_{ji}$.

C _{ij}	Assessment criterion C_1	Assessment criterion C_2	 Assessment criterion C_n
Assessment criterion C_1	- <u><u>C</u>14</u>	C ₁₂	 C _{1n} _
Assessment criterion C_2	C ₂₁	τ_{22}	 <i>C</i> _{2n}
:	÷	÷	 : :
Assessment criterion C_n	C _{n1}	<i>C</i> _{<i>n</i>2}	 <i>c_{nn}</i>

Table 7.21. Pairwise comparison matrix for individual judgment

The form for collecting individual judgment is in Appendix D-1. With the collected individual judgments, the matrix of individual judgments can be constructed as Equation 7.1.

$$C = \begin{bmatrix} c_{11} & c_{12} & \dots & c_{1n} \\ c_{21} & c_{22} & \dots & c_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ c_{n1} & c_{n2} & \dots & c_{nn} \end{bmatrix}$$
Equation 7.1.

For example, Table 7.17 shows that there are four assessment criteria in the retrofitting stage of the environmental dimension, and the matrix of the individual judgments can be constructed as Equation 7.2.

$$C = \begin{bmatrix} c_{11} & c_{12} & c_{13} & c_{14} \\ c_{21} & c_{22} & c_{23} & c_{24} \\ c_{31} & c_{32} & c_{33} & c_{34} \\ c_{41} & c_{42} & c_{43} & c_{44} \end{bmatrix}$$
Equation 7.2.

7.5.2. Combine individual judgment matrix to reach group decision

The geographic mean is used to combine individual judgments into a group decision. Assuming there are k participants who compare the level of importance of n assessment criteria. With collected individual judgment in the last step, the group judgment matrix can be built, as Equation 7.3 shows.

$$C = \begin{bmatrix} \sqrt[k]{c_{11}^{1} \times c_{11}^{2} \times \dots \times c_{n1}^{k}} & \sqrt[k]{c_{12}^{1} \times c_{12}^{2} \times \dots \times c_{n2}^{k}} & \dots & \sqrt[k]{c_{1n}^{1} \times c_{1n}^{2} \times \dots \times c_{1n}^{k}} \\ \sqrt[k]{c_{21}^{1} \times c_{21}^{2} \times \dots \times c_{21}^{k}} & \sqrt[k]{c_{22}^{1} \times c_{22}^{2} \times \dots \times c_{22}^{k}} & \dots & \sqrt[k]{c_{2n}^{1} \times c_{2n}^{2} \times \dots \times c_{2n}^{k}} \\ \vdots & \vdots & \ddots & \vdots \\ \sqrt[k]{c_{n1}^{1} \times c_{n1}^{2} \times \dots \times c_{n1}^{k}} & \sqrt[k]{c_{n2}^{1} \times c_{n2}^{2} \times \dots \times c_{n2}^{k}} & \dots & \sqrt[k]{c_{nn}^{1} \times c_{2n}^{2} \times \dots \times c_{nn}^{k}} \end{bmatrix}$$

$$=\lambda_{max}W$$
 Equation 7.3.

Where,

- C the group judgment matrix
- k the number of participants who determine the weights
- λ_{max} the largest eigenvalue of a comparison matrix, which can be calculated using MATLAB
- W the corresponding eigenvector, the components in W are the weightings for each of the assessment criteria

Taking the assessment criteria in the retrofitting stage of the environmental dimension as an example, the matrix of group judgment can be illustrated in Table 7.22.

	Energy consumption	Carbon emissions	Use of recyclable/reusable materials	Waste generation
Energy consumption	1	1.013	0.964	1.021
Carbon emissions	0.987	1	0.967	0.9871
Use of recyclable/reusable materials	1.037	1.034	1	0.925
Waste generation	0.980	1.013	1.082	1

 Table 7.22. Group judgment matrix on the environmental assessment criteria in the retrofitting stage

In this study, five matrices of group judgment were constructed for the level of importance of the environmental assessment criteria in the retrofitting stage, the environmental assessment criteria in the operation stage, the social assessment criteria in the retrofitting stage, the social assessment criteria in the operation stage, and the three sustainability dimensions. The details of establishing the matrix of group judgment for assessment criteria and the three sustainability dimensions are in Appendix D-2.

7.5.3. Consistency test

After the group judgment matrices are constructed, the consistency of the group judgment needs to be checked. The pairwise comparison may lead to inconsistent importance logic among all compared items. For example, A, B and C items are compared. In pairwise comparison, A is regarded as more important than B, B is more important than C, and C is more important than A. The importance loop among the three items is inconsistent, as the importance logic is A>B>C>A. In addition, inconsistency may arise from inappropriate relationships among compared items. For instance, A is twice as important as B, B is twice as important as C, and A is also regarded as twice as important as C in the pairwise comparison. However, based on the first two comparisons, A should be four times more important than C.

To avoid inconsistency, the consistency ratio is calculated based on Equations 7.4 and 7.5 to check the consistency of the group judgment matrix.

$$CR = \frac{CI}{RI}$$
 Equation 7.4.

$$CI = \frac{\lambda_{max} - n}{n - 1}$$
 Equation 7.5.

Where,

CR – consistency ratio CI – consistency index RI – the random index, referring to Table 7.23 λ_{max} – the largest eigenvalue of a comparison matrix n – the number of compared items in total

Table 7.23. Random index in AHP

n	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
RI	0	0	0.58	0.90	1.12	1.24	1.32	1.41	1.45	1.49	1.51	1.48	1.56	1.57	1.59
Note: n is the number of compared items $1.24 + 1.52 + 1.41 + 1.45 + 1.45 + 1.51 + 1.46 + 1.50 + 1.57 + 1.57$															

Source: Saaty 1990

Saaty (1990) stated that a judgment matrix can be regarded as consistent if the calculated CR is less than 0.1. Otherwise, the group judgment matrix needs to be rebuilt based on new individual judgment.

Before the participants of focus group discussions started the questionnaire form for pairwise comparison (referring to Section 7.4.1), the reciprocal relation between the objects of pairwise comparison and possible situations of inconsistency were explained to them. As a result, all the calculated consistency ratios (CR) of the constructed matrices of group judgment (see Section 7.4.2) are less than 0.1 (see Table 7.24). Therefore, these group judgments can be regarded as consistent.

Matrix of group	No. of compared items	λ_{max}	RI	CI	CR	Consistent or not	
Assessment criteria in the	Retrofitting stage	4	4.002	0.9	0.001	0.001	Consistent
environmental dimension	Operation stage	6	6.078	1.24	0.016	0.013	Consistent
Assessment criteria in the social	Retrofitting stage	6	6.017	1.24	0.003	0.002	Consistent
dimension	Operation stage	5	5.007	1.12	0.002	0.002	Consistent
The three sustainabili	ity dimensions	3	3 001	0.58	0.0005	0.001	Consistent

Table 7.24. Consistency of the constructed matrices of group judgment

Note: CI and CR are calculated based on Equations 7.4 and 7.5

7.5.4. Outcome of weights generation

After the valid group judgment matrix is formulated, the weights of compared items can be obtained by normalising the column and line weights of the matrix accordingly. In this way, the weights of each assessment can be attained. For the three sustainability dimensions, the weights can be obtained by repeating the above process but making the pairwise comparison only among these three dimensions. The detailed process of generating weights for assessment criteria and sustainability dimensions is in Appendices D-1 to D-3. The generated weights for assessment criteria and the three sustainability dimensions are summarised in Table 7.25 and Table 7.26.

		Assessment criteria	Weights
	ng	Energy consumption	0.25
ension	fitti 1ge	Carbon emissions	0.245
	etro sta	Use of recyclable/reusable materials	0.25
	R	Waste generation	0.255
me		Total	1
di		Energy consumption	0.219
tal	uc	Carbon emissions	0.155
en	atic	Use of recyclable/reusable materials	0.155
uu	per sta	Waste generation	0.154
rot	0	Indoor environmental quality	0.189
IVI.		Water consumption	0.128
Er		Total	1
		Noise impact on neighbourhood	0.145
	ing	Emission impact on neighbourhood	0.18
	fitt ıge	Safety of retrofitting construction	0.227
	tro sta	Impact on cultural heritage	0.148
ion	Re	Impact on surrounding traffic and pedestrians	0.163
susi		Impact on relocating tenants	0.137
me		Total	1
l di		Accessibility to building facilities for people with special needs	0.259
cia	uo	Accessibility to building services	0.167
So	perati stage	Impact on maintenance and maintainability from newly added building fabrics and building systems	0.163
	Ō	Safety and security of users	0.222
		Room flexibility for different demands	0.189
		Total	1

Table 7.25. Weights of assessment criteria for the case study

Table 7.26. Weights of the three sustainability dimensions for the case study

Sustainability dimensions	Weights
Environmental	0.437
Economic	0.324
Social	0.239
Total	1

Based on the above discussion, the conceptual model has been converted to an operating model, which can be used to develop retrofitting strategies for office buildings in China. The next chapter illustrates the detailed process of using the operating model to develop retrofitting strategies for the case building.

7.6. Summary

This chapter illustrated how to convert the conceptual model to an operating model for dealing with locational variations. The conceptual model developed in Chapter 6 contained most

information on deciding retrofitting strategies for office buildings. However, the retrofitting activities and assessment criteria in the conceptual model were identified based on the literature review. Before applying it to develop retrofitting strategies for an actual office building, they need to be adapted to suit the specific situation of the case building.

To verify the conceptual model, a case study was conducted in Hangzhou, China. Therefore, the conceptual model needed to be converted to an operating model based on the specific situation of the case building. First, a survey was conducted in China to investigate whether the identified retrofitting activities and assessment criteria are applicable for retrofitting projects in China. The current situation regarding sustainable retrofitting in China can also be understood through the survey. Then, three focus group discussions with local professionals and key stakeholders were conducted to consolidate the survey results and further modify the retrofitting activities and assessment criteria to be applicable for the case building and other office buildings in China. Based on the information collected from the survey and the focus group discussions, suitable retrofitting activities and assessment criteria were finalised. In this way, the operating model for retrofitting projects in China is established. In the focus group discussions, apart from asking participants' opinions about retrofitting activities and assessment criteria using the AHP method.

With the converted operating model and the determined weights of the three sustainability dimensions and assessment criteria under them, the second part of the case study is to use the operating model to develop retrofitting strategies for the case building, which is described in Chapter 8.

Chapter 8. Case study – Develop retrofitting strategies 8.1. Introduction

The last chapter converted the conceptual model to an operating model by modifying the retrofitting activities and assessment criteria to suit the specific situation of the local case building. In this chapter, the operating model is used to develop retrofitting strategies for the case building, according to the steps outlined in the conceptual model (Figure 6.2 in Chapter 6). Following the illustration of developing retrofitting strategies for the case building is an analysis discussing the outcomes of the case study and its limitations. Combining the process described in Chapters 7 and 8, the detailed process of using the conceptual model to develop retrofitting strategies for an office building is fully demonstrated, and it can be regarded as a logically and methodologically correct framework for the work done and to be done in the future.

8.2. Developing retrofitting strategies based on the operating

model

The case building is an office building located in the centre of Hangzhou. Since its construction in 1985, it has never undergone a comprehensive evaluation of its performance or a well-planned retrofitting. Only some upgrades have been undertaken from time to time. Detailed information about the case building can be found in Section 7.3 of Chapter 7. The following sections illustrate how to use the operating model to develop retrofitting strategies for the case building.

8.2.1. Set retrofitting goals and identify problems of the case building

The preliminary demands of retrofitting from the main stakeholders are acquired by consulting the building owners, facility managers, and tenants. Building owners expressed that they tend to conduct sustainable retrofitting mainly to reduce current annual operation costs, and the target for reduction is 30% or more. Moreover, they would also like to obtain the national green building label, based on the Assessment Standard for Green Retrofitting of Existing Building (GB/T 51141-2015) (MOHURD 2015), as a way of advertising to attract higher rents. The building owners proposed a budget of CNY 3 million for the retrofitting project, which is about AUD 660,000 based on the currency rate in September 2022 (AUD 1 is equal to CNY 4.55). Speaking with the facility managers, they were concerned that the building was more than 35 years old and not equipped with a building management control system (BMCS), so the air conditioning and lighting can only be controlled manually. Therefore, they need to patrol after hours to ensure that the air conditioning and lights had been turned off. As for opinions from tenants, there is a suggestion box in the hall where tenants can put their opinions about the building and retrofitting. The major complaint was that they were unable to see images on computer monitors due to intensive sunlight at some times. Retrofitting was seen as an opportunity to solve this problem.

To further identify the building's problems, the finalised assessment criteria in the operating model (see Table 7.25 in Chapter 7) are used as a guideline for diagnosing problems of the case building. The records of electricity and water use in the past three years were also collected. Combining the issues raised by tenants and facility managers mentioned above, the problems are identified for the case building, as discussed below.

8.2.1.1. High energy consumption and carbon emissions

Electricity is the primary energy source for the case building. During the operation stage, electricity consumption and carbon emissions are mainly caused by operating different building

service systems, such as HVAC and lighting. Based on the electricity bills in the past three years, the average annual electricity consumption is about 1,836,262 kWh or 93.29 kWh/m².

According to the study by Yu, Wu and Can (2020), there are 1,057 office buildings in Hangzhou, which can be categorised based on the "green" level and annual electricity consumption, as Table 8.1 shows. For conventional office buildings in Hangzhou, the average electricity consumption is about 64 kWh/m², and 51 kWh/m² for green buildings, resulting in a potential energy saving of 20% (Yu, Wu & Can 2020). Regarding this situation, the case building can be described as a conventional office building at electricity consumption level 2 as shown in Table 8.1 (93.29 kWh/m² per year). Compared with green office buildings with average electricity consumption of 51 kWh/m², it has an energy saving potential of about 45%.

Office buildings	Electricity consumption level 1 (less than 50 kWh/m ²)	Electricity consumption level 2 (between 50 and 100 kWh/m ²)	Electricity consumption level 3 (more than 100 kWh/m ²)	Total
All office buildings (both conventional and green office buildings)	41%	47%	12%	100%
Conventional office buildings	35%	48%	17%	100%
Green office buildings	51%	39%	10%	100%

Table 8.1. Categories of office buildings in Hangzhou

Source: Yu, Wu & Can 2020

Figure 8.1 shows the usage allocation of electricity by the case building based on the electricity bills in the past three years. Lighting, office equipment and the HVAC system are the three largest electricity consumers, with 40%, 34% and 22% of total yearly consumption, respectively. They should have the biggest potential for energy saving. However, retrofitting in this study does not involve upgrading office equipment. Therefore, performance improvement of the lighting system and HVAC system is the focus of the retrofitting project.



To identify the problems causing high energy consumption and carbon emissions, the retrofitting team audits the performance of building elements and service systems relating to electricity consumption. It is then compared with the benchmark by local regulations. The below problems are identified.

(1) Poor insulation of building envelopes

As discussed in Section 3.4.6 of Chapter 3, the insulation performance of building envelopes is highly related to the energy demand of the HVAC system. The energy saving from the HVAC system can only reach an optimum if the building is well-insulated. Therefore, the performance of the building insulation, windows and HVAC is discussed together.

Heating loss of buildings occurs mainly through external walls, roofs and windows. Therefore, the heat transfer coefficient (or U-value) of these three parts is calculated based on the local standard, Design Standard for Energy Efficiency of Public Buildings (DB33/1036-2007) (UAD, ZIAD & CMA 2007). The calculation results are illustrated in Table 8.2, and the calculation details are in Table E1-1 and E1-2 in Appendix E1.

		Heat transfer coefficient before retrofitting (W/m ² •K)	Standard for level II public buildings in Zheijang (W/m ² •K)				
External walls		2.029	≤1.0				
Roofs		3.356	≤0.7				
Windows	Steel frames	5.000					
	Plastic steel frames	6.600	\leq 4.7 for east- and north-facing windows \leq 3.5 for west- and south-facing windows				
Note		The calculation can be found	Source: Design Standard for Energy				
		in Table E1-1 and E1-2 in	Efficiency of Public Buildings (DB33/1036-				
		Appendix E1	2007) (UAD, ZIAD & CMA 2007)				

Tab	le 8.2	. Heat	transfer	coeffic	cient of	f walls	, roofs	and	wind	lows	of t	he	case	buil	dir	ıg
																_

The existing external walls are built with mortar, clay bricks, and tiles without insulation. Some parts of the tiles have been found to be hollow or peeling, leading to fast heat loss. The current heat transfer coefficient of external walls is 2.029 W/(m²•K), which is over the benchmark of the local standard (UAD, ZIAD & CMA 2007) of no more than 1.0 W/(m²•K). For existing roofs, there is a prefabricated insulation panel installed. However, it has not been upgraded since the building was built, and is broken and does not perform as insulation anymore. Currently, the U-value of existing roofs is about 3.356 W/(m²•K), which is much over the benchmark of 0.7 W/(m²•K) (UAD, ZIAD & CMA 2007). For windows, most existing windows are single glazing (3 mm thickness) with steel frames. Some of them were changed to PVC frames when the leaking problem was noticed. The U-value of existing windows with steel frames is 6.6 W/(m²•K), and 5 W/(m²•K) for those with PVC frames. However, both are beyond the value of the local standard as Table 8.2 shows.

The consequence of poor insulation and windows is the over-dependency on HVAC. A study by Zhu et al. (2009) analysed the relation between HVAC types and the energy consumption of public buildings in Hangzhou. It showed that, for public buildings of a similar scale, the electricity consumption by reverse cycle and varied refrigerant volume air conditioning system is similar, and the central air conditioning system may cause more consumption. The reverse cycle air conditioning system was only installed about five years ago for the case building. The power of cooling and heating is enough to adjust for indoor thermal comfort. Therefore, the HVAC system has little room for improvement to achieve energy efficiency. Based on the above discussion, the massive energy consumption by the HVAC system is mainly because the building does not have insulation leading to fast heating loss. To maintain indoor thermal comfort, the air conditioning must work with full power during its whole working period in winter and summer. The climate in Hangzhou city is hot in summer and cool in winter. The indoor temperatures in spring and autumn are good enough without using HVAC (Zhu et al. 2009). For the case building, the average indoor temperature is about 28 °C in summer and 18 °C in winter. However, based on the local standard (UAD, ZIAD & CMA 2007), the comfortable indoor temperature with intervention by HVAC should be in the range of 24 °C to 26 °C in summer and 22 °C to 24 °C in winter. To reach the target temperature, HVAC is needed to make the indoor temperature 2 to 4 °C cooler in summer and 4 to 6 °C warmer in winter.

(2) Inefficient lighting system

The current bulbs are T8 fluorescent with a power of 32W (Jappha 2018). There are 2,900 bulbs in the building. Assuming 8 hours in a working day and 250 working days per year, the lighting system consumes about 185,600 kWh electricity per year. The local standard (UAD, ZIAD & CMA 2007) regulates that, for office buildings, the lighting power density should be equal to or less than 11 W/m², and the illuminance value should be around 300 lx. There are other bulb types that can provide enough illuminance but with less electricity demand. Due to the large number of bulbs in use, even if little electricity can be saved by each bulb, the potential for electricity saving is still great by changing the bulb type to a more energy efficient one, such as T5 and LED. Moreover, the lighting can only be controlled manually. People often forget to turn off lights when they leave the room, resulting in the waste of electricity.

(3) Inefficient lifts

There are four lifts in the building. They were installed when the building was first built, which makes the lifts energy consuming and slow compared to new types. Records show that the annual electricity consumption of these four lifts is 18,363 kWh. There is a lack of regulation or standard that can provide a benchmark of acceptable electricity consumption by lifts. However, according to the study by Zhang, Ni and Nu (2009), the application of energy-efficient lifts in China may achieve 20% to 50% energy saving compared to conventional lifts.

(4) No renewable energy in use

Nowadays, solar energy is encouraged by the government and is widely used in China. The local government in Hangzhou also encourages the use of solar energy because of the adequate solar light and relatively long daytime compared to northern China. However, there is no renewable energy in use in this building, leading to the heavy reliance on electricity energy.

(5) Lack of building management control system (BMCS)

The air conditioning in the case building can only be controlled manually, so waste can be easily caused by people forgetting to turn the system off when they leave the room or when the indoor temperature is the same as the outdoor temperature. Moreover, without an energy consumption monitoring system installed, which is one type of BMCS, it is difficult to notice an abnormal use, leading to waste as well.

8.2.1.2. Excessive water consumption

Over the previous three years, the average annual water consumption of the building is about 23,970 kL for the 1000 tenants in the building. Therefore, yearly water consumption is about 24 kL per person, which is beyond the design limit for office buildings of 11~18 kL per person (MOHURD 2019). By checking the current water system, the problems below are identified, which lead to excessive water consumption.

- (1) Old water fixtures, with 86 old flush toilets in total. Water tanks provide no pressure, and 9 L of water is needed for each use. There are 200 old taps (screw type lifting cast iron faucet) with slow water output.
- (2) No water treatment system is installed. About 30 kL tap water is used for car washing per month, and 2 kL tap water is used to irrigate plants. In addition, tap water is also used for regular cleaning, such as floor cleaning. If a water treatment system is installed for treating greywater and stormwater, treated water can be used for car washing and irrigation, contributing to water saving.

8.2.1.3. Poor indoor visual comfort

The indoor thermal, acoustic and visual comfort of the case building are investigated in the case study. There were no complaints about indoor thermal performance collected from tenants, mainly because the climate in this region (hot-summer-cold-winter) is relatively mild compared to the climate in northern China. As discussed in Section 8.2.1.1, the U-value of the existing external walls, roofs and windows cannot meet the requirement of the local standard. However, the operation of HVAC can adjust the indoor temperature to a comfortable level. Therefore, it is expected that with the improvement of insulation performance of building envelopes, the indoor comfort level will not be reduced with less or no HVAC intervention.

To test indoor acoustic performance, a test of background noise was conducted at an office room and a meeting room at a low (floor level 2), medium (floor level 5), and high (floor level 12) level respectively. A real-time signal noise sensor (Model type: AWA2691) was used to test background noise. The duration of each test is 20 minutes with 0.1s sampling interval. The average value of all the samplings was taken as the background noise of the tested room. The test results are shown in Table 8.3. The standard value of background noise for office rooms and meeting rooms is 45 dB based on the national standard, Code for Design of Sound Insulation of Civil Buildings (GB50118-2010), which should not be exceeded (MOHURD
2010). All the test results meet the requirement. Therefore, the indoor acoustic performance of the case building can be regarded as acceptable.

Floor level	Room type	Background noise (dB)	Standard value (dB)			
Level 2	Office room	40.3				
	Meeting room	38.4				
Level 5	Office room	43.5	- 15			
	Meeting room	43.1	≤ 43			
Level 12	Office room	42.6				
	Meeting room	39.2				
Note	Measurement is based on the Code for Design of Sound Insulation of Civil Buildings					
	(GB50118-2010), which should not be exceeded (MOHURD 2010)					

Table 8.3. Test of background noise of the case building

The case building is constructed with single-glazing windows $(1.8m \times 1.8m)$ and no internal blinds or external shading devices are installed. In addition, based on the information provided by building managers, the designed illuminance of existing lighting bulbs meets the required value (300 lx) by the standard of Zhejiang province, Design Standard for Energy Efficiency of Public Buildings (UAD, ZIAD & CMA 2007). Regarding the tenant complaints about the intensive sunlight (see the beginning of Section 8.3.1), a problem of the case building is the intensive sunlight shining from the south at certain moments. Therefore, retrofitting is an opportunity to solve this problem.

8.2.1.4. High operation cost

The operation cost of the case building is mainly for electricity and water usage. The operation cost is highly related to the electricity and water efficiency of the building. Due to the diagnosed problems of high energy consumption and excessive water consumption, there is a big potential for saving operation cost by improving the efficiency of electricity and water. Based on the electricity and water bills in the past three years, the annual operation cost of the building is about CNY 1,186,953, which is about AUD 262,869 based on the currency rate in 2022 (AUD $1 \approx \text{CNY 4.55}$).

Based on the identified problems and demands of building owner, facility managers, and tenants, the retrofitting goals and project constraints of the case study are identified:

- Two retrofitting goals are defined:
 - (1) to achieve at least a 30% reduction in annual operation cost
 - (2) to obtain a national green building label
- Two retrofitting objectives are identified:
 - to reduce operation cost by increasing the energy efficiency of building service systems
 - (2) to reduce operation cost by decreasing the energy demand of the case building
 - (3) to reduce water cost by increasing the water efficiency of water fixtures
 - (4) to reduce water consumption by reusing treated greywater and stormwater for car washing, irrigation and cleaning
 - (5) to improve indoor visual comfort by glare control
- Project constraint is:
 - (1) The budget of the retrofitting project is CNY 3 million (AUD 659,341).

8.2.2. Develop retrofitting activities

Based on the problems identified, reference to locally applicable retrofitting activities (in Table 7.18 of Chapter 7), and consultation with local professionals in construction and property management sector, the following ten retrofitting activities are identified for the case building (see Table 8.4).

8.2.2.1. Improve insulation performance of external walls and roofs

The local standard, Design Standard for Energy Efficiency of Public Buildings (DB33/1036-2007) (UAD, ZIAD & CMA 2007), includes information about recommended insulation

design for local buildings, which is usually used as a design guide. By consulting with local professionals in construction and property management and combining the information from the local standard, an EPS system is selected as insulation for external walls, an XPS system is selected for roofs, and low-e double-glazing windows (5 mm thick low-e glass plus 5 mm thick normal glass with air filling in between) with aluminium frames are selected. These insulation materials are also selected because they are locally available, which can save cost and time on transport. The initial cost of upgrading insulation of external walls and roofs is about CNY 771,376, and it is CNY 710,088 to replace existing windows with low-e double-glazing windows. Detailed information about type, thickness, and amount of insulation materials for external walls, roofs and windows is in Table E2-1 in Appendix E-2. The calculation of the heat transfer coefficient of the selected insulation materials of external walls, roofs and windows shows that they can meet the requirement of the local standard, DB33/1036-2007 (UAD, ZIAD & CMA 2007), as Table E1-1 and Table E1-2 (in Appendix E-1) show.

In addition, adopting a green roof can also improve the insulation performance of roofs. Extensive roofs can mitigate the cooling energy of buildings by 1.3% to 13.3% (Aboelata 2021). However, the initial cost of a green roof of about CNY 500/m² based on quotes from local companies engaging in green roofing¹ is too high. It may cost CNY 650,000 (about 22% of the project budget) if the whole roof area of about 1300 m² is constructed as a green roof. In addition to the high initial cost, the maintenance cost of green roofs is also very high. The high construction and maintenance cost is the biggest obstacle to implementing green roofs in China (Dong, Zuo & Luo 2020). Considering the limited amount of energy saving, and the relatively high costs, a green roof is not considered for the case building.

8.2.2.2. Improve energy efficiency of lighting system

Two activities can be adopted to improve the energy efficiency of the lighting system: installing motion sensors and changing existing bulbs to more energy efficient ones. Installing motion

sensors can turn off lights automatically when motion cannot be detected for a certain time, which helps achieve about a 30% energy reduction on the lighting system (Riyanto et al. 2018). In addition, the cost of motion sensors is about CNY 78 each, and 900 motion sensors are needed (see Table E2-1 in Appendix E-2). Therefore, the retrofitting activity may cost about CNY 702,200 at the retrofitting stage.

The current bulb type is T8 fluorescent. There are several bulb types that are more energy efficient than T8 fluorescent, such as T5 and LED. However, the life of T5 bulbs is normally shorter than T8 bulbs. Moreover, by removing existing T8 fluorescent bulbs, T8 LED tube can be easily installed and operated since existing electronic ballast of T8 fluorescent is also applicable for T8 LED. The cost of one T8 LED bulb is about CNY 28.9 according to local suppliers. Therefore, T8 LED is selected to replace the existing T8 fluorescent bulbs. The power of a T8 LED is about 17W, while it is 32W for a T8 fluorescent bulb (Jappha 2018). About 2900 bulbs need to be replaced. Therefore, changing existing bulbs to T8 LED is expected to achieve substantial energy saving.

8.2.2.3. Improve energy efficiency of lifts

The current lifts are energy consuming and slow. New types of lifts with a regeneration system can convert motion power to electricity to support the operation of lifts, resulting in electricity saving. Based on studies by Al-Kodmany (2015), Ali et al. (2021), Carrillo et al. (2013) and Nguyen (2017), changing traditional lifts to energy efficient ones can help reach energy saving of lifts by 30%. There are several lift companies in the local area that can provide transport, installation and maintenance service. The cost of one energy efficient lift is about CNY 656,348 based on the quote by the local suppliers. Even though the initial cost of energy efficient lift is relatively higher than other proposed retrofitting activities, the brought benefits in environmental and social dimensions such as improved energy efficiency and more efficient

traffic flow (see Section 3.4.5 in Chapter 3) still make it worthy to be considered for the case building.

8.2.2.4. Install PV panels

China is a world leader in PV module production, accounting for approximately 64.5% of global output in 2018 (Yu & Tong 2021). In 2019, the cumulative PV power generation installed capacity in China reached 204.7 GW, accounting for 32.6% of the world's total installed capacity (Yu & Tong 2021). The technology and practice of installing PV panels to reduce reliance on fossil fuel is available and common in China. There are several local suppliers who can provide transport, installation and maintenance service, making installing PV panels a feasible retrofitting activity for the case building. Based on the quote by local suppliers, about CNY 698 is needed to install one PV panel. Considering the relatively high price of per unit and limited roof area available, it is designed that 50 PV panels are installed by the retrofitting. The power of each PV panel is about 300W, and the efficiency rate is 75% according to the information provided by the local suppliers. The designed annual electricity generation is about 20,531 kWh.

8.2.2.5. Improve energy efficiency of HVAC

As discussed in Section 8.3.1.1, there is no BMCS installed in the case building, leading to additional energy consumption due to waste. There is an energy consumption monitoring platform available in Zhejiang province (Li et al. 2020). By installing monitoring meters, the monitoring platform can automatically record energy consumption by different building service systems and report any abnormal use in real-time to avoid waste (Li et al. 2020). The initial cost of the monitoring system is about CNY 270,000 based on the quotes by local suppliers, which is about 9% of the project budget. Therefore, it can be considered for the case building.

8.2.2.6. Improve indoor visual comfort

There is no external shading panels or internal blinds installed in the case building, leading to poor indoor visual comfort. Retrofitting is an opportunity to fix the problem. The service life of window blinds is estimated to be about 10 years (EATS 2015), while it is about 25 to 30 years for external shading devices (Penny 2015). In addition, regular maintenance such as cleaning and repair is needed for internal blinds, but is barely required for external shading panels if the fixed shading devices (without automatic angle adjustment) are adopted. In addition, sun shading devices can also contribute to a reduction of about 9% of energy consumption by the HVAC system (Valladares-Rendón, Schmid & Lo 2017). The common practice in the local area is adding aluminium sun shading slats (as in Figure 8.2) for windows facing south which get intensive sunlight. The cost of aluminium sun shading slats is about CNY 550 per metre square. To install sun shading slats on windows facing south, about CNY 249,480 is needed. Therefore, aluminium sun shading slats will be added on windows facing south to control indoor glare.



Figure 8.2. Aluminium sun shading slats

Source: Achiexpo 2022

8.2.2.7. Improve water efficiency

Three retrofitting activities are available to improve water efficiency of the case building: installing water control sensors, replacing existing water fixtures with water efficient ones, and installing a water treatment system to recycle greywater and stormwater. Automatic control

sensors for taps can turn off taps automatically when the handle is released, or in a set time (such as 30 seconds) to avoid waste (Nguyen 2017). However, the existing taps and toilets in the building are outdated, and cannot provide the required flow rate. The prices of water-saving taps and toilets are reasonable, at about CNY 55 for one tap and CNY 550 for one toilet. Therefore, replacing existing water fixtures with water efficient ones is a better solution to the inefficient water use compared to installing water control sensors.

Treating and recycling greywater and stormwater by installing a water treatment system can save up to 50% of water use in an office building (Friedler & Alfiya 2010; Nguyen 2017). There is an above-ground box-type integrated water treatment plant available in the local area which can be easily installed and operated to recycle greywater and stormwater. The initial cost of the water treatment plant is about CNY 31,900 (about 1% of the budget) based on the local suppliers. Therefore, installing a water treatment system can be considered for the case building.

Based on the above discussion, ten retrofitting activities are identified for the case building as Table 8.4 summarises. To save time, expense and energy caused by transport, the materials and products needed for the retrofitting will be purchased locally or from neighbouring provinces, such Shanghai and Suzhou. Therefore, the specification information of required materials and products provided by local suppliers is used in the case study. Details and references for the adopted materials and products in this case study are in Table E2-1 in Appendix E-2.

Problems	Possible reasons	Retrofitting activities
High energy	Poor insulation of building	 Install/upgrade insulation of building envelopes
consumption	envelopes	(EPS for walls and XPS for roofs)
		 Replace existing windows with double glazed
		windows with low-e coat and aluminium frames
	Inefficient lighting system	Install motion sensors
		Change from T8 fluorescent to T8 LED light bulbs
	Inefficient lifts	Change existing lifts to energy efficient lifts
	No renewable energy in use	• Install photovoltaic system (PV panels)
	Lack of energy consumption	Install building energy consumption monitoring
	monitoring system	system
Excessive water	Inefficient sanitary fixtures	• Replace existing taps and flush toilets with water-
consumption		saving ones
	No water treatment system	• Add water treatment system for recycling greywater
		and stormwater
Poor indoor visual	No sun shading devices	 Add aluminium sun shading slats
comfort	installed	
High operation	Same as high energy	Same as high energy consumption and water
cost	consumption and water	consumption
	consumption	

Table 8.4. Proposed retrofitting activities for the case building

8.2.3. Develop assessment criteria

The assessment criteria suitable for local use are also developed in the operating model, as Table 7.19 in Chapter 7 shows. The identified assessment criteria are used to estimate the performance of potential retrofitting activities. However, some of the identified assessment criteria are not included in this case study for the following reasons.

First, the embodied energy and embodied carbon emissions of newly added building materials and building service systems need to be measured. Plenty of studies have measured the embodied energy and embodied carbon emissions of building materials, such as concrete, cement, glass, steel, etc., for which the associated embodied energy and embodied carbon coefficients are available. As for building service systems, such as building management control system, lifts etc., they are made of various materials and have a complicated manufacturing process, leading to complex computations. However, compared to building materials, the embodied energy and embodied carbon emissions of building service system are minor. For example, the weight of insulation materials of external walls added in the case study is about 291 tonnes, but the weight of installed lifts is only 1,344 kg. Therefore, the embodied energy and embodied carbon emissions of lighting motion sensors, lifts, the building energy consumption monitoring system, and the water treatment plant is not calculated in this study.

Second, as discussed in Section 8.2.2, local suppliers of building materials and building service systems are considered over those in other provinces, so there will be very little energy consumption by transporting building materials and products from the supplier to the construction site.

Third, as discussed in Section 6.7.1.1 (in Chapter 6), due to the small amount and uncertain estimation of waste generation by operating and maintaining the upgraded building materials and service system, only waste generation by replacing broken or obsolescent building materials and service systems is considered in this case study.

Fourth, maintenance cost and cleaning cost need to be calculated in LCC. However, the purpose of LCC in this study is to measure and compare the improvement level of economic performance by the proposed retrofitting activities, which can be used to select suitable ones. There are regular expenditures on maintaining and cleaning the case building, even though retrofitting is not conducted. Therefore, the difference in maintenance cost and cleaning cost by retrofitting activities is expected to be minor compared to the difference in operation cost.

Based on the above discussion, it is determined that the embodied energy and carbon emissions of lighting motion sensors, lifts, the building energy consumption monitoring system, the water treatment plant, and waste generation by operating and maintaining building service systems are not considered in this case study. Moreover, if one estimation is less than 1% of the total amount for one assessment criterion, it is neglected in the case study. For example, the energy consumption by transporting lifts from the supplier to the construction site is less than 1% of

the total energy consumption by lifts in the retrofitting and operation stage. Therefore, it will not be included. The finalised assessment criteria for this case study are listed in Table 8.5.

Sustainability dimension	Life cycle stage		Assessment criteria				
Environmental	Retrofitting	Energy	Embodied energy of added building materials				
dimension	stage	consumption	Energy consumption by retrofitting construction				
		Carbon	Embodied carbon emissions of added building materials				
		emissions	Carbon emissions by retrofitting construction				
		Potential for r	reuse/recycling of materials				
	Operation	Energy	Recurrent embodied energy of added building materials				
	stage	consumption	Energy consumption by operation activities				
		Carbon	Recurrent embodied carbon emissions of added building materials				
		emissions	Carbon emissions by operation activities				
		Potential for r	euse/recycling of materials				
		Waste genera	tion from replacement				
		Indoor visual	comfort				
		nption by operation activities					
Economic	Retrofitting	Initial cost					
dimension	stage						
	Operation	Operation cos	Operation cost				
	stage	Replacement	Replacement cost				
Social	Retrofitting	Noise impact	on neighbourhood				
dimension	stage	Emission imp	act on neighbourhood				
		Safety of retro	ofitting construction				
		Impact on cul	tural heritage				
		Impact on sur	rounding traffic and pedestrians				
		Impact on rele	ocating tenants				
	Operation	Accessibility	to building facilities for people with special needs				
	stage	Accessibility	to building services (such as automatically adjusted indoor				
		temperature, l	numidity, and dimming level by BMCS)				
		Impact on ma	intenance and maintainability from newly added building fabrics				
		and building s	systems				
		Impact on saf	ety and security of users				
		Room flexibil	ity for different users				

Table 8.5. Assessment criteria for the case study

8.2.4. Evaluate performance of retrofitting activities

The environmental, economic and social performance of the proposed retrofitting activities is evaluated against the developed assessment criteria.

8.2.4.1. Environmental impact assessment

The cradle-to-use boundary system is adopted, in which the production stage, retrofitting construction stage, and operation stage are included. The 50-year remaining service life is assumed as the study period. For environmental assessment, two kinds of data were collected:

- Material data (see Table 8.6), such as the quantity of materials used in the retrofitting stage. Quantities are measured based on drawings of the case building, regulations from related national or local standards, and consulting with local professionals. Details are in Table E2-1 in Appendix E-2.
- Equipment data (see Table 8.7), including quantity, type and capacity of equipment in the construction and operation stages. This information is determined based on the amount of construction and consulting with local professionals. As noted, energy and carbon for the manufacturing of construction tools, equipment and plants are excluded from the study. Details about equipment data are in Table E2-2 in Appendix E-2.

External walls and roofsA1Install/upgrade insulation of building envelopesExternal walls (EPS system)Cement mortar with interface agent (cement:sand = 1:3)275,223.6kgRoof (XPS system)EVS boardI7,487piecesRoof (XPS system)Roof (XPS system)Concrete (30MPA (30MPA store)Steel mesh1,500kgRoof (XPS system)Softer (Containing o4@200 cold-drawn steel wire)Steel mesh1,500kgWindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass29,59m³A3Add aluminium sun shading slats for windows facing south ramesAluminium sun shading slats for windows facing southAluminium sun shading slats for windows facing southAluminium sun shading slats for WindowsA2Change the existing filts to more energy efficient ones280panelsLights A4Install motion sensors900pieces100piecesMarce ASChange the existing filts to more energy efficient ones4liftsA6Change the existing filts to more energy efficient ones4liftsA7Install building energy consumption monitoring system1systemWater systemA9Change existing taps and to liets to water-saving onesChange existing taps and to liets to water-saving onesChange existing to south saving to liets to water-saving laps	Part of building		Retrofitting a	ctivity	М	aterials	Amount	Unit
walls and roofsinsulation of building envelopeswalls (EPS system)agent (cement:sand = 1:3) 2^{15} , 2^{15	External	A1	Install/upgrade	External	Cement morta	ar with interface	275 223 6	ka
$ \begin{array}{ c c c c c } \hline \mbox{roofs} & \mbox{bilding} & \mbox{system} & \mbox{system} & \mbox{EPS board} & \mbox{17,487} & \mbox{pieces} & \mbox{Alkali resistant fibreglass mesh} & \mbox{37,77} & \mbox{m}^3 & \mbox{Waterproof paint} & \mbox{18,89} & \mbox{m}^3 & \mbox{Waterson} & \mbox{System} & Syste$	walls and		insulation of	walls (EPS	agent (cement:sand = 1:3)		273,223.0	K5
	roofs		building	system)	EPS board		17,487	pieces
			envelopes		Alkali resista	nt fibreglass mesh	37.77	m ³
WindowsA2Change existing single glazing with diminum framesGassSecond SystemSecond Strene <th< td=""><td></td><td></td><td></td><td>-</td><td>Waterproof pa</td><td>aint</td><td>18.89</td><td>m³</td></th<>				-	Waterproof pa	aint	18.89	m ³
WindowsA2 existing single glazing windows to low-ce double glazing with aluminium framesGlass anterials29.59 m³WindowsA2 change existing single glazing windows to low-ce double glazing with aluminium framesGlass coment existing single glazing modified bituminous sheet materials29.59 m³WindowsA2 change existing single glazing windows to low-ce double glazing with aluminium framesAluminium sun shading slats for windows facing southAluminium sun shading slats south29.00 piecesLightsA4 A4 nstall motion sensorsAluminium sun shading slats for windows facing southAluminium sun shading slats south280 panelsLiftsA6 A6 Change existing single glazing windows facing southAluminium sun shading slats south280 panelsLiftsA6 Change existing lifts to more upplyA8 A1 Install PV panelsComere energy efficient ones4 LiftsMater systemA9 to ilets to water-saving onesChange existing to ilets to water- saving toilets with high pressure water tanksChange existing toilets to water- saving toilets with high pressure water tanks2000 taps				Roof (XPS	Concrete	Steel mesh	1,500	kg
Windows WindowsA2 R Change existing single glazing with aluminium sun shading slats for windows facing southGlass R Add aluminium sun shading slats for windows facing south29.59 r r ramesm³WindowsA2 Change existing single glazing windows to low-e double framesGlass29.59m³A3 LightsAdd aluminium sun shading slats for windows facing southAluminium sun shading slats28.00panelsLightsA6 Change the existing lifts to more energy efficient onesA1Install PV panels900pieceesLiftsA6 Change existing lifts to more energy efficient onesA1lifts111systemBMCSA8 Change existing taps and toilets to water-saving onesChange existing taps and toilets to water-saving onesChange existing taps and toilets to water-saving tapsChange existing taps and toilets to water-saving onesChange existing taps200taps				system)	(30MPa	Concrete (C30)		
kinki					with fine			
WindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass Aluminium frames29.59m³WindowsA2Change existing single glazing slats for windows facing southGlass29.59m³LightsA4Install motion sensorsAluminium sun shading slats for windows facing southAluminium sun shading slats280panelsLightsA4Install motion sensors900piecces118118LightsA6Change texisting lifts to more nergy efficient ones4liftsBMCSA8Install building energy consumption monitoring system50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing taps and toilets with high pressure86toiletsWater systemA9Change existing taps and toilets to water-saving onesChange existing taps and toilets with high pressure86toilets					stone)		52	3
Windows Image: Aspect of the second se					(Containing		52	m
Windows WindowsA2 A BChange existing single glazing windows to low-e double glazing with aluminium framesGlass Aluminium frames29.59m³WindowsA2 BChange existing single glazing windows to low-e double glazing with aluminium framesGlass29.59m³WindowsA2 BChange existing single glazing windows to low-e double glazing with aluminium framesAluminium frames913framesA3 Energy supplyAd BInstall motion sensorsAluminium sun shading slats for windows facing southAluminium sun shading slats alts for windows facing southP000piecesLifts Energy supplyA4Install motion sensors900pieces50panelsBMCS SystemA8Install PV panelsT8 LED2,900bulbs50panelsBMCS SystemA9 toilets to water-saving onesChange existing taps and toilets to water-saving tapsChange existing toilets to water- saving toilets with high pressure water tanks200taps					04(0)200			
WindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass Aluminium frames29.59m³WindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass29.59m³A3Add aluminium sun shading slats for windows facing southAluminium sun shading slats280panelsLightsA4Install motion sensors900piecesLiftsA6Change the existing lifts to more energy efficient ones4liftsBMCSA8Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing taps and toilets with high pressure6toiletsWater systemA9Change existing taps and toilets to water-saving toilets with high pressure200taps					steel wire)			
WindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass29.59m³A3Add aluminium sun shading slats for windows facing southAluminium sun shading slats slats for windows facing southPanelsLightsA4Install motion sensors900piecesA5Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing toilets to water- saving toilets with high pressure water tanks200taps					Asphalt felt		65	rolls
WindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass29.59m³A3Add aluminium sun shading slats for windows facing southAluminium sun shading slats280panelsLightsA4Install motion sensors900piecesA5Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install building energy consumption monitoring system500panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing taps and toilets to water-saving onesChange existing toilets to water- saving toilets with high pressure water tanks200taps					Cement morta	ar (cement:sand =	02	ions
$ \begin{array}{ c c c c } & A2 & Change existing single glazing windows to low-c double glazing with aluminum frames 4.3 Add aluminium sun shading slats for windows facing south 1.30 Add aluminium sun shading slats for windows facing south 1.30 Add aluminium sun shading slats for windows facing south 1.30 Add aluminium sun shading slats for windows facing south 1.30 K Change from T8 fluorescent bulbs to T8 LED 2.900 bulbs 1.1fts A6 Change the existing lifts to more energy efficient ones 4 lifts 1.1fts 1.36 Change the existing lifts to more energy efficient ones 4 lifts 1.37 BMCS A8 Install building energy consumption monitoring system 1.1 system 1.30 Mater saving toilets to water-saving ones 1.30 Mater Saving toilets to water-saving taps and toilets to water-saving ones 1.30 Mater Saving toilets to water-saving taps 1.30 Mater Saving toilets with high pressure Saving toilets wit$					1:3)		52,000	kg
WindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass29.59m³A3Add aluminium sun shading slats for windows facing southAluminium sun shading slats913framesLightsA4Install motion sensors900piecesA5Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing taps and toilets to water-saving toilets with high pressure water tanks200taps					XPS board		1,806	pieces
WindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass29.59m³A3Add aluminium sun shading slats for windows facing southAluminium sun shading slats913framesLightsA4Install motion sensors900piecesA5Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing taps and toilets to water-saving toilets with high pressure water tanks200taps					Styrene butad	liene styrene (SBS)		
WindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass29.59m³A3Add aluminium sun shading slats for windows facing southAluminium sun shading slats913framesLightsA4Install motion sensors900piecesA5Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing taps and toilets to water-saving toilets to water- saving toilets with high pressure water tanks200taps					modified bituminous sheet		130	rolls
WindowsA2Change existing single glazing windows to low-e double glazing with aluminium framesGlass29.59m³A3Add aluminium sun shading slats for windows facing southAluminium sun shading slats913framesLightsA4Install motion sensors900piecesA5Change from T8 fluorescent bulbs to T8 LED2,900bulbsLiftsA6Change the existing lifts to more energy efficient ones4liftsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing toilets to water- saving toilets with high pressure water tanks200taps					materials			
windows to low-e double glazing with aluminium framesAluminium frames913framesA3Add aluminium sun shading slats for windows facing southAluminium sun shading slats280panelsLightsA4Install motion sensors900piecesA5Change from T8 fluorescent bulbs to T8 LED2,900bulbsLiftsA6Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing toilets to water- saving toilets with high pressure200tapsKater supplyA9Change existing taps and toilets to water-saving onesChange existing toilets to water- saving toilets with high pressure86toilets	Windows	A2	Change existing	single glazing	Glass		29.59	m ³
Image: space of transmission o			windows to low	-e double	Aluminium fr	ames	0.1.0	
A3Add aluminium sun shading slats for windows facing southAluminium sun shading slats280panelsLightsA4Install motion sensors900piecesA5Change from T8 fluorescent bulbs to T8 LED2,900bulbsLiftsA6Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving tapsChange existing toilets to water- saving toilets with high pressure water tanks200taps			glazing with alu	minium			913	frames
A3Add atuminitum sun snading slats for windows facing southAtuminitum sun snading slats280panelsLightsA4Install motion sensors900piecesA5Change from T8 fluorescent bulbs to T8 LED2,900bulbsLiftsA6Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing taps200tapsChange existing toilets to water- saving toilets with high pressure water tanks86toilets		A 2	trames	ann abadina	A luminium a	un abadina alata		
LightsA4Install motion sensors900piecesA5Change from T8 fluorescent bulbs to T8 LED2,900bulbsLiftsA6Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing taps200tapsChange existing toilets to water-saving toilets with high pressure water tanks86toilets		AS	slats for window	sull shading vs facing south	Aluminum se	un shaung stats	280	panels
A5Change from T8 fluorescent bulbs to T8 LED2,900bulbsLiftsA6Change the existing lifts to more energy efficient ones4liftsEnergy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing taps200tapsChange existing toilets to water-saving toilets to water- saving toilets with high pressure water tanks86toilets	Lights	A4	Install motion se	ensors	•		900	pieces
$\begin{array}{c c c c c c c } \begin{tabular}{ c c c c c c } \hline Lifts & A6 & Change the existing lifts to more energy efficient ones & 4 & lifts \\ \hline Energy & A7 & Install PV panels & 50 & panels \\ \hline supply & & & & & \\ \hline Supply & & & & & \\ \hline BMCS & A8 & Install building energy consumtor monitoring system & 1 & system \\ \hline BMCS & A8 & Install building energy consumtor monitoring system & 1 & system \\ \hline Water & A9 & Change existing taps and toilets to water-saving ones & Change existing taps & 200 & taps \\ \hline Change existing toilets to water taps to toilets to water taps & Change existing toilets to water saving toilets to water taps & & & \\ \hline Change existing toilets to water taps & & & & \\ \hline Change existing toilets to water taps & & & & \\ \hline Saving toilets with high pressure & & & & & & \\ \hline Water tanks & & & & & & & \\ \hline \end{array}$	-	A5	Change from T8	fluorescent bul	bs to T8 LED		2,900	bulbs
Energy supplyA7Install PV panels50panelsBMCSA8Install building energy consumption monitoring system1systemWater systemA9Change existing taps and toilets to water-saving onesChange existing water taps to water-saving taps200tapsChange existing toilets to water-saving onesChange existing toilets to water- saving toilets with high pressure water tanks86toilets	Lifts	A6	Change the exist	ting lifts to more	e energy efficie	nt ones	4	lifts
supply Solution BMCS A8 Install building energy consumption monitoring system 1 system Water A9 Change existing taps and toilets to water-saving ones Change existing water taps to water-saving taps 200 taps Change existing taps and toilets to water-saving ones Change existing toilets to water saving toilets with high pressure water tanks 86 toilets	Energy	A7	Install PV panel	s			50	nanels
BMCS A8 Install building energy consumption monitoring system 1 system Water system A9 Change existing taps and toilets to water-saving ones Change existing water taps to water-saving taps 200 taps Change existing toilets to water-saving ones Change existing toilets to water- saving toilets with high pressure water tanks 86 toilets	supply							Puireis
Water A9 Change existing taps and toilets to water-saving ones Change existing water taps to water-taps to water-saving taps 200 taps System Change existing toilets to water-saving ones Change existing toilets to water-saving taps 200 taps Water Change existing toilets to water-saving toilets to water-saving toilets with high pressure water tanks 86 toilets	BMCS	A8	Install building	energy consump	otion monitoring	g system	l	system
system tollets to water-saving ones water-saving taps Change existing toilets to water-saving toilets with high pressure 86 water tanks 6	Water	A9	Change existing	taps and	Change existi	ng water taps to	200	taps
saving toilets with high pressure 86 toilets water tanks	system		toilets to water-s	saving ones	water-saving	taps		-
water tanks					Change existi	ng toilets to water-	96	tailata
water talks					saving tonets	with high pressure	80	tonets
A 10 Install water treatment system to recycle greywater and		Δ10	Install water tree	atment system to	recycle greyny	rater and		
stormwater stormwater		AIU	stormwater	ament system a	, recycle greyw		86	item

Table 8.6. Materials data for retrofitting the case building

(1) Energy consumption and carbon emissions

In the retrofitting stage, the quantity, power and running hours of required construction equipment for different retrofitting activities are estimated based on the amount of construction work shown in Table E2-1 in Appendix E2. Most of the required equipment consumes electricity, but some equipment consumes diesel, such as the autocrane and concrete pump truck. Based on the local regulation, General Rules for Calculation of the Comprehensive Energy Consumption (GB/T2589-2020) (SAMR & SAC 2020), the coefficient of converting diesel to electricity is 11.86 kWh/kg, and this figure is adopted to convert diesel consumption to electricity. As for the carbon emissions, the carbon emission intensity of electricity generation is 0.81 kg CO₂/kWh for Zhejiang province in 2016 (Ding & Ying 2019). According to Cheng and Yao (2021) and Zheng, Song and Shen (2021), carbon emission intensity in China decreases by 0.028% to 0.043% for every 1% increase in the innovation level of renewable energy development. Therefore, the value in the current year (2022) is supposed to be less than 0.81 kg CO₂/kWh. However, no later data than 2016 can be identified for Zhejiang province. Therefore, the figure in 2016 is adopted in this case study to calculate carbon emissions caused by the proposed retrofitting activities, and the calculated carbon emissions are expected to be the maximum value. The detailed calculation can be found in Table E2-2 in Appendix E-2. The type of required construction equipment and calculation results are illustrated in Table 8.7.

Part of building		Retrofitting ac	tivities	Electrical equipment	Energy consumption (MJ)	Carbon emissions (kg CO2)
External walls and roofs	A1	Install/upgrade insulation of building envelopes	External walls (EPS system)	Electric suspended platform Rotary hammer Mortar mixer	65,201.00	14,670.00
			Roof (XPS system)	Rebar cutting machine Bar straightening machine Tapered reverse tilting concrete mixer Mortar mixer Autocrane Concrete pump truck	21,656.00	4,872.61
Windows	A2	Change existing single glazing windows to low-e double glazing with aluminium frames		Electric suspended platform	4,140.00	931.50
Sun shading	A3	Add aluminium sun shading slats for windows facing south		Autocrane AC arc welder	42,822.60	9,635.10
Lights	A4	Install motion set	nsors	Mobile elevator	2,430.00	546.75
	A5	Change from T8 bulbs to T8 LED	fluorescent	Mobile elevator	3,920.00	882.09
Lifts	A6	Change the exist more energy efficiency	ing lifts to cient ones	Electric winch	2,484.00	558.90
Energy supply	A7	Install PV panels		Autocrane AC arc welder	21,275.78	4,764.55
BMCS	A8	Install building e consumption mot	nergy nitoring system		/	/
Water system	A9	Change existing to water-saving c	taps and toilets		/	/
	A10	Install water treat to recycle greywa stormwater	tment system ater and	Autocrane Electric single stage centrifugal water pump	634.80	142.83

Table 8.7. Energy consumption and carbon emissions by operating construction equipment

In addition, embodied energy (EE) and embodied carbon (EC) emissions of the used building materials are also calculated for the retrofitting stage. The source of data for energy intensity and carbon emission coefficient of building materials in China is adopted from research studies by Wang (2009), Yan (2011), Li et al. (2013), Li (2015), Zhang and Wang (2016), Cang et al. (2020), Zhu et al. (2020) and Chen et al. (2022). However, these studies cannot provide all required data for this case study. In China, there is not a comprehensive database for embodied energy and embodied carbon, which can cover all types of construction materials. Therefore, data from outside of China is widely used as proxies for research purposes. In this case study, the database Inventory of Carbon and Energy (ICE) (Hammond & Jones 2011) and several other studies from other countries including Alstone, Mills and Jacobson (2011), AUSTEP

Lighting (2015), Koezjakov et al. (2018), Robati, Daly and Kodogiannakis (2019), Resalati, Kendrick and Hill (2020) and Ahmed et al. (2021) are used as supplements. The detailed calculation and source of data adoption are in Table E2-3 in Appendix E-2. The calculation results for embodied energy and embodied carbon emissions are presented in Table 8.8.

Part of EE **EC** emissions **Retrofitting activities** building (MJ) (kg CO₂) Install/upgrade insulation of External A1 External walls 1,205,186.18 104,351.23 walls and external walls and roofs Roofs roofs 629,312.44 80,951.33 Windows A2 Change existing single-glazing windows to low-e 1,886,920.94 232,414.01 double glazing with aluminium frames A3 Add aluminium sun shading slats for windows 324,800.00 106,288.00 facing south A5 Change from T8 fluorescent bulbs to T8 LED Lights 124,700.00 6,690.00 A7 Install PV panels Energy 394,871.40 24,157.98 supply Water A9 Change existing taps and toilets to Taps 40,000.00 2,240.00 system water-saving ones Toilets 15,609.00 1,006.20

Table 8.8. Embodied energy (EE) and embodied carbon (EC) emissions of installed building materials

In the operation stage, the electricity consumption by operating electricity-related service systems is estimated. It can be evaluated by using simulation tools. However, it is time-consuming and expensive. Considering the time constraint of this study, the saving potential by proposed retrofitting activities in existing research studies is adopted in this study. As for carbon emissions in the operation stage, according to the study by Ding and Ying (2019), the carbon intensity of electricity generation is 0.81 kg CO₂/kWh for Zhejiang province. This figure is adopted to evaluate the carbon emissions in the operation stage in this case study. The detailed calculations are in Table E2-4 in Appendix E-2, and the estimation results are illustrated in Table 8.9.

Upgraded			Annual electricity o building ser	consumption by the vice system	Annual carbon emissions by the building service system	
building service systems		Retrofitting activities	Before retrofitting (MJ)	After retrofitting (MJ)	Before retrofitting (kg CO ₂)	After retrofitting (kg CO2)
HVAC	A1	Install/upgrade insulation for external walls and roofs		581,727.80		206,432.53
A2		Change existing single-glazing windows to low-e double glazing with aluminium frames	1,454,319.50	1,289,981.40	402,765.66	365,789.59
	A3 Add aluminium sun shading slats for windows facing south			271,876.90		373,315.69
Lights	A4	Install motion sensors	2 644 217 20	1,850,952.10	722 201 20	553,816.53
	A5	Change from T8 fluorescent bulbs to T8 LED	2,044,217.30	2,331,017.28	752,501.20	661,831.20
Lifts	A6	Change the existing lifts to more energy efficient ones	66,105.43	46,273.80	18,307.53	13,845.41
Energy supply	A7	Install PV panels	/	-73,377.03	/	-16,509.83
BMCS	A8	Install energy consumption monitoring system	/	-158,653.04	/	-35,696.93
Water system	A9	Change existing taps and toilets to water-saving ones	1	/	/	/
	A10	Install water treatment system to recycle greywater and stormwater	/	6750.00	/	1,518.75

Table 8.9. Evaluation results of annual energy consumption and carbon emissions of building service system after retrofitting

Note: The negative symbol "-" represents energy saving or carbon emission reduction by retrofitting activities.

In addition, the recurrent embodied energy (REE) and embodied carbon (REC) emissions by replacing building materials during the operation stage is estimated based on initial EE and EC emissions of newly added building materials and required replacement frequency. The detailed calculation is in Table A2-5 in Appendix A2, and the estimation results are presented in Table 8.10.

Part of building		Retrofitting activities	REE (MJ)	REC emissions (kg CO2)	
External walls and	A1	Install/upgrade insulation of external walls and roofs	External walls	2,410,372.36	208,702.46
roofs			Roofs	1,384,624.88	161,902.66
Windows	A2	Replace existing single-glazing win low-e double glazing with aluminiu	3,773,841.88	464,828.00	
	A3	Add aluminium sun shading slats		324,800.00	106,288.00
Lights	A5	Change from T8 fluorescent bulbs t	to T8 LED	124,700.00	6,960.00
Energy supply	A7	Install PV panels		394,871.40	24,157.98
Water	A9	Change existing taps and toilets to	Taps	/	/
systems		water-saving ones	Toilets	31,218.00	2,012.40

Table 8.10. Recurrent embodied energy (REE) and embodied carbon (REC) emissions in the operation stage

By summarising the estimation undertaken above, the difference (extra consumption or saving compared to the consumption before retrofitting) on energy consumption and carbon emissions by the proposed retrofitting activities can be obtained as Table 8.11 shows. More detailed calculation of total energy consumption and carbon emissions by the proposed retrofitting activities is in Tables E2-6 and E2-7 in Appendix E-2.

Dant of			Difference on r	etrofitting stage	Difference on operation stage (50 years)		Total difference by retrofitting activities (50 years)	
building		Retrofitting activities	Energy consumption (MJ)	Carbon emissions (kg CO2)	Energy consumption (MJ)	Carbon emissions (kg CO2)	Energy consumption (MJ)	Carbon emissions (kg CO2)
External walls and roofs	A1	Install/upgrade insulation for external walls and roofs	1,984,335.62	204,845.56	-39,834,587.80	-9,446,051.40	-37,850,232.14	-9,241,206.21
Windows	A2	Change existing single-glazing windows to low-e double glazing with aluminium frames	1,891,060.90	233,345.51	-4,443,063.00	-1,383,975.50	-2,552,002.20	-1,150,629.99
	A3	Add aluminium sun shading slats for windows facing south	367,622.60	115,923.10	-6,219,638.00	-1,366,210.50	-5,852,015.40	-1,250,287.40
Lights	A4	Install motion sensors	2,430.00	546.75	-39,663,259.00	-8,924,233.50	-39,660,829.00	-8,923,686.75
	A5	Change from T8 fluorescent bulbs to T8 LED	128,620.00	7,842.09	-15,535,300.00	-3,516,540.00	-15,406,680.00	-3,508,697.91
Lifts	A6	Change existing lifts to more energy efficient ones	2,484.00	558.90	-991,581.50	-223,106.00	-989,097.50	-222,547.00
Energy supply	A7	Install PV panels	416,047.18	28,922.53	-3,273,980.10	-801,333.50	-2,857,932.92	-772,411.99
BMCS	A8	Install energy consumption monitoring system	/	/	-7,932,652.00	-1,784,846.50	-7,932,652.00	-1,784,846.50
Water system	A9	Change existing taps and toilets to water-saving ones	55,609.00	3,246.20	31,218.00	2,012.40	86,827.00	5,258.60
	A10	Install water treatment system to recycle greywater and stormwater	634.80	142.83	337,500.00	75,937.50	338,134.80	76,080.33

Table 8.11. Difference in energy consumption and carbon emissions by reporting activities	Table 8.11. [Difference in	energy	consumption	and carbon	emissions	by retrofitting	g activities
---	---------------	---------------	--------	-------------	------------	-----------	-----------------	--------------

Note: symbol "-" represents energy saving or carbon emissions reduction

Table 8.11 shows that the proposed retrofitting activities cause extra energy consumption and carbon emissions in the retrofitting stage, while contributing to energy saving and carbon emission reduction in the operation stage. By compensating for the consumption and emissions in the retrofitting stage, the total energy saving and carbon emissions by the proposed retrofitting activities can be attained and are used to develop retrofitting strategies in later steps (in Section 8.3.6).

(2) Water consumption

Water consumption is evaluated for the operation stage only, including daily office use, car washing, and irrigation. The proposed retrofitting activities that may achieve water savings include A9 – Replace existing taps and toilets with water-saving taps (hob mount infrared sensor taps) and toilets (siphonix toilet with dual flush tank), and A10 – Install water treatment system for recycling greywater and stormwater.

The flow rate of existing taps is about 6 L/minute. The designed flow rate of the water-saving taps is about 4.5 L/minute. There are 200 taps installed in the building which need to be replaced. Assuming each tap works for about 10 minutes per working day and 250 working days per year, about 750 kL water can be saved annually by replacing water taps with water-saving ones. In addition, the existing toilets need about 9 L water for each flush, but the designed water use of the water-saving toilets is about 4.5 L water each flush. There are 86 toilets in total, and each toilet is used about 5 times per working day on average. Therefore, about 483.75 kL water can be saved annually by replacing existing toilets with water-saving toilets.

With water-saving taps installed, about 9 kL greywater will be generated per day based on the above assumptions. Assuming about 9 kL stormwater can be collected per month, there will be about 189 kL greywater and stormwater generated per month. According to Table A2-1 in Appendix A2, the designed water treatment speed of the water treatment plant is about 40 m³/hour, and the treatment rate is 60%. Therefore, it is capable of recycling the 189 kL of greywater and stormwater per month, resulting in 1,361 kL annual water saving. The recycled water can be used for car washing, irrigation and cleaning as discussed in Section 8.2.1.2. The water saved by the retrofitting activities A9 and A10 is illustrated in Table 8.12.

Retrofitting activities		Number	Original flow rate	Flow rate after retrofitting	Original annual water consumption by taps and toilets (kL)	Annual water consumption by water- saving taps and toilets (kL)	Annual water saving (kL)	Total water saving (in 50 years) (kL)	
A9	Change existing	Taps	200	6 L/minute	4.5 L/minute	3,000.00	2,250.00	750.00	
	taps and toilets to water- saving ones	Toilets	86	9 L/flush	4.5 L/flush	967.50	483.75	483.75	61,687.50
A10	Install wate treatment s to recycle greywater stormwater	er system and r	/	/	/	/	/	1,361.00	68,050.00

Table 8.12. Evaluation results of water saving

(3) Potential reuse or recycling of materials

The percentage of material use of each retrofitting activity that can be recycled for future use is estimated. The steel and concrete of insulation systems (retrofitting activity A1), glass and aluminium of windows (A2), aluminium of sun shading slats (A3), brass of water taps, and ceramics of toilets (A9) are materials that can be recycled or reused. The proportion of recyclable or reusable materials can be obtained by dividing the weight of the recyclable or reusable materials by the total weight of the added materials by the retrofitting activity. Moreover, based on the study by Franz and Wenzl (2017), about 20% of the weight of a LED bulb can be recycled. This figure is adopted to estimate the proportion of recyclable materials by retrofitting activity A5. For lifts and the water treatment plant, no data about recycling or reuse of them is available. However, removed lifts and water treatment plants are often disassembled, and most components can be recycled or reused, such as using them as spare components for repairs. Considering the finishes and lights in lift cars, and broken components, it is assumed that 90% of lifts and water treatment plants can be recycled or reused. As for the building energy consumption monitoring system, no information can be found to calculate the proportion of recyclable or reusable materials. The main components of the monitoring system are electricity meters, which are very light compared to other building materials and service systems applied in the case study. Therefore, it is neglected in this estimation. The estimation results are shown in Table 8.13. The detailed calculation is in Table E2-8 in Appendix E-2.

Part of building		Retrofitting activities	Recyclable/ reusable materials	% of recyclable/ reusable materials
External walls and roofs	A1	Install/upgrade insulation for external walls and roofs	Steel wire Concrete	27%
Windows	A2	Change existing single-glazing windows to low-e double glazing with aluminium frames	• Glass • Aluminium	100%
	A3	Install aluminium sun shading slats for windows facing south	• Aluminium	100%
Lights	A4	Install motion sensors		0%
	A5	Change from T8 fluorescent bulbs to T8 LED	• Glass	20%
Lifts	A6	Change existing lifts to energy efficient ones	Whole lifts	90%
Energy supply	A7	Install PV panels		0%
BMCS	A8	Install energy consumption monitoring system		0%
Water system	A9	Change existing taps and toilets to water- saving ones	Brass Ceramics	100%
	A10	Install water treatment system to recycle greywater and stormwater	Whole water treatment plant	90%

Table 8.13. Evaluation results on the use of reusable/recyclable materials

(4) Waste generation

Waste generation in the operation stage is mainly from replacing broken or obsolescent building materials. They are estimated based on the designed service life, quantity, and weight of replaced building materials. Based on the above estimation of recyclable or reusable materials, the total waste generation by the proposed retrofitting activities can be estimated by multiplying the frequency of replacement during the operation stage with the amount of waste generation by replacement which the total weight of the added material minus the weight of the recyclable or reusable material by each retrofitting activity. The detailed calculation is in Table E2-9 in Appendix E-2, and evaluation results on waste generation are in Table 8.14.

Part of building	Retrofitting activities		Retrofitting activities		Total weight added by a retrofitting activity (kg)	Weight of recyclable or reusable materials (kg)	Total waste generation (kg)
External walls and	A1	Install/upgrade insulation for	External walls	20	478 197	127 340	701 713
roofs		external walls and roofs	Roofs	20	470,197	127,540	701,715
Windows	A2	Change existing single- glazing windows to low-e double glazing with aluminium frames		20	98,224	98,224	0
	A3	Install aluminium sun shading slats for windows facing south		30	112,200	112,200	0
Lights	A4	Install motion sensors		15	450	0	1,350
	A5	Change from T8 fluorescent bulbs to T8 LED		30	667	133	534
Lifts	A6	Change existing l energy efficient o	ifts to nes	30	1,344	1210	134
Energy supply	A7	Install PV panels		30	1,125	0	1,125
BMCS	A8	Install energy con monitoring system	sumption n	15	0	0	0
Water	A9	Change existing	Taps	50	400	400	0
system		taps and toilets to water-saving ones	Toilets	20	1,376	1,376	0
	A10	Install water treat recycle greywater stormwater	ment to and	30	150	135	15

Table 8.14. Evaluation results on waste generation

(5) Indoor environmental quality

Based on the discussion in Section 8.2.2.6, the main problem of indoor environmental quality is the poor indoor visual comfort due to intensive daylight. By changing single-glazing windows to double glazing with low-e coat and installing sun shading slats for windows facing the south, the situation can be improved. However, it is possible that the low-e coat and sun shading slats lead to insufficient interior daylight. Based on the Standard for Daylight Design of Buildings (MOHURD 2013), the daylight factor for office rooms and meeting rooms should be equal to or more than 2%. Daylight factor is designed as the standard value when meeting the requirements of visual function at the specified design illuminance of exterior daylight (450 lx for office buildings located in Zhejiang province) (MOHURD 2013).

Therefore, a simulation of interior daylight is conducted by using the simulation tool Ecotect, developed by Autodesk (Liu & Wang 2019). Ecotect is a commonly used simulation tool which

can provide detailed environmental analysis of buildings, such as daylight radiation and spatial visibility, while providing visualisation analysis, such as images and line charts (Liu & Wang 2019).

According to the floor layout of the case building, the layout of Levels 1 to 4 is basically the same, and the layout of Levels 5 to 13 is basically the same, so in the modelling, Level 2 and Level 10 were selected for simulation, as Figure 8.3 and Figure 8.4 illustrate.







The simulation was conducted with the parameter settings illustrated in Table 8.15, according to the Chinese national standard, Standard for Daylighting Design of Buildings (GB 50033-2013) (MOHURD 2013).

Parameter	Definition	Input value
Phytoclimatic zone	Based on the range of total illuminance, China is divided into different phytoclimatic zone. Hangzhou is in phytoclimatic zone IV, where the design illuminance of exterior daylight is 13,500 lx.	IV
Critical illuminance of exterior daylight	Illuminance of exterior daylight when artificial lighting is required indoors	4500 lx
Reference surface	A surface in which illumination is measured or specified	0.8 m (measured from the floor)
Transmission ratio of windows	Ratio of transmitted luminous flux to incident luminous flux	0.8 (the value for low-e double-glazing windows with aluminium frames)
Standard value of interior daylight illuminance	Illuminance values on a reference surface corresponding to the specified design illuminance values for exterior daylight and the corresponding standard values for the daylight factor	450 lx (the value for office rooms and meeting rooms)
Reflectance ratio	The ratio of the reflected energy to the incident energy.	Internal wall (white painting): 0.75 Ceiling (white painting): 0.75 Floor (tile): 0.4 Glass: 0.08

Table 8.15. Parameters for interior daylight simulation in Ecotect

Source: MOHURD 2013

The simulation results are shown in Figure 8.5, Figure 8.6 and Table 8.16. The results show that, with the implementation of retrofitting activities, changing window types and installing external sun shading slats, the daylight factor of the building is more than 2%, which can meet the requirement of the national standard (GB 50033-2013) (MOHURD 2013).



Figure 8.5. Ecotect simulation results of Level 2





Level	Room (in Figures 8.3 and 8.4)	Range of daylight factor	Limited value
Level 2	2-A	3.2%-23.2%	
	2-B	2.4%-22.4%	
	2-C	2.1%-22.1%	
	2-D	2.0%-21.4%	
Level 10	10-A	2.5%-21.8%	
	10-В	2.2%-21.8%	$\geq 2\%$
	10-C	2.1%-21.5%	
	10-D	4.5%-24.5%	
	10-E	2.0%-21.2%	
	10-F	2.1%-22.1%	

Table 8.16. Simulation results of interior daylight

As for indoor thermal comfort, it can be adjusted in an acceptable range by the use of HVAC. However, the retrofitting activity of install or upgrade insulation of building envelopes may contribute to keeping or improving indoor thermal comfort with less use of HVAC. The U-value of upgraded insulation for external walls, roofs and windows can be found in Tables A1-1 and A1-2 in Appendix A1, which are within the limit value of the local standard DB33/1036-2007 (UAD, ZIAD & CMA 2007). As for indoor acoustic performance, the background noise of the case building was tested (see Section 8.3.1.3), and it is within the limit value of the local standard DB30/118-2010) (MOHURD 2010), and no tenants complained about it. Therefore, no retrofitting activities are proposed for indoor acoustic performance.

Based on the discussion in Section 6.7.1.3 (in Chapter 6), only a pre-assessment of indoor environmental quality was conducted in this study to identify improvement areas for which potential retrofitting activities can be proposed. Combining the discussion in Section 8.2.1.3, it can be concluded that the problem of indoor environmental quality is mainly about indoor visual comfort due to intensive sunlight from the south. Two retrofitting activities may help mitigate the issue, changing existing single-glazing windows to low-e double glazing windows, and installing external sun shading slats. The simulation tool Ecotect was used to simulate the interior daylight of the case building with these two retrofitting activities implemented. The simulation results show that the daylight factor of the case building can meet the requirement of the Chinese national standard (GB 50033-2013) (MOHURD 2013). As for indoor thermal comfort and indoor acoustic comfort, the current performance is regarded as acceptable since no request for improvement was made. However, the two retrofitting activities, install or upgrade insulation of external walls and roof, and change existing windows to low-e double glazing windows, which were proposed to achieve energy saving, may contribute to keeping or improving indoor thermal comfort with less use of HVAC.

Based on the above estimation, the environmental impacts of the proposed retrofitting activities are quantified and converted into numerical data. In the retrofitting stage, the environmental impacts caused by retrofitting construction are assessed, including embodied energy and embodied carbon emissions of building materials and service systems needed by proposed retrofitting activities, energy consumption and carbon emissions by operating construction equipment for implementing retrofitting activities, and the proportion of recyclable or reusable materials used in the proposed retrofitting activities. In the operation stage, the effects (improvement or decline compared to the performance before retrofitting) of the proposed retrofitting activities are assessed, including recurrent embodied energy and carbon emissions, difference (saving or additional amount by retrofitting activities) on energy consumption, carbon emissions, and water consumption by operating upgraded building service systems, proportion of recyclable or reusable materials and waste generation by replacing broken or obsolescent building materials or building service systems. As for indoor environmental quality, the pre-assessment was conducted to evaluate indoor thermal, visual and acoustic comfort to identify improvement areas for which potential retrofitting activities were proposed. It is not included in the decision making of retrofitting strategies in the following steps.

The assessment results that are used in later steps to develop retrofitting strategies, combining the assessment results of economic and social impacts, are summarised in Table 8.17, with all the assessment results rounded to whole numbers.

Assessment criteria		Ilu:4					Retrofittin	g activities				
		Unit	A1	A2	A3	A4	A5	A6	A7	A8	A9	A10
<u>a</u> d	Energy consumption	MJ	1,984,356	1,891,061	367,623	2,340	128,620	2,484	416,047	/	55,609	635
ge	Carbon emissions	kg CO ₂	204,845	233,346	115,923	547	7,842	559	28,923	/	3,246	143
Retrof	Potential for reuse/recycling of materials	%	27	100	100	0	20	90	0	0	100	90
	Potential savings on energy consumption	MJ	-39,834,588	-4,443,063	-6,219,638	-39,663,259	-15,535,300	-991,582	-3,273,980	-7,932,652	31,218	337,500
age	Potential savings on carbon emissions	kg CO ₂	-9,446,051	-1,383,976	-1,366,211	-8,924,234	-3,516,540	-223,106	-801,334	-1,784,847	2,012	75,938
tion st	Potential savings on water consumption	kL	/	/	/	/	/	/	/	/	-61,688	-68,050
Operat	Potential for reuse/recycling of materials	%	27	100	100	0	20	90	0	0	100	90
	Waste generation from replacement	kg	701,713	0	0	1,350	534	134	1,125	0	0	15

Table 8.17 Assessment results of environmental impacts by proposed retrofitting activities

Note: The negative symbol "-" represents energy saving, carbon emissions reduction, or water consumption by retrofitting activities

A1. Install/upgrade insulation for external wall and roofs

A2. Change existing windows to low-e double glazing with aluminium frames

A3. Install aluminium sun shading slats for windows facing south

A4. Install motion sensors for lights

A5. Change T8 fluorescent bulbs to T8 LED

A6. Change existing lifts to energy efficient lifts

A7. Install PV panels

A8. Install building energy consumption monitoring system

A9. Change existing taps and toilets to water-saving ones

A10. Install water treatment system to recycle greywater and stormwater

8.2.4.2. Economic assessment – LCC method

The LCC method is used to calculate the equivalent annual cost (EAC) of the energy and water consumption after implementing the retrofitting activity. Based on Equation 4.3 (in Chapter 4), the initial costs in the retrofitting stage, and operation costs and replacement costs in the operation stage are converted to an equivalent annual value based on the current real discount in China. In this study, a current real discount rate of 8% is taken in the calculation, since it is often adopted for economic analysis in China in the current year, 2022 (Clark, Benoit & Walters 2022; Jin et al. 2022; Li 2022; Xu et al. 2022). The expected service life span of involved building components and service systems are adopted based on Australian Cost Management Manual – Volume 3 (Austalian Institute of Quantity Surveyors 2000), and studies by Kubba (2010), EATS (2015), Penny (2015), Kono et al. (2016), Alam et al. (2017), RICS (2018), and Tavares, Silva and de Brito (2020). The initial cost is estimated based on the relevant study and price provided by local suppliers of building materials and service systems. The sources are included in Appendix E-3. In addition, the electricity price for business use in China in 2022 is CNY 0.64/kWh, and the water price for business use in China in 2022 is CNY 0.49/kL according to CEIC database (CEIC 2022a; CEIC 2022b).

Based on the above assumption and information, the EACs of upgraded or newly added building service systems in a 50-year study period are calculated. The detailed calculation is in Table E3-1 in Appendix E-3. The calculation results are listed in Table 8.18.

				Retrofitting stage				Operatio	n stage				EAC of the upgraded
Upgraded building systems	Ret a	crofitting ctivity	Life expectancy	Initial cost	Annual operation cost	Replacement cost (¥)							building service system after implementing the retrofitting activity
			years	¥	¥	At year 15	At year 20	At year 25	At year 30	At year 35	At year 40	At year 45	¥
HVAC	A1	External walls	20	771 276	102 411		771 276				771 276		155 672
		Roofs	20	//1,3/0	105,411		//1,5/0				//1,5/0		155,075
	A2		20	710,088	299,323		710,088				710,088		302,493
	A3		30	249,480	235,270				249,480				295,341
Lights	A4		15	70,200	329,058	70,200			70,200			70,200	337,356
	A5		30	83,810	414,403				83,810				421,935
Lifts	A6		30	2,625,390	8,226				2,625,390				244,160
PV panels	A7		30	34,900	-13,045				34,900				-9,908
BMCS	A8		15	270,000	-28,205	270,000			270,000			270,000	3,708
Water	A9	Taps	50	11,000 2,724			17 200				47 200		7 1 1 2
system		Toilets	20	47,300	2,734		47,300				47,500		/,115
	A10		30	31,900	533				31,900				3,400

Table 8.18. Assessment of economic performance

Note: The negative symbol "-" represents cost saving.

A1. Install/upgrade insulation for external wall and roofs

A2. Change existing windows to low-e double glazing with aluminium frames

A3. Install aluminium sun shading slats for windows facing south

A4. Install motion sensors for lights

A5. Change T8 fluorescent bulbs to T8 LED

A6. Change existing lifts to energy efficient lifts

A7. Install PV panels

A8. Install building energy consumption monitoring system

A9. Change existing taps and toilets to water-saving ones

A10. Install water treatment system to recycle greywater and stormwater

The EACs in Table 8.18 can only compare the cost saving ability of retrofitting activities when they contribute to electricity saving on the same building service system. For example, retrofitting activities A1, A2 and A3 can reduce electricity saving by HVAC. By implementing A1, A2 and A3, the EAC of HVAC will be CNY 155,675, CNY 302,493 and CNY 295,341, respectively. Therefore, A1 is the best one among them due to the least EAC achieved. However, the cost saving ability of A1, A4 and A6 cannot be directly compared based on Table 8.17, since they contribute electricity saving on different building service systems. In this study, the cost saving abilities of all the proposed retrofitting activities need to be assessed to compare their economic performance. Therefore, the cost and saving in the retrofitting stage and operation stage caused by each retrofitting activity are assessed, and they are converted as the annual equivalent cost or saving as Table 8.19 shows. The detailed calculation is in Table E3-2 in Appendix E-3.

		Retrofitting stage				Operatio	on stage				Equivalent annual
]	Retrofitting activity	Initial cost	Potential savings on annual operation cost	Replacement cost (¥)						cost or saving by retrofitting activities	
		¥	¥	At year 15	At year 20	At year 25	At year 30	At year 35	At year 40	At year 45	¥
A1	External walls Roofs	771,376	-155,127		771,376				771,376		-75,642
A2		710,088	-29,216		710,088				710,088		43,954
A3		249,480	-23,269				249,480				-849
A4		70,200	-141,025	70,200			70,200			70,200	-132,728
A5		83,810	-55,680				83,810				-48,148
A6		2,625,390	-3,526				2,625,390				232,408
A7		34,900	-13,045				34,900				-9,908
A8		270,000	-28,205	270,000			270,000			270,000	3,708
A9	Taps	11,000	(05		47 200				47 200		5 1(0
	Toilets	47,300	-003		47,300				47,500		5,109
A10		31,900	533				31,900				3,400

Table 8.19. Equivalent annual cost or saving by retrofitting activities

Note: The negative symbol "-" represents cost saving.

A1. Install/upgrade insulation for external wall and roofs

A2. Change existing windows to low-e double glazing with aluminium frames

A3. Install aluminium sun shading slats for windows facing south

A4. Install motion sensors for lights

A5. Change T8 fluorescent bulbs to T8 LED

A6. Change existing lifts to energy efficient lifts

A7. Install PV panels

A8. Install building energy consumption monitoring system

A9. Change existing taps and toilets to water-saving ones

A10. Install water treatment system to recycle greywater and stormwater

Based on Table 8.19, retrofitting activity A4 – Install motion sensors for lights has the best economic performance among the proposed retrofitting activities, by creating the largest equivalent annual saving. Changing existing lifts to energy-efficient lifts (A6) has the worst economic performance due to the highest EAC. In addition, considering the retrofitting goal is to reduce at least 30% of operation cost (see Section 8.3.1), the savings on operation cost by each proposed retrofitting activity are calculated and illustrated in Table 8.20. The EAC or saving illustrated in Table 8.19 and saving percentage of operation cost in Table 8.20 are used to develop retrofitting strategies in later steps.

	1		e	
	Retrofitting activity	Original operation cost (¥)	Potential savings on operation cost (¥)	Percentage of saving (%)
A1	Install/upgrade insulation for external walls and roofs		155,127	13.07
A2	Change existing single-glazing windows to low-e double glazing with aluminium frames		29,216	2.46
A3	Install aluminium sun shading slats for windows facing south		23,269	1.96
A4	Install motion sensors		141,025	11.88
A5	Change T8 fluorescent bulbs to T8 LED	1,186,953	55,680	4.69
A6	Change existing lifts to energy efficient ones		3,526	0.30
A7	Install PV panels		13,045	1.10
A8	Install energy consumption monitoring system		28,205	2.38
A9	Change existing taps and toilets to water- saving ones		605	0.05
A10	Install water treatment system to recycle greywater and stormwater		-533	-0.04

Table 8.20. Operation cost saving by retrofitting activities

8.2.4.3. Social assessment – rating score

Value score is adopted to quantify the social impacts of the proposed retrofitting activities. Five professionals of the focus group discussions, including one architect, two facility managers, and two engineers, were invited to assess the social impact of the retrofitting activities. They have all worked on the case building before. Based on their expertise in construction and property management and knowledge of the case building, they may give a proper social score for the proposed retrofitting activities. A questionnaire form was developed and sent to the five professionals by email. Based on the provided rating scale (see Section 6.7.1.2 and Table 6.4 in Chapter 6), they gave rating scores to the proposed retrofitting activities for each identified social criterion. It took about two weeks (17 to 31 August 2020) to collect all five completed questionnaires. The averages of the rating scores given by the five professionals are calculated

based on Equation 6.1 in Chapter 6. The estimation results are listed in Table 8.21, and the detailed calculation is in Appendix E-4.

	Social assessment criteria	A1	A2	A3	A4	A5	A6	A7	A8	A9	A10
Retrofitting	Noise impact on neighbourhood	3.6	3.8	3.6	1	1	1	3	2	1	2.4
stage	Emission impact on neighbourhood	3.8	2.6	3.4	1	1	1	2.8	1.6	1	2.6
	Impacts on safety of undertaking the retrofitting activity	3.6	3.6	3.8	1.6	2.6	2.2	3.2	2	1.6	1.8
	Impacts on cultural heritage	3.8	3.8	4	1.4	1.6	2	3.8	2.4	2.6	3.6
	Impacts on surrounding traffic and pedestrians	4.2	3.4	3.4	1	1	1.4	2	2.2	1.2	2.4
	Impacts on relocating tenants	3.4	3.6	2.8	2.2	3.4	3.2	1.6	1.8	3.6	1.2
Operation stage	Accessibility to building facilities for people with additional needs	3.4	2.4	3.2	3.2	3	4.4	3.2	2.6	4.2	2.4
	Accessibility to building services (such as automatically adjusted indoor temperature and humidity by BMCS)	4.4	3.6	4.4	2.2	3.6	4	3.2	2.8	4.4	3.4
	Impacts on maintenance and maintainability from newly added building component or building system	3.4	2	2.6	2	3.6	3.6	2.2	3.4	3.6	2.4
	Impacts on safety and security of tenants	3.8	3.8	1.8	2	2.8	4.2	2.8	2.6	3.2	2.6
	Impacts on room flexibility for different demands	2.6	2.2	2.8	2	2.6	2.2	2	2.6	2.4	2.2

Table 8.21. Assessment of social impacts by retrofitting activities

Note:

1. Rating score for assessing social performance of retrofitting activities

Impact range on retrofitting stage	Score	Performance range in operation stage
Minimum impact	5	Excellent performance
Minor impact	4	Very good performance
Average impact	3	Good performance
Significant impact	2	Same as current performance
Very significant impact	1	Unsatisfactory performance

2. Retrofitting activities

A1. Install/upgrade insulation for external wall and roofs

A2. Change existing windows to low-e double glazing with aluminium frames

A3. Install aluminium sun shading slats for windows facing south

A4. Install motion sensors for lights

A5. Change T8 fluorescent bulbs to T8 LED

A6. Change existing lifts to energy efficient lifts

A7. Install PV panels

A8. Install building energy consumption monitoring system

A9. Change existing taps and toilets to water-saving ones

A10. Install water treatment system to recycle greywater and stormwater

8.2.4.4. Determine weighted scores of retrofitting activities

To reach an aggregated overall score for the proposed retrofitting activities, the estimation in the environmental, economic and social dimensions is converted into unitless value by using the normalisation method, maximum/minimum ratio (Equations 6.5 and 6.6 in Chapter 6). This normalisation method compares the estimation results of all the proposed retrofitting activities in one assessment criterion. Then, the gap between one retrofitting activity and the best performed one can be measured. The gap is used to compare the performance range of all proposed retrofitting activities (the gap between the best and the worst). In this way, the estimation in different units can be converted to unitless. Moreover, the negative direction of measurement can be converted to positive in which the bigger normalised score will always present the better performance. The detailed process of normalising environmental, economic and social estimations can be found in Table E5-1 to Table E5-3 respectively in Appendix E-5.

Taking the environmental estimation in retrofitting stage as an example as Table 8.17 shows, retrofitting activity A8 has the best performance in energy consumption due to the least energy consumption (0 MJ), while retrofitting activity A1 has the worst performance with the most energy consumption (1,984,356 MJ). Therefore, the performance range of all the proposed retrofitting activities is 1,984,356 MJ. Since the measure direction is negative, the less energy consumption the better the performance. Therefore, Equation 6.6 (in Chapter 6) is used to normalise the estimation of energy consumption in the retrofitting stage of all proposed retrofitting activities. Two decimals are saved in normalisation results, and to have a clear and concise expression, the normalisation results are multiplied with 100 to make them integer. For example, the energy consumption in the retrofitting activity A3 is 367,623 MJ as shown in Table 8.22. Therefore, the normalised score of it can be calculated as:

$$a_{ij}^* = \frac{max_{a_{ij}} - a_{ij}}{max_{a_{ij}} - min_{a_{ij}}} \times 100$$
$$= \frac{1984355 - 367623}{1984356 - 0} \times 100$$
$$= 81$$
Assessment criteria		Energy consumption		Carbon	emissions	Use of recyclable or reusable materials	
Retrofitting activities		Max=1,984,355 MJ; Min=0 MJ		Max=233,345 kg CO ₂ ; Min=0 kg		Max=100%; Min=0%	
		Estimation (MJ)	Normalised score	Estimation (kg CO2)	Normalised score	Estimation (%)	Normalised score
A1	Install/upgrade insulation for external walls and roofs	1,984,355	0	204,845	12	27	27
A2	Change existing single-glazing windows to low-e double glazing with aluminium frames	1,891,061	5	233,345	0	100	100
A3	Install aluminium sun shading slats for windows facing south	367,623	81	115,923	50	100	100
A4	Install motion sensors	2,430	100	547	100	0	0
A5	Change T8 fluorescent bulbs to T8 LED	128,620	94	7,842	97	20	20
A6	Change existing lifts to energy efficient ones	2,484	100	559	100	100	100
A7	Install PV panels	416,047	79	28,923	79	0	0
A8	Install energy consumption monitoring system	0	100	0	100	0	0
A9	Change existing taps and toilets to water-saving ones	55,609	97	3,246	97	100	100
A10	Install water treatment to recycle greywater and stormwater	635	100	143	100	90	90

Table 8.22. Normalisation of environmental estimation in retrofitting stage

The weights of the three sustainability dimensions and environmental and social assessment criteria for the operating model were determined at the focus group meeting (see Table 7.25 and Table 7.26 in Chapter 7). As for the economic dimension, the LCC method was employed to evaluate equivalent annual cost and saving by the retrofitting activities. In this method, the costs in different stages are regarded as equal. Therefore, their weights are equal to 1. By multiplying the normalised score with associated weights, the weighted scores of each proposed retrofitting activity in the environmental and social dimensions can be attained. Due to the equal importance of calculated costs, the weighted scores in one sustainability dimension, the total weighted score of the ten proposed retrofitting activities in the three sustainability dimensions can be achieved. The detailed calculation are in Table E5-4 in Appendix E-5, and the results are illustrated in Table 8.23.

	Detwofitting estimities	Total weighted score after normalisation				
	Retrolitting activities	Environmental	Economic	Social		
A1	Install/upgrade insulation for external walls and roofs	51.3	84.0	82.0		
A2	Change existing single-glazing windows to low-e double glazing with aluminium frames	62.0	52.0	48.8		
A3	Install aluminium sun shading slats for windows facing south	94.5	64.0	64.3		
A4	Install motion sensors	101.3	100.0	106.5		
A5	Change T8 fluorescent bulbs to T8 LED	85.1	77.0	134.1		
A6	Change existing lifts to energy efficient ones	102.5	0.0	159.7		
A7	Install PV panels	60.0	66.0	69.8		
A8	Install energy consumption monitoring system	72.5	63.0	114.6		
A9	Change existing taps and toilets to water- saving ones	116.3	62.0	157.2		
A10	Install water treatment to recycle greywater and stormwater	114.1	63.0	86.1		

Table 8.23. The total weighted scores in the three sustainability dimensions

8.2.5. Develop retrofitting strategies

A binary linear mathematical model (Equation 6.9 in Chapter 6) is established to develop retrofitting strategies. Three elements need to be specified to establish the mathematical model: retrofitting goals, project constraints, and weights of the three sustainability dimensions. As discussed in Section 8.2.1, two retrofitting goals are defined. One is a must-achieve goal: to achieve at least a 30% reduction of operation cost, and another is an expectation: to obtain a national green building label. Therefore, the Chinese national green retrofitting standard:

Assessment Standard for Green Retrofitting of Existing Buildings (GB/T 51141-2015) (MOHURD 2015) is used to assess the performance of the developed retrofitting strategies to check whether the developed retrofitting strategies can help the case building achieve the green building label. Meanwhile, the effectiveness of the decision model can further confirm if the green label can be achieved.

The project constraint is the proposed budget by the building owners of CNY 3 million. The weights of the three sustainability dimensions were developed in focus group meetings (see Table 7.26 in Chapter 7). By substituting these three figures into Equation 6.9 in Chapter 6, the binary linear mathematical model for developing retrofitting strategies for the case building can be established, as Equation 8.1 shows.

$$\max Z = 0.437 \sum_{j=1}^{10} P_j^{en} x_j + 0.324 \sum_{j=1}^{10} P_j^{ec} x_j + 0.239 \sum_{j=1}^{10} P_j^s x_j \quad \text{Equation 8.1.}$$

$$s.t. \begin{cases} \sum_{j=1}^{j=1} a_{1j} x_j \ge 30\% \\ \sum_{j=1}^{j=1} a_{2j} x_j \le 3,000,000 \\ x_j = 0,1 (j = 1, 2, ..., 10) \end{cases}$$

Where,

- Z the overall performance of a retrofitting strategy
- P_j^{en} the total weighted environmental score of retrofitting activity *j*, which can be found in Table 8.21
- P_j^{ec} the total weighted economic score of retrofitting activity *j*, which can be found in Table 8.21
- P_j^s the total weighted social score of retrofitting activity *j*, which can be found in Table 8.21
- $\sum_{j=1} a_{1j} x_i \ge 30\%$ the constraint function for satisfying the retrofitting goal, to achieve at least 30% saving of operation cost, and a_{1j} is the saved operation cost by each proposed retrofitting activity, which can be found in Table 8.20
- $\sum_{j=1} a_{2j} x_i \leq 3,000,000$ the constraint function for subjecting to the project constraint, CNY 3,000,000 project budget, and a_{2j} is the initial cost of each proposed retrofitting activity, which can be found in Table 8.18.
- $x_j = 0,1$ whether retrofitting activity x_j is selected. When $x_j = 0$, it is not selected. When $x_j = 1$, it is selected.

Excel is used to formulate Equation 8.1 as Table 8.24. By undertaking "SUMPRODUCT" formula and "SOLVER" order in Excel, the "optimal" retrofitting strategy can be developed, which can maximise overall improvement in environmental, economic and social dimensions within the CNY 3 million project budget, and reach the retrofitting goal, to achieve at least a 30% saving on operation cost. The selected retrofitting activities by running the program will be noted as "1" in the "select or not" column. The developed results by Excel using Solver are presented in Table 8.25.

Retrofitting activity (a_j) Environmental dimension (P_j^{em}) (weight=0.437)		Economic dimension (P _j ^{ec}) (weight=0.324)	Social dimension (P_j^s) (weight=0.239)	% of operation cost saving by retrofitting activities (a_{1j})	Initial cost of retrofitting activities (a _{2j})	Select or not (x _j)
A1	51.3	84.0 82.0 13.07%		771,376		
A2	62.0	52.0	48.8	2.46%	710,088	
A3	94.5	64.0	64.3	1.96%	249,480	
A4	101.3	100.0	106.5	11.88%	70,200	
A5	85.1	77.0	134.1	4.69%	83,810	
A6	102.5	0.0	159.7	0.30%	2,625,390	
A7	60.0	66.0	69.8	1.10%	34,900	
A8	A8 72.5		114.6	2.38%	270,000	
A9 116.3		62.0	157.2	0.05%	58,300	
A10	114.1	63.0	86.1	-0.01%	31,900	
The overall w	eighted score of the	e developed retrofit	ting strategy (Z):		•	
max Z = 0	.437 $\sum_{j=1}^{10} P_j^{en} x_j +$	$0.324 \sum_{j=1}^{10} P_j^{ec} x_j +$				
Operation cos	st saving by the dev	eloped retrofitting	strategy (%):			
	$\sum_{j=1} a_{1j} x_j \ge$	30%				
Total initial cost of the developed retrofitting strategy (\mathbf{Y}):						
	$\sum_{j=1} a_{2j} x_j \le$	3,000,000				

Table 8.24. Establishing Equation 8.1 in Excel

Retrofitting activity (a_j)	rofitting ity (a_j) Environmental dimension (P_j^{en}) (weight=0.437)Economic dimension (P_j^{ec}) (weight=0.324)Social dimension (P_j^S) (weight=0.239)% of operat saving by ref activities (% of operation cost saving by retrofitting activities (a_{1j})	Initial cost of retrofitting activities (a _{2j})	Select or not (x _j)	
A1	51.3	84.0	82.0	13.07%	771,376	1
A2	62.0	52.0	48.8	2.46%	710,088	1
A3	94.5	64.0	64.3	1.96%	249,480	1
A4	101.3	100.0	106.5	11.88%	70,200	1
A5	85.1	77.0	134.1	4.69%	83,810	1
A6	102.5	0.0	159.7	0.30%	2,625,390	0
A7	60.0	66.0	69.8	1.10%	34,900	1
A8	72.5	63.0	114.6	2.38%	270,000	1
A9	116.3	62.0	157.2	0.05% 58,300		1
A10	114.1	63.0	86.1	-0.01%	31,900	1
The overall w $max Z = 0$	reighted score of the .437 $\sum_{j=1}^{10} P_j^{en} x_j +$	e developed retrofit 0.324 $\sum_{j=1}^{10} P_j^{ec} x_j +$		741.65		
Operation cos	st saving by the dev $\sum_{j=1}^{n} a_{1j} x_j \ge$	eloped retrofitting	37.58%			
Total initial cost of the developed retrofitting strategy (¥): $\sum_{j=1}^{2} a_{2j} x_j \le 3,000,000$				2,280,054		

Table 8.25. The developed retrofitting strategy for the case building

According to Table 8.25, nine out of the ten proposed retrofitting activities are selected to amalgamate the "optimal" retrofitting strategy, except retrofitting activity A6 (change existing lifts to energy efficient lifts). The developed retrofitting strategy (consists of retrofitting A1– A5 and A7–A10) can achieve about a 38% saving on operation cost with CNY 2,280,054 initial cost. The overall weighted score of the retrofitting strategy regarding environmental, economic and social dimensions is 741.65. Indeed, changing existing lifts to energy efficient ones can only achieve about 0.3% saving on operation cost with the highest initial cost among all proposed retrofitting activities.

Based on estimation of proposed retrofitting activities, the national green building label for retrofitting (GB/T 51141-2015) (MOHURD 2015) is used to check whether the case building can achieve a green building label by using the developed retrofitting strategy. In the standard, assessment is conducted in two stages. The first stage is the pre-assessment, which is conducted after the retrofitting design is finished. Five assessment categories are identified for the pre-assessment, planning and architecture, structure and materials, performance of HVAC,

performance of water system, and electricity use. The second stage is to assess the performance of the building after retrofitting construction is finished. In addition to the above five categories, another two assessment categories are identified for this stage – construction management, and operation management. The pre-assessment is conducted in this case study to assess the performance of the developed retrofitting strategy.

Same as most of GBRS, the scores assigned in the standard (GB/T 51141-2015) (MOHURD 2015) are also unitless. GB/T 51141-2015 classifies the greenness or sustainability level into three grades – one star, two stars or three stars, based on the achieved overall score:

- $50 \le \text{overall score} < 60$: \bigstar
- $60 \leq \text{overall score} < 80: \bigstar \bigstar$
- Overall score ≥ 80 : $\star \star \star$

The national standard is a self-assessment system; by ticking off the items that have been met, the associated score assigned by the standard can be achieved. The overall score is the sum of the product of the achieved score in each category and the corresponding weight (given in the standard). By conducting the self-assessment based on this national standard (GB/T 51141-2015) (MOHURD 2015), the building can be labelled as a two-star building if using the developed retrofitting strategy. The assessment results are illustrated in Table 8.26.

Assessment criteria	Weight (given in the standard)	Achieved score
Plan and architecture	0.21	73
Structure and materials	0.19	70
Performance of HVAC	0.27	60
Performance of water system	0.13	41
Electricity use	0.20	52
Overall score	60.56	
Grade of green building	**	

Table 8.26. Assessment results based on the national green building standard

8.2.6. Conduct sensitivity analysis

The sensitivity analysis is conducted to identify uncertainties that may affect decision results, and to check the scale of effects that may be caused by identified uncertainties. The OAT method is adopted to check whether uncertainties affect decision making results. In OAT, one

parameter is changed at a time to check how much it may affect the decision-making result. In this case study, the discount rate, study period, and energy saving potential of proposed retrofitting activities are assumed based on existing studies, which may cause uncertainties. The discount rate may be more or less than the assumed one in the case study (8%). Therefore, $\pm 5\%$ based on the adopted discount rate (8%) is used to check how much effect may be caused by adopting different discount rates. A 50-year remaining service life is assumed in this case study, which is a common assumption in life cycle studies, and ± 20 years (30 years and 70 years) based on the adopted study period (50 years) is used to check how much effect may be caused by adopting different study periods. In addition, the potential energy savings by different retrofitting activities are adopted from existing studies, which may be sensitive to assessment results and decision making of retrofitting strategies. Based on Table E2-4 in Appendix E-2, retrofitting activities A1 (install/upgrade insulation for external walls and roofs) and A4 (change existing lifts to energy efficient lifts) have the largest potential for energy saving, which is about 12% for each. Therefore, \pm 5% based on the adopted figures is used to check how much effect the adopted energy saving potential for retrofitting A1 and A4 may have.

Based on the above discussion, the sensitivity analysis is conducted by repeating the process from Section 8.2.4 to Section 8.2.5 with one parameter changed at a time. The analysis results are illustrated in Table 8.27, and changed parameters are summarised as below:

- Discount rate: 3% or 13%
- Study period: 30 years or 70 years
- Energy saving potential on HVAC by retrofitting activity A1: 55% or 65%
- Energy saving potential on lighting system by retrofitting activity A4: 25% or 35%

Changed parameter		Selected retrofitting activities	% of reduced annual operation cost	Total energy saving achieved in operation stage (MJ)	Equivalent annual saving by selected retrofitting activities (¥)	Overall weighted score (Z)
Discount	i=3%		37.58%	116,533,762	282,519	719.29
rate (i)	i=13%		37.58%	116,533,762	120,497	757.53
Study	30 years		37.58%	71,202,495	204,890	740.35
period	70 years		37.58%	161,840,638	209,639	741.97
Energy saving potential	A1: 55% saving on HVAC system	A1–A5,	36.37%	112,897,965	195,007	771.53
of retrofitting activity	A1: 65% saving on HVAC system	A7–A10	38.67%	120,169,561	224,417	748.08
A1 and A4	A4: 25% saving on lights		35.6%	109,923,219	187,986	753.25
	A4: 35% saving on lights		39.56%	123,144,306	234,994	727.46

Table 8.27. Results of sensitivity analysis

Based on the results of the sensitivity analysis shown in Table 8.25, the changed parameters have no effect on the strategy development for the case study, in which A1, A2, A3, A4, A5, A7, A8, A9 and A10 are always selected and amalgamated as a retrofitting strategy in each analysis. However, they do have an effect on energy consumption and costs. Based on Table 8.25, the percentage of reduced annual operation cost by the developed retrofitting strategy does not change with a different discount rate or study period since these changes have equal impacts on operation cost. As for the changed energy saving potential of retrofitting activities A1 and A4, the bigger potential of them may contribute to more saving on annual operation cost, and the strength of impact by the change in retrofitting activity A4 is bigger than it is in retrofitting activity A1. In addition, the changes in Table 8.25 mainly impact the energy consumption in the operation stage. The longer study period is assumed, the more energy saving can be achieved in the operation stage. All the changes have an effect on the equivalent annual saving, and the change of discount rate has the biggest impact.

By changing these parameters, different overall weighted scores are achieved. It is noticed that the bigger energy saving potential of retrofitting activities A1 and A4 contributes to a smaller overall weighted score, which sounds wrong. In fact, it is correct. This situation is caused by the normalisation method, maximum/minimum ratio. In this normalisation method, by changing to a bigger energy saving potential, the performance range regarding energy saving $((max_{a_{ii}} - min_{a_{ii}})$ in note 1 and 2) used to be 40,172,688 MJ (337,500 – (-39,834,588))

referring to Table A4-1 in Appendix A4. By changing the energy saving potential of retrofitting activity A1 from 60% to 65%, it achieves the largest energy saving compared to others. Accordingly, the performance range increases to 43,807,887 MJ (337,500 – (-43,479,387)). With the bigger denominator, even though the numerator remains unchanged (for other retrofitting activities), the normalised score will become smaller, resulting in a smaller overall weighted score. Therefore, the overall weighted scores can only be used to compare the overall sustainability performance of different amalgamations of a certain group of retrofitting activities. For example, there are ten retrofitting activities proposed for the case building. Among these ten retrofitting activities, the overall weighted score of retrofitting activities A1, A2 and A4 is bigger than of retrofitting strategy consisting of A1, A2 and A4 is better than the one consisting of A1, A2 and A4. If another retrofitting activities cannot be used to compare with the scores when only ten activities are proposed.

Based on the above discussion, the changes of $\pm 5\%$ on the discount rate and energy saving potential of retrofitting activities A1 and A4, and ± 20 years on the study period adopted in the case study do not affect decision-making results. In reality, the chance that these parameters change over these change ranges is small. Therefore, no further steps are needed to check how much effect they may have.

However, if results by the OAT method indicate that the decision results will be changed by changing selected parameters, it is necessary to check the effect on solutions that may be caused by these parameters. In this case, the range of changes need to be estimated first based on assumed probability distribution, such as a normal distribution. Then, based on the estimated range of changes, simulation tools for sensitivity analysis can be adopted to check the effects of these parameters, individually or jointly.

8.2.7. Determine the most preferred retrofitting strategy

The retrofitting strategy developed in Section 8.2.5 is the preferable solution regarding the identified retrofitting goal and the given project constraint. By relaxing or tightening the project

constraint of the CNY 3 million project budget, four other options are developed for the building owners, as shown in Table 8.28.

Retrofitting strategies			Initial cost		% of		
		Selected retrofitting activities	¥	% of the original budget	reduced annual operation cost	Overall weighted score	Grade of green building
1	The preferable strategy (the one developed in Section 8.3.6)	A1–A5, A7–A10	2,280,054	76%	37.58%	741.65	**
2	Including all proposed retrofitting activities	A1–A10	4,905,444	164%	37.88%	818.59	**
3	Reducing budget to 90% of the initial cost of Strategy 1	A1, A3–A5, A7–A10	1,569,966	52%	35.12%	681.16	*
4	Reducing budget to 60% of the initial cost of Strategy 1	A1, A4–A5, A7–A10	1,320,486	44%	33.16%	604.42	*
5	Reducing budget to 50% of the initial cost of Strategy 1	A1, A4–A5, A7, A9–A10	1,050,486	35%	30.78%	525.44	/

Table 8.28. Alternative strategies by altering the project budget

The preferable retrofitting strategy has included nine of the ten identified retrofitting activities. To amalgamate all the ten retrofitting activities as a retrofitting strategy (retrofitting strategy 2), the project budget has to increase by 64% (to CNY 4,905,444 as Table 2028 shows). Retrofitting strategy 2 may contribute to a 37.88% reduction in annual operation cost and attain the highest overall weighted score (818.59) among all alternatives and a two-star green building label.

The preferable retrofitting strategy developed in Section 8.3.6 has an initial cost of CNY 2,280,054, which is 24% below the budget. Therefore, by tightening the budget to 90% of the initial cost of retrofitting strategy 1, the retrofitting strategy (retrofitting strategy 3) amalgamated by retrofitting activities A1, A3, A4, A5, A7, A8, A9 and A10 is developed by running Equation 8.1 in Excel. Retrofitting strategy 3 can achieve a 35.12% operation cost saving with CNY 1,569,966 initial cost (69% of the initial cost of retrofitting strategy 1), and a one-star green building label, as shown in Table 8.28. By tightening the budget to 60% of the initial cost of retrofitting activities A1, A4, A5 and A7–A10 is developed. It can achieve 33.16% of operation cost saving with

CNY 1,320,486 initial cost (58% of the initial cost of retrofitting strategy 1), and a one-star green building label. Finally, by tightening the budget to 50% of the initial cost of the "optimal" solution, alternative 5 can be developed, which contributes to 30.78% of operation cost saving with CNY 1,050,486 initial cost, and no green building label is achieved.

To conclude the results of retrofitting strategy development, there are five options for the building owners. Three options (retrofitting strategies 1, 3 and 4) can satisfy the retrofitting goals within the project budget with different levels of achievement. Retrofitting strategy 2 can best meet the retrofitting goals compared to others, but the initial cost may be over the budget by 64%. Retrofitting strategy 5 can meet the retrofitting goal for operation cost saving within the project budget, but no green building label can be achieved. Moreover, the sensitivity analysis indicates no significant effect on the decision-making results by changing the discount rate, study period or energy saving potential of retrofitting activities. With the given options, the building owners can make the final decision: spending the least money just to meet the retrofitting goals, spending more money to better meet the retrofitting goals, or spending more than half of the budget to "best" meet the retrofitting goals.

8.3. Outcomes and discussions

This chapter and the previous one conducted a case study to verify the validity of the conceptual model and illustrate the process of quantifying the conceptual model to develop retrofitting strategies for existing office buildings. In the case study, five retrofitting strategies are developed (see Table 8.28) regarding the identified retrofitting goals and project constraints, and the building owners can choose the preferred one to carry out. The validity and flexibility of the conceptual model, as well as the limitations, are discussed in the following sections.

8.3.1. Validity of the conceptual model

One purpose of the case study is to illustrate the validity of the conceptual model, by which a retrofitting strategy can be developed for the case building, which can maximise improvement in the environmental, economic and social dimensions within project constraints, and meet retrofitting goals at the same time.

The developed conceptual model can be adapted to suit office buildings in different climate zones. The case building is an office building in Hangzhou, Zhejiang, China. Therefore, it needs to be adapted to the Chinese retrofitting market before using it to develop retrofitting strategies for the case building. A two-stage data collection process from the broad to the specific was conducted in this case study to collect opinions from local professionals in the construction and property sector and key stakeholders, including building owners and tenants. Based on the collected opinions, the retrofitting activities and assessment criteria are revised to suit the situation of sustainable retrofitting in China, in which the conceptual model is converted to an operating model for Chinese retrofitting projects. By applying the operating model to develop retrofitting strategies for the case building, the retrofitting goals, problems of the case building, and suitable retrofitting activities are identified based on consultation with building owners, facility managers, and tenants, as well as investigation of the situation of the case building via a comprehensive audit. In addition, local data, such as applicable construction equipment for retrofitting construction, locally available building materials and service systems, and associated initial cost based on information provided by local suppliers, is used to measure the performance of the proposed retrofitting activities based on the finalised assessment criteria. In addition, the Chinese national standard for green retrofitting is used to confirm the effectiveness of the developed retrofitting strategies. Based on Table 8.26, the preferable retrofitting strategy, which maximises overall sustainability improvement within project constraints, can achieve two green stars. With a smaller overall weighted score achieved by retrofitting strategies 3 and 4, one green star can be achieved. The retrofitting strategy 5, having the lowest overall weighted score, cannot achieve a green building label.

To conclude, the retrofitting strategies developed for the case building are based on opinions from local professionals and key stakeholders of retrofitting, deep investigation of the case building's problems, and proper estimation of the environmental, economic and social performance of the proposed retrofitting activities. The effectiveness of developed retrofitting strategies is also confirmed based on the Chinese national standard of sustainable retrofitting. Therefore, it can be stated that the case study illustrates the validity of the developed conceptual model. It provides a detailed illustration of using the conceptual model to develop retrofitting strategies for a real office building.

8.3.2. Applicability and flexibility of the conceptual model

China is a large country with different climate zones, resulting in different retrofitting demands. Therefore, it is a good opportunity to test the flexibility of the conceptual model. Due to historical energy scarcity and relevant heating standards, central heating is only available in the north for varying lengths of time. In the south, no district heating substations and longdistance heating pipelines have ever been provided. As a result of economic development over the past decades people who do not access central heating have begun to use air conditioners or heaters. For now, central heating is still available in the north, but mainly for residential buildings. For public buildings, especially office buildings, HVAC is becoming the main heat supply just as in the south. Based on the study by Li et al. (2018), apart from office buildings built before the 1980s, which were fitted with different cooling and heating facilities, such as fans and central heating supply, the newer office buildings built after that are air conditioned across the whole country. It means office buildings in China work in a similar way for cooling and heat supply. Consequently, the applicable retrofitting activities and suitable assessment criteria for measuring performance of retrofitting activities are similar in both regions. The difference of demands for retrofitting office buildings in the north and the south becomes small, such as different thickness of insulation materials needed in the two regions, which will not reduce the applicability of the operating model.

In addition, the iterative process of modifying retrofitting activities and assessment criteria to be suitable for use in China (illustrated in Figure 8.7) contributes to the increased flexibility of the operating model. In this case study, the assessment criteria and retrofitting activities are refined twice. To convert the conceptual model to an operating model, the retrofitting activities and assessment criteria are modified to suit the local situation based on a questionnaire survey and three focus group discussions. Based on the result of the survey, the identified retrofitting activities have been used by participants from both the north and the south, but with different proportions. It means all the provided retrofitting activities are applicable in northern and southern China. In addition, the added assessment criteria by the survey and focus groups, including impacts on surrounding traffic and pedestrians from retrofitting construction and the adaptability for different users, are actual impacts and expected performance from retrofitting activities, no matter where they are conducted. The added retrofitting activity, adding or upgrading insulation for the walls that subdivide the tenancy areas, is also suitable for both

regions. Therefore, there is nothing to indicate the operating model is not applicable and flexible to be used for retrofitting strategies for office buildings in China.



Figure 8.7. An iterative process of information refining

In addition, the model is able to be adapted again based on the identified problems and specified retrofitting activities for the retrofitted building by removing irrelevant assessment criteria, adding new ones, and specifying required ones. For example, in this case study, due to the small amount expected, the waste generation in the retrofitting stage is not considered. According to the requirement by the local standard, the heat transfer coefficient of external walls should not be over 1 W/m²•K. Due to the flexibility of the model, the actual situation of the retrofitting project can be considered, so that the developed retrofitting strategies can effectively improve the building's sustainability performance and meet the identified retrofitting goals.

However, two considerations about this case study need to be noted. First, the diversity of building materials and service systems cannot be considered in the developed model. For

example, there are different insulation materials, which have different embodied energy, life span, insulation performance, initial costs etc. Different selection of insulation materials may result in different retrofitting strategies being developed. In principle, several different materials for each retrofitting activity can be included, but it may be time consuming, and even lead to an infinite decision-making process if the number of proposed retrofitting activities is large. Therefore, to develop suitable retrofitting strategies, it is important to consult with professionals who are able to identify retrofitting activities that have desirable traits and reasonable costs, and available construction technologies to implement them. In addition, there is a lack of post-occupancy evaluation, which can be used to obtain feedback on the effect of the developed retrofitting strategy and also to test the accuracy of estimation. A post-occupancy evaluation may need data on operating the building for at least one year after retrofitting, which cannot be finished in the study period of the thesis study. A post survey may be conducted as a follow-up study after the thesis study.

8.4. Summary

In summary, this chapter gave a detailed description of a case study, which demonstrates the process of using the operating model to develop retrofitting strategies for an office building in China. In the case study, five retrofitting strategies are developed with different levels of achieving retrofitting goals and costs. Combined with the discussion in Chapters 6 and 7, these three chapters present how the research question is answered: how can we develop a retrofitting strategy that can effectively improve a building's environmental, economic and social performance in a balanced way? With developed retrofitting strategies for the case building, the research proposition was justified, which it is possible to improve sustainability – which could be sufficient to allow economic growth and improved social wellbeing – by retrofitting existing buildings; and that this can be done by using the triple-bottom line approach. In the next chapter, the whole study is discussed and the thesis concluded.

Chapter 9. Summary and conclusions

9.1. Introduction

This chapter presents a summary of the study. It encapsulates the findings from the literature review, the development of a conceptual model for deciding retrofitting strategies for office buildings, and a case study that verifies the conceptual model and illustrates the process of implementing the model. The conclusions in this chapter link and integrate the research findings with the research aims and objectives. Meanwhile, based on the contribution and limitation of this study, suggestions for future research are also provided.

This thesis critically examines the current performance of existing office buildings in the environmental, economic and social dimensions, investigates common retrofitting activities for improving existing office buildings' performance in these three dimensions, and reviews existing assessment methods and decision-making frameworks of sustainability.

In acknowledging the substantial negative environmental, economic and social impacts from existing office buildings, and the effectiveness of sustainable retrofitting, this study develops a conceptual model for deciding retrofitting strategies for office buildings based on a comprehensive literature review. Then, it converts the conceptual model to an operating model to suit the local situation of a sustainable retrofitting based on a survey and three focus group discussions. In the end, the operating model is implemented to develop retrofitting strategies for the case building. The results show that it is possible to improve sustainability while potentially allowing economic growth and improved social wellbeing by retrofitting existing buildings using the triple-bottom line approach.

Based on the research process and achieved outcomes, this chapter discusses and summarises the study by providing an overview of research aims and objectives (Section 9.2), a summary of the whole research (Section 9.3), summary of contribution to knowledge (Section 9.4), discussion of the study limitations (Section 9.5), and recommendation for future studies (Section 9.6).

9.2. Overview of aims and objectives

This study has satisfied the research aim which is to develop a model for developing retrofitting strategies for office buildings to improve their sustainability performance (see Chapter 1). This model is able to develop retrofitting strategies for existing office buildings based on estimation of impacts of potential retrofitting activities in the environmental, economic and social dimensions. In addition, the long-term perspective is taken in the model which considers impacts of retrofitting activities in the retrofitting stage and operation stage. Ultimately, the model can help improve performance of existing buildings by covering all the sustainability dimensions from a long-term perspective. This study has satisfied several objectives to realise the research aim, as summarised below.

9.2.1. Current performance of existing office buildings in triple-bottom line

aspects

The first objective is to examine the current performance of existing office buildings in the environmental, economic and social dimensions. The literature review in Chapters 2 and 3 confirmed the significance of sustainable development and the necessity of sustainable retrofitting to reduce negative impacts from the building sector. The literature revealed that the construction sector is consuming many resources and generating pollution of the natural environment, and it is a crucial field for realising sustainable development for the whole of society. To achieve overall sustainable construction, the impacts of construction activities in the triple-bottom line aspects – the environmental, economic and social dimensions – need to be considered.

The literature shows that most existing office buildings have poorer environmental sustainability performance than new buildings because of the stricter newer regulations about buildings' environmental impacts and increasing demands for indoor environmental quality. Due to the poor energy efficiency and outdated building materials and service systems, existing office buildings have higher operation and maintenance costs. On the social dimension, aging office buildings may lead to poor indoor comfort and less job satisfaction for building occupants. Considering the large quantity of existing buildings, their long service life, and the identified massive negative impacts, retrofitting is recognised as being a better remedy than

demolition and new construction. Sustainable retrofitting can quickly improve existing buildings' environmental, economic and social performance, and can also retain the embodied energy and embodied carbon emissions in existing buildings and avoid more of them being created by new construction. There is a clear call for the construction and property management sector to adopt sustainable retrofitting to reduce their negative impacts on the environmental, economic and social dimensions.

9.2.2. Existing assessment methods for building retrofitting

The second research objective is to review literature about assessment methods for sustainable buildings and suggest ways to realise comprehensive assessment for sustainability and to aid developing sustainability strategies. In Chapter 4, the common assessment methods for evaluating sustainability performance of constructions were examined with the analysis of their benefits and limits. The literature reveals that most existing assessment methods fail to embrace all the three sustainability dimensions by focusing on one or combining the environmental and economic dimensions with the absence of the social dimension. Therefore, the assessment results of them cannot provide a complete picture of the sustainability performance of potential sustainability strategies.

The literature identifies two methods that can include all the three sustainability dimensions in assessment and the decision-making process: life cycle sustainability assessment and multicriteria decision making. In life cycle sustainability assessment, the three sustainability dimensions are assessed based on the same methodology of life cycle assessment, but the mutual relations between the three dimensions are neglected in this method. In multi-criteria decision making, appropriate assessment methods in the three dimensions can be integrated and trade-offs between different assessment criteria can be analysed and balanced. Therefore, it is adopted in this study to build the conceptual model for deciding on retrofitting strategies.

The literature has revealed that existing sustainability assessments are insufficient to include all the three sustainability dimensions and analyse trade-offs between them. There is a need for a multi-dimensional framework that can provide a comprehensive assessment of sustainability and be an aid to decision making.

9.2.3. Identifying assessment criteria for sustainable retrofitting

The third and fourth research objectives are to identify suitable assessment criteria and tradeoffs between the three sustainability dimensions, which need to be considered for retrofitting projects. Chapter 4 reviewed existing assessment methods of sustainability. Most methods are for new buildings, and there is a lack of an assessment method for sustainable retrofitting which can include all the three sustainability dimensions. In Chapter 3, the potential improvements by sustainable retrofitting are identified, and common retrofitting activities for office buildings are discussed. Based on information derived from the literature review, the desired attributes of sustainable retrofitting and existing office buildings in the three sustainability dimensions are identified and associated assessment criteria from relevant studies are selected in Chapter 6 as assessment criteria in the developed model.

Chapter 6 reviewed the general retrofitting process and showed that retrofitting is a complicated process with the involvement of different stakeholders at different life stages of the retrofitting building. Identifying trade-offs between different sustainability dimensions from a long-term perspective is key to weighing up the perspectives of different stakeholders. Chapter 6 defined the measurement of point-to-point trade-offs between assessment criteria. Since the three sustainability dimensions are considered for sustainable retrofitting in the developed model, the derivative of the point-to-point measurement by considering trade-offs between the three sustainability dimensions is investigated and represented in Appendix A.

In the literature, there is a clear call for a comprehensive model for sustainable retrofitting that can provide specified assessment criteria for each sustainability dimension and consider tradeoffs between them by which suitable retrofitting strategies can be developed.

9.2.4. Developing a conceptual model for deciding retrofitting strategies for office buildings

The fifth research objective is to develop a conceptual model for developing retrofitting strategies for office buildings using the triple-bottom line approach. The literature review in Chapters 2 to 4 recognised that humans contribute to climate change. Based on the theory and information derived from the literature review, a research proposition was proposed in this

study: it is possible to improve sustainability while potentially allowing economic growth and improved social wellbeing by retrofitting existing buildings using the triple-bottom line approach. This research proposition was justified based on a process from conceptualisation to operationalisation. The conceptualisation is realised by developing a conceptual model for developing retrofitting strategies for office buildings, which has been developed and illustrated in Chapter 6.

In the conceptual model, eight steps are identified for deciding retrofitting strategies: Step 1 – Set retrofitting goals and objectives; Step 2 – Identify problems; Step 3 – Propose retrofitting activities; Step 4 – Develop assessment criteria; Step 5 – Evaluate performance of retrofitting activities; Step 6 – Develop retrofitting strategies; Step 7 – Conduct sensitivity analysis; and Step 8 – Choose the "best" retrofitting strategy. In Step 3, the common retrofitting activities for retrofitting activities for proposed projects. In Step 4, suitable environmental, economic and social assessment criteria for assessing the performance of existing buildings are summarised based on the literature review. They need to be modified to suit the local situation of sustainable retrofitting before using the model to develop retrofitting strategies for an office building. In Step 6, a binary linear mathematical model is established in which a preferable retrofitting activities. The developed retrofitting strategy can maximise the overall improvement of the existing building in environmental, economic and social dimensions within project constraints, while meeting retrofitting goals.

9.2.5. Verifying the conceptual model and illustrating the implementation of

the model

The last research objective is to verify the conceptual model and illustrate how to use the model to develop retrofitting strategies for an office building. To satisfy this research objective, a case study was conducted in Hangzhou, China. Since the conceptual model intends to be general, it needs to be adapted to suit the specific situation of sustainable retrofitting in China. In Chapter 7, the conceptual model was converted to an operating model (operationalisation) for use in China by revising the retrofitting activities and assessment criteria. China is the world's largest construction market, but the performance of existing buildings does not meet the requirements

of sustainable buildings. The large number of buildings built in the past 30 years are now due for retrofit.

In China, the difference between buildings in the north and the south is mainly the different methods of heat supply. In the north, there is central heat supply available for varying lengths of time, but HVAC or heaters are the only heat supply in the south. However, the difference exists mainly in residential buildings in the north and the south. As a result of economic development over the past decades in China, office buildings built after the 1980s are air conditioned across the whole country. HVAC has become the major intervention for indoor thermal comfort for office buildings in China by providing heating in winter and cooling in summer. Therefore, even though China is a big country with various climate conditions, the office buildings across the country have a similar way of operating, and it provides a good opportunity to test the flexibility of the developed model.

To convert the conceptual model to an operating model for use in China, opinions about suitable retrofitting activities and assessment criteria for sustainable retrofitting in China were collected from professionals in the construction and property management sector and key stakeholders of retrofitting. A two-stage strategy was adopted to collect data from the broad to the specific. First, a survey was conducted in the north and the south to collect broad and general opinions. Then, three discussion groups were organised to confirm the results of the survey, and to further modify the retrofitting activities and assessment criteria to be suitable for use in China. At the end of focus group discussions, participants were invited to decide weights for the three sustainability dimensions and the assessment criteria under them using the Analytic Hierarchy Process. In this way, the conceptual model was converted to an operating model that can be used to develop retrofitting strategies for office buildings in China.

In Chapter 8, the operating model was used to develop retrofitting strategies for the case building. By following the steps in the conceptual model, ten potential retrofitting activities that may solve the problems of the building were proposed. Based on the evaluation of environmental, economic and social impacts on the retrofitting stage and operation stage, the preferable retrofitting strategy that can maximise the overall improvement on the three sustainability dimensions while meeting retrofitting goals was developed by running the binary linear mathematical model in Excel. By relaxing or tightening the project constraints, four other retrofitting strategies were developed. The building owner can simply choose the strategy that can best meet their demands as the final retrofitting strategy. To verify the effectiveness of the conceptual model, the Chinese national standard of green retrofit is used to assess whether the national green building label can be achieved by implementing the developed retrofitting strategies. The result shows that the building can achieve two green stars if the preferable retrofitting strategy is implemented. Based on the developed retrofitting strategies for the case building, and the verified effectiveness of the conceptual model, Chapter 8 illustrated the validity of the conceptual model and provided a detailed process for using the conceptual model to develop retrofitting strategies for an office building.

9.2.6. Conclusion

In conclusion, this study has achieved the research aim and objectives by developing a model that can be used to develop retrofitting strategies for office buildings by integrating the environmental, economic and social dimensions. In addition, the developed model has been tested and verified based on a case study, indicating that by implementing the retrofitting strategies developed by the model, the sustainability performance of existing office buildings can be effectively improved.

9.3. Summary of the research

This research aimed to develop a model for deciding retrofitting strategies for office buildings to improve their sustainability performance. It firstly investigated the significance of sustainable development of construction by emphasising the necessity and challenges facing sustainable construction. Based on the literature review, the necessity of sustainable construction was confirmed, and four points were identified which are crucial to overcome those challenges:

- Environmental, economic and social dimensions should be integrated and balanced in an optimal manner.
- A long-term perspective should be adopted for investigating impacts on the three sustainability dimensions.

- The assessment method and assessment indicators should be thoroughly studied and determined for each sustainability dimension.
- A universal unit should be adopted for comparison and decision making.

To achieve genuine sustainable retrofitting, these four points are considered and carefully integrated in the developed model.

By investigating the current performance of existing office buildings around the world, it was shown that most existing office buildings have poor performance on environmental, economic and social dimensions, not only because of aging building materials and service systems, but also increasing demands for indoor environmental quality. With the large quantity of existing buildings in the world, the rate of new buildings added to the whole building sector is only around 1% a year. Therefore, compared to demolition and then building new buildings, retrofitting existing buildings can deliver a faster improvement on sustainability performance of the whole building sector. Moreover, by largely reusing existing structure and materials, retrofitting can save embodied energy and embodied carbon emissions embedded in existing buildings. Therefore, sustainable retrofitting is regarded as an effective way to improve the sustainability performance of existing buildings. Ultimately, the overall sustainability performance of the whole building sector may be improved.

Based on existing studies about building retrofitting, 15 common retrofitting activities for office buildings were identified with the emphasis on potential improvement on the three sustainability dimensions. In addition, the level of interruption to existing tenants by implementing these retrofitting activities was discussed and categorised. Based on the potential contribution and different level of interruption to existing tenants, the identified retrofitting activities were categorised, which may help select suitable retrofitting activities.

Suitable retrofitting strategies cannot be developed if the performance of proposed retrofitting activities is not assessed appropriately and comprehensively. This study reviewed common assessment methods in each sustainability dimension. Life cycle impact and green building rating systems are two common methods to assess environmental impact. For the economic dimension, life cycle costing and cost benefit analysis are widely used to evaluate income and

costs throughout a product's life cycle. Social cost benefit analysis and social life cycle impacts are two methods that can be adopted to assess social impact. Often environmental and economic assessments are integrated as a two-pillar model for assessing the sustainability performance of construction. As an indispensable component of sustainability, the social dimension conveys both the tangible and intangible benefits of sustainability. Therefore, it is desirable for assessment models or frameworks to integrate all the sustainability dimensions and generate an overall assessment result to aid in decision making.

Life cycle sustainability assessment is a three-pillar model built on the principle of the triplebottom line. In life cycle sustainability assessment, environmental impact is assessed using environmental life cycle assessment, the economic dimension is assessed using environmental life cycle costing, and social life cycle assessment is used to assess social impacts. These three techniques have the same methodology based on the standard ISO 14040 (ISO 2006a). Life cycle sustainability assessment is praised for providing a whole picture for sustainability assessment by including all the three sustainability dimensions. However, it models each pillar of sustainability independently; thus, the assessment results in each pillar do not facilitate the decision-making process unless they are amalgamated into a decision-making support system.

Deciding retrofitting strategies is a complex process that needs careful assessment in different sustainability dimensions, clear understanding and managing trade-offs between different assessment criteria, and the ability to deal with competing retrofitting objectives. Multi-criteria decision making is believed to be able to fulfill these requirements. Multi-objective decision making and multi-attribute decision making are two branches of multi-criteria decision making. Multi-objective decision making can generate optimal solutions within project constraints, while multi-attribute decision making is usually used to select the "best" solution from a pool of predetermined alternatives. Due to the nature of retrofitting projects, the possibilities for the site should be compared to one another rather than to strict goals. The optimisation model is more suitable for generating retrofitting strategies since it can provide the best overall and balanced performance regarding the three sustainability pillars. Therefore, multi-objective decision making was adopted in this study to build the conceptual model for deciding on retrofitting strategies.

Based on the information derived from the literature review, a conceptual model for deciding retrofitting strategies for office buildings was developed. In the conceptual model, the environmental, economic and social impacts of potential retrofitting activities are assessed, and the assessment outcomes are normalised to unitless values, so they can be aggregated as one to support decision making. By using the conceptual model, the developed retrofitting strategy is able to maximise the overall improvement in the environmental, economic and social dimensions within project constraints, while meeting retrofitting objectives. Following the model development, a case study was conducted to verify the validity of the conceptual model and illustrate the detailed process of using the conceptual model to develop retrofitting strategies for an office building, as discussed in Section 9.3.5.

9.4. Contribution to knowledge

The main contribution of this research is to develop a model for deciding retrofitting strategies for office buildings. With the development and verification of the model, this research successfully addresses the research gaps identified in Chapter 1:

- The developed model is able to deal with environmental, economic and social issues of existing office buildings, and provide appropriate assessment of impacts of potential retrofitting activities in the three sustainability dimensions.
- The developed model provides a comprehensive framework of sustainable retrofitting from assessing impacts of retrofitting activities in the three sustainability dimensions to using the resulting assessment outcomes to develop suitable retrofitting strategies.
- The developed model is able to analyse trade-offs between different assessment criteria and sustainability dimensions from a long-term perspective. Therefore, the developed retrofitting strategies may maximise benefits among different stakeholders, contributing to effective promotion of sustainable retrofitting.

The developed model consists of two parts – the conceptual and the operational. The conceptual model is developed based on the literature review. It outlines general steps for developing retrofitting strategies, and provides common retrofitting activities and assessment criteria for sustainable retrofitting for office buildings. By modifying retrofitting activities and assessment criteria, the conceptual model can be adapted to an operating model to suit the specific situation of sustainable retrofitting in any location with any climate condition.

In this research, a case study was conducted to illustrate the process of the conceptualisation to operationalisation. First, a survey and three focus group discussions were conducted with local professionals in the construction and property management sector and key stakeholders to modify retrofitting activities and assessment criteria to be locally suitable, which enabled the conceptual model to be converted to an operating model. The operating model was then used to develop retrofitting strategies for the case building. The conceptual model considers all the various strategies for retrofitting, and the operating model is a quantified set for a specific location or region. The whole process can test whether the research proposition is correct or not. Then the process from converting the conceptual model to an operating model using a case study can be recognised as a logically and methodologically correct framework for the work done and to be done in the future. By copying the process, the conceptual model can be adapted according to the specific situation of each retrofitting project and used to develop retrofitting strategies.

9.5. Limitations of this research

The research carried out in this thesis is significant and the developed model can be adapted and used to develop retrofitting strategies for office buildings. However, there are some limits of this research which need to be recognised.

First, the diversity of building materials and service systems cannot be considered in the developed model. However, different materials have different embodied energy, life span, insulation performance, and initial costs that may result in different retrofitting strategies being developed. The model assumes that the determined type and composition of building materials and service systems in each retrofitting activity is the optimum for the retrofitted building. In the case study, the suitable and locally available building materials and service systems for the proposed retrofitting activities were determined based on consultation with local professionals in the construction and property management sector. Without considering impacts of different types and composition of building materials and service systems, even though the developed retrofitting strategies can meet identified retrofitting goals within the given project constraints, the model cannot ensure that it is the optimal solution for meeting the retrofitting goals under the given project constraints.

Second, there is a lack of post-occupancy evaluation in the case study, which could be used to obtain feedback on the effect of the developed retrofitting strategy, especially on the indoor environmental quality. In the case study, only a pre-assessment of indoor environmental quality was conducted to identify improvement areas for which potential retrofitting activities could be proposed. Therefore, a post evaluation is necessary to understand how the retrofitting strategy developed by the model improves indoor environmental quality. A post evaluation could also test the accuracy of estimation. The results of a post survey could be used to calibrate the assessment and decision-making process in the developed model. However, for a post evaluation, data is needed on operating the building for at least one year after retrofitting, which cannot be achieved within the thesis study period. A post-occupancy evaluation may be conducted as a follow-up study after the thesis study.

Thirdly, demands of commercial office buildings may differ to some extent after the COVID-19 pandemic. Since the emergence of COVID-19 in late 2019, the format of people's daily life and work has changed. The decrease in social activities and a new form of work, working from home, make it uncertain what the demands for office buildings may be in the future. Meanwhile, likely future pandemics also lead to uncertainty about demand for offices, even new construction. In this study, the conceptual model was developed, and the survey and focus group discussion were conducted before the emergence of COVID; thus, the impact of the COVID pandemic on demands for office buildings could not be reflected in this study. However, when applying this model to develop retrofitting strategies, infectious disease pandemics need to be considered, especially in the steps to identify retrofitting activities and develop assessment criteria.

9.6. Recommendations for future research

As indicated above, this research has developed a model for deciding retrofitting strategies for office buildings to improve their sustainability performance. During the research, it was noted that there is scope for future study. The following research is desirable to extend and modify the findings in this study.

9.6.1. Develop a decision model to select suitable building materials for retrofitting

As discussed in Section 9.5, the developed model does not consider the diversity of building materials, which may have significant impacts on the insulation performance of external walls, roofs and windows. In fact, the model can be adapted to a decision model to select suitable building materials for retrofitting. By focusing on building materials, the assessment criteria need to be adapted to suit the context of impacts of building materials instead of retrofitting activities. There are multiple studies about evaluating the environmental and economic performance of building materials. Similar to most studies about sustainable construction, the social dimension does not get sufficient attention. The definition and assessment of the social performance of different building materials is relatively vague. Therefore, more efforts are needed to identify social assessment criteria for selecting suitable building materials.

Once the assessment criteria for the three sustainability dimensions are identified, the binary linear mathematical model in the conceptual model can be adapted to select suitable building materials for retrofitting, such as revising the objective function according to the expected U-value of insulation materials. The expected outcomes are that several building materials can be identified to meet different decision rules, such as the best insulation performance, the most cost effectiveness, or the best overall sustainability performance. In this way, a limited, manageable number of types of building materials can be identified. However, the selection of building materials is sensitive to the climate condition. Therefore, the model should be adapted to select suitable retrofitting materials for regions with different climate conditions.

9.6.2. Include a post-occupancy evaluation in the model

Due to the time constraint, post-occupancy evaluation is not included in this study. However, the feedback obtained by a post-occupancy evaluation could help verify the effectiveness of the developed model. The results of a post-occupancy evaluation could test the accuracy of assessment and be used to calibrate the assessment and decision-making process in the developed model, especially when the converted operating model is used for a region instead of a single building. By improving the operating model based on the results of post-occupancy

evaluation in previous retrofitting projects in the same region, great improvement in the accuracy and effectiveness of the developed model is likely.

9.6.3. Adapt the model to be used for the public sector

The model developed in this study only considers the internal factors which are directly relevant to individuals, such as building owners, tenants, facility managers, and people visiting the buildings. However, the external factors that are relevant to the whole community or society, such as how the retrofitted building fits into the streetscape, and whether sustainable retrofitting can contribute to less public investment in energy production, are not considered. External factors of sustainable retrofitting are normally considered by governments to plan the development of communities or the whole of society. Therefore, it is worth adapting the model by incorporating external factors to develop retrofitting strategies. Moreover, with the support of government, it is more possible to convert the conceptual model to operating models for different regions in a country and use them as a guideline for retrofitting office buildings. In this way, the overall sustainability performance of existing buildings may be improved effectively in line with the direction of the government's macro planning.

9.6.4. Extend the model

This study has concentrated on investigating ways to improve the sustainability performance of existing office buildings by sustainable retrofitting. The outcome is a conceptual model and an illustration of converting the conceptual model to an operating model to develop retrofitting strategies for an office building in China. The outcome provides many opportunities for future research.

First, the literature review on sustainable retrofitting reveals there are many studies about the environmental and economic performance of office buildings, but limited study about the social performance of office buildings. In this study, only the social impacts of implementing retrofitting activities and operating retrofitted office buildings are considered. In fact, there are other social impacts caused by retrofitting activities, such as corporate governance and productivity of workers in the building. However, as discussed in Chapter 3, the split benefit, in which tenants receive the benefits of retrofitting paid for by building owners, is a barrier to

sustainable retrofitting. Therefore, the building owners are less likely to care whether retrofitting can contribute to improved corporate social responsibility and productivity. However, it is different if the building owners occupy the office building. In this case, the improved social performance may become the driver of sustainable retrofitting. Therefore, future research could either investigate more comprehensive sustainable retrofitting by identifying and incorporating other applicable social impact assessment criteria, or analysing retrofitting outcomes from different perspectives, such as adapting the model to develop retrofitting strategies for office buildings that are occupied by owners,

Second, this research only investigated sustainable retrofitting for office buildings. There is huge potential for sustainability improvement in other building types, such as residential buildings, school buildings, hospitals, shopping malls, and retail stores. Therefore, future studies can investigate each type of building to identify possible retrofitting activities and suitable assessment criteria in the three sustainability dimensions. By doing this, the rate of enhancing sustainability performance of the whole building sector can be accelerated.

Third, this research only considered buildings themself, and the outdoor landscape was not included. In future research, the outdoor landscape of buildings can be included to maximise the effect of sustainable retrofitting. In this case, the retrofitting activities that can improve outdoor environment, such as planting more greenery around the building and adding access to the building, should be considered. Accordingly, the related environmental and social assessment criteria, such as the increased rate of outdoor greenery and improved accessibility to the building should also be considered and evaluated.

Fourth, the case study in this research only investigated retrofitting activities and assessment criteria of sustainable retrofitting of office buildings in China. Climate change is a global crisis. The existing buildings across the world have negative impacts on the natural environment. It is necessary to conduct sustainable retrofitting for office buildings throughout the world. International application of the developed model will provide interesting opportunities for comparisons and information exchange. These opportunities will strengthen the methodology for practising sustainable retrofitting of office buildings, thereby reducing the environmental impact of existing office buildings.

9.7. Conclusion

This research aims to develop a model for deciding retrofitting strategies for office buildings to improve their sustainability performance. A series of research objectives were identified and met to realise this research aim, including an extensive review on sustainable construction, current sustainability performance of existing office buildings, common retrofitting activities for office buildings, applicable assessment criteria for sustainable retrofitting of office buildings, and common assessment methods of sustainability performance; development of a conceptual model for deciding retrofitting strategies for office buildings; and a case study for verifying the conceptual model and illustrating the process of implementation of the model.

The research outcomes indicate that the model is able to develop retrofitting strategies that can maximise the overall sustainability performance of an existing building in the environmental, economic and social dimensions within project constraints, while meeting retrofitting goals. Moreover, the conceptual model is flexible and can be adapted to suit the local situation of sustainable retrofitting, in a way that locally suitable and applicable retrofitting strategies can be developed. The process from conceptualisation to operationalisation in the research provides a logically and methodologically correct framework for the work done and to be done in the future. By retrofitting existing office buildings using the retrofitting strategies developed by the model, the sustainability performance of the whole building sector is expected to improve effectively.

Appendices

Appendix A. Trade-off between three attributes

The point-to-point trade-offs can be shown in a two-dimensional (2D) graph as Figure A-1. According to Equation 6.7, trade-off between attributes X^p and X^p equals the cotangent of angle α . Therefore, the smaller angle α , the bigger the trade-off. The scenario of trade-offs with different angle α is summarised as below:

- Scenario 1: When $\alpha = 45^{\circ}$, $T_{ji}^{pq} = cot\alpha = 1$, meaning the amount of increase on X^p is the same as the amount of decrease on X^q . Therefore, the total evaluation values regarding X^p and X^q of alternative a_j and a_i are the same.
- Scenario 2: When $0^{\circ} < \alpha < 45^{\circ}$, $T_{ji}^{pq} = cot\alpha > 1$, meaning the amount of increase on X^p is more than the amount of decrease on X^q when alternative a_i is replaced by alternative a_j . Therefore, the total evaluation value regarding X^p and X^q of alternative a_i is bigger than that of a_i .
- Scenario 3: When $45^{\circ} < \alpha < 90^{\circ}$, $T_{ji}^{pq} = cot\alpha < 1$, meaning the amount of increase on X^p is less than the amount of decrease on X^q . Therefore, the total evaluation value regarding X^p and X^q of alternative a_j is smaller than that of a_i .

Figure A-1. A 2D expression of point-to-point trade-offs



Among the above three scenarios, Scenario 2 is the most preferable since the "price" to reach the same amount of improvement on attribute X^p is the smallest – the least decrease on attribute X^q . In this case, the total evaluation value of attribute X^p and X^q is the biggest among the three scenarios. The extreme case of Scenario 2 is when angle α is close to 0°, so there is almost no decrease on attribute X^q , but the same amount of increase on attribute X^p as in Scenarios 1 and 3. In this case, the line between a_i and a_j approximately parallels axis X in Figure A-1.

When there are three attributes that need to be considered at the same time, such as the three sustainability pillars, the process for looking for an alternative that can maximise trade-offs between the three is complicated. A three-dimensional (3D) graph as Figure A-2 shows is used to intuitively illustrate the trade-offs between three attributes.

Figure A-2. A 3D expression of trade-offs between three attributes



Figure A-2 illustrates a trade-off occurs between three attributes when the alternative a_i is analysed and it is realised that the evaluation regarding attribute X should be improved while

the evaluation regarding attribute *Y* and attribute *Z* can be decreased. Therefore, there is no doubt that another alternative a''_j is preferable if the evaluation $X_{j''} > X_i$, $Y_{j''} \ge Y_i$, and $Z_{j''} \ge Z_i$. If evaluation of alternative a'_j regarding attribute *Y* or *Z* is less than that of alternative a_i , it is a point-to-point trade-off as Figure A-1 shows. If no a''_j or a'_j can be found, alternative a_j should be looked for which can maximise the trade-off between attributes *X*, *Y* and *Z*, as Equation A-1 shows.

$$\begin{cases} T_{ji}^{XY} = \frac{X_j - X_i}{Y_i - Y_j} \\ T_{ji}^{XZ} = \frac{X_j - X_i}{Z_i - Z_j} \end{cases}$$
 Equation A-1

Where,

- T_{ji}^{XY} trade-off between a pair of attributes X and Y when alternative a_i is replaced by the alternative a_j
- T_{ji}^{XZ} trade-off between a pair of attributes X and Z when alternative a_i is replaced by the alternative a_i
- $Y_i Y_j$ value decrease in attribute Y when alternative a_i is replaced by the alternative a_j
- $Z_i Z_j$ value decrease in attribute Z when alternative a_i is replaced by the alternative a_j

Based on the analysis of point-to-point trade-offs, the desired trade-off between three attributes should be the one where both angle α and β in Figure A-2 are between 0° and 45°, making the increase in attribute *X* bigger than the decrease in attributes *Y* and *Z* respectively. The extreme case is that the line between a_i and a_j approximately parallels axis X in Figure A-2. Therefore, the optimum solution is that the total evaluation value of attributes *X*, *Y* and *Z* is the biggest among all possible trade-off situations with three attributes considered.

Appendix B-1. Information letter and consent form for online questionnaire survey

INFORMATION SHEET AND CONSENT FORM FOR ONLINE SURVEYS

ETH 18-2810 – Developing a strategic assessment model of retrofitting for existing office buildings – A triple-bottom line approach

What is the research study about?

The purpose of this research is to develop a strategic assessment model of retrofitting for existing office buildings to improve their sustainability performance. The developed assessment model has the ability to assess the performance of existing office buildings from environmental, economic and social perspectives and identify suitable retrofitting activities. For this purpose, I will be conducting the questionnaire survey with seven different roles related to professionals in the construction industry, including building owners, contractors, project managers, architects, engineers, facility managers and tenants, to identify assessment variables. Then, based on the resulting assessment outcomes, suitable retrofitting strategies can be generated by the developed model.

You have been invited to participate because you are recognised having related life experience or professional experience of using office buildings or/and retrofitting buildings.

Who is conducting this research?

My name is Chenyang Li, and I am a PhD student at UTS. My supervisor are Associate Professor Grace Ding (<u>Grace.Ding@uts.edu.au</u>), Dr. Goran Runeson (<u>Karl.Runeson@uts.edu.au</u>) and Associate Professor Xiaoyu Ying (<u>yingxiaoyu@zucc.edu.cn</u>).

Inclusion/Exclusion Criteria

Before you decide to participate in this research study, we need to ensure that it is ok for you to take part.

If you satisfy at least one of the below criteria, you are included in this study.

- You are used to work/currently working as a contractor of building construction projects.
- You are used to work/currently working as a project manager of building constructions projects.
- You are used to work/currently working as a facility manager of an office building.
- You are used to work/currently working as an architect.
- You are used to work/currently working as a construction engineer.
- You are used to own/currently owning an office building.
- You are used to use/currently using at least one office building.

Do I have to take part in this research study?

Participation in this study is voluntary. It is completely up to you whether or not you decide to take part.

If you decide to participate, I will invite you to

- Read the information carefully;
- Complete an online questionnaire.

You can change your mind at any time and stop completing the surveys without consequences.

Are there any risks/inconvenience?

We don't expect this questionnaire to cause any harm or discomfort, however if you experience feelings of distress as a result of participation in this study you can let the researcher know and they will provide you with assistance.

What will happen to information about me?

Access to the online questionnaire is via a generic web link. Submission of the online questionnaire is an indication of your consent. By clicking the web link you consent to the
research team collecting and using personal information about you for the research project. All this information will be treated confidentially. The collected information will be stored using codes, and all names or identity information will be represented by codes as well like participant A. The collected information and analysis document will be separately stored in hard drives, and the hard drives will also be locked in different lockers. Only the researcher will access the provided information.

We would like to store your information for future use in research projects that are an extension of this research project. In all instances your information will be treated confidentially. In any publication, information will be provided in such a way that you cannot be identified.

What if I have concerns or a complaint?

If you have concerns about the research that you think I or my supervisor can help you with, please feel free to contact us on <u>Chenyang.Li@student.uts.edu.au</u>, <u>Grace.Ding@uts.edu.au</u>, <u>Karl.Runeson@uts.edu.au</u>, <u>yingxiaoyu@zucc.edu.cn</u>.

If you would like to talk to someone who is not connected with the research, you may contact the Research Ethics Officer on 02 9514 9772 or <u>Research.ethics@uts.edu.au</u> and quote this number ETH 18-2810.

Appendix B-2. Questionnaire survey (English version)

Part I. General information	
1. What is your gender?	
□ Male	Female
2. Your age fits in which of the following groups?	
\square < 25 years old	26-35 years old
36-45 years old	\Box 46 – 55 years old
$\square > 55$ years old	
3. How many years have you been in the property	management industry?
\Box < 5 years	6 - 10 years
□ 11 – 15 years	\Box 16 – 20 years
$\square > 20$ years	
4. Which one(s) of the following best describes ye	our role? (You can choose more than one)
Building owner	Project manager
Facilities manager	Architect
Engineer	Tenant
Other	
5. How often do you participate in the retrofitting	of projects?
Always	Very often
Sometimes	Rarely
Never	

6. In the past three years, in how many retrofitting projects have you been involved in?

0	1-3
□ 4 − 7	8-11
12 - 14	More than 14

7. What kinds of building have you retrofitted?

Residential building	Office building
Industrial building	Shopping mall
School building	Hospital building
Other	

Part II. Development of sustainable retrofitting

ones

8. What are the activities you have been involved in the retrofitting of buildings? (You can choose more than one)

Installing/upgrading insulation of	Upgrading HVAC system
building envelopes	Upgrading lighting system
Adopting extensive green roof	Installing PV panel
Changing windows to energy efficient	Installing water control sensors
windows	Replacing existing water fixtures with
Installing sun shading devices	water efficient ones
Installing building management control	Installing water treatment system
system (BMCS)	Other
Upgrading lifts to more energy efficient	

9. Do you use any tools to assess the performance of existing buildings after retrofitting? (You can choose more than one)

NABERS (AU)	BREEAM (UK)
LEED (USA)	GreenStar (AU)
ASGB (CN)	None
Other	
10. In your opinion, who is responsible for impro	oving sustainability? (You can choose more
than one)	
Government through regulation	Government through financial
Government and industry jointly	incentives
Other	Individual owners and managers
11. In your opinion, what are the drivers of retro	fitting of buildings? (You can choose more
than one)	

Tenants' requirement	Government
Economic payback	Images/Branding
Market request	Support and promotion from related
Other	organisations

Part III. Importance level of different aspects of sustainability

12. Could you rate the level of importance of the following aspects that could be considered in the retrofitting of buildings?

	Not important	Slightly important	Moderately important	Important	Very important
Environmental					
Economic					
Social					

Please specify if there are other aspects you think should be added.

13. When retrofitting an office building, what environmental concerns will you consider?

	Not important	Slightly important	Moderately important	Important	Very important
1. Reduce energy consumption					
2. Reduce carbon emissions					
3. Use reusable or recyclable materials or components					
4. Reduce waste generation					
5. Improve indoor environmental quality					
6. Reduce impacts on water resources					

Please specify if there are other aspects you think should be added.

14. Please rate the below social parameters for sustainable assessment of retrofitting.

	Not important	Slightly important	Moderately important	Important	Very important
1. Noise impact on the neighbourhood					
2. Emission impact on the neighbourhood (e.g., particulates, odour, water and heat)					
3. Impact from glare or overshadowing neighbourhood					
4. Safety and security					
5. Impact on cultural heritage					
6. Accessibility to building facilities for people with special needs					
7. Accessibility to building services					
8. Impact on maintenance and maintainability from newly added building fabrics or building systems					

Please specify if there are other aspects you think should be added.

Part IV. Further discussion about improvement of sustainable retrofitting

15. Based on your experience as a user/owner of office building(s), what do you expect to be improved by sustainable retrofitting? (If it is not relevant to your experience, please skip this question)

16. Based on your professional experience as a participant who works on a building retrofitting project, which parts of the office building should be included for improving the overall sustainability? (If it is not relevant to your experience, please skip this question)

17. In addition to retrofitting, are there other actions that should be taken to improve the performance of existing buildings?

Appendix B-3. Questionnaire survey (Chinese version)

- 一、基本信息
- 1. 您的性别是什么?

□男	□女
2. 您的年龄在以下那个年龄区间?	
□ < 25 岁	□ 26-35岁
□ 36-45 岁	□ 45 – 55 岁
□>55岁	
3. 您已在建筑与房屋管理领域从业多少年	?
□<5年	0-10年
□ 11 – 15 年	□ 16-20 年
□>20年	
4. 您从事过或正在从事以下哪个领域? (可以选择一个以上的选项)
□ 建筑拥有者	□ 项目经理

□ 建筑师	□ 工程师
□ 物业管理	🗌 租户
其他	

5. 您参与房屋翻新项目的频率大概是什么?

□ 很频繁	🗌 很少
□ 偶尔	
□ 从来没有	
□经常	

6. 在过去三年里,您参与过多少个房屋翻新项目?

0	1-3
□ 4 – 7	8-11
12 - 14	3 多于 14

7. 您参与过以下哪种类型建筑的翻新项目? (可以选择一个以上的选项)

□ 居住建筑	🗌 办公建筑
□ 工业建筑	□ 商场
□ 教学建筑	□ 医院
□ 其他	

二、对于绿色翻新的认知

8.	您曾经参与过的房屋翻新项目中,	采用过以下哪些翻新措施?	(可以选择一个以上
	的选项)		

□ 安装/更换外围护保温层	□ 更换照明系统
□ 使用绿色屋顶	□ 使用太阳能
□ 更换为节能窗户	□ 安装水控制传感器
□ 安装遮阳装置	🗌 更换小便池、马桶和水龙头等为节
□ 安装建筑管理控制系统	水装置
□ 更换为节能电梯	□ 安装水处理系统
□ 更换空调通风系统	□ 其他

9. 您用过以下哪些房屋评价系统? (可以选择一个以上的选项)

□ NABERS (澳洲)	DBREEAM (英国)
□ LEED (美国)	OreenStar (澳洲)
□ ASGB (中国)	🗌 没有使用过
□其他	

10. 您认为以下哪些对象有责任推广绿色建筑?	(可以选择一个以上的选项)
🗌 政府通过颁布和执行相关规范从而	🗌 政府和企业共同负责
提高绿色建筑的表现力	□ 个人和管理者负责
□ 政府通过经济刺激推广绿色建筑业	□其他

11. 您认为以下哪些方面主导着房屋翻新行业? (可以选择一个以上的选项)

□ 租户的需求	🗌 相关机构和组织
□ 经济回报	🗌 政府
□ 市场需求	□其他
□ 企业形象	

三、绿色翻新整修的评价参数

12. 请您为以下关于绿色建筑的几个方面进行重要性等级划分。

	不重要	有一些重 要	中等重要	重要	非常重要
环境					
经济					
社会					

除了以上 5 个方面,如果您认为还有关于绿色建筑的其他方面需要考虑,请在此说明。

13. 在实施房屋翻新时,需要考虑的环境问题有哪些?并请为它们的重要性进行等级划分。

	不重要	有 一 些 重要	中 等 重 要	重要	非 常 重 要
1.减少能耗					
2.减少碳排放					
3.使用可重复利用或可回收使用的 材料或部件					
4.减少垃圾的产生					
5.提高室内环境质量					
6.减少对水资源的影响					

除了以上 7 个方面,在实施房屋翻新时,如果您认为还有其他环境问题需要考虑,请 在此说明。

14. 在实施房屋翻新时,需要考虑的社会因素有哪些?并请为它们的重要性进行等级划分。

	不重要	有 一 些 重要	中等重 要	重要	非常重要
1.噪声对毗邻建筑的影响					
2.排放物对毗邻建筑的影响(例如:颗 粒物、气味、水喝热量的排放)					
3. 对毗邻建筑的光污染或过分遮挡					
4. 安全性					
5. 对历史文化的影响					
6.是否有为特殊人群(残疾人、孕妇、 老人和孩子)提供的建筑设施					
7.是否有相应的建筑物设施以供使用					

8.新安装建筑材料或建筑系统对建筑维 修和围护的频率和时间的影响(包括 建筑翻新)					
---	--	--	--	--	--

除了以上 10 个社会因素,如果您认为在房屋翻新时,还有其他社会因素需要考虑,请 在此说明。

四、对提高绿色翻新整修的进一步讨论

15. 根据您作为建筑拥有者或者使用者的经验,您期望建筑翻新时可以改善建筑的哪些方面? (如果您不是建筑拥有者或使用者,请忽略此题。)

16. 根据您相关的从业经验,您认为在建筑翻新过程中需要考虑哪些方面去提高建筑的可持续性?(如果您不是建筑师、工程师、项目经理或者物业从业人员,请忽略此题。)

17. 除了建筑翻新,您认为还可以采取哪些措施来提高既有建筑的可持续性?

Appendix B-4. Chi-Square test for association between region and

commonly used retrofitting activities

(1) Region and installing/upgrading insulation of external walls

Crosstab				
Retrofitting	Installing/upgrading			
activity	insulation of external walls			
Region	Count	% of Total		
Northern China	46	35.9%		
Southern China	20	15.6%		
Total	66	51.6%		

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	10.012ª	1	.002		
Continuity Correction ^b	8.915	1	.003		
Likelihood Ratio	10.140	1	.001		
Fisher's Exact Test				.002	.001
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 27.13.

(2) Region and adopting extensive green roof

Crosstab			
Retrofitting	Adopting extensive green roof		
activity	Count	% of Total	
Region			
Northern China	26	20.3%	
Southern China	19	14.8%	
Total	45	35.2%	

Chi-Square Tests			
			Asymptotic
	Value	df	Significance
			(2-sided)
Pearson Chi-Square	1.329ª	2	.514
Likelihood Ratio	1.697	2	.6428
N of Valid Cases	128		

a. 2 cells (33.3%) have expected count less than 5. The minimum expected count is .44.

(3) Region and changing windows to energy efficient windows

Crosstab				
Retrofitting	Changing windows to energy			
activity	efficient windows			
Region	Count	% of Total		
Northern China	25	19.5%		
Southern China	14	10.9%		
Total	39	30.5%		

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	1.405 ^a	1	.236		
Continuity Correction ^b	0.984	1	.321		
Likelihood Ratio	1.421	1	.233		
Fisher's Exact Test				.253	.161
N of Valid Cases	128				

a. 2 cells (50.0%) have expected count less than 5. The minimum expected count is 17.06.

(4) Region and installing sun shading devices

Crosstab			
Retrofitting	Installing sun shading devices		
activity Region	Count	% of Total	
Northern China	4	3.1%	
Southern China	15	11.7%	
Total	19	14.8%	

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	11.232ª	1	.001		
Continuity Correction ^b	9.615	1	.002		
Likelihood Ratio	11.526	1	.001		
Fisher's Exact Test				.001	.001
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 8.31.

(5) Region and installing building management control system

Crosstab				
Retrofitting	Installing	building		
activity	management control system			
Region	Count	% of Total		
Northern China	7	5.5%		
Southern China	10	7.8%		
Total	17	13.3%		

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	1.810 ^a	1	.179		
Continuity Correction ^b	1.173	1	.279		
Likelihood Ratio	1.795	1	.180		
Fisher's Exact Test				.199	.140
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 7.44.

(6) Region and upgrading lifts to more energy efficient ones

Crosstab				
Retrofitting	Upgrading	lifts	to	more
activity	energy efficient ones			
Region	Count	% of '	Total	
Northern China	37	28.9%	0	
Southern China	35	27.3%	0	
Total	72	56.3%	0	

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	1.580ª	1	.209		
Continuity Correction ^b	1.161	1	.281		
Likelihood Ratio	1.588	1	.208		
Fisher's Exact Test				.281	.141
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 24.50.

(7) Region and upgrading HVAC

Crosstab			
Retrofitting	Upgrading HVAC		
activity Region	Count	% of Total	
Northern China	12	9.4%	
Southern China	25	19.5%	
Total	37	28.9%	

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	11.997 ^a	1	.001		
Continuity Correction ^b	10.674	1	.001		
Likelihood Ratio	12.066	1	.001		
Fisher's Exact Test				.001	.001
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 16.19.

(8) Region and upgrading lighting system

Crosstab					
Retrofitting	Upgrading lighting system				
Region	Count	% of Total			
Northern China	22	17.2%			
Southern China	28	21.9%			
Total	50	39.1%			

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	5.003 ^a	1	.025		
Continuity Correction ^b	4.220	1	.040		
Likelihood Ratio	5.007	1	.025		
Fisher's Exact Test				.030	.020
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 21.88.

(9) Region and installing PV panels

Crosstab				
Retrofitting	Installing PV panels			
activity Region	Count	% of Total		
Northern China	36	28.1%		
Southern China	20	15.6%		
Total	56	43.8%		

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	2.612 ^a	1	.106		
Continuity Correction ^b	2.064	1	.151		
Likelihood Ratio	2.631	1	.105		
Fisher's Exact Test				.151	.075
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 24.50.

(10) Region and installing water control system

Crosstab				
Retrofitting	Installing	water control		
activity	system			
Region	Count	% of Total		
Northern China	8	6.3%		
Southern China	7	5.5%		
Total	15	11.7%		

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	.059ª	1	.809		
Continuity Correction ^b	.000	1	1.000		
Likelihood Ratio	.059	1	.809		
Fisher's Exact Test				1.000	.510
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 6.56.

(11) Region and replacing existing water fixtures with water efficient ones

Crosstab		
Retrofitting	Changing	existing water
activity	fixtures wit	h water efficient
	ones	
Region	Count	% of Total
Northern China	23	18.0%
Southern China	21	16.4%
Total	44	34.4%

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	.431ª	1	.512		
Continuity Correction ^b	.220	1	.639		
Likelihood Ratio	.430	1	.512		
Fisher's Exact Test				.575	.319
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 19.25.

(12) Region and installing a water treatment system

Crosstab						
Retrofitting	Installing a water treatment					
activity	system					
Region	Count	% of Total				
Northern China	10	7.8%				
Southern China	16	12.5%				
Total	26	20.3%				

Chi-Square Tests					
	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	4.195 ^a	1	.041		
Continuity Correction ^b	3.337	1	.068		
Likelihood Ratio	4.175	1	.041		
Fisher's Exact Test				.048	.034
N of Valid Cases	128				

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 11.38.

Appendix B-5. Chi-Square test for association between

importance level of environmental assessment criteria and

whether respondents have ever conducted projects involving

office buildings

(1) Retrofitted building types and importance level of reduce energy consumption

Crosstab							
Importance level		Reduce energy consumption					
	Slightly	Moderately	Important	Very	Total		
Building type	important	important		important			
Office buildings	1	2	22	19	44		
Others	2	8	34	40	84		
Total	3	10	56	59	128		

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	1.639ª	3	.650
Likelihood Ratio	1.715	3	.634
Linear-by-linear Association	.003	1	.955
N of Valid Cases	128		

a. 3 cells (37.5%) have expected count less than 5. The minimum expected count is 1.03.

11	• •	D (C (1	1 1111		1	1	1 C 1	1	• •
1	23	Retrotiffed	huulding	tunes and	1 1mnort	ance lev	el of reduc	e carbon	emission
14	- 1	Renomineu	ounung	types and	i mporu		ci oi icuuc		CHIISSION
· ·			0	21					

Crosstab								
Importance level	Reduce carbon emission							
Devilding	Not	Slightly	Moderately	Important	Very	Total		
Building type	important	important	important		important			
Office buildings	0	1	4	11	28	44		
Others	1	1	8	30	44	84		
Total	1	2	12	41	72	128		

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	2.431 ^a	4	.657
Likelihood Ratio	2.770	4	.597
Linear-by-linear Association	.795	1	.372
N of Valid Cases	128		

a. 5 cells (50%) have expected count less than 5. The minimum expected count is .34.

(3) Retrofitted building types and importance level of use reusable or recyclable materials or components

Crosstab						
Importance level Use reusable or recyclable materials or components						
	Not	Slightly	Moderately	Important	Very	Total
Building type	important	important	important		important	
Office buildings	1	1	10	22	10	44
Others	1	3	22	35	23	84
Total	2	4	32	57	33	128

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	1.204 ^a	4	.877
Likelihood Ratio	1.200	4	.878
Linear-by-linear Association	.013	1	.910
N of Valid Cases	128		

a. 4 cells (40.0%) have expected count less than 5. The minimum expected count is .69.

Crosstab								
Importance level	Reduce waste generation							
DUIT	Not	Slightly	Moderately	Important	Very	Total		
Building type	important	important	important		important			
Office buildings	0	2	6	22	14	44		
Others	2	2	10	45	25	84		
Total	2	4	16	67	39	128		

(4) Retrofitted building types and importance level of reduce waste generation

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	1.660 ^a	4	.798
Likelihood Ratio	2.275	4	.685
Linear-by-linear Association	.041	1	.840
N of Valid Cases	128		

a. 4 cells (40.0%) have expected count less than 5. The minimum expected count is .69.

	->	D (C) 11 111	1 • .	1 1 0 '	• 1	• 1	1.
1	``	Ratrotittad building tunge on	d importance	LOVAL OF 1m	nrova indoor	anuranmontal	ang 14437
	, ,	KULUHILLU DUHUHUS LVDUS AH				CHVHOHHCHLAF	uuantv
· · ·	~)		# mp =		pro / c	••••••	

Crosstab							
Importance level	Improve indoor environmental quality						
	Slightly	Moderately	Important	Very	Total		
Building type	important	important		important			
Office buildings	0	3	11	30	44		
Others	1	8	26	49	84		
Total	1	11	37	79	128		

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	1.577 ^a	3	.665
Likelihood Ratio	1.907	3	.592
Linear-by-linear Association	1.364	1	.243
N of Valid Cases	128		

a. 3 cells (37.5%) have expected count less than 5. The minimum expected count is .34.

Crosstab					
Importance level Reduce impacts on water resources					
	Slightly	Moderately	Important	Very	Total
Building type	important	important		important	
Office buildings	1	2	11	24	44
Others	1	9	34	40	84
Total	2	11	51	64	128

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	1.797 ^a	3	.616
Likelihood Ratio	1.926	3	.588
Linear-by-linear Association	.682	1	.409
N of Valid Cases	128		

a. 3 cells (37.5%) have expected count less than 5. The minimum expected count is .69.

Appendix B-6. Chi-Square test for association between importance level of social assessment criteria and whether respondents have ever conducted projects involving office buildings

(1) Retrofitted building types and importance level of noise impact on the neighbourhood

Crosstab				
Importance level Noise impact on the neighbourhood				
Building type	Moderately important	Important	Very important	Total
Office buildings	6	26	12	44
Others	11	38	35	84
Total	17	64	47	128

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	2.744 ^a	2	.254
Likelihood Ratio	2.798	2	.247
Linear-by-linear Association	1.436	1	.231
N of Valid Cases	128		

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 5.84.

(2) Retrofitted building types and importance level of emission impact on the neighbourhood

Crosstab				
Importance level Emission impact on the neighbourhood				
Building type	Moderately important	Important	Very important	Total
Office buildings	5	24	15	44
Others	10	34	40	84
Total	15	58	55	128

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	2.498 ^a	2	.287
Likelihood Ratio	2.511	2	.285
Linear-by-linear Association	1.076	1	.300
N of Valid Cases	128		

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 5.16.

(3) Retrofitted building types and importance level of impact from glare or overshadowing neighbourhood

Crosstab					
Importance level Impact from glare or overshadowing neighbourhood					
Building type	Moderately important	Important	Very important	Total	
Office buildings	4	29	11	44	
Others	9	40	35	84	
Total	13	69	46	128	

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	4.099 ^a	2	.129
Likelihood Ratio	4.185	2	.123
Linear-by-linear Association	1.644	1	.200
N of Valid Cases	128		

a. 1 cells (16.7%) have expected count less than 5. The minimum expected count is 4.47.

(4) Retrofitted building types and importance level of safety and security

Crosstab					
Importance level		Sa	afety and secur	ity	
Building type	Slightly important	Moderately important	Important	Very important	Total
Office buildings	1	3	16	24	44
Others	1	3	27	53	84
Total	2	6	43	77	128

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	1.370 ^a	3	.713
Likelihood Ratio	1.330	3	.722
Linear-by-linear Association	1.278	1	.258
N of Valid Cases	128		

a. 4 cells (50%) have expected count less than 5. The minimum expected count is .69.

(5)	Retrofitted build	ing types and	importance	level of impact of	on cultural he	eritage
-----	-------------------	---------------	------------	--------------------	----------------	---------

Crosstab				
Importance level	Impact on cultural heritage			
	Moderately	Important	Very	Total
Building type	important		important	
Office buildings	10	22	12	44
Others	10	42	32	84
Total	20	64	44	128

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	3.148 ^a	2	.207
Likelihood Ratio	3.077	2	.215
Linear-by-linear Association	2.888	1	.089
N of Valid Cases	128		

a. 0 cells (0.0%) have expected count less than 5. The minimum expected count is 6.88.

(6) Retrofitted building types and importance level of accessibility to building facilities for people with special needs

Crosstab						
Importance	Accessibility to building facilities for people with special needs					
	Not	Slightly	Moderately	Important	Very	Total
Building type	important	important	important		important	
Office	1	0	4	20	19	44
buildings						
Others	1	2	11	32	38	84
Total	2	2	15	52	57	128

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	2.072 ^a	4	.723
Likelihood Ratio	2.708	4	.608
Linear-by-linear Association	.049	1	.825
N of Valid Cases	128		

a. 4 cells (40.0%) have expected count less than 5. The minimum expected count is .69.

(7) Retrofitted building types and importance level of accessibility to building services

Crosstab					
Importance level	Accessibility to building services				
Building type	Slightly	Moderately	Important	Very	Total
Building type	important	important		important	
Office buildings	2	7	27	8	44
Others	2	14	43	25	84
Total	4	21	70	33	128

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	2.491ª	3	.477
Likelihood Ratio	2.549	3	.466
Linear-by-linear Association	1.205	1	.272
N of Valid Cases	128		

a. 2 cells (25%) have expected count less than 5. The minimum expected count is 1.38.
(8) Retrofitted building types and importance level of impact on maintenance and maintainability from newly added building fabrics or building systems

Crosstab								
Importance level	mportance level Impact on maintenance and maintainability from newly added building							
	fabrics or building systems							
Building type	Slightly	Moderately	Important	Very	Total			
	important	important		important				
Office buildings	0	8	26	10	44			
Others	1	14	40	29	84			
Total	1	22	66	39	128			

	Value	df	Asymptotic Significance (2-sided)
Pearson Chi-Square	2.618 ^a	3	.454
Likelihood Ratio	2.987	3	.384
Linear-by-linear Association	.693	1	.405
N of Valid Cases	128		

a. 2 cells (25.0%) have expected count less than 5. The minimum expected count is .34.

Appendix C-1. Information letter for focus group discussions

(English version)

PARTICIPANT INFORMATION SHEET

Developing a strategic assessment model of retrofitting for existing office buildings – A triplebottom line approach (UTS NUMBER ETH-2810)

WHO IS DOING THE RESEARCH?

My name is Chenyang Li and I am a PhD student at UTS. My supervisor is Associate Professor Grace, Dr. Goran Runeson and Associate Professor Xiaoyu Ying.

WHAT IS THIS RESEARCH ABOUT?

This research is to find out about developing a strategic assessment model of retrofitting for existing office buildings to improve their sustainability performance. The developed assessment model has the ability to assess the performance of existing office buildings from environmental, economic and social perspectives and identify suitable retrofitting activities. For this purpose, I will be conducting the questionnaire survey with seven different roles related to professionals in the construction industry, including building owners, contractors, project managers, architects, engineers, facility managers and tenants, to identify assessment variables. Then, based on the resulting assessment outcomes, suitable retrofitting strategies can be generated by the developed model.

FUNDING

There is no funding received for this research.

WHY HAVE I BEEN ASKED?

You have been invited to participate in this study because you are a professional or stakeholder of building industry, and you are able to provide the information based on your knowledge and experience that I need to complete my research.

IF I SAY YES, WHAT WILL IT INVOLVE?

If you decide to participate, I will invite you to participate in a focus group that will take approximately 60 minutes of your time at your convenience.

ARE THERE ANY RISKS/INCONVENIENCE?

Yes, there are some risks/inconvenience. They are:

- Inconvenience of time taken to undertake this focus group
- May leave the workplace during work time for undertaking this focus group
- Feeling uncomfortable being recorded
- Feeling uncomfortable being speaking in a group

DO I HAVE TO SAY YES?

Participation in this study is voluntary. It is completely up to you whether or not you decide to take part.

WHAT WILL HAPPEN IF I SAY NO?

If you decide not to participate, it will not affect your relationship with the researchers or the University of Technology Sydney. If you wish to withdraw from the study once it has started, you can do so at any time without having to give a reason, by contacting Chenyang Li (Chenyang.Li@student.uts.edu.au).

If you withdraw from the study, your audio tape will be erased, the transcripts will be destroyed and all your information will be destroyed as well. CONFIDENTIALITY

By signing the consent form you consent to the research team collecting and using personal information about you for the research project. All this information will be treated confidentially. The collected information will be stored using codes, and all names or identity information will be represented by codes as well like participant A. The collected information and analysis document will be separately stored in hard drives, and the hard drives will also be locked in different lockers. Your information will only be used for the purpose of this research project.

We would like to store your information for future use in research projects that are an extension of this research project. In all instances your information will be treated confidentially. In any publication, information will be provided in such a way that you cannot be identified.

WHAT IF I HAVE CONCERNS OR A COMPLAINT?

If you have concerns about the research that you think I or my supervisor can help you with, please feel free to contact us on <u>Chenyang.Li@student.uts.edu.au</u>, <u>Grace.Ding@uts.edu.au</u>, <u>Karl.Runeson@uts.edu.au</u>, <u>yingxiaoyu@zucc.edu.au</u>.

You will be given a copy of this form to keep.

NOTE:

This study has been approved by the University of Technology Sydney Human Research Ethics Committee [UTS HREC]. If you have any concerns or complaints about any aspect of the conduct of this research, please contact the Ethics Secretariat on ph.: +61 2 9514 2478 or email: Research.Ethics@uts.edu.au], and quote the UTS HREC reference number. Any matter raised will be treated confidentially, investigated and you will be informed of the outcome.

Appendix C-2. Consent form for focus group discussions (English version)

CONSENT FORM

Developing a strategic assessment model of retrofitting for existing office buildings – A triplebottom line approach (ETH18-2810)

I ______ (participant's name) agree to participate in the research project "Developing a strategic assessment model of retrofitting for existing office buildings – A triplebottom line approach" (UTS HREC reference number: ETH18-2810) being conducted by Chenyang Li (15 Broadway, Ultimo, NSW 2007, contact telephone: _____).

I have read the Participant Information Sheet or someone has read it to me in a language that I understand.

I understand the purposes, procedures and risks of the research as described in the Participant Information Sheet.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I freely agree to participate in this research project as described and understand that I am free to withdraw at any time without affecting my relationship with the researchers or the University of Technology Sydney.

I understand that I will be given a signed copy of this document to keep.

I agree to be:

Audio recorded

I agree that the research data gathered from this project may be published in a form that:

Does not identify me in any way

May be used for future research purposes

I am aware that I can contact Chenyang Li if I have any concerns about the research.

Name and Signature [participant]

____/___/____

Date

Name and Signature [researcher or delegate]

____/___/____

Date

* Witness to the consent process

If the participant, or if their legally acceptable representative, is not able to read this document, this form must be witnessed by an independent person over the age of 18. In the event that an interpreter is used, the interpreter may not act as a witness to the consent process. By signing the consent form, the witness attests that the information in the consent form and any other written information was accurately explained to, and apparently understood by, the participant (or representative) and that informed consent was freely given by the participant (or representative).

Appendix D-1. AHP survey sample

Please use the below scale to define the relative importance of the element A compared to element B

Intensity of Importance	Definition	Explanation
1	Equal importance	Two criteria contribute equally to the objective
2	Weak or slight	
3	Moderate importance	Experience and judgement slightly favour one over another
4	Moderate plus	
5	Strong importance	Experience and judgement strongly favour one over another
6	Strong plus	
7	Very strong or demonstrated importance	A criterion is favoured very strongly over another; its dominance demonstrated in practice
8	Very, very strong	
9	Extreme importance	The evidence favouring one over another is of the highest possible order of affirmation
Reciprocals of above	If criterion <i>i</i> has one of the above non- zero numbers assigned to it when compared with criterion <i>j</i> , then <i>j</i> has the reciprocal value when compared with <i>i</i>	A reasonable assumption
1.1 – 1.9	If the criteria are very close	It may be challenging to assign the best value, but when compared with other contrasting criteria, the size of the small numbers would not be too noticeable, yet they can still indicate the relative importance of the criteria.

The lower left hand matrix triangle is the reciprocal of the upper right hand, so only the upper right hand needs to be filled in.

Part 1. The three sustainability dimensions

Tuble 1. The three subtainability anneholoris						
A/B	Environmental	Economic	Social			
Environmental	1					
Economic		1				
Social			1			

Table 1. The three sustainability dimensions

Part 2. The assessment criteria in the environmental dimension

A/B	Energy consumption	Carbon emissions	Use of recyclable/reusable materials	Waste generation
Energy consumption	1			
Carbon emissions		1		
Use of recyclable/reusable materials			1	
Waste generation				1

Table 2. Assessment criteria in retrofitting stage of environmental dimension

Table 3. Assessment criteria in operation stage of environmental dimension

A/B	Energy consumption	Carbon emissions	Use of recyclable/reusable materials	Waste generation	Indoor environmental quality	Water consumption
Energy consumption	1					
Carbon emissions		1				
Use of recyclable/reusable materials			1			
Waste generation				1		
Indoor environmental quality					1	
Water consumption						1

Part 3. The assessment criteria in the social dimension

Table 4. Assessment criteria in retrofitting stage of social dimension

A/B	Noise impact on neighbourhood	Emission impact on neighbourhood	Safety of retrofitting construction	Impact on cultural heritage	Impact on surrounding traffic and pedestrians	Impact on relocating tenants
Noise impact						
on	1					
neighbourhood						
Emission						
impact on		1				
neighbourhood						
Safety of						
retrofitting			1			
construction						
Impact on						
cultural				1		
heritage						
Impact on						
surrounding					1	
traffic and					1	
pedestrians						
Impact on						
relocating						1
tenants						

			Ŭ.	~ ~ 1	~ ~ ~ ~ ~
A/B	Accessibility to	Accessibility to	Impacts on	Safety and	Room flexibility
	building facilities	building services	maintenance and	security of users	for different
	for people with		maintainability		demands
	special needs		from newly		
			added building		
			fabrics and		
			building systems		
Accessibility to					
building facilities	1				
for people with	1				
special needs					
Accessibility to		1			
building services		1			
Impacts on					
maintenance and					
maintainability					
from newly			1		
added building					
fabrics and					
building systems					
Safety and				1	
security of users				I	
Room flexibility					
for different					1
demands					

Table 5. Assessment criteria in operation stage of social dimension

Appendix D-2. Establishing the matrix of group judgment

Based on Equation 7.3 in Chapter 3, the matrix of group judgment on the three sustainability dimensions and assessment criteria is established as Table D2-1 to Table D2-5 illustrate.

Tuble D2 1. The matrix of group judgment on the three sustainationity unitensions							
Sustainability	Environmental	Economic	Social				
dimensions	Environmentar	Economie					
Environmental	1	1.309	1.891				
Economic	0.764	1	1.321				
Social	0.529	0.757	1				

Table D2-1. The matrix of group judgment on the three sustainability dimensions

Table D2-2. The matrix of group judgment on assessment criteria in the retrofitting stage of the environmental dimension

Environmental assessment criteria in retrofitting stage	Energy consumption	Carbon emissions	Potential for recyclable/reusable materials	Waste generation
Energy consumption	1	1.013	0.964	1.021
Carbon emissions	0.987	1	0.967	0.987
Potential for recyclable/reusable materials	1.037	1.034	1	0.925
Waste generation	0.980	1.013	1.082	1

Table D2-3. The matrix of group judgment on assessment criteria in operation stage of the environmental dimension

Environmental assessment criteria in operation stage	Energy consumption	Carbon emissions	Potential for recyclable/reusable materials	Waste generation	Indoor environmental quality	Water consumption
Energy consumption	1	1.523	1.301	1.409	1.182	1.830
Carbon emissions	0.657	1	1.026	1.082	0.818	1.265
Potential for recyclable/reusable materials	0.768	0.974	1	1.051	0.844	1.145
Waste generation	0.768	1.090	0.952	1	0.784	1.160
Indoor environmental quality	0.766	1.406	1.185	1.484	1	1.484
Water consumption	0.541	0.790	0.873	0.952	0.674	1

Social assessment criteria in retrofitting stage	Noise impact on neighbourhood	Emission impact on neighbourhood	Safety of retrofitting construction	Impact on cultural heritage	Impact on surrounding traffic and pedestrians	Impact on relocating tenants
Noise impact on neighbourhood	1	0.910	0.604	1.026	0.891	0.952
Emission impact on neighbourhood	1.099	1	0.699	1.330	1.229	1.379
Safety of retrofitting construction	1.657	1.430	1	1.364	1.261	1.727
Impact on cultural heritage	0.974	0.752	0.733	1	0.976	1.037
Impact on surrounding traffic and pedestrians	1.122	0.813	0.793	1.024	1	1.275
Impact on relocating tenants	1.051	0.725	0.579	0.964	0.784	1

Table D2-4. The matrix of group judgment on assessment criteria in the retrofitting stage of the social dimension

Table D2-5.	The matrix o	of group j	udgment or	assessment	criteria i	n the o	peration	stage o	of the
social dimen	sion								

A/B	Accessibility to building facilities for people with special needs	Accessibility to building services	Impacts on maintenance and maintainability from newly added building fabrics and building systems	Safety and security of users	Room flexibility for different demands
Accessibility to building facilities for people with special needs	1	1.492	1.644	1.203	1.340
Accessibility to building services	0.670	1	0.993	0.774	0.845
Impacts on maintenance and maintainability from newly added building fabrics and building systems	0.608	1.008	1	0.795	0.807
Safety and security of users	0.381	1.292	1.258	1	1.343
Room flexibility for different demands	0.746	1.183	1.240	0.745	1

Appendix D-3. Weight generation

	/		
	Environmental	Economic	Social
Environmental	1.000	1.309	1.891
Economic	0.764	1.000	1.321
Social	0.529	0.757	1.000
Sum	2.293	3.066	4.212

Table D3-1. The column normalised weights for the three sustainability dimensions

Table D3-2. The line normalised weights for the three sustainability dimensions

	Environmental	Economic	Social	Sum	Normalised weight
Environmental	0.436	0.427	0.449	1.312	0.437
Economic	0.333	0.326	0.314	0.973	0.324
Social	0.231	0.247	0.237	0.715	0.239
Sum	1	1	1	3	1

Environmental dimension									
Retrofitting st	tage								
	Energy consumption	Carbon emissions	Potential for recyclable/reusable materials	Waste generation					
Energy consumption	1	1.013	0.964	1.021					
Carbon emissions	0.987	1	0.967	0.987					
Potential for recyclable/ reusable materials	1.037	1.034	1	0.925					
Waste generation	0.980	1.013	1.082	1					
Sum	4.004	4.060	4.013	3.932					
Operation sta	ge		Г	I	r	1			
	Energy consumption	Carbon emissions	Potential for recyclable/reusable materials	Waste generation	Indoor environmental quality	Water consumption			
Energy consumption	1	1.523	1.301	1.409	1.182	1.830			
Carbon emissions	0.657	1	1.026	1.082	0.818	1.265			
Potential for recyclable/ reusable materials	0.768	0.974	1	1.051	0.844	1.145			
Waste generation	0.768	1.090	0.952	1	0.784	1.160			
Indoor environmental quality	0.766	1.406	1.185	1.484	1	1.484			
Water consumption	0.541	0.790	0.873	0.952	0.674	1			
Sum	4.500	6.783	6.337	6.806	5.302	7.884			
Deter Cttierer		Sc	ocial dimension						
Retrontting st									
	Noise impact on neighbourhood	Emission impact on neighbourhood	Safety of retrofitting construction	Impact on cultural heritage	Impact on surrounding traffic and pedestrians	Impact on relocating tenants			
Noise impact on neighbourhood	1	0.910	0.604	1.026	0.891	0.952			
Emission impact on neighbourhood	1.099	1	0.699	1.330	1.229	1.379			
Safety of retrofitting construction	1.657	1.430	1	1.364	1.261	1.727			
Impact on cultural heritage	0.974	0.752	0.733	1	0.976	1.037			
Impact on surrounding traffic and pedestrians	1.122	0.813	0.793	1.024	1	1.275			
Sum	6.903	5.630	4.408	6.708	6.141	7.370			

Table D3-3. The column normalised weights for assessment criteria

Operation sta	ge				
A/B	Accessibility to building facilities for people with special needs	Accessibility to building services	Impacts on maintenance and maintainability from newly added building fabrics and building systems	Safety and security of users	Room flexibility for different demands
Accessibility to building facilities for people with special needs	1	1.492	1.644	1.203	1.340
Accessibility to building services	0.670	1	0.993	0.774	0.845
Impacts on maintenance and maintainability from newly added building fabrics and building systems	0.608	1.008	1	0.795	0.807
Safety and security of users	0.381	1.292	1.258	1	1.343
Room flexibility for different demands	0.746	1.183	1.240	0.745	1
Sum	3.855	5.975	6.135	4.517	5.335

Environmental dimension									
Retrofitting stag	je								
	Energy consumption	Carbon emissions	Potential for recyclable/ reusable materials	Waste generation	Sum	Normalised weight			
Energy consumption	0.250	0.250	0.240	0.260	0.999	0.25			
Carbon emissions	0.247	0.246	0.241	0.251	0.985	0.245			
Potential for recyclable/ reusable materials	0.259	0.255	0.249	0.235	0.998	0.25			
Waste	0.245	0.250	0.270	0.254	1.018	0.255			
Sum	1	1	1	1	4	1			
Operation stage									
7	Energy consumption	Carbon emissions	Potential for recyclable/ reusable materials	Waste generation	Indoor environmental quality	Water consumption	Sum	Normalised weight	
Energy consumption	0.222	0.225	0.205	0.207	0.223	0.232	1.314	0.219	
Carbon emissions	0.146	0.147	0.162	0.159	0.154	0.161	0.929	0.155	
Potential for recyclable/ reusable materials	0.171	0.144	0.158	0.154	0.159	0.145	0.931	0.155	
Waste generation	0.171	0.161	0.150	0.147	0.148	0.147	0.924	0.154	
Indoor environmental quality	0.170	0.207	0.187	0.193	0.189	0.188	1.134	0.189	
Water consumption	0.120	0.116	0.138	0.140	0.127	0.127	0.768	0.128	
C									
Sum	1	1	1	1	1	1	6	1	
Sum		1	1 Social	1 dimension	1	1	6	1	
Retrofitting stag	ge		1 Social	1 dimension			6		
Retrofitting stag	Re Noise impact on neighbourhood	Emission impact on neighbourhood	1 Social Safety of retrofitting construction	1 dimension Impact on cultural heritage	1 Impact on surrounding traffic and pedestrians	I Impact on relocating tenants	6 Sum	1 Normalised weight	
Retrofitting stag	Re Noise impact on neighbourhood 0.145	Emission impact on neighbourhood 0.162	1 Social Safety of retrofitting construction 0.137	1 dimension Impact on cultural heritage 0.153	I Impact on surrounding traffic and pedestrians 0.145	Impact on relocating tenants 0.129	6 Sum 0.871	Normalised weight 0.145	
Retrofitting stag	Re Noise impact on neighbourhood 0.145 0.159	Emission impact on neighbourhood 0.162 0.178	1 Social Safety of retrofitting construction 0.137 0.159	1 dimension Impact on cultural heritage 0.153 0.198	Impact on surrounding traffic and pedestrians 0.145 0.200	Impact on relocating tenants 0.129 0.187	6 Sum 0.871 1.081	1 Normalised weight 0.145 0.180	
Retrofitting stag Retrofitting stag Noise impact on neighbourhood Emission impact on neighbourhood Safety of retrofitting construction	voise impact on neighbourhood 0.145 0.159 0.240	Emission impact on neighbourhood 0.162 0.178 0.254	I Social Safety of retrofitting construction 0.137 0.159 0.227	1 dimension Impact on cultural heritage 0.153 0.198 0.203	Impact on surrounding traffic and pedestrians 0.145 0.200 0.205	Impact on relocating tenants 0.129 0.187 0.234	6 Sum 0.871 1.081 1.364	1 Normalised weight 0.145 0.180 0.227	
Retrofitting stag	I Re Noise impact on neighbourhood 0.145 0.159 0.240 0.141	Emission impact on neighbourhood 0.162 0.178 0.254 0.134	I Social Safety of retrofitting construction 0.137 0.159 0.227 0.166	1 dimension Impact on cultural heritage 0.153 0.198 0.203 0.149	Impact on surrounding traffic and pedestrians 0.145 0.200 0.205 0.159	Impact on relocating tenants 0.129 0.187 0.234 0.141	6 Sum 0.871 1.081 1.364 0.890	1 Normalised weight 0.145 0.180 0.227 0.148	
Retrofitting stag Retrofitting stag Noise impact on neighbourhood Emission impact on neighbourhood Safety of retrofitting construction Impact on cultural heritage Impact on surrounding traffic and pedestrians	I se Noise impact on neighbourhood 0.145 0.159 0.240 0.141 0.163	I Emission impact on neighbourhood 0.162 0.178 0.254 0.134 0.143	1 Social Safety of retrofitting construction 0.137 0.159 0.227 0.166 0.180	1 dimension Impact on cultural heritage 0.153 0.198 0.203 0.149 0.153	Impact on surrounding traffic and pedestrians 0.145 0.200 0.205 0.159 0.163	Impact on relocating tenants 0.129 0.187 0.234 0.141 0.163	6 Sum 0.871 1.081 1.364 0.890 0.975	1 Normalised weight 0.145 0.180 0.227 0.148 0.163	
Retrofitting stag Retrofitting stag Noise impact on neighbourhood Emission impact on neighbourhood Safety of retrofitting construction Impact on cultural heritage Impact on surrounding traffic and pedestrians Impact on relocating tenants	I re Noise impact on neighbourhood 0.145 0.159 0.240 0.141 0.163 0.152	1 Emission impact on neighbourhood 0.162 0.178 0.254 0.134 0.143 0.143	1 Social Safety of retrofitting construction 0.137 0.159 0.227 0.166 0.180 0.131	1 dimension Impact on cultural heritage 0.153 0.198 0.203 0.149 0.153	Impact on surrounding traffic and pedestrians 0.145 0.200 0.205 0.159 0.163 0.128	I Impact on relocating tenants 0.129 0.187 0.234 0.141 0.163 0.136	6 Sum 0.871 1.081 1.364 0.890 0.975 0.819	1 Normalised weight 0.145 0.180 0.227 0.148 0.163 0.137	
Retrofitting stag Retrofitting stag Noise impact on neighbourhood Emission impact on neighbourhood Safety of retrofitting construction Impact on cultural heritage Impact on surrounding traffic and pedestrians Impact on relocating tenants Sum	I Re Noise impact on neighbourhood 0.145 0.159 0.240 0.141 0.163 0.152 1	1 Emission impact on neighbourhood 0.162 0.178 0.254 0.134 0.143 0.129 1	1 Social Safety of retrofitting construction 0.137 0.159 0.227 0.166 0.180 0.131 1	1 dimension Impact on cultural heritage 0.153 0.198 0.203 0.149 0.153 0.144 1	Impact on surrounding traffic and pedestrians 0.145 0.200 0.205 0.159 0.163 0.128 1	Impact on relocating tenants 0.129 0.187 0.234 0.141 0.163 0.136 1	6 Sum 0.871 1.081 1.364 0.890 0.975 0.819 6	1 Normalised weight 0.145 0.180 0.227 0.148 0.163 0.137 1	
Retrofitting stag Retrofitting stag Noise impact on neighbourhood Emission impact on neighbourhood Safety of retrofitting construction Impact on cultural heritage Impact on surrounding traffic and pedestrians Impact on relocating tenants Sum Operation stage	I se Noise impact on neighbourhood 0.145 0.159 0.240 0.141 0.163 0.152 1	1 Emission impact on neighbourhood 0.162 0.178 0.254 0.134 0.134 0.143 0.129 1	1 Social Safety of retrofitting construction 0.137 0.159 0.227 0.166 0.180 0.131 1	1 dimension Impact on cultural heritage 0.153 0.198 0.203 0.149 0.153 0.144 1	1 Impact on surrounding traffic and pedestrians 0.145 0.200 0.205 0.159 0.163 0.128 1	1 Impact on relocating tenants 0.129 0.129 0.187 0.234 0.141 0.163 0.136 1	6 Sum 0.871 1.081 1.364 0.890 0.975 0.819 6	1 Normalised weight 0.145 0.180 0.227 0.148 0.163 0.137 1	
Retrofitting stag Retrofitting stag Noise impact on neighbourhood Emission impact on neighbourhood Safety of retrofitting construction Impact on cultural heritage Impact on surrounding traffic and pedestrians Impact on relocating tenants Sum Operation stage Accessibility to	Re Noise impact on neighbourhood 0.145 0.159 0.240 0.141 0.163 0.152 1 Accessibility to building facilities for people with special needs	I Emission impact on neighbourhood 0.162 0.162 0.178 0.254 0.134 0.143 0.129 1 Accessibility to building services	I Social Safety of retrofitting construction 0.137 0.137 0.159 0.227 0.166 0.180 0.131 Impacts on maintenance and maintainability from newly added building fabrics and building systems	1 Impact on cultural heritage 0.153 0.198 0.203 0.149 0.153 0.144 1 Safety and security of users	Impact on surrounding traffic and pedestrians 0.145 0.200 0.205 0.159 0.163 0.128 1 Room flexibility for different demands	1 Impact on relocating tenants 0.129 0.187 0.234 0.141 0.163 0.136 1 Sum	6 Sum 0.871 1.081 1.364 0.890 0.975 0.819 6 Normalised weight	1 Normalised weight 0.145 0.145 0.180 0.227 0.148 0.163 0.137 1	

Table D3-4. The column normalised weights for assessment criteria

facilities for people with special needs							
Accessibility to building services	0.174	0.167	0.162	0.171	0.158	0.833	0.167
Impacts on maintenance and maintainability from newly added building fabrics and building systems	0.158	0.169	0.163	0.176	0.151	0.817	0.163
Safety and security of users	0.216	0.216	0.205	0.221	0.252	1.110	0.222
Room flexibility for different demands	0.194	0.198	0.202	0.165	0.187	0.946	0.189
Sum	1	1	1	1	1	5	1

Appendix E-1. Insulation performance

		Insulation materials	Thickness ⁽¹⁾ (mm)	Thermal conductivity ⁽²⁾ (W/m•K)	Heat transfer resistance ⁽³⁾ (m ² •K/W)	Heat transfer coefficient ⁽⁴⁾ (U value) (W/m ² •K)	
Ext	ernal walls (f	from interior to exterior) (standard value:	$U \le 1.0 \text{ W/m}^2 \cdot 1$	K)			
Bef	ore	Cement mortar for internal finishes	20	0.93	0.022		
retro	ofitting	Solid clay brick	240	0.81	0.296	2 020	
		Cement mortar for fixing tiles to brick wall	20	0.93	0.022	2.029	
		Tiles on the external surface of walls	8	2.3	0.003		
	Remaining	Cement mortar for internal finishes	20	0.93	0.022		
ing	original	Solid clay brick	240	0.81	0.296		
ofitt	walls	Cement mortar	20	0.93	0.022		
etro	Newly	Cement mortar with interface agent	10	0.8	0.013	0.692	
t 1	added	EPS board	45	0.048	0.938		
Afi	insulation	Alkali resistant fibreglass mesh	3	0.8	0.004		
	materials	Waterproof paint	0.7	0.002			
Roc	ofs (from top	to bottom) (limit value: U≤0.7 W/m ² •K)					
Bef	ore	Prefabricated insulation panel	60	1.74	0.034		
retro	ofitting	Concrete	40	1.74	0.023		
		Waterproof layer	2	/	/	3.356	
		Cement mortar	20	0.93	0.022		
		Structural plate	120	1.74	0.069		
	Newly	Concrete (30MPa with fine stone)	40	1.74	0.023		
	added	Asphalt felt	1	/	/		
ing	insulation	Cement mortar	20	0.93	0.022		
ofitt	materials	XPS board	50	0.033	1.515		
etro		SBS modified bituminous sheet materials	2	/	/	0.549	
er r	Remainedor	Concrete	40	1.74	0.023		
Aft	iginal roofs	Waterproof layer	2	/	/		
		Cement mortar	20	0.93	0.022		
		Structural plate	120	1.74	0.069		

Calculation:

(4)=1÷(sum of (3) +0.15) (3)=1÷(2)×(1)÷1000

Note:

Thermal conductivity and thickness of each material is designed based on the local standard, Design Standard for Energy Efficiency of Public Buildings (DB33/1036-2007) (UAD, ZIAD & CMA 2007).

	Window types	Heat transfer coefficient of the whole window (W/m ² •K)	Visible light transmittance	Shading coefficient of glazing ⁽¹⁾	Shading coefficient of external sun shading device ⁽²⁾ (SD _H)	Overall shading coefficient (SC) ⁽³⁾
Limited Value		 • U≤4.7 for east- and north-facing windows • U≤3.5 for west- and south-facing windows 	/	/	/	SC≤0.55
Before retrofitting	Steel frame with single glazing (3mm)	5	0.83	1	/	/
	Plastic steel frame with single glazing (3mm)	6.6	0.83	1	/	/
After retrofitting	Aluminium frame with double glazing (5mm low-e+9mm air+5mm normal glazing)	2.57	≥0.72	0.62	0.72	0.446

Table E1-2. Heat transfer coefficient of windows

(3)=(1)×(2)

Note:

The heat transfer coefficient of the whole window is adopted from the local standard, Design Standard for Energy Efficiency of Public Buildings (DB33/1036-2007) (UAD, ZIAD & CMA 2007).

Appendix E-2. Environmental impact assessment

Part of building		Retrofitting	fitting activity Materials		als	Thickness (mm)	Specification of materials	Quantity	Unit	Total weight (kg)	Reference
External walls and	A1	Install/upgrade insulation for external	External walls (Area of externall walls	Cement mortar wit agent (cement:sand	h interface l = 1:3)	10	Weight ratio of cement and sand = 1:3 Density: about 2186 kg/m ³	275,223.60	kg	275,224	
roofs		walls and roofs	excluding windows and doors is about 12590 2m ²)	EPS board		45	Area of one board: 1200mm*600mm Density: about 15.5kg/m ³	17,487.00	pieces	8,782	
			12390.200	Alkali resistant fib	erglass mesh	3	Density: about 160g/m ²	37.77	m3	2,014	
				Waterproof paint		1.5	Weight: 0.2kg/m ² for one coat (double coats are needed)	18.89	m3	5,036	
			Roofs (Area of roofs	Concrete (30MPa	Steel mesh		Type: ø4@200 cold-drawn steel wire	1,500.00	kg		
			is about 1300m ²)	with fine stone) (Containing ø4@200 cold- drawn steel wire)	Concrete (C30)	40	Density: about 2420 kg/m ³	52.00	m3	127,340	[1]
				Asphalt felt		1	Type: Label 200 Each roll weights about 15kg	65.00	rolls	975	
				Cement mortar (cer 1:3)	ment:sand =	20	The weight ratio of cement and sand = 1:3 Density: about 2000 kg/m ³	52,000.00	kg	52,000	
				XPS board		50	Area of one board=1200mm*600mm Density: about 35kg/m ³	1,806.00	pieces	2,276	
				Styrene butadiene s modified bituminos materials	styrene(SBS) us sheet	3	Area of each roll=10m ² Weight: 35kg/roll	130.00	rolls	4,550	
Windows	A2	A2 Change existing single-glazing windows to low- e double glazing with aluminium frames (Area of windows = 2958.7m ²)		Glass		5	Type: 5mm low-e glazing + 9mm air + 5mm normal glazing Density: 2500 kg/m ³	29.59	m3	73,968	
				Aluminium frames			Depth: 80mm Thickness:1.4mm For this type of frame, each window area needs about 8.2kg aluminum Area of each window: 1800mm*1800mm	913.00	frames	24,257	[1]
	A3	Install aluminium sun sh facing south	ading slats for windows	Aluminium sun sh	ading slats		Area of each panel =900mm*1800mm Weight: 40kg/each	280.00	panels	11,200	[1,2]
Lighting	A4	Install motion sensors					Weight: 0.5kg/each	900.00	nieces	450	[3]
	A5	Change T8 fluorescent b	oulbs to T8 LED				Weight: 0.23/each	2,900.00	bulbs	667	[4]
Lifts	A6	Change existing lifts to	energy efficient lifts				Type: REGEN-M Weight: 336kg Load capacity of each lift: 1000kg	4.00	lifts	1,344	[5]
Energy supply	A7	Install PV panels					Area of each panel: 1956mm*992mm Weight: 22.5kg/panel Electricity power: 300W/panel Efficiency rate is 75% Designed yearly electricity generation: about 20531.3kWh	50.00	panels	1,125	[6]
BMCS	A8	Install building energy c	onsumption monitoring s	system				1.00		/	[7]
Water system	A9	Change existing taps and ones	d toilets to water-saving	Change existing w water-saving taps	ater taps to		Type: hob mount infrared sensor taps Material: brass Weight: about 2kg/each Duration of each use: 30 seconds Pressure: 0.06 - 0.6 MPa Flow rate: 4.5L/minute	200.00	taps	400	[8]
				Change existing to saving toilets with water tanks	ilets to water- high pressure		Type: Siphonix toilet with dual flush tank Material: ceramics Weight: about 16kg/each Flow rate: 4.5L/flush	86.00	toilets	1,376	[9]
	A10	Install water treatment s greywater and rainwater	ystem for recycling	Above-ground box integrated water tre	atment plant		Type: HSHA/O Size of water tank: 3500mm*2200mm*1600mm Weight: 150 kg Power: 2.5kW Flow rate: 10m ³ /hour Speed of treatment: 40m ³ /hour Treatment rate: 60%	1.00	item	/	[10]

Table E2-1. Materials and products used for the retrofitting of the case study building

Note:

- Insulation for external walls, roofs, windows, and sun shading devices are designed based on the local standard (reference [1]).
- The selection of materials and products is in consultation with two local professionals in the construction sector.
- The specifications of required materials and products based on local suppliers are cited as below.

References

[1] SAMR & SAC 2020

[2]

https://b2b.baidu.com/land?url=https%3A%2F%2Fb2bwork.baidu.com%2Fland%3Flid%3D171494018366 6943133&query=建筑水平遮阳板

&lattr=&xzhid=33639358&pi=b2b.s.main.2..1707346666575081&category=建材家装%3B 顶篷材料%3B 阳篷%2F 雨篷

&fid=67174400%2C1661959827567&iid=2b4886b8f0728dbed7a7b3dad3825d2e&miniId=8469&jid=3793 341849&prod_type=0

- [3] https://item.jd.com/1781588062.html
- [4] https://search.jd.com/Search?keyword=t5%20灯管&enc=utf-

8&suggest=1.def.0.SAK7|MIXTAG_SAK7R,SAK7_M_AM_L5407,SAK7_M_COL_R,SAK7_S_AM_R,S AK7_SC_PD_R,SAK7_SM_PB_R,SAK7_SS_PM_R,SAK7_CS_PRF_R,tsabtest_base64_U2VhcmNobGlz dF80MzkyfG1pZGRsZUhhdmU_tsabtest|&wq=T5%20d&pvid=339673900f4b4c9cb14378eab85df364 [5] https://magicbook.otiselectric.com.cn/Regen-M/

[6]

https://b2b.baidu.com/land?url=https%3A%2F%2Fwww.zhaosw.com%2Fproduct%2Fdetail%2F244424340 &query=太阳能板&lattr=&xzhid=31415470&pi=b2b.s.main.19..6101857362735708&category=水暖电 工%3B 电池%3B 太阳能电池板%28 组

件%29&fid=67174400%2C1661959827567&iid=e6df38a9958c3bf419625c944d327845&miniId=8469&jid=1612908224&prod_type=0

- [7] http://www.china-antin.com/index.php?ac=article&at=read&did=98
- [8] https://b2b.baidu.com/s?q=节水水龙头

&from=search&fid=67174400%2C1661959827567&pi=b2b.s.search...8839339489235226

[9] https://b2b.baidu.com/land?url=http%3A%2F%2Fwww.51sole.com%2Ftp%2F337584751.htm&query=商用 节水马桶&lattr=&xzhid=37326029&pi=b2b.s.main.7..0779386989916883&category=建材家装%3B 卫浴 洁具%3B 普通坐便器

&fid=67174400%2C1661959827567&iid=f644c437ac88500d9becd0ed93ef1e7e&miniId=8469&jid=1649070388&prod_type=0

[10]

https://b2b.baidu.com/land?url=https%3A%2F%2Fwww.china.cn%2Fwushuichulichengtsb%2F5032117108 .html&query=一体式水处理系统

&lattr=ot&xzhid=35501229&pi=b2b.s.main.5..6263857252096584&category=机械工业%3B 污水处理设备%3B 污水处理成套设备

&fid=67174400%2C1661958457006&iid=1b8cf047a5f3e65a6baf33c4fae02791&miniId=8469&jid=151168 900&prod_type

Part of building		Retrofitting activity		Equipment	Specification of equipment	Power ⁽¹⁾ (kW)	Number ⁽²⁾	Operation duration per equipment ⁽³⁾ (hours)	Amount of diesel needed per hour ⁽⁴⁾ (kg)	Energy consumption of each type of equipment ⁽⁵⁾ (MJ)	Energy consumption by each retrofitting activity ⁽⁶⁾ (MJ)	Carbon emissions by each type of equipment ⁽⁷⁾ (kg)	Carbon emissions by each retrofitting activity ⁽⁸⁾ (kg)
External	A1	Install/upgrade insulation for	External	Electric suspended platform	Load capacity: 0.63 ton	2.5	8	480	/	34,560		7,776	
walls and		external walls and roofs	walls	Rotary hammer		0.98	16	480	/	27,095	65,201	6,096	14,670
roofs				Mortar mixer	Volume: 200 L	2.5	2	197	/	3,546		798	
			Roofs	Rebar cutting machine		2.2	1	1.5	/	12		3	
				Bar straightening machine		1.5	1	4.5	/	24		5	
				Tapered reverse tilting concrete mixer	Output capacity: 500 L	18.5	1	42	/	2,797	21,656	629	4,873
				Mortar mixer	Volume: 200 L	2.5	1	42	/	378		85	
				Autocrane	Load capacity: 50 ton		1	64	6.5	17,762		3,996	
				Concrete pump truck	Pump capacity: 70 m3/hour		1	2	8	683		154	
Windows	A2	Change existing single-glazing to low-e double glazing with al frames	y windows luminium	Electric suspended platform	Load capacity: 0.63 ton	2.5	4	115	/	4,140	4,140	932	932
	A3	Aluminium sun shading slats f	for	Autocrane	Load capacity: 50ton		2	51	6.5	28,307	42,922	6,369	0.(25
		windows facing south		AC arc welder (32KVA)		12	2	168	/	14,515	42,823	3,266	9,035
Lighting	A4	Install motion sensors		Mobile elevator	Load capacity: 300kg	1.5	2	225	/	2,430	2,430	547	547
	A5	Change T8 fluorescent bulbs to	o T8 LED	Mobile elevator	Load capacity: 300kg	1.5	2	363	/	3,920	3,920	882	882
Lifts	A6	Change existing lifts to energy lifts	efficient	Electric winch	JK1.5T with Single drum Speed: 7-12m/minute	7.5	1	92	/	2,484	2,484	559	559
Energy	A7	Install PV panels		Autocrane (50ton)	Load capacity: 50 ton		1	24	6.5	6,661	21.176	1,499	4 765
supply				AC arc welder (32KVA)		12	2	168		14,515	21,170	3,266	4,705
BMCS	A8	Install building energy consumption monitoring system		/	/	/	/	/	/	/	/	/	/
Water	A9	Change existing taps and	Taps	/	/	/	/	/	/	/	/	/	/
system		toilets to water-saving ones	Toilets	/	/	/	/	/	/	/	/	/	/
	A10	Install water treatment system	for	Autocrane	Load capacity: 8 ton	/	1	4	3.6	615		138	
		Install water treatment system for recycling greywater and rainwater		Electric single stage centrifugal water pump	Pump height: 100 m	3.7	1	1.5	/	20	635	4	143

Table E2-2. Energy consumption and CO ₂ emiss	sions by operating construct	ction equipment in the re	etrofitting stage
--	------------------------------	---------------------------	-------------------

• For equipment whose energy source is electricity, energy consumption $(5) = (1) \times (2) \times (3) \times 3.6$

• For equipment whose energy source is diesel, energy consumption $(5) = (2) \times (3) \times (4) \times 11.86 \times 3.6$

Where,

- 11.86 kWh/kg is the coefficient of converting diesel consumption to electricity consumption based on reference [1].
- 3.6 is coefficient for converting kWh to MJ

• Carbon emissions $(7) = (5) \div 3.6 \times 0.81$

Where,

- 0.81 kg CO₂/kWh is carbon intensity of electricity generation in Zhejiang Province, China according to reference [2]

References

[1] SAMR & SAC 2020

[2] Ding & Ying 2019

Part of building		Retrofitting activity Materials Install/ungrade External walls (Area of Interface compent mortar (compents and = 1:3)				Amount ⁽¹⁾	Unit	EE coefficient ⁽²⁾ (MJ/unit)	EC coefficient ⁽³⁾ (kg CO ₂ /unit)	EE(4) (MJ)	Total EE of each retrofitting activity (MJ)	EC emission ⁽⁵⁾ (kg CO ₂)	Total EC emission of each retrofitting activity (kg CO2)	Reference of EE	Reference of EC emission
External	A1	Install/upgrade	External walls (Area of	Interface cement mortar (cem	ent:sand = 1:3)	275,223.6	kg	1.33	0.21	366,047.39		57,246.51		[6]	[6]
walls and		insulation for	external walls excluding	EPS board		566.58	m ³	1,329.6	46.34	753,324.77	1 205 196	26,255.32	104 251	[10]	[10]
roofs		external walls	windows and doors is	Alkali resistant fiberglass me	sh	2,014.4	kg	28	1.35	56,403.20	1,205,180	2,719.44	104,551	[6]	[6]
		and roofs	about 12590.2m ²)	Waterproof paint		5,036.1	kg	5.84	3.6	29,410.82		18,129.96		[8, 13]	[12]
			Roofs (Area of roofs is	Concrete (Containing	Steel mesh	1,500	kg	29	2.2	43,500		3,300	0	[5]	[4]
			about 1300m ²)	ø4@200 cold-drawn steel wire)	Concrete (C30)	52	m ³	2,841	297	147,732		15,444		[15]	[4]
				Asphalt felt	sphalt felt		kg	3	0.16	2,925	(02.212	157.95	90.051	[5]	[5]
				Cement mortar (cement:sand	= 1:3)	52,000	kg	1.33	0.21	69,160	692,312	10,816	80,951	[6]	[6]
				XPS board		65.02	m ³	3,022	669	196,490.44		43,498.38		[10]	[4]
	Styrene butadiene styrene(SI sheet		BS) bituminous	4,550	kg	51.1	1.7	232,505		7,735		[6, 7]	[1]		
Windows	A2	A2 Change existing single-glazing windows to low-e double glazing with aluminium frames (Area of windows = 2958.7m ²) Aluminium frames		73,967.5	kg	16	0.03	1,183,480	1 886 021	2,219.03	232,414	[5, 9, 13]	[15]		
				Aluminium frames		24,256.58	kg	29	9.49	703,440.94	1,000,021	230,194.98		[6]	[14]
	A3	Install aluminit windows facin	um sun shading slats for g south	Aluminium slats		11,200	kg	29	9.49	324,800	324,800	106,288.0	106,288	[6]	[14]
Lighting	A4	Install motion s	ensors			/	/	/	/	/	/	/	/	/	/
	A5	Change T8 flue	prescent bulbs to T8 LED			2,900	tubes	43	2.4	124,700	124,700	6,960	6,960	[2]	[3]
Lifts	A6	Change existin	g lifts to energy efficient lift	S		/	/	/	/	/	/	/	/	/	/
Energy supply	A7	Install PV pane	ls			97.02	m ²	4,070	249	394,871.4	394,871	24,158	24,158	[6]	[11]
BMCS	A8	Install building	energy consumption monit	oring system		/	/	/	/	/	/	/	1	/	/
Water system	A9	Change existing taps	Change to hob mount infrared sensor taps	Brass		2000	kg	20	1.12	40,000	40,000	2240	2240	[6]	[6]
		and toilets to water-saving ones		1290	kg	12.1	0.78	15,609	15,609	1006	1006	[7]	[14]		
	A10	Install water tre recycling greyv	atment system for vater and rainwater	Above-ground box-type integ treatment plant	grated water	/	/	/	/	/	/	/	/	/	/

Table E2-3. Embodied energy and embodied carbon emissions in the retrofitting stage

 $(4)=(1)\times(2)$

 $(5)=(1)\times(3)$

Where,

- Data of column (1) is the amount of building materials and service system, which can be found in Table A2-1

References

[1] Ahmed et al. 2021
 [2] Alstone, Mills & Jacobson 2011
 [3] AUSTEP Lighting 2015
 [4] Cang et al. 2021
 [5] Chen et al. 2022
 [6] Hammond & Jones 2011
 [7] Koezjakov et al. 2018
 [8] Li 2015

[9] Li et al. 2013
[10] Resalati, Kendrick & Hill 2020
[11] Robati, Daly & Kokogiannakis 2019
[12] Wang 2009
[13] Yan 2011
[14] Zhang & Wang 2016
[15] Zhu et al. 2020

Building service system			Electricity cons	sumption and car	bon eimssion befe	ore retrofitting			Electricity con retro	sumption after fitting	Carbon emission		
		Retrofitting activities	% of the total electricity consumption/ carbon emission by the building	Amount of electricity consumption of the building service system		Amount of carbon emissions by the building service system ⁽⁴⁾ (kg	Amount of carbon emissions by the building service system ⁽⁶⁾		Difference on annual electricity consumption by retrofitting	Annual electricity consumption by the service system after retrofitting ⁽⁸⁾	Difference on annual carbon emissions by retrofitting activities ⁽⁹⁾ (kg	Annual carbon emission by the upgraded service system after retrofitting ⁽¹⁰⁾	Reference
			service system ⁽¹⁾	(kWh) ⁽²⁾	(MJ) ⁽³⁾	CO ₂)			activities ⁽⁷⁾ (MJ)	(MJ)	CO ₂)	(kg CO ₂)	
HVAC	A1	Install/upgrade insulation for external walls and roofs			1,454,319.50		60%	13.20%	-872,592	581,728	-196,333	206,433	[7, 10]
	A2	Change existing single-glazing windows to low-e double glazing with aluminium frames	22%	403,977.64		402,765.66	11.3%	2.49%	-164,338	1,289,981	-36,976	365,790	[12]
	A3	Aluminium sun shading slats for windows facing south					9%	1.98%	-130,889	271,877	-29,450	373,316	[11]
Lighting	A4	Install motion sensors		734,504.8	2,644,217.3	732,301.2	30%	12%	-793,265	1,850,952	-178,485	553,817	[9]
system	A5	Change T8 fluorescent bulbs to equivalent T8 LED	40%				/	/	-313,200	2,331,017	-70,470	661,831	[5]
Lifts	A6	Change existing lifts with energy efficient lifts	1%	18,362.62	66,105.43	18,307.53	30%	0.3%	-19,832	46,274	-4,462	13,845	[1, 2, 3, 8]
Energy supply	A7	Install PV panels	/	/	/	/	/	1.11%	-73,377	-73,377	-16,510	-16,510	[14]
BMCS	A8	Install building energy consumption monitoring system	/	/	/	/	/	2.4%	-158,653	-158,653	-35,697	-35,697	[4, 8]
Water system	A9	Change existing taps and toilets to water-saving ones	/	/	/	/	/	/	/	1	/	/	/
	A10	Install water treatment system for	/	/	/	/	/	/	6,750	6,750	1,519	1,519	[13]

Table E2-4. Operation energy consumption and CO₂ emissions

Note: The negative symbol "-" represents energy saving or carbon emissions reduction

Calculation:

- (1) Electricity consumption
- Difference on annual electricity consumption by retrofitting activity A1, A2, A3, A4, A6, A7 and A8 is calculated as:

(7)=(6)×1,836,262×3.6

 $(6)=(1)\times(5)$

Where,

- Data (5) are cited from below references.
- 1,836,262 kWh is the total annual energy consumption by the case building before retrofitting, shown in Table 7.1.
- Data of (1) are based on electricity bills of the building during the past three years, shown in Figure 8.1.
- For retrofitting activity A1, A2, A3, A4 and A6, difference on annual electricity consumption can also be calculated as:

$(7)=(3)\times(5)$

(3)=(2)×3.6

(2)=(1)×1,836,262

Where,

- Data of (2) and (3) can be achieved by multiplying (1) with total annual energy consumption (or carbon emissions) of the case building before retrofitting, shown in Table 7.1.
- 3.6 is coefficient for converting kWh to MJ
- Difference on annual electricity saving by retrofitting activity A5 is calculated as:

(32-17)÷1000×2900×8×250×3.6=313,200 MJ

Where,

- 32W is the power of one T8 fluorescent bulb based on reference [5].
- 17W is the power of one T8 LED bulb based on reference [5].
- There are 2,900 bulbs that need to be replaced.
- Assuming a working day of 8 hours, and 250 working days per year.

- 3.6 is coefficient for converting kWh to MJ
- Annual electricity consumption by upgraded building service systems by retrofitting activity A1 A8 is calculated as:

(8)=(3)+(7)

Where,

- (8) is the annual electricity consumption by the upgraded building service system after implementing the retrofitting activity
- (3) is the annual electricity consumption (in MJ) by the building service system before retrofitting
- (7) is the difference on the annual energy consumption of the building service system by the retrofitting activity
- Annual electricity consumption by retrofitting activity A10 is calculated as:
 - The power of the above-ground box-type integrated water treatment plant is 2.5 kW, and treatment speed is 40 kL/hour based on reference [12].
 - Based on the water bill during the past three years, annual water consumption is 23,970 kL (shown in Table 7.10).
 - Assuming 250 working days per year, about 95.88 kL water is consumed per working day by the case building before retrofitting.
 - Based on above information, the time that the water treatment plant works about 3 (≈95.88/40) hour maximum per working day.
 - Therefore, the electricity consumption by operating the water treatment plant is: 2.5×3×250×3.6=6750 MJ.

(2) Carbon emissions

• Difference in annual carbon emissions by retrofitting activities is calculated as:

(9)=(7)÷3.6×0.81

Where,

- (7) is the difference on annual energy consumption of the building service system by the retrofitting activity
- 0.81 kg CO₂/kWh is carbon intensity of electricity generation in Zhejiang Province, China according to reference [6]
- Annual carbon emissions of building service systems after implementing the retrofitting activity are calculated as:

(10)=(4)+(9)

Where,

- (4) is annual carbon emissions of the building service system before retrofitting
- (9) is the difference on the annual carbon emissions of the building service system by the retrofitting activity

References

- [1] Ali et al. 2021
- [2] Al-Kodmany 2015
- [3] Carrillo et al. 2013
- [4] Chen et al. 2020
- [5] Jappha 2018
- [6] Grace & Ying 2019
- [7] Kumar et al. 2020
- [8] Nguyen 2017
- [9] Riyanto et al. 2018
- [10] Tsang et al. 2022
- [11] Valladares-Rendón, Schmid & Lo 2017
- [12] Zhuo & Zhao 2013
- [13] https://b2b.baidu.com/land?url=https%3A%2F%2Fwww.china.cn%2Fwushuichulichengtsb%2F5032117108.html&query=一体式水处理系统 &lattr=ot&xzhid=35501229&pi=b2b.s.main.5..6263857252096584&category=机械工业%3B 污水处理设备%3B 污水处理成套设备 &fid=67174400%2C1661958457006&iid=1b8cf047a5f3e65a6baf33c4fae02791&minId=8469&jid=151168900&prod type
- [14] https://b2b.baidu.com/land?url=https%3A%2F%2Fwww.zhaosw.com%2Fproduct%2Fdetail%2F244424340&query=太阳能板 &lattr=&xzhid=31415470&pi=b2b.s.main.19..6101857362735708&category=水暖电工%3B 电池%3B 太阳能电池板%28 组 件%29&fid=67174400%2C1661959827567&iid=e6df38a9958c3bf419625c944d327845&miniId=8469&jid=1612908224&prod type

Part of building		Retrofitting activity		Service life ⁽¹⁾ (years)	EE ⁽²⁾ (MJ)	EC emission ⁽³⁾ (kg CO2)	Recurrent EE ⁽⁴⁾ (MJ)	Recurrent EC emission ⁽⁵⁾ (kg CO ₂)
External walls	A1	Install/upgrade insulation for	External walls	20	1,205,186.18	104,351.23	2,410,372	208,702
and roofs		external walls and roofs	Roofs	20	692,312.44	80,951.33	1,384,625	161,903
Windows	A2	Change existing single-glazing windo double glazing with aluminium frame	ows to low-e es	20	1,886,920.94	232,414.01	3,773,842	464,828
	A3	Install aluminium sun shading slats for facing south	30	324,800.00	106,288.00	324,800	106,288	
Lighting	A5	Change T8 fluorescent bulbs to T8 L	ED	30	124,700.00	6,960.00	124,700	6,960
Energy supply	A7	Install PV panels	30	394,871.40	24,157.98	394,871	24,158	
Water system	A9	Change existing taps and toilets to	Taps	50	40,000.00	2,240.00	/	/
		water-saving ones	20	15,609.00	1,006.20	31,218	2,012	

Table E2-5. Recurrent embodied energy and re	ecurrent carbon emissions in the operation stage
--	--

(4)=(2)×integer of [the study period/(1)]

(5)=(3)×integer of [the study period/(1)]

Where,

- Data of (2) and (3) can be found in Table A2-3
- Assuming a 50-year remaining service life as the study period for this case study

Note:

Service life of building materials and service systems (in column (1)) can be found in Table 4.5 in Chapter 4, adopted from Australian Cost Management Manual, Volume 3; Kubba 2010; EATS 2015; Penny 2015; Kono et al. 2016; Alam et al. 2017; RICS 2018; and Tavares, Silva & de Brito 2020

				-	Retrofitting stag	e				
Part of building		Retrofitting activity		EE ⁽¹⁾ (MJ)	Energy consumption by operating construction equipment ⁽²⁾ (MJ)	Total consumption on retrofitting stage ⁽³⁾ (MJ)	Recurrent EE ⁽⁴⁾ (MJ)	Difference on energy Total difference consumption by retrofitting stage ⁽⁶⁾ (MJ)		Total difference on energy consumption by retrofitting activities ⁽⁷⁾ (MJ)
External	A1	Install/upgrade insulation for	External walls	1,205,186	65,201		2,410,372			
walls and roofs		external walls and roofs	Roofs	692,312	21,656	1,984,356	1,384,625	-43,629,585	-39,834,588	-37,850,232
Windows	A2 Change existing single-glazing windows to low-e double glazing with aluminium frames		1,886,921	4,140	1,891,061	3,773,842	-8,216,905	-4,443,063	-2,552,002	
	A3	A3 Install aluminium sun shading slats for windows facing south		324,800	42,823	367,623	324,800	-6,544,438	-6,219,638	-5,852,015
Lighting	A4	Install motion sensors		/	2,430	2,430	/	-39,663,259	-39,663,259	-39,660,829
	A5	Change T8 fluorescent bulbs to	124,700	3,920	128,620	124,700	-15,660,000	-15,535,300	-15,406,680	
Lifts	A6	Change existing lifts to energy	efficient lifts	/	2,484	2,484		-991,582	-991,582	-989,098
Energy supply	A7	Install PV panels		394,871	21,176	416,047	394,871	-3,668,852	-3,273,980	-2,857,933
BMCS	A8	Install building energy consumption monitoring system		/	/	/	/	-7,932,652	-7,932,652	-7,932,652
Water	A9	Change existing taps and	Taps	40,000	/	55 609	/	/	31 218	86 827
system		toilets to water-saving ones	Toilets	15,609	/	55,009	31,218	1	51,218	80,827
	A10	Install water treatment system f greywater and rainwater	for recycling	/	635	635	/	337,500	337,500	338,135

Table E 2-6. Total energy consumption by retrofitting activities

Note: symbol "-" represents energy saving by retrofitting activities

(3)=(1)+(2)

(6)=(4)+(5)

(7)=(3)+(6)

Where,

- Data in column (1) can be found in Table A2-3
- Data in column (2) can be found in Table A2-2
- Data in column (4) can be found in Table A2-5
- Data in column (5) is calculated as: (5)= the study period (50 years)×Column (6) in Table A2-4

Table E2-7. Total carbon emissions by retrofitting activities

Part of building				R	etrofitting stage	e		je	Total		
		Retrofitting activity	y	EC emissions ⁽¹⁾ (kg)	Carbon emissions by operating construction equipment ⁽²⁾ (kg)	Total emissions on retrofitting stage ⁽³⁾ (kg)	REC emissions ⁽⁴⁾ (kg)	Difference on carbon emissions by retroiftting activities ⁽⁵⁾ (kg)	Total difference on operation stage ⁽⁶⁾ (kg)	difference on carbon emissions by retrofitting activities ⁽⁷⁾ (kg)	
External	A1	Install/upgrade insulation for	External walls	104,351	14,670	201015	208,702		0.446.051	0.041.000	
walls and roofs		external walls and roofs	Roofs	80,951	4,873	204,845	161,903	-9,816,657	-9,446,051	-9,241,206	
Windows	Windows A2 Change existing single-glazing wir low-e double glazing with alumining		g windows to minium frames	232,414	932	233,346	464,828	-1,848,804	-1,383,976	-1,150,630	
	A3	Install aluminium sun shading windows facing south	106,288	9,635	115,923	106,288	-1,472,499	-1,366,211	-1,250,287		
Lighting	A4	Install motion sensors		/	547	547	/	-8,924,234	-8,924,234	-8,923,687	
	A5	Change T8 fluorescent bulbs t	o T8 LED	6,960	882	7,842	6,960	-3,523,500	-3,516,540	-3,508,698	
Lifts	A6	Change existing lifts to energy	v efficient lifts	/	559	559	/	-223,106	-223,106	-222,547	
Energy supply	A7	Install PV panels		24,158	4,765	28,923	24,158	-825,492	-801,334	-772,411	
BMCS	A8	B Install building energy consumption monitoring system		/	/	/	/	-1,784,847	-1,784,847	-1,784,847	
Water	A9	Change existing taps and	Taps	2,240	/	2 246	/	/	2 012	5 250	
system		toilets to water-saving ones	Toilets	1,006	/	5,240	2,012	1	2,012	5,259	
А	A10	Install water treatment system greywater and rainwater	for recycling	/	143	143	/	75,938	75,938	76,080	

Note: symbol "-" represents reduction of carbon emission by retrofitting activities

(3)=(1)+(2)

(6)=(4)+(5)

(7)=(3)+(6)

Where,

- Data in column (1) can be found in Table A2-3
- Data in column (2) can be found in Table A2-2
- Data in column (4) can be found in Table A2-5
- Data in column (5) can be calculated as: (5)= the study period (50 years)×Column (8) in Table A2-4

Table E2-8. Use of recyclable/reusable materials

Part of building	Retrofitting activity		Materials	Weight ⁽¹⁾ (kg)	Total weight added by a retrofitting activity ⁽²⁾ (kg)	Recyclable or reusable materials	Recyclable or reusable rate ⁽³⁾ (%)	Weight of reusable or recyclable materials ⁽⁴⁾ (kg)	% of recyclable or reusable materials ⁽⁵⁾	Note	
External	A1	Install/upgrade insulation for	External	Mortar (cement:sand = 1:3)	275,223.6		• Steel				
walls and		building envelops	walls	EPS board	8,781.97	-	Concrete				
roofs				Alkali resistant fiberglass mesh	2,014.43	-					
				Waterproof paint	5,036.08						
			Roofs	Concrete Steel wire	1,500						
				Concrete (C30)	125,840	478,196.64		/	127,340	27%	/
				Asphalt felt	975						
				Mortar (cement:sand = 1:3)	52,000	-					
	XPS bo Styrene bitumino		XPS board	2,275.56							
			Styrene butadiene styrene (SBS) bituminous sheet	4,550							
Windows	A2 Change existing single-glazing windows		Glass	73,967.5		• Glass					
		to low-e double glazing with aluminium frames		Aluminium frames	24,256.58	98,224.08	• Aluminium	/	98,224.08	100%	/
	A3	Install aluminium sun shading windows facing south	slats for	Aluminium sun shading slats	11,200	11,200	• Aluminium	/	11,200	100%	/
Lighting	A4	Install motion sensors			450	450	/	/	0	0%	/
	A5	Change T8 fluorescent bulbs t equivalent T8 LED	0	LED bulbs	667	667	• Glass	20%	133	20%	[1]
Lifts	A6	6 Change existing lifts to energy efficient lifts		Lifts	1,344	1,344	Whole lifts	90%	1,210	90%	[2]
Energy supply	A7	A7 Install PV panels			1,125	1,125	/	/	0	0%	/
BMCS	A8	8 Install energy consumption monitoring system			/	/	/	/	0	0%	[3]
Water	A9	9 Change existing taps and Taps		Brass	400	1 776	• Brass	1	1 776	100%	/
system		toilets to water-saving ones	Toilets	Ceramics	1,376	1,770	Ceramics	/	1,770	10070	/
	A10	Install water treatment system recycling greywater and rainw	for ater	Above-ground box-type integrated water treatment plant	150	150	Whole treatment plant	90%	135	90%	[4]

Note: highlighted materials are recyclable or reusable materials

• For retrofitting activities A1, A2, A3, A4, A7 and A9, the proportion of recyclable or reusable materials for future use is calculated as:

(5)=(4)÷(2)×100%

Where,

- (4) is sum of the weight of recyclable or reusable materials from column (1) (the highlighted items)
- (2) is sum of (1) contained in the retrofitting activity
- Data of (1) can be found in Table A2-1
- For retrofitting activities A5 and A6, proportion of recyclable or reusable materials for future use is calculated as:

$(5)=(3)\times(2)$

Where,

- (3) is reuse or recycle rate, referring to below notes
- (2) is sum of (1) contained in the retrofitting activity
- Data of (1) can be found in Table A2-1

Note:

[1] Based on the study by Franz and Wenzi (2017), about 20% of the weight of a LED bulb can be recycled. This figure is adopted in this case study as the recyclable rate of retrofitting activity A5.

[2] Removed lifts are often disassembled, and most components can be recycled or reused. Considering the finishing and lights in lift cars, it is assumed that 90% of removed lifts can be recycled or reused.

[3] There is no data about weight or recycling or reuse information available for the building energy consumption monitoring system. The main components of the monitoring system are electricity meters, which are very light compared to other added building materials or service system. Therefore, it is neglected in this estimation.
[4] Similar to lifts, most components of the water treatment plant can be recycled or reused. Assuming 10% of a removed water treatment plant is broken and will go to disposal, 90% of the water treatment plant can be recycled or reused.

Table E2-9. Waste generation

Part of building		Retrofitting activit	y	Service life ⁽¹⁾ (years)	Total weight added by a retrofitting activity ⁽²⁾ (kg)	Weight of recyclable or reusable materials ⁽³⁾ (kg)	Total waste generation ⁽⁴⁾ (kg)
External	A1	Install/upgrade insulation for	External walls	20	178 107	127 340	701 713
walls and		external walls and roofs	Roofs	20	470,177	127,540	701,715
Windows	A2	Change existing single-glazing e double glazing with aluminiu	windows to low- im frame	20	98,224	98,224.08	0
	A3	Install aluminium sun shading windows facing south	slats for	30	112,200	112,200	0
Lighting	A4	Install motion sensors		15	450	0	1,350
	A5	Change T8 fluorescent bulbs to	o T8 LED	30	667	133	534
Lifts	A6	Change existing lifts to energy	efficient lifts	30	1,344	1,210	134
Energy supply	A7	Install PV panels		30	1,125	0	1,125
BMCS	A8	Install building energy consum system	nption monitoring	15	0	0	0
Water	A9	Change existing taps and	Taps	50	400	400	0
system		toilets to water-saving ones	Toilets	20	1,376	1,376	0
	A10	Install water treatment system greywater and rainwater	for recycling	30	150	135	15

Calculation:

 $(4)=[(2)-(3)]\times$ integer of [the study period/(1)]

Where,

- (2) is the total weight of added materials by retrofitting activities, which can be found in Table A2-1
- (3) is weight of recyclable or reusable materials, which can be found in Table A2-8
- Assuming a 50-year remaining service life as the study period for this case study

Note:

Service life of building materials and service systems (in column (1)) can be found in Table 4.5 in Chapter 4, adopted from Australian Cost Management Manual, Volume 3; Kubba 2010; EATS 2015; Penny 2015; Kono et al. 2016; Alam et al. 2017; RICS 2018; and Tavares, Silva & de Brito 2020

Appendix E-3. Economic performance assessment

Building service					Retrofitting stage					Ope	ration stage							EAC of the upgraded	
Building service system		Retrofitting activities		Life expectancy	Initial cost ⁽¹⁾	Annual o consumption service sys retrot	electricity 1 by building stem after fitting	Annual water consumption by building service system after retrofitting ⁽⁴⁾	Unit price of electricity/ water ⁽⁵⁾	Annual operation cost ⁽⁶⁾			Rep	lacement cos	t ⁽⁷⁾ (¥)			building service system after implementing the retrofitting activity ⁽⁸⁾	Reference of initial cost
				years	¥	MJ ⁽²⁾	kWh ⁽³⁾	kL	¥/kWh or ¥/kL	¥	year 15	year 20	year 25	year 30	year 35	year 40	year 45	¥	
HVAC	A1	Add insulation for external walls and roofs	External walls	20	438,340	581.728	161.591	/	0.64	103.418.28		771.376				771.376		155.680	[1]
			Roofs	20	333,036	,.=0	,											,	1-1
	A2	Change single-glazing windows double glazing with aluminium	s to low-e frames	20	710,088	1,289,981	358,328	/	0.64	229,330.03		710,088				710,088		302,500	[2]
	A3	Install aluminium sun shading s windows facing south	lats for	30	249,480	271,877	75,521		0.64	48,333.67				249,480				108,405	[3]
Lighting	A4	Install motion sensors		15	70,200	1,850,952	514,153	/	0.64	329,058.15	70,200			70,200			70,200	337,355	[4]
system	A5	Change T8 fluorescent bulbs to	T8 LED	30	83,810	2,331,017	647,505	/	0.64	414,403.07				83,810				421,935	[5]
Lifts	A6	Change existing lifts to energy e	efficient lifts	30	2,625,390	46,274	12,854	/	0.64	8,226.45				2,625,390				244,160	[6]
Energy	A7	Install PV panels		30	34,900	-73,377	-20,383	/	0.64	-13,044.81				34,900				-9908	[7]
BMCS	A8	Install building energy consump monitoring system	otion	15	270,000	-158,653	-44,070	1	0.64	-28,204.98	270,000			270,000			270,000	3,708	[8]
Water	A9	Change existing taps and	Taps	50	11,000	/	/	2 734	0.49	1 339 54		47 300				47 300		7 113	[9 10]
system		toilets to waer-saving ones	Toilets	20	47,300	1	,	2,754	0.47	1,557.54		47,500				-7,500		7,115	[2, 10]
	A10 Install water treatmet system for recycling greywater and rainwater			30	31,900	6,750	1,875	-1,361	0.64 or 0.49	533.11				31,900				3,400	[11]

Table E3-1. EAC of building service system after implementing proposed retrofitting activities

Note: The negative symbol "-" represents cost saving.

Calculation:

• Calculate operation cost: $(6)=(3)\times(5)$

(3)=(2)÷3.6

Where,

- (2) can be found in column (8) in Table A2-4.

- For retrofitting activity A10, the operation of the water treatment system consumes electricity and results in water saving. Therefore, the annual cost of the water system by implementing retrofitting activity A10 is the annual electricity cost with compensation of cost saving by annual reduced water consumption.
- Calculation EAC of upgraded building service system:

Step 1. Convert replacement cost (column (7) to NPV based on equation:

$$NPV = \frac{C}{(1+i)^d}$$

Where,

- C is replacement cost occurs at year d
- *i* is current real discount rate, 8% in this study

Step 2. Convert initial cost (1) and NPV of replacement cost (from step 1) to EAC based on equation:

$$EAC_{(1,7)} = \frac{P \times i \times (1+i)^n}{(1+i)^n - 1}$$

Where,

- $EAC_{(1,7)}$ is the EAC when only considering initial cost and replacement cost
- P is sum of initial cost (column (1)) and NPV of replacement cost from step 1
- n is the study period, 50 years

Step 3. Calculate total EAC (column (8)) of the upgraded building service system by a retrofitting activity:

 $(8) = EAC_{(1,7)} + (6)$

Note:

- The life expectancy of building components and service systems is adopted from Australian Cost Management Manual, Volume 3 (2000), and studies by Kubba (2010), EATS (2015), Penny (2015), Kono et al. (2016), Alam et al. (2017), RICS (2018), and Tavares, Silva and de Brito 2020.
- According to CEIC database (CEIC 2022a), the electricity price for business use in China in 2022 is ¥0.64/kWh.
- According to CEIC database (CEIC 2022b), the water price for business use in China in 2022 is ± 0.49 /kL.
- The initial cost of retrofitting activity A1 is adopted from the study about retrofitting by Chow, Li and Darkwa (2013), which is conducted in the same region and climate zone as this case study. For the capital cost of other building components and service systems, they are adopted from local procurement websites (accessed in 2022). The URLs are listed in the below references.

References

- [1] Chow, Li & Darkwa (2013)
- [2] https://www.to8to.com/yezhu/v70425.html
- [3] https://b2b.baidu.com/land?url=https%3A%2F%2Fb2bwork.baidu.com%2Fland%3Flid%3D1714940183666943133&query=建筑水平遮阳板 &lattr=&xzhid=33639358&pi=b2b.s.main.2..1707346666575081&category=建材家装%3B 顶篷材料%3B 阳篷%2F 雨篷 &fid=67174400%2C1661959827567&iid=2b4886b8f0728dbed7a7b3dad3825d2e&miniId=8469&jid=3793341849&prod type=0
- [4] https://item.jd.com/1781588062.html
- [5] https://item.jd.com/11476147694.html
- [6] https://magicbook.otiselectric.com.cn/Regen-M/
- [7] https://b2b.baidu.com/land?url=https%3A%2F%2Fwww.zhaosw.com%2Fproduct%2Fdetail%2F244424340&query=太阳能板 &lattr=&xzhid=31415470&pi=b2b.s.main.19..6101857362735708&category=水暖电工%3B 电池%3B 太阳能电池板%28 组 件%29&fid=67174400%2C1661959827567&iid=e6df38a9958c3bf419625c944d327845&miniId=8469&jid=1612908224&prod_type=0
- [8] http://www.china-antin.com/index.php?ac=article&at=read&did=98
- [9] https://b2b.baidu.com/land?url=https%3A%2F%2Fb2bwork.baidu.com%2Fland%3Flid%3D1735054231525472959&query=节水水龙头 &lattr=&xzhid=40649129&pi=b2b.s.main.9..5414118005152788&category=建材家装%3B水龙头%3B 感应龙头 &fid=67174400%2C1661959827567&iid=6af775f890a8412191db555aab7065f4&miniId=8469&jid=2403942115&prod_type=0
- [10] https://b2b.baidu.com/land?url=http%3A%2F%2Fwww.51sole.com%2Ftp%2F337584751.htm&query=商用节水马桶 &lattr=&xzhid=37326029&pi=b2b.s.main.7..0779386989916883&category=建材家装%3B 卫浴洁具%3B 普通坐便器 &fid=67174400%2C1661959827567&iid=f644c437ac88500d9becd0ed93ef1e7e&miniId=8469&jid=1649070388&prod_type=0
- [11] https://b2b.baidu.com/land?url=https%3A%2F%2Fwww.china.cn%2Fwushuichulichengtsb%2F5032117108.html&query=一体式水处理系统 &lattr=ot&xzhid=35501229&pi=b2b.s.main.5..6263857252096584&category=机械工业%3B 污水处理设备%3B 污水处理成套设备 &fid=67174400%2C1661958457006&iid=1b8cf047a5f3e65a6baf33c4fae02791&miniId=8469&jid=151168900&prod_type

		-		Retrofitting stage					Opera	tion stage								
	Retrofitting activities		Life expectancy	Initial cost ⁽¹⁾	Difference (electricity co by retrofittin	on annual onsumption ng activities	Difference on water consumption by retrofitting activities ⁽⁴⁾	Unit price of electricity/ water ⁽⁵⁾	Difference on annual electricity/water cost by retrofitting activity ⁽⁶⁾			Repl	acement cost	. ⁽⁷⁾ (¥)			NPV of initial cost and replacement cost ⁽⁸⁾ (¥)	Equivalent annual cost or saving by retrofitting activities ⁽⁹⁾
			years	¥	MJ ⁽²⁾	kWh ⁽³⁾	kL	¥/kWh or ¥/kL	¥	at year 15	at year 20	at year 25	at year 30	at year 35	at year 40	at year 45	¥	¥
A1	Add insulation for external walls and roofs	External walls	20	438,340	-872,591.7	-242,387	/	0.64	-155,127.41		771,376				771,376		972,380	-75,642
		Roofs	20	333,036														
A2	Change single-glazing window e double glazing with aluminiu	rs to low- m frames	20	710,088	-164,338.1	-45,649	/	0.64	-29,215.66		710,088				710,088		895,122	43,954
A3	Install aluminium sun shading windows facing south	slats for	30	249,480	-130,888.76	-36,358	/	0.64	-23,269.11				249,480				274,273	-849
A4	Install motion sensors		15	70,200	-661,054.32	-183,626	/	0.64	-117,520.77	70,200			70,200			70,200	101,505	-109,223
A5	Change T8 fluorescent bulbs to	T8 LED	30	83,810	-313,200	-87,000	/	0.64	-55,680.00				83,810				92,139	-48,148
A6	Change existing lifts to energy lifts	efficient	30	2,625,390	-19,831.63	-5,509	/	0.64	-3,525.62				2,625,390				2,886,294	232,408
A7	Install PV panels		30	34,900	-73,377.03	-20,383	/	0.64	-13,044.81				34,900				38,368	-9,908
A8	Install building energy consum monitoring system	ption	15	270,000	-158,653.04	-44,070	/	0.64	-28,204.98	270,000			270,000			270,000	390,406	3,708
A9	Change existing taps and toilets	s Taps	50	11,000	1	,	1 222 75	0.40	(04.54		47 200				47 200		70 (25	5.1(0
	to waer-saving ones	Toilets	20	47,300	/	/	-1,233.75	0.49	-604.54		47,300				47,300		/0,625	5,169
A10	Install water treatment system f recycling greywater and rainwa	for ater	30	31,900	6,750	1,875	-1,361	0.49	533.11				31,900				35,070	3,400

Table E3-2. Equivalent annual cost or saving by retrofitting activities

Note: The negative symbol "-" represents cost saving.

Calculation:

(3)=(2)÷3.6

Where,

- (2) can be found in column (7) in Table A2-4

(6)=(3) (and/or (4))×(5)

$(8)=(1)+\sum[(7)\div(1+i)^{d}]$

Where,

- (1) can be found in column (1) in Table A3-1
- i is the current real discount rate, 8%
- *d* is the year that replacement cost occurs

$(9)=(6)+(8)\times i\times (1+i)^{n} \div [(1+i)^{n}-1]$

Where,

- n is study period in the case study, 50 years

Appendix E-4. Social performance assessment

Table E4-1. Social performance assessment

Life stage	Assessmen	t criteria	A1	A2	A3	A4	A5	A6	A7	A8	A9	A10
Retrofitting stage	Noise impact on neighbourhood	Professional 1	4	3	4	1	1	1	2	1	1	3
8-		Professional 2	2	4	3	1	1	1	3	2	1	1
		Professional 3	4	4	4	1	1	1	3	2	1	3
		Professional 4	4	3	3	1	1	1	4	3	1	2
		Professional 5	4	5	4	1	1	1	3	2	1	3
		Final score (average)	3.6	3.8	3.6	1	1	1	3	2	1	2.4
	Emission impact on neighbourhood	Professional 1	5	2	4	1	1	1	3	1	1	3
		Professional 2	3	3	3	1	1	1	2	1	1	2
		Professional 3	4	3	4	1	1	1	3	2	1	3
		Professional 4	4	2	3	1	1	1	3	2	1	2
		Professional 5	3	3	3	1	1	1	3	2	1	3
		Final score (average)	3.8	2.6	3.4	1	1	1	2.8	1.6	1	2.6
	Impacts on safety of undertaking the	Professional 1	4	4	4	2	3	3	4	3	2	2
	retrofitting activity	Professional 2	3	3	4	1	2	1	3	1	2	1
		Professional 3	4	3	4	1	3	3	3	2	2	2
		Professional 4	4	4	4	2	3	2	3	2	1	2
		Professional 5	3	4	3	2	2	2	3	2	1	2
		Final score (average)	3.6	3.6	3.8	1.6	2.6	2.2	3.2	2	1.6	1.8

	Impacts on cultural	Professional 1	4	4	4	1	1	1	3	1	2	3
	nernage	Professional 2	4	4	3	1	2	2	3	2	1	4
		Professional 3	5	4	4	1	1	2	4	3	4	4
		Professional 4	2	3	5	2	2	2	5	3	3	3
		Professional 5	4	4	4	2	2	3	4	3	3	4
		Final score (average)	3.8	3.8	4	1.4	1.6	2	3.8	2.4	2.6	3.6
	Impacts on surrounding traffic	Professional 1	5	3	3	1	1	1	2	3	1	2
	and pedestrians	Professional 2	4	4	3	1	1	2	3	1	1	1
		Professional 3	4	4	4	1	1	1	2	2	1	3
		Professional 4	4	3	3	1	1	1	2	2	2	3
		Professional 5	4	3	4	1	1	2	1	3	1	3
	Impacts on	Final score (average)	4.2	3.4	3.4	1	1	1.4	2	2.2	1.2	2.4
	Impacts on relocating tenants	Professional 1	2	3	2	2	4	4	1	1	4	2
	C C	Professional 2	4	4	3	3	3	3	1	2	4	1
		Professional 3	4	4	3	3	3	3	2	2	3	1
		Professional 4	3	3	3	2	4	3	2	2	3	1
		Professional 5	4	4	3	1	3	3	2	2	4	1
		Final score (average)	3.4	3.6	2.8	2.2	3.4	3.2	1.6	1.8	3.6	1.2
Operation stage	Accessibility to building facilities	Professional 1	2	2	2	3	2	3	2	2	4	2
	for people with additional needs	Professional 2	4	2	3	4	3	5	3	3	5	2
		Professional 3	3	3	3	3	2	4	3	3	4	2
		Professional 4	4	2	4	3	4	5	4	2	4	3
		Professional 5	4	3	4	3	4	5	4	3	4	3

	Final score (average)	3.4	2.4	3.2	3.2	3	4.4	3.2	2.6	4.2	2.4
Accessibility to building services	Professional 1	5	3	5	2	2	3	2	2	4	2
C	Professional 2	4	3	4	2	5	4	3	3	5	5
	Professional 3	4	4	4	3	3	4	3	2	4	3
	Professional 4	5	4	5	2	4	5	4	4	5	4
	Professional 5	4	4	4	2	4	4	4	3	4	3
	Final score (average)	4.4	3.6	4.4	2.2	3.6	4	3.2	2.8	4.4	3.4
Impacts on maintenance and	Professional 1	4	2	2	2	3	4	2	5	3	2
maintainability from newly added	Professional 2	2	2	3	2	4	3	2	4	3	3
building	Professional 3	3	2	3	2	3	4	2	3	4	2
service systems	Professional 4	4	2	3	2	4	4	3	3	4	3
ervice systems P ₁	Professional 5	4	2	2	2	4	3	2	2	4	2
	Final score (average)	3.4	2	2.6	2	3.6	3.6	2.2	3.4	3.6	2.4
Impacts on safety and security of	Professional 1	4	3	2	2	2	4	2	2	2	2
users	Professional 2	2	4	1	2	2	4	3	2	4	2
	Professional 3	4	4	2	2	3	5	3	3	3	3
	Professional 4	4	4	2	2	4	4	3	3	4	3
	Professional 5	5	4	2	2	3	4	3	3	3	3
F (Final score (average)	3.8	3.8	1.8	2	2.8	4.2	2.8	2.6	3.2	2.6
Impacts on room flexibility for	Professional 1	2	2	2	2	2	2	2	2	2	2
different demands	Professional 2	3	2	4	2	2	2	2	2	4	3
	Professional 3	2	3	3	2	3	2	2	3	2	2

	Professional 4	3	2	2	2	3	2	2	3	2	2
	Professional 5	3	2	3	2	3	3	2	3	2	2
	Final score (average)	2.6	2.2	2.8	2	2.6	2.2	2	2.6	2.4	2.2

Note:

• The final score regarding each assessment criterion is the average score of the given scores by the five professionals, which can be calculated based on the Equation:

$$S_{ij} = \frac{\sum S_{ij}^n}{5}$$

Where,

- S_{ij} social assessment score of retrofitting activity *i* regarding assessment criterion *j*
- S_{ij}^{n} the rating score of retrofitting activity *i* regarding assessment criterion *j* given by the nth professional
- The rating scores are given based on the provided rating scale as the below table shows:

Impact range on retrofitting stage	Value score	Performance range in operation stage
Very significant impact	5	Excellent performance
Significant impact	4	Very good performance
Average impact	3	Good performance
Minor impact	2	Same as current performance
Minimum impact	1	Unsatisfied performance

Appendix E-5. Normalisation

														Retrofittin	g activities									
Ass	essment criteria	Unit	Max	Min	Α	1	A	12	A	13	Α	4	А	5	1	16	Α	.7	А	.8	Α	.9	Α	10
					Est.	Norm.	Est.	Norm.	Est.	Norm.	Est.	Norm.	Est.	Norm.	Est.	Norm.	Est.	Norm.	Est.	Norm.	Est.	Norm.	Est.	Norm.
ge	Energy consumption	MJ	1,984,355	0	1,984,355	0	1,891,061	5	367,623	81	2,430	100	128,620	94	2,484	100	416,047	79	0	100	55,609	97	635	100
ting sta	Carbon emission	kg CO ₂	233,345	0	204,845	12	233,345	0	115,923	50	547	100	7,842	97	559	100	28,923	88	0	100	3,246	99	143	100
Retrofit	Potential for reusable/recycl able materials	%	100%	0%	27%	27	100%	100	100%	100	0%	0	20%	20	90%	90	0%	0	0%	0	100%	100	90%	90
	Potential savings on energy consumption	MJ	337500	-39,834,588	-39,834,588	100	-4,443,063	12	-6,219,638	16	-39,663,259	100	-15,535,300	40	-991,582	3	-3,273,980	9	-7,932,652	21	31,218	1	337,500	0
stage	Potential savings on carbon emission	kg CO ₂	75938	-9,446,051	-9,446,051	100	-1,383,976	15	-1,366,211	15	-8,924,234	95	-3,516,540	38	-223,106	3	-801,334	9	-1,784,847	20	2,012	1	75,938	0
Operation stage	Poential savings on water consumption	kL	0	-68,050	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	-61,688	91	-68,050	100
	Waste generation	kg	701,731	0	701,731	0	0	100	0	100	1,350	100	534	100	134	100	1,125	100	0	100	0	100	15	100
	Potential for reusable/ recyclable materials	%	100%	0%	27%	27	100%	100	100%	100	0%	0	20%	20	90%	90	0%	0	0%	0	100%	100	90%	90

Table E5-1. Normalising the estimation of environmental performance

Note:

- The negative symbol "-" represents energy saving, carbon emissions, or water saving.
- Est. = Estimation; Norm. = Normalised score
- For impacts measured in positive direction, including use of reusable/recyclable materials in the retrofitting stage and operation stage, Equation 6.5 (in Chapter 6, $(N_{ij} = \frac{x_{ij} x_{minj}}{x_{maxj} x_{minj}})$ is used to normalise the estimation. Two decimals are saved in normalisation results, and to have a clear and concise expression, the normalisation results are multiplied by 100 to make them integer. It is done for other normalisations.
- For impacts measured in negative direction, including energy consumption and carbon emissions in the retrofitting stage, difference in energy consumption, difference in carbon emissions, difference in water consumption, and waste generation in the operation stage, Equation 6.6 (in Chapter 6, $N_{ij} = \frac{x_{max} x_{ij}}{x_{maxj} x_{minj}}$) is used to normalise the estimation.

	Detrofitting potivities	EA	AC
	Ketrontung activities	Estimation (¥)	Normalised score
A1	Install/upgrade insulation for external walls and roofs	-75,642	84
A2	Change single-glazing windows to low-e double glazing with aluminium frames	43,954	52
A3	Install aluminium sun shading slats for windows facing south	-849	64
A4	Install motion sensors to lighting system	-132,728	100
A5	Change T8 fluorescent bulbs to T8 LED	-48,148	77
A6	Change existing lifts to energy efficient lifts	232,408	0
A7	Install PV panels	-9,908	66
A8	Install energy consumption monitoring system	3,708	63
A9	Change existing taps and toilets to water-saving ones	5,169	62
A10	Install water treatment system to recycle greywater and stormwater	3,400	63

Note:

- The negative symbol "-" represents cost saving.
- The maximum estimation is ¥232,408 by retrofitting activity A6, and the minimum estimation is -¥ 132,728 by retrofitting activity A4.
- Since the measure direction is negative, the equation $N_{ij} = \frac{x_{ij} x_{minj}}{x_{maxj} x_{minj}} \times 100$ is used to normalise equivalent annual cost or saving by the proposed retrofitting activities.

	Social assessment criteria		NC	A	.1	Α	2	Α	3	A	4	Α	5	Α	6	Α	7	A	8	Α	9	Α	10
8	ocial assessment criteria	Max	Min	Est.	Norm.																		
	Noise impact on neighbourhood	3.80	1.00	3.60	7	3.80	0	3.60	7	1.00	100	1.00	100	1.00	100	3.00	29	2.00	64	1.00	100	2.40	50
	Emission impact on neighbourhood	3.80	1.00	3.80	0	2.60	43	3.40	14	1.00	100	1.00	100	1.00	100	2.80	36	1.60	79	1.00	100	2.60	43
g stag	Impacts on safety of undertaking the	3.80	1.60	3.60	9	3.60	9	3.80	0	1.60	100	2.60	55	2.20	73	3.20	27	2.00	82	1.60	100	1.80	91
ofittin	Impacts on cultural heritage	4.00	1.40	3.80	8	3.80	8	4.00	0	1.40	100	1.60	92	2.00	77	3.80	8	2.40	62	2.60	54	3.60	15
Retr	Impacts on surrounding traffic and pedestrians	4.20	1.00	4.20	0	3.40	25	3.40	25	1.00	100	1.00	100	1.40	88	2.00	69	2.20	63	1.20	94	2.40	56
	Impacts on relocating tenants	3.60	1.20	3.40	8	3.60	0	2.80	33	2.20	58	3.40	8	3.20	17	1.60	83	1.80	75	3.60	0	1.20	100
	Accessibility to building facilities for people with additional needs	4.40	2.40	3.40	50	2.40	0	3.20	40	3.20	40	3.00	30	4.40	100	3.20	40	2.60	10	4.20	90	2.40	0
	Accessibility to building services	4.40	2.20	4.40	100	3.60	64	4.40	100	2.20	0	3.60	64	4.00	82	3.20	45	2.80	27	4.40	100	3.40	55
Operation stage	Impacts on maintenance and maintainability from newly added building component or building system	3.60	2.00	3.40	88	2.00	0	2.60	38	2.00	0	3.60	100	3.60	100	2.20	13	3.40	88	3.60	100	2.40	25
	Impacts on safety and security of tenants	4.20	1.80	3.80	83	3.80	83	1.80	0	2.00	8	2.80	42	4.20	100	2.80	42	2.60	33	3.20	58	2.60	33
	Impacts on room flexibility for different demands	2.80	2.00	2.60	75	2.20	25	2.80	100	2.00	0	2.60	75	2.20	25	2.00	0	2.60	75	2.40	50	2.20	25

Table E5-3. Normalising the estimation of social performance

Note:

- Est. = Estimation; Norm. = Normalised score ٠
- ٠
- For assessment criteria of the retrofitting stage, the measure direction is negative, the equation $N_{ij} = \frac{x_{ij} x_{minj}}{x_{maxj} x_{minj}} \times 100$ is used to normalise the estimations. For assessment criteria of the operation stage, the measure direction is positive, the equation $N_{ij} = \frac{x_{maxj} x_{minj}}{x_{maxj} x_{minj}} \times 100$ is used to normalise the estimations. ٠

	Life	Assessment criteria	Weight ⁽¹⁾			1		1		1			Retrofitti	ng activity						1			
Pillar	s stage			A1		A2		A3		A4		A5		A6		A7		A8		A9		A10	
				Norm. ⁽²⁾	Weig. ⁽³⁾																		
Environmental dimension	<u>5</u> 0	Energy consumption	0.250	0	0.00	5	1.18	81	20.37	100	24.97	94	23.38	100	24.97	79	19.76	100	25.00	97	24.30	100	24.99
	ittin Se	Carbon emissions	0.245	12	2.99	0	0.00	50	12.33	100	24.44	97	23.68	100	24.44	88	21.46	100	24.50	99	24.16	100	24.50
	Retrof	Potential for reusable/recyclable materials	0.250	27	6.75	100	25.00	100	25.00	0	0.00	20	5.00	90	22.50	0	0.00	0	0.00	100	25.00	90	22.50
		Potential savings on energy consumption	0.219	100	21.90	12	2.61	16	3.57	100	21.81	40	8.65	3	0.72	9	1.97	21	4.51	1	0.17	0	0.00
	age	Potential savings on carbon emissions	0.155	100	15.50	15	2.33	15	2.33	95	14.73	38	5.89	3	0.49	9	1.40	20	3.10	1	0.12	0	0.00
	tion s	Potential savings on water consumption	0.128	0	0.00	0	0.00	0	0.00	0	0.00	0	0.00	0	0.00	0	0.00	0	0.00	91	11.65	100	12.80
	pera	Waste generation	0.154	0	0.00	100	15.40	100	15.40	100	15.37	100	15.39	100	15.40	100	15.38	100	15.40	100	15.40	100	15.40
	ō	Potential for reusable/recyclable materials	0.155	27	4.19	100	15.50	100	15.50	0	0.00	20	3.10	90	13.95	0	0.00	0	0.00	100	15.50	90	13.95
	Total	weighted environmen	tal score ⁽⁴⁾	51	.33	62.	.01	94	.50	101	1.31	85	.09	102	.47	59.	.96	72.	.51	116	5.29	114	.14
Social dimension		Noise impact on	0.145	7	1.04	0	0.00	7	1.04	100	14.50	100	14.50	100	14.50	29	4.14	64	9.32	100	14.50	50	7.25
		Emission impact on	0.180	0	0.00	43	7.71	14	2.57	100	18.00	100	18.00	100	18.00	36	6.43	79	14.14	100	18.00	43	7.71
	litting stage	Safety of retrofitting	0.227	9	2.06	9	2.06	0	0.00	100	22.70	55	12.38	73	16.51	27	6.19	82	18.57	100	22.70	91	20.64
		Impacts on cultural	0.148	8	1.14	8	1.14	0	0.00	100	14.80	92	13.66	77	11.38	8	1.14	62	9.11	54	7.97	15	2.28
	Retro	Impacts on surrounding traffic and pedestrians	0.163	0	0.00	25	4.08	25	4.08	100	16.30	100	16.30	88	14.26	69	11.21	63	10.19	94	15.28	56	9.17
		Impacts on relocating tenants	0.137	8	1.14	0	0.00	33	4.57	58	7.99	8	1.14	17	2.28	83	11.42	75	10.28	0	0.00	100	13.70
	Operation stage	Accessibility to building facilities for people with additional needs	0.259	50	12.95	0	0.00	40	10.36	40	10.36	30	7.77	100	25.90	40	10.36	10	2.59	90	23.31	0	0.00
		Accessibility to building services	0.167	100	16.70	64	10.63	100	16.70	0	0.00	64	10.63	82	13.66	45	7.59	27	4.55	100	16.70	55	9.11
		Impacts on maintenance and maintainability from newly added building component or building system	0.163	88	14.26	0	0.00	38	6.11	0	0.00	100	16.30	100	16.30	13	2.04	88	14.26	100	16.30	25	4.08
		Impacts on safety and security of	0.222	83	18.50	83	18.50	0	0.00	8	1.85	42	9.25	100	22.20	42	9.25	33	7.40	58	12.95	33	7.40
		Impacts on room flexibility for different demands	0.189	75	14.18	25	4.73	100	18.90	0	0.00	75	14.18	25	4.73	0	0.00	75	14.18	50	9.45	25	4.73
		Total weighted social score ⁽⁴⁾			81.97		48.84		.32	106.50		134	.11	159.73		69.76		114	.59	157.16		86.	.06

Table E5-4. Weighted environmental and social scores

Note:

• Norm. = Normalised score; Weig. = Weighted score

Calculation:

• Weighted score regarding one assessment criterion: $(3)=(1)\times(2)$

Where,

- is generated in focus group meeting, can be found in Table 7.25 (in Chapter 7)
- can be found in Tables A4-1 and A4-3
- Total weighted score of one sustainability pillar: $(4)=\sum(3)$ of one retrofitting activity

References

ABCB, Australia Building Codes Board. 2019, National Construction Code Series, Australia.

Aboelata, A. 2021, 'Assessment of green roof benefits on buildings' energy-saving by cooling outdoor spaces in different urban densities in arid cities', *Energy*, vol. 219, p. 119514.

Abu-Rayash, A. & Dincer, I. 2019, 'Sustainability assessment of energy systems: A novel integrated model', *Journal of Cleaner Production*, vol. 212, pp. 1098–116.

Aghimien, D.O., Aigbavboa, C.O. & Thwala, W.D. 2019, 'Microscoping the challenges of sustainable construction in developing countries', *Journal of Engineering, Design and Technology*, vol. 17, no. 6, pp. 1110–28.

AIQS, Austalian Institute of Quantity Surveyors. 2002, *Australian Cost Management Manual* - *Life cycle costing*, vol. 3, Canberra, p. 52.

Akhimien, A., Adamolekun, A.M. & Isiwele, A.A. 2017, 'Adaptability and Sustainability of Office Buildings', *Prieiga per internetq*.

Al-Kodmany, K. 2015, 'Tall buildings and elevators: A review of recent technological advances', *Buildings*, vol. 5, no. 3, pp. 1070–104.

Alamu, S.O., Wemida, A., Tsegaye, T. & Oguntimein, G. 2021, 'Sustainability assessment of municipal solid waste in Baltimore USA', *Sustainability*, vol. 13, no. 4, p. 1915.

Aldiabat, K.M. & Navenec, L. 2018, 'Data saturation: The mysterious step in grounded theory method', *The Qualitative Report*, vol. 23, no. 1, pp. 245–61.

Ali, S.B.M., Hasanuzzaman, M., Rahim, N., Mamun, M. & Obaidellah, U.H. 2021, 'Analysis of energy consumption and potential energy savings of an institutional building in Malaysia', *Alexandria Engineering Journal*, vol. 60, no. 1, pp. 805–20.

Alinezhad, A. & Khalili, J. 2019, New methods and applications in multiple attribute decision making (MADM), vol. 277, Springer.

Álvarez, A., Suárez, S.F., Joaristi, N., Lee, V., Tam, M. & Woodcock, E. 2022, *How can corporate functions become more agile?*, McKinsey & Company.

Amaratunga, D., Baldry, D., Sarshar, M. & Newton, R. 2002, 'Quantitative and qualitative research in the built environment: application of "mixed" research approach', *Work Study*, vol. 51, no. 1, pp. 17–31.

Amstalden, R.W., Kost, M., Nathani, C. & Imboden, D.M. 2007, 'Economic potential of energy-efficient retrofitting in the Swiss residential building sector: The effects of policy instruments and energy price expectations', *Energy Policy*, vol. 35, no. 3, pp. 1819–29.

Anand, C.K. & Amor, B. 2017, 'Recent developments, future challenges and new research directions in LCA of buildings: A critical review', *Renewable and Sustainable Energy Reviews*, vol. 67, pp. 408–16.

Andersen, R., Jensen, L.B. & Ryberg, M. 2021, 'Using digitized public accessible building data to assess the renovation potential of existing building stock in a sustainable urban perspective', *Sustainable Cities and Society*, vol. 75, p. 103303.

Anderson, K., Ryan, B., Sonntag, W., Kavvada, A. & Friedl, L. 2017, 'Earth observation in service of the 2030 Agenda for Sustainable Development', *Geo-spatial Information Science*, vol. 20, no. 2, pp. 77–96.

Andrew, R. & Peters, G. 2021, 'The Global Carbon Project's fossil CO2 emissions dataset', 2019-2020, figshare, <<u>https://doi.org/10.6084/m9.figshare.16729084.v1</u>>.

Ardente, F., Beccali, M., Cellura, M. & Mistretta, M. 2011, 'Energy and environmental benefits in public buildings as a result of retrofit actions', *Renewable and Sustainable Energy Reviews*, vol. 15, no. 1, pp. 460–70.

Arjunan, K.C. & Kannapiran, K. 2017, 'Cost-benefit analysis and the controversial reinvestment assumption in IRR and NPV estimates: Some new evidence against reinvestment assumption', *Economic Papers: A journal of applied economics and policy*, vol. 36, no. 3, pp. 351–63.

Asadi, E., da Silva, M.G., Antunes, C.H., Dias, L. & Glicksman, L. 2014, 'Multi-objective optimization for building retrofit: A model using genetic algorithm and artificial neural network and an application', *Energy and Buildings*, vol. 81, pp. 444–56.

Ascione, F., Bianco, N., De Masi, R.F., Mauro, G.M. & Vanoli, G.P. 2017, 'Resilience of robust cost-optimal energy retrofit of buildings to global warming: A multi-stage, multi-objective approach', *Energy and Buildings*, vol. 153, pp. 150–67.

ASHRAE, American Society of Heating, Refrigerating and Air-Conditioning Engineers 2009, *ASHRAE Handbook Fundamentals*.

Asmone, A. & Chew, M. 2018, 'Merging building maintainability and sustainability assessment: A multicriteria decision making approach', *IOP Conference Series: Earth and Environmental Science*, vol. 117, IOP Publishing, p. 012029.

Atanda, J.O. 2019, 'Developing a social sustainability assessment framework', *Sustainable Cities and Society*, vol. 44, pp. 237–52.

Atkinson, B. 1988, 'Urban ideals and the mechanism of renewal', *RAIA Conference, June, Sydney*.

Australian Government, Department of Climate Change, Energy, the Environmentl and Water 2010, *Case Study:Commercial Building Energy Efficiency Retrofit – 4 Mort Street, Canberra*, Department of Agriculture, Water and the Environment, Australia.

Australian Gorvernment, Department of Climate Change, Energy, the Environmentl and Water 2022, *HVAC*, Department of Climate Change, Energy, the Environmentl and Water, Australia

Ayres, R., Van den Berrgh, J. & Gowdy, J. 2001, 'Strong versus weak sustainability: Economics, natural sciences, and consilience', *Environmental Ethics*, vol. 23, no. 2, pp. 155–68.

Balachandran, V., Mahanta, N.R. & Samuel, A.K. 2020, 'Strategic framework in retrofitting existing buildings using sustainable design techniques in the UAE', *2020 Advances in Science and Engineering Technology International Conferences (ASET)*, IEEE, pp. 1–6.

Ball, R. 1999, 'Developers, regeneration and sustainability issues in the reuse of vacant industrial buildings', *Building Research & Information*, vol. 27, no. 3, pp. 140-8.

Barmparesos, N., Assimakopoulos, M.N., Assimakopoulos, V.D., Loumos, N., Sotiriou, M.A. and Koukoumtzis, A., 2018. Indoor air quality and thermal conditions in a primary school with a green roof system, *Atmosphere*, vol. 9, no. 2, pp.75.

Bateman, N. 2017, Operating model: an exploration of the concept. Loughborough University. Report. https://hdl.handle.net/2134/26654>

Baumhof, R., Decker, T., Röder, H. & Menrad, K. 2017, 'Which factors determine the extent of house owners' energy-related refurbishment projects? A Motivation-Opportunity-Ability Approach', *Sustainable Cities and Society*, vol. 36, pp. 33-41.

Beccali, M., Bonomolo, M., Brano, V.L., Ciulla, G., Di Dio, V., Massaro, F. & Favuzza, S. 2019, 'Energy saving and user satisfaction for a new advanced public lighting system', *Energy Conversion and Management*, vol. 195, pp. 943–57.

BEEX, 2017, Lighting the way – Upgrading lighting systems for commercial offices, Building Energy Exchange.

Belussi, L., Barozzi, B., Bellazzi, A., Danza, L., Devitofrancesco, A., Fanciulli, C., Ghellere, M., Guazzi, G., Meroni, I. & Salamone, F. 2019, 'A review of performance of zero energy buildings and energy efficiency solutions', *Journal of Building Engineering*, vol. 25, p. 100772.

Benayoun, R., De Montgolfier, J., Tergny, J. & Laritchev, O. 1971, 'Linear programming with multiple objective functions: Step method (STEM)', *Mathematical Programming*, vol. 1, no. 1, pp. 366–75.

Benoît, C., Norris, G.A., Valdivia, S., Ciroth, A., Moberg, A., Bos, U., Prakash, S., Ugaya, C. & Beck, T. 2010, 'The guidelines for social life cycle assessment of products: just in time!', *The International Journal of Life Cycle Assessment*, vol. 15, no. 2, pp. 156–63.

Berardi, U. 2018, 'Aerogel-enhanced systems for building energy retrofits: Insights from a case study', *Energy and Buildings*, vol. 159, pp. 370–81.

Berawi, M.A., Miraj, P., Windrayani, R. & Berawi, A.R.B. 2019, 'Stakeholders' perspectives on green building rating: A case study in Indonesia', *Heliyon*, vol. 5, no. 3, p. e01328.

Bertone, E., Sahin, O., Stewart, R.A., Zou, P., Alam, M. & Blair, E. 2016, 'State-of-the-art review revealing a roadmap for public building water and energy efficiency retrofit projects', *International Journal of Sustainable Built Environment*, vol. 5, no. 2, pp. 526–48.

Bertone, E., Sahin, O., Stewart, R.A., Zou, P.X., Alam, M., Hampson, K. & Blair, E. 2018, 'Role of financial mechanisms for accelerating the rate of water and energy efficiency retrofits in Australian public buildings: Hybrid Bayesian Network and System Dynamics modelling approach', *Applied Energy*, vol. 210, pp. 409–19.

Bertone, E., Stewart, R.A., Sahin, O., Alam, M., Zou, P.X., Buntine, C. & Marshall, C. 2018, 'Guidelines, barriers and strategies for energy and water retrofits of public buildings', *Journal of Cleaner Production*, vol. 174, pp. 1064–78.

Besir, A.B. & Cuce, E. 2018, 'Green roofs and facades: A comprehensive review', *Renewable and Sustainable Energy Reviews*, vol. 82, pp. 915–39.

Bhattacherjee, A. 2012, Social science research: Principles, methods, and practices, Anol Bhattacheree

Bianchini, F. & Hewage, K. 2012, 'How "green" are the green roofs? Lifecycle analysis of green roof materials', *Building and Environment*, vol. 48, pp. 57–65.

Biely, K., Maes, D. & Van Passel, S. 2018, 'The idea of weak sustainability is illegitimate', *Environment, Development and Sustainability*, vol. 20, no. 1, pp. 223–32.

Birgisdottir, H. & Rasmussen, F.N. 2016, Introduction to LCA of Buildings, Trafik-og Byggestyrelsen.

Bleyl, J.W., Bareit, M., Casas, M.A., Chatterjee, S., Coolen, J., Hulshoff, A., Lohse, R., Mitchell, S., Robertson, M. & Ürge-Vorsatz, D. 2019, 'Office building deep energy retrofit: Life cycle cost benefit analyses using cash flow analysis and multiple benefits on project level', *Energy Efficiency*, vol. 12, no. 1, pp. 261–79.

Bogach, S. & Wang, X. 2013, 'Applying abatement cost curve methodology for low-carbon strategy in Changning district, Shanghai', *World Bank Other Oper. Stud.* 84068v1, pp. 1–86.

Braulio-Gonzalo, M. & Bovea, M.D. 2017, 'Environmental and cost performance of building's envelope insulation materials to reduce energy demand: Thickness optimisation', *Energy and Buildings*, vol. 150, pp. 527–45.

Brewer, J. & Hunter, A. 2006, Foundations of multimethod research: Synthesizing styles, Sage.

Brown, G.M. & Teernstra, T. 2008, 'Examining investor perceptions of obsolescence and value through a behavioural economics lens', *Journal of European Real Estate Research*, vol. 1, no. 3, pp. 267–90.

Brownley, C.W. 2013, *Multi-objective decision analysis: Managing trade-offs and uncertainty*, Business Expert Press.

Bruce, T., Zuo, J., Rameezdeen, R. & Pullen, S. 2015, 'Factors influencing the retrofitting of existing office buildings using Adelaide, South Australia as a case study', *Structural Survey*, vol. 33, no. 2, pp. 150–66.

Brundtland, G.H. 1987, Our Common Future: Report of the World Commission on Environment and Development, Oxford University Press, Oxford, UK.

BSI 2014, 'EN 16309:2014+A1:2014 Sustainability of construction works – Assessment of social performance of buildings – Calculation methodology', *EC 16309:2014+A1*.

Bulut, M.B., Wilkinson, S., Khan, A., Jin, X.H. and Lee, C.L., 2022, 'Thermal performance of retrofitted secondary glazed windows in residential buildings–two cases from Australia', *Smart and Sustainable Built Environment*, vol. 11, no. 4, pp.1182-92.

Burrell, G. & Morgan, G. 1979, Sociological paradigms and organisational analysis: Elements of the sociology of corporate life, Ashgate, Great Britain.

Campbell, A. & Gutierrez, M. 2021, 'Why you need an operating model: To align your people and deliver your strategy', *Management and Business Review*, vol. 1, no. 2.

Cang, Y., Luo, Z., Yang, L. & Han, B. 2020, 'A new method for calculating the embodied carbon emissions from buildings in schematic design: Taking "building element" as basic unit', *Building and Environment*, vol. 185, p. 107306.

Cao, C. 2022, 'Comparative analysis of German DGNB and Chinese Green Building Evaluation Standards', *Journal of Physics: Conference Series*, vol. 2202, IOP Publishing, p. 012033.

Carrillo, C., Díaz-Dorado, E., Cidrás, J. & Silva-Ucha, M. 2013, 'A methodology for energy analysis of escalators', *Energy and Buildings*, vol. 61, pp. 21–30.

Carvalho, J.P., Bragança, L. & Mateus, R. 2021, 'Sustainable building design: Analysing the feasibility of BIM platforms to support practical building sustainability assessment', *Computers in Industry*, vol. 127, pp. 103400.

CEN 2011, Sustainability of construction works – Assessment of environmental performance of buildings – Calculation method, European Committee for Standardization.

Cetiner, I. & Edis, E. 2014, 'An environmental and economic sustainability assessment method for the retrofitting of residential buildings', *Energy and Buildings*, vol. 74, pp. 132–40.

Chahidi, L.O., Fossa, M., Priarone, A. & Mechaqrane, A. 2021, 'Energy saving strategies in sustainable greenhouse cultivation in the mediterranean climate–A case study', *Applied Energy*, vol. 282, pp. 116156.

Charmaz, K. & Belgrave, L.L. 2007, 'Grounded theory', *The Blackwell Encyclopedia of Sociology*, pp. 2023–27, https://doi.org/10.1002/9781405165518.wbeosg070>.

Chau, C.-K., Tse, M. & Chung, K. 2010, 'A choice experiment to estimate the effect of green experience on preferences and willingness-to-pay for green building attributes', *Building and Environment*, vol. 45, no. 11, pp. 2553–61.

Chau, C., Leung, T. & Ng, W. 2015, 'A review on life cycle assessment, life cycle energy assessment and life cycle carbon emissions assessment on buildings', *Applied Energy*, vol. 143, pp. 395–413.

Chaves, F.J., Tzortzopoulos, P., Formoso, C.T. & Biotto, C.N. 2016, 'Building information modelling to cut disruption in housing retrofit', *Proceedings of the Institution of Civil Engineers-Engineering Sustainability*, vol. 170, Thomas Telford Ltd, pp. 322–33.

Che, W.W., Tso, C.Y., Sun, L., Ip, D.Y., Lee, H., Chao, C.Y. & Lau, A.K. 2019, 'Energy consumption, indoor thermal comfort and air quality in a commercial office with retrofitted heat, ventilation and air conditioning (HVAC) system', *Energy and Buildings*, vol. 201, pp. 202–15.

Chen, C.-f., De Simone, M., Yilmaz, S., Xu, X., Wang, Z., Hong, T. & Pan, Y. 2021, 'Intersecting heuristic adaptive strategies, building design and energy saving intentions when facing discomfort environment: A cross-country analysis', *Building and Environment*, vol. 204, p. 108129.

Chen, S., Zhang, G., Xia, X., Setunge, S. & Shi, L. 2020, 'A review of internal and external influencing factors on energy efficiency design of buildings', *Energy and Buildings*, vol. 216, pp. 109944.

Chen, Z., Hammad, A.W., Kamardeen, I. & Akbarnezhad, A. 2020, 'Optimising embodied energy and thermal performance of thermal insulation in building envelopes via an automated building information modelling (BIM) tool', *Buildings*, vol. 10, no. 12, pp. 218.

Cheng, Y. & Yao, X. 2021, 'Carbon intensity reduction assessment of renewable energy technology innovation in China: A panel data model with cross-section dependence and slope heterogeneity', *Renewable and Sustainable Energy Reviews*, vol. 135, p. 110157.

Cheong, K.H., Teo, Y.H., Koh, J.M., Acharya, U.R. & Yu, S.C.M. 2020, 'A simulation-aided approach in improving thermal-visual comfort and power efficiency in buildings', *Journal of Building Engineering*, vol. 27, pp. 100936.

Cherryholmes, C.H. 1992, 'Notes on pragmatism and scientific realism', *Educational Researcher*, vol. 21, no. 6, pp. 13–7.

Chew, M. & Das, S. 2008, 'Building grading systems: a review of the state-of-the-art', *Architectural Science Review*, vol. 51, no. 1, pp. 3–13.

Chidiac, S., Catania, E., Morofsky, E. & Foo, S. 2011, 'Effectiveness of single and multiple energy retrofit measures on the energy consumption of office buildings', *Energy*, vol. 36, no. 8, pp. 5037–52.

Chinneck, J. 2004, Practical optimization: A gentle introduction, Ottawa, Ontario, Canada.

Ciroth, A., Finkbeiner, M., Traverso, M., Hildenbrand, J., Kloepffer, W., Mazijn, B., Prakash, S., Sonnemann, G., Valdivia, S. & Ugaya, C.M.L. 2011, *Towards a life cycle sustainability assessment: making informed choices on products*, UNEP.

Cohen, N. & Arieli, T. 2011, 'Field research in conflict environments: Methodological challenges and snowball sampling', *Journal of Peace Research*, vol. 48, no. 4, pp. 423–35.

The Consulting Engineer Survivor. 2022, *How much does building maintenance cost?*, viewed 29 June 2022, <<u>https://www.consultengsurvivor.com/the-cost-of-building-maintenance#h.6azshjs12tso</u>>.

Conte, E. 2018, 'The era of sustainability: Promises, pitfalls and prospects for sustainable buildings and the built environment', *Sustainability*, vol. 10, no. 6, p. 2092.

Cook, S., Sharma, A.K. & Gurung, T.R. 2014, 'Evaluation of alternative water sources for commercial buildings: A case study in Brisbane, Australia', *Resources, Conservation and Recycling*, vol. 89, pp. 86–93.

Cordero, A.S., Melgar, S.G. & Márquez, J.M.A. 2019, 'Green building rating systems and the new framework level (s): A critical review of sustainability certification within Europe', *Energies*, vol. 13, no. 1, pp. 66.

Cotter, T.S. 2022, *Engineering managerial economic decision and risk analysis*, Springer, Norfolk, VA, USA.

Creswell, J. 2003, *Research design: qualitative, quantitative, and mixed methods approaches*, 2nd edn, Sage.

Cresswell, J.W. 2012, *Qualitative inquiry and research design: Choosing among five traditions*, 3rd edn, Sage Publications, California.

Creswell, J.W. & Creswell, J.D. 2018, *Research design: Qualitative, quantitative, and mixed methods approaches*, 5th edn, Sage Publications, London.

Creswell, J.W. & Poth, C.N. 2018, *Qualitative inquiry and research design: Choosing among five approaches*, 5th edn, Sage Publications.

Crotty, M. 1998, *The foundations of social research: Meaning and perspective in the research process*, Sage.

Cuce, E. 2018, 'Accurate and reliable U-value assessment of argon-filled double glazed windows: A numerical and experimental investigation', *Energy and Buildings*, vol. 171, pp. 100–6.

Cupido, A.F., Baetz, B.W., Pujari, A. & Chidiac, S. 2010, 'Evaluating institutional green building policies: A mixed-methods approach', *Journal of Green Building*, vol. 5, no. 1, pp. 115–31.

Curtis, B. & Curtis, C. 2011, Social research: A practical introduction, Sage.

D'Oca, S., Chen, C.-F., Hong, T. & Belafi, Z. 2017, 'Synthesizing building physics with social psychology: An interdisciplinary framework for context and occupant behavior in office buildings', *Energy Research & Social Science*, vol. 34, pp. 240–51.

da Cunha, S.R.L. & de Aguiar, J.L.B. 2020, 'Phase change materials and energy efficiency of buildings: A review of knowledge', *Journal of Energy Storage*, vol. 27, pp. 101083.

Dabbagh, M. & Krarti, M. 2021, 'Optimal control strategies for switchable transparent insulation systems applied to smart windows for US residential buildings', *Energies*, vol. 14, no. 10, pp. 2917.

De Almeida, A., Hirzel, S., Patrão, C., Fong, J. & Dütschke, E. 2012, 'Energy-efficient elevators and escalators in Europe: An analysis of energy efficiency potentials and policy measures', *Energy and Buildings*, vol. 47, pp. 151–8.

de Nooij, M. 2011, 'Social cost-benefit analysis of electricity interconnector investment: A critical appraisal', *Energy Policy*, vol. 39, no. 6, pp. 3096–105.

de Rocquigny, E., Devictor, N. & Tarantola, S. 2008, Uncertainty in industrial practice: A guide to quantitative uncertainty management, John Wiley & Sons.

de Vries, M., van der Merwe, A., Kotzé, P. & Gerber, A. 2011, 'A method for identifying process reuse opportunities to enhance the operating model', *2011 IEEE International Conference on Industrial Engineering and Engineering Management*, IEEE, pp. 1005–9.

Dendena, B. & Corsi, S. 2015, 'The environmental and social impact assessment: a further step towards an integrated assessment process', *Journal of Cleaner Production*, vol. 108, pp. 965–77.

Deng, W., Peng, Z. & Tang, Y.-t. 2019, 'A quick assessment method to evaluate sustainability of urban built environment: Case studies of four large-sized Chinese cities', *Cities*, vol. 89, pp. 57–69.

Denzin, N.K. 1978, *The research act: A theoretical orientation to sociological methods*, 2nd edn, McGraw-Hill, New York.

Desideri, U. & Asdrubali, F. 2018, *Handbook of energy efficiency in buildings: A life cycle approach*, Butterworth-Heinemann.

Deuble, M.P. & de Dear, R.J. 2012, 'Green occupants for green buildings: the missing link?', *Building and Environment*, vol. 56, pp. 21–7.

DEWHA, Department of the Environment, Water, Heritage and the Arts 2008, *Energy use in the Australian Residential Sector*, Canberra.

Dincer, I. & Zamfirescu, C. 2018, 'Environmental dimensions of energy', *Comprehensive Energy Systems*, vol. 1, pp. 49–100.

Ding, G. & Ying, X. 2019, 'Embodied and operating energy assessment of existing buildings– Demolish or rebuild', *Energy*, vol. 182, pp. 623–31.

Ding, G.K. 2005, 'Developing a multicriteria approach for the measurement of sustainable performance', *Building Research & Information*, vol. 33, no. 1, pp. 3–16.

Department of Climate Change, Energy, the Environment, and Water 2022, *HVAC*, Department of Industry, Science, Energy and Resources, viewed 20 June 2022, <<u>https://www.energy.gov.au/business/equipment-and-technology-guides/hvac</u>>.

Dodgson, J.S., Spackman, M., Pearman, A. & Phillips, L.D. 2009, *Multi-criteria analysis: A manual*, Department for Communities and Local Government, London.

Dong, J., Zuo, J. & Luo, J. 2020, 'Development of a management framework for applying green roof policy in urban China: A preliminary study', *Sustainability*, vol. 12, no. 24, p. 10364.

Dong, Y., Miraglia, S., Manzo, S., Georgiadis, S., Sørup, H.J.D., Boriani, E., Hald, T., Thöns, S. & Hauschild, M.Z. 2018, 'Environmental sustainable decision making–The need and obstacles for integration of LCA into decision analysis', *Environmental Science & Policy*, vol. 87, pp. 33–44.

Dong, Y.H. & Ng, S.T. 2016, 'A modeling framework to evaluate sustainability of building construction based on LCSA', *The International Journal of Life Cycle Assessment*, vol. 21, no. 4, pp. 555–68.

Dougherty, T. & Hall, A. 1995, *Environmental impact assessment of irrigation and drainage projects*, vol. 53, Food & Agriculture Organisation.

Douglas-Smith, D., Iwanaga, T., Croke, B.F. & Jakeman, A.J. 2020, 'Certain trends in uncertainty and sensitivity analysis: An overview of software tools and techniques', *Environmental Modelling & Software*, vol. 124, p. 104588.

Dowson, M., Poole, A., Harrison, D. & Susman, G. 2012, 'Domestic UK retrofit challenge: Barriers, incentives and current performance leading into the Green Deal', *Energy Policy*, vol. 50, pp. 294–305.

Drosou, V., Giama, E., Oxizidis, S., Papadopoulos, A. & Santamouris, M. 2018, *Guide to Low Carbon Commercial Buildings – Retrofit*, Low Carbon Living CRC, Sydney, Australia.

Dubois, M.-C. & Blomsterberg, Å. 2011, 'Energy saving potential and strategies for electric lighting in future North European, low energy office buildings: A literature review', *Energy and Buildings*, vol. 43, no. 10, pp. 2572–82.

Dudovskiy, J. 2016, *The ultimate guide to writing a dissertation in business studies: a step-by-step assistance*, Pittsburgh, USA.

Durdyev, S., Zavadskas, E.K., Thurnell, D., Banaitis, A. & Ihtiyar, A. 2018, 'Sustainable construction industry in Cambodia: Awareness, drivers and barriers', *Sustainability*, vol. 10, no. 2, pp. 392.

Dwaikat, L.N. & Ali, K.N. 2016, 'Green buildings cost premium: A review of empirical evidence', *Energy and Buildings*, vol. 110, pp. 396–403.

Dwaikat, L.N. & Ali, K.N. 2018, 'Green buildings life cycle cost analysis and life cycle budget development: Practical applications', *Journal of Building Engineering*, vol. 18, pp. 303–11.

EATS, Eastern Area Tenants Service 2015, Blinds and Curtains.

Eberhardt, L.C.M., Birgisdottir, H. & Birkved, M. 2019, 'Potential of circular economy in sustainable buildings', IOP Conference Series: Materials Science and Engineering, vol. 471, IOP Publishing, p. 092051.

Eichholtz, P., Kok, N. & Quigley, J.M. 2013, 'The economics of green building', *Review of Economics and Statistics*, vol. 95, no. 1, pp. 50–63.

Ekel, P., Pedrycz, W. & Pereira Jr, J. 2020, *Multicriteria decision-making under conditions of uncertainty: A fuzzy set perspective*, 1st edition, Wiley, NewJersey.

Eldomiaty, T., Saeed, Y., Hammam, R. and AboulSoud, S., 2020, 'The associations between stock prices, inflation rates, interest rates are still persistent: Empirical evidence from stock duration model', *Journal of Economics, Finance and Administrative Science*, vol. 25, no. 49, pp.149-61.

Elkington, J. 1997, *Cannibals with forks: the triple bottom line of 21st century business*, New Society Publishers.

EPA, United States Environmental Protection Agency. 2022, 'Emissions & Generation Resource Integrated Database (eGRID)', https://www.epa.gov/egrid.

Epstein, M.J., Buhovac, A.R. & Yuthas, K. 2015, 'Managing social, environmental and financial performance simultaneously', *Long Range Planning*, vol. 48, no. 1, pp. 35–45.

Escrig-Olmedo, E., Muñoz-Torres, M.J., Fernández-Izquierdo, M.Á. & Rivera-Lirio, J.M. 2017, 'Measuring corporate environmental performance: A methodology for sustainable development', *Business Strategy and the Environment*, vol. 26, no. 2, pp. 142–62.

Esteghamati, M.Z., Sharifnia, H., Ton, D., Asiatico, P., Reichard, G. & Flint, M.M. 2022, 'Sustainable early design exploration of mid-rise office buildings with different subsystems using comparative life cycle assessment', *Journal of Building Engineering*, pp. 104004.

Esteves, A.M., Franks, D. & Vanclay, F. 2012, 'Social impact assessment: the state of the art', *Impact Assessment and Project Appraisal*, vol. 30, no. 1, pp. 34–42.

Evangelisti, L., Guattari, C., Asdrubali, F. & de Lieto Vollaro, R. 2020, 'An experimental investigation of the thermal performance of a building solar shading device', *Journal of Building Engineering*, vol. 28, pp. 101089.

Fan, L., Pang, B., Zhang, Y., Zhang, X., Sun, Y. & Wang, Y. 2018, 'Evaluation for social and humanity demand on green residential districts in China based on SLCA', *The International Journal of Life Cycle Assessment*, vol. 23, no. 3, pp. 640–50.

Far, C. & Far, H. 2019, 'Improving energy efficiency of existing residential buildings using effective thermal retrofit of building envelope', *Indoor and Built Environment*, vol. 28, no. 6, pp. 744–60.

Fatourehchi, D. & Zarghami, E. 2020, 'Social sustainability assessment framework for managing sustainable construction in residential buildings', *Journal of Building Engineering*, vol. 32, p. 101761.

Fauzi, R.T., Lavoie, P., Sorelli, L., Heidari, M.D. & Amor, B. 2019, 'Exploring the current challenges and opportunities of life cycle sustainability assessment', *Sustainability*, vol. 11, no. 3, p. 636.

Favi, C., Di Giuseppe, E., D'Orazio, M., Rossi, M. & Germani, M. 2018, 'Building retrofit measures and design: A probabilistic approach for LCA', *Sustainability*, vol. 10, no. 10, p. 3655.

Fellows, R.F. & Liu, A.M. 2015, Research methods for construction, John Wiley & Sons.

Filippi, M. & Sirombo, E. 2015, 'Green rating of existing school facilities', *Energy Procedia*, vol. 78, pp. 3156–61.

Finkbeiner, M., Schau, E.M., Lehmann, A. & Traverso, M. 2010, 'Towards life cycle sustainability assessment', *Sustainability*, vol. 2, no. 10, pp. 3309–22.

Finlay, D. 2014, 'The creative behind the design: A phenomenological investigation into the lived experience of leading creative practitioners', Master dissertation thesis, University of New South Wales.

Fisher, I. 1907, The rate of interest - Its nature, determination and relation to economic phenomena, Macmillan Company, New York.

Fleurbaey, M. & Abi-Rafeh, R. 2020, 'The use of distributional weights in benefit-cost analysis: Insights from welfare economics', *Review of Environmental Economics and Policy*.

Fouche, M. & Crawford, R.H. 2017, 'Towards an integrated approach for evaluating both the life cycle environmental and financial performance of a building: A review', *Procedia Engineering*, vol. 180, pp. 118–27.

Francis, A. & Thomas, A. 2022, 'A framework for dynamic life cycle sustainability assessment and policy analysis of built environment through a system dynamics approach', *Sustainable Cities and Society*, vol. 76, pp. 103521.

Franz, M. & Wenzl, F.P. 2017, 'Critical review on life cycle inventories and environmental assessments of LED-lamps', *Critical Reviews in Environmental Science and Technology*, vol. 47, no. 21, pp. 2017–78.

Frey, P., Dunn, L., Cochran, R., Spataro, K., McLennan, J.F., DiNola, R. & Heider, B. 2012, 'The greenest building: quantifying the environmental value of building reuse', The Preservation Green Lab of the National Trust for Historic Preservation.

Friedler, E. & Alfiya, Y. 2010, 'Physicochemical treatment of office and public buildings greywater', *Water Science and Technology*, vol. 62, no. 10, pp. 2357–63.

Friedlingstein, P., Jones, M.W., O'Sullivan, M., Andrew, R.M., Bakker, D., C. E., Hauck, J., Le Quéré, C., Peters, G.P., Peters, W., Pongratz, J., Sitch, S., Canadell, J.G., Ciais, P., Jackson, R.B., Alin, S.R., Anthoni, P., Bates, N.R., Becker, M., Bellouin, N., Bopp, L., Chau, T.T.T., Chevallier, F.d.r., Chini, L.P., Cronin, M., Currie, K.I., Decharme, B., Djeutchouang, L.M., Dou, X., Evans, W., Feely, R.A., Feng, L., Gasser, T., Gilfillan, D., Gkritzalis, T., Grassi, G., Gregor, L., Gruber, N., Gürses, O.z.r., Harris, I., Houghton, R.A., Hurtt, G.C., Iida, Y., Ilyina, T., Luijkx, I.T., Jain, A., Jones, S.D., Kato, E., Kennedy, D., Goldewijk, K.K., Knauer, J.r., Korsbakken, J.I., Körtzinger, A., Landschützer, P., Lauvset, S.K., Lefèvre, N., Lienert, S., Liu, J., Marland, G., McGuire, P.C., Melton, J.R., Munro, D.R., Nabel, J.E.M.S., Nakaoka, S.-I., Niwa, Y., Ono, T., Pierrot, D., Poulter, B., Rehder, G., Resplandy, L., Robertson, E., Rödenbeck, C., Rosan, T.M., Schwinger, J.r., Schwingshackl, C., Séférian, R., Sutton, A.J., Sweeney, C., Tanhua, T., Tans, P.P., Tian, H., Tilbrook, B., Tubiello, F., Werf, G.v.d., Vuichard, N., Wada, C., Wanninkhof, R., Watson, A.J., Willis, D., Wiltshire, A.J., Yuan, W., Yue, C., Yue, X., Zaehle, S.n. & Zeng, J. 2021, 'The Global Carbon Budget 2021', 2019-2020, Earth System Science Data.

Fulton, L., Beauvais, B., Brooks, M., Kruse, C.S. & Lee, K. 2020, 'A publicly available cost simulation of sustainable construction options for residential houses', *Sustainability*, vol. 12, no. 7, pp. 2873.

Gajjar, D. 2013, 'Ethical consideration in research', Education, vol. 2, no. 7.

GBCA, Green Building Council of Australia 2020, *Introducing Green Star - Together, we can build a sustainable future*, pamphlet.

Ge, J., Weng, J., Zhao, K., Gui, X., Li, P. & Lin, M. 2018, 'The development of green building in China and an analysis of the corresponding incremental cost: A case study of Zhejiang Province', *Lowland Technology International*, vol. 20, no. 3, Dec, pp. 321-30.

Geddes, A., Parker, C. & Scott, S. 2018, 'When the snowball fails to roll and the use of 'horizontal' networking in qualitative social research', *International Journal of Social Research Methodology*, vol. 21, no. 3, pp. 347–58.

Geneletti, D. 2019, Multicriteria analysis for environmental decision-making, Anthem Press.

Gerring, J. 2011, Social science methodology: A unified framework, Cambridge University Press.

Ghisellini, P., Ji, X., Liu, G. & Ulgiati, S. 2018, 'Evaluating the transition towards cleaner production in the construction and demolition sector of China: A review', *Journal of Cleaner Production*, vol. 195, pp. 418–34.

Giebeler, G., Krause, H., Fisch, R., Musso, F., Lenz, B. & Rudolphi, A. 2009, *Refurbishment manual: Maintenance, conversions, extensions*, Walter de Gruyter.

Gimenez, C., Sierra, V. & Rodon, J. 2012, 'Sustainable operations: Their impact on the triple bottom line', *International Journal of Production Economics*, vol. 140, no. 1, pp. 149–59.

Gimeno-Frontera, B., Mainar-Toledo, M.D., de Guinoa, A.S., Zambrana-Vasquez, D. & Zabalza-Bribián, I. 2018, 'Sustainability of non-residential buildings and relevance of main environmental impact contributors' variability. A case study of food retail stores buildings', *Renewable and Sustainable Energy Reviews*, vol. 94, pp. 669–81.

Goh, C.S., Chong, H.-Y., Jack, L. & Faris, A.F.M. 2020, 'Revisiting triple bottom line within the context of sustainable construction: A systematic review', *Journal of Cleaner Production*, vol. 252, pp. 119884.

Gou, Z. & Xie, X. 2017, 'Evolving green building: triple bottom line or regenerative design?', *Journal of Cleaner Production*, vol. 153, pp. 600–7.

Goulouti, K., Padey, P., Galimshina, A., Habert, G. & Lasvaux, S. 2020, 'Uncertainty of building elements' service lives in building LCA & LCC: What matters?', *Building and Environment*, vol. 183, pp. 106904.

Görener, A., 2012, 'Comparing AHP and ANP: an application of strategic decisions making in a manufacturing company', *International Journal of Business and Social Science*, vol. 3, no. 11, pp. 194–208.

Gray, D.E. 2014, Doing research in the real world, 3rd edn, Sage.

Guinee, J.B., Heijungs, R., Huppes, G., Zamagni, A., Masoni, P., Buonamici, R., Ekvall, T. & Rydberg, T. 2011, *Life cycle assessment: past, present, and future*, ACS Publications.

Gunduz, M. & Almuajebh, M. 2020, 'Critical success factors for sustainable construction project management', *Sustainability*, vol. 12, no. 5, pp. 1990.

Guo, C., Bian, C., Liu, Q., You, Y., Li, S. & Wang, L. 2022, 'A new method of evaluating energy efficiency of public buildings in China', *Journal of Building Engineering*, vol. 46, pp. 103776.

Gustafsson, J. 2017, 'Single case studies vs. multiple case studies: A comparative study', *Engineering and Science*, vol. 12, no.1.

Hammond, G. & Jones, C. 2011, *Inventory of Carbon and Energy: ICE*, Bath: Sustainable Energy Research Team, Department of Mechanical Engineering, University of Bath.

Han, H.J., Mehmood, M.U., Ahmed, R., Kim, Y., Dutton, S., Lim, S.H. & Chun, W. 2019, 'An advanced lighting system combining solar and an artificial light source for constant illumination and energy saving in buildings', *Energy and Buildings*, vol. 203, pp. 109404.

Hartwig, S. 2020, 'Sustainability and future generations: How can their interests be preserved?', *Jean Monnet Papers*.

Hauashdh, A., Jailani, J. & Rahman, I.A. 2022, 'Strategic approaches towards achieving sustainable and effective building maintenance practices in maintenance-managed buildings: A combination of expert interviews and a literature review', *Journal of Building Engineering*, vol. 45, pp. 103490.

Hauck, M., Giezen, C., Calvert, S., Keijzer, E., de Vos-Effting, S., Van Vliet, D. & Voskuilen, J. 2016, 'Economic, environmental and social cost-benefit analysis of road construction works', *Life-Cycle of Engineering Systems*, CRC Press, pp. 901–6.

Haule, E.M., Chaiwiwatworakul, P. & Chirarattananon, S. 2020, 'The performance of external vertical slats for shading west windows in office buildings', *Naresuan University Journal: Science and Technology (NUJST)*, vol. 28, no. 3, pp. 69–78.

Hauschild, M.Z., Rosenbaum, R.K. & Olsen, S.I. 2018, *Life Cycle Assessment: Theory and Practice*, Spinger.

Hawken, P., Lovins, A.B. & Lovins, L.H. 2013, Natural capitalism: The next industrial revolution, Routledge.

He, J., Yue, Q., Li, Y., Zhao, F. & Wang, H. 2020, 'Driving force analysis of carbon emissions in China's building industry: 2000–2015', *Sustainable Cities and Society*, vol. 60, pp. 102268.

He, Y., Kvan, T., Liu, M. & Li, B. 2018, 'How green building rating systems affect designing green', *Building and Environment*, vol. 133, pp. 19–31.

Heale, R. & Twycross, A. 2018, 'What is a case study?', *Evidence-based nursing*, vol. 21, no 1, pp. 7–8.

Hellweg, S., Hofstetter, T.B. & Hungerbuhler, K. 2003, 'Discounting and the environment should current impacts be weighted differently than impacts harming future generations?', *The International Journal of Life Cycle Assessment*, vol. 8, no. 1, pp. 8–18.

Hennink, M.M. 2014, Focus Group Discussions, Oxford University Press, New York.

Hennink, M., Hutter, I. & Bailey, A. 2010, Qualitative research methods, Sage.

Heo, Y., Choudhary, R. & Augenbroe, G. 2012, 'Calibration of building energy models for retrofit analysis under uncertainty', *Energy and Buildings*, vol. 47, pp. 550–60.

Hobbs, B.F. & Meier, P. 2000, *Energy Decisions and the Environment: A Guide to the Use of Multicriteria Methods*, Springer Science & Business Media, New York.

Hong, T., Piette, M.A., Chen, Y., Lee, S.H., Taylor-Lange, S.C., Zhang, R., Sun, K. & Price, P. 2015, 'Commercial building energy saver: an energy retrofit analysis toolkit', *Applied Energy*, vol. 159, pp. 298–309.

Hong, Y., Deng, W., Ezeh, C.I. & Peng, Z. 2019, 'Attaining sustainability in built environment: Review of green retrofit measures for existing buildings', *IOP Conference Series: Earth and Environmental Science*, vol. 227, IOP Publishing, pp. 042051.

Hoogmartens, R., Van Passel, S., Van Acker, K. & Dubois, M. 2014, 'Bridging the gap between LCA, LCC and CBA as sustainability assessment tools', *Environmental Impact Assessment Review*, vol. 48, pp. 27–33.

Horr, Y.A., Arif, M., Kaushik, A., Mazroei, A., Katafygiotou, M. & Elsarrag, E. 2016, 'Occupant productivity and office indoor environment quality: A review of the literature', *Building and Environment*, vol. 105, pp. 369–89.

Hossain, M.U., Ng, S.T., Antwi-Afari, P. & Amor, B. 2020, 'Circular economy and the construction industry: Existing trends, challenges and prospective framework for sustainable construction', *Renewable and Sustainable Energy Reviews*, vol. 130, p. 109948.

Huertas-Valdivia, I., Ferrari, A.M., Settembre-Blundo, D. & García-Muiña, F.E. 2020, 'Social life-cycle assessment: A review by bibliometric analysis', *Sustainability*, vol. 12, no. 15, pp. 6211.

Hughes, C. 2012, *Qualitative and quantitative approaches*, <URL: <u>http://tinyurl</u>. com/bmztxp8>.

Huijbregts, M.A., Steinmann, Z.J., Elshout, P.M., Stam, G., Verones, F., Vieira, M., Zijp, M., Hollander, A. & van Zelm, R. 2017, 'ReCiPe2016: a harmonised life cycle impact assessment method at midpoint and endpoint level', *The International Journal of Life Cycle Assessment*, vol. 22, no. 2, pp. 138–47.

Hunkeler, D. & Rebitzer, G. 2005, 'The future of life cycle assessment', *The International Journal of Life Cycle Assessment*, vol. 10, pp. 305–8.

Hunkeler, D., Lichtenvort, K. & Rebitzer, G. 2008, *Environmental life cycle costing*, CRC Press, Florida.

Huo, T., Cai, W., Ren, H., Feng, W., Zhu, M., Lang, N. & Gao, J. 2019, 'China's building stock estimation and energy intensity analysis', *Journal of Cleaner Production*, vol. 207, pp. 801–13.

IEA 2021, Tracking Buildings 2021, IEA, Paris.

IESNA 2011, The IESNA Lighting Handbook, Illuminating Engineering Society.

Illankoon, C.S., Tam, V.W. & Le, K.N. 2018, 'Analysis on life-cycle costing for insulated external walls in Australia', *International Journal of Innovation, Management and Technology*, vol. 9, no. 1.

Ilter, D. & Ergen, E. 2015, 'BIM for building refurbishment and maintenance: current status and research directions', *Structural Survey*, vol. 33, no. 3, pp. 228–56.

Ingrao, C., Messineo, A., Beltramo, R., Yigitcanlar, T. & Ioppolo, G. 2018, 'How can life cycle thinking support sustainability of buildings? Investigating life cycle assessment applications for energy efficiency and environmental performance', *Journal of Cleaner Production*, vol. 201, pp. 556–69.

Invidiata, A., Lavagna, M. & Ghisi, E. 2018, 'Selecting design strategies using multi-criteria decision making to improve the sustainability of buildings', *Building and Environment*, vol. 139, pp. 58–68.

ISO 2006a, Environmental management - Life cycle assessment - Principles and framwork, International Standard Organisation, UK.

ISO 2006b, Environmental management - Life cycle assessment - Requirements and guidelines, International Standard Organisation, UK.

ISO 2012, Environmental management – Eco-efficiency assessment of product systems – Principles, requirements and guidelines, British Standards Institute.

ISO 2017, Buildings and constructed assets – Service life planning – Part 5: Life-cycle costing (ISO 15686-5: 2017), International Standardisation Organisation, UK.

Jafari, A. & Valentin, V. 2015, 'Decision-making life-cycle cost analysis model for energyefficient housing retrofits', *International Journal of Sustainable Building Technology and Urban Development*, vol. 6, no. 3, pp. 173–87.

Jafari, A., Valentin, V. & Russell, M. 2014, 'Probabilistic life cycle cost model for sustainable housing retrofit decision-making', *Computing in Civil and Building Engineering (2014)*, pp. 1925–33.

Jagarajan, R., Asmoni, M.N.A.M., Mohammed, A.H., Jaafar, M.N., Mei, J.L.Y. & Baba, M. 2017, 'Green retrofitting–A review of current status, implementations and challenges', *Renewable and Sustainable Energy Reviews*, vol. 67, pp. 1360–8.

James, P. 2015, Urban sustainability in theory and practice: circles of sustainability, Routledge.

Jappha, Y. 2018, Replacing fluorescents with T8 LED light bulbs to power up and save, Conservation Mart, viewed 2 October 2022, <https://www.conservationmart.com/blog/index.php/replacing-fluorescents-with-t8-led-lightbulbs-to-power-up-and-save/>.

Jeswani, H.K., Azapagic, A., Schepelmann, P. & Ritthoff, M. 2010, 'Options for broadening and deepening the LCA approaches' *Journal of Cleaner Production*, vol. 18, no. 2, pp. 120–7.

Jiang, Y., Zhao, D., Wang, D. & Xing, Y. 2019, 'Sustainable performance of buildings through modular prefabrication in the construction phase: A comparative study', *Sustainability*, vol. 11, no. 20, pp. 5658.

Jin, Y., Yan, D., Zhang, X., An, J. & Han, M. 2021, 'A data-driven model predictive control for lighting system based on historical occupancy in an office building: Methodology development', *Building Simulation*, vol. 14, Springer, pp. 219–35.

Jones, D. & Tamiz, M. 2010, Practical goal programming, vol. 141, Springer.

Joro, T., Korhonen, P. & Wallenius, J. 1998, 'Structural comparison of data envelopment analysis and multiple objective linear programming', *Management Science*, vol. 44, no. 7, pp. 962–70.

Joumard, R. & Nicolas, J.-P. 2010, 'Transport project assessment methodology within the framework of sustainable development', *Ecological Indicators*, vol. 10, no. 2, pp. 136–42.

Juan, Y.-K., Gao, P. & Wang, J. 2010, 'A hybrid decision support system for sustainable office building renovation and energy performance improvement', *Energy and Buildings*, vol. 42, no. 3, pp. 290–7.

Juliardi, R., Misnan, M., Khalid, A. & Haron, L. 2019, 'The problems in renovation & retrofitting works to achieve green building performance: a review', *International Graduate Conference of Built Environment & Surveying*, vol. 1, pp. 295–303.

Kalverkamp, M. & Karbe, N. 2019, 'Comparability of life cycle assessments: modelling and analyzing LCA using different databases', *Cascade use in technologies 2018*, Springer, pp. 51–63.

Kanazawa, M. 2017, *Research methods for environmental studies: A social science approach*, Routledge.

Karaca, F., Guney, M., Kumisbek, A., Kaskina, D. & Tokbolat, S. 2020, 'A new stakeholder opinion-based rapid sustainability assessment method (RSAM) for existing residential buildings', *Sustainable Cities and Society*, vol. 60, pp. 102155.

Kassab, M. 2013, 'An integrated approach of AHP and NFRs framework', *IEEE 7th International Conference on Research Challenges in Information Science (RCIS)*, IEEE, pp. 1–8.

Kats, G. 2003, *Green building costs and financial benefits*, Massachusetts Technology Collaborative Boston.

Kaushik, A., Arif, M., Tumula, P. & Ebohon, O.J. 2020, 'Effect of thermal comfort on occupant productivity in office buildings: Response surface analysis', *Building and Environment*, vol. 180, pp. 107021.

Keef, S.P. & Roush, M.L. 2001, 'Discounted cash flow methods and the fallacious reinvestment assumption: a review of recent texts', *Accounting Education*, vol. 10, no. 1, pp. 105–16.

Kibert, C.J. 1994, 'Establishing principles and a model for sustainable construction', *Proceedings of the first international conference of CIB TG 16 on sustainable construction*, pp. 3–12.

Kibert, C.J. 2016, *Sustainable construction: Green building design and delivery*, John Wiley & Sons.

Kiker, G.A., Bridges, T.S., Varghese, A., Seager, T.P. & Linkov, I. 2005, 'Application of multicriteria decision analysis in environmental decision making', *Integrated Environmental Assessment and Management: An international journal*, vol. 1, no. 2, pp. 95–108.

Kim, A.A., Wang, S. & McCunn, L.J. 2019, 'Building value proposition for interactive lighting systems in the workplace: Combining energy and occupant perspectives', *Journal of Building Engineering*, vol. 24, pp. 100752.

Kim, J.J. 2017, 'Economic analysis on energy saving technologies for complex manufacturing building', *Resources, Conservation and Recycling*, vol. 123, pp. 249–54.

Klöpffer, W. 2003, 'Life-cycle based methods for sustainable product development', *The international journal of life cycle assessment*, vol. 8, no 3, pp. 157–9.

Klöpffer, W. 2008, 'Life cycle sustainability assessment of products', *The International Journal of Life Cycle Assessment*, vol. 13, no. 2, pp. 89–95.

Kneifel, J. 2010, 'Life-cycle carbon and cost analysis of energy efficiency measures in new commercial buildings', *Energy and Buildings*, vol. 42, no. 3, pp. 333–40.

Kohler, N., König, H., Kreissig, J. & Lützkendorf, T. 2010, *A life cycle approach to buildings: Principles-Calculations-Design tools*, Walter de Gruyter.

Kok, N., Miller, N. & Morris, P. 2012, 'The economics of green retrofits', *Journal of Sustainable Real Estate*, vol. 4, no. 1, pp. 4–22.

Kovacic, I. & Zoller, V. 2015, 'Building life cycle optimization tools for early design phases', *Energy*, vol. 92, pp. 409–19.

Krajnc, D. & Glavič, P. 2005, 'A model for integrated assessment of sustainable development', *Resources, Conservation and Recycling*, vol. 43, no. 2, pp. 189–208.

Krarti, M. & Deneuville, A. 2015, 'Comparative evaluation of optimal energy efficiency designs for French and US office buildings', *Energy and Buildings*, vol. 93, pp. 332–44.

Kremling, A., Bruns, F. & Schildmann, J. 2019, 'Ceteris paribus" and morally relevant facts', *The American Journal of Bioethics: AJOB*, vol. 19, no. 3, pp. 66–7.

Krstić-Furundžić, A., Vujošević, M. & Petrovski, A. 2019, 'Energy and environmental performance of the office building facade scenarios', *Energy*, vol. 183, pp. 437–47.

Kubba, S. 2010, *Green construction project management and cost oversight*, Architectural Press (an imprint of ELsevier), Oxford, UK.

Kuhlman, T. & Farrington, J. 2010, 'What is sustainability?', *Sustainability*, vol. 2, no. 11, pp. 3436–48.

Kühnen, M. & Hahn, R. 2017, 'Indicators in social life cycle assessment: A review of frameworks, theories, and empirical experience', *Journal of Industrial Ecology*, vol. 21, no. 6, pp. 1547–65.

Kumar, D., Alam, M., Zou, P.X., Sanjayan, J.G. & Memon, R.A. 2020, 'Comparative analysis of building insulation material properties and performance', *Renewable and Sustainable Energy Reviews*, vol. 131, pp. 110038.

la Cruz-Lovera, D., Perea-Moreno, A.-J., la Cruz-Fernández, D., Alvarez-Bermejo, J.A. & Manzano-Agugliaro, F. 2017, 'Worldwide research on energy efficiency and sustainability in public buildings', *Sustainability*, vol. 9, no. 8, pp. 1294.

Lai, L.S. & To, W.M. 2015, 'Content analysis of social media: a grounded theory approach', *Journal of Electronic Commerce Research*, vol. 16, no. 2, pp. 138.

Landuyt, L., De Turck, S., Laverge, J., Steeman, M. & Van Den Bossche, N. 2021, 'Balancing environmental impact, energy use and thermal comfort: Optimizing insulation levels for The Mobble with standard HVAC and personal comfort systems', *Building and Environment*, vol. 206, pp. 108307.

Langston, C., Wong, F.K., Hui, E.C. & Shen, L.-Y. 2008, 'Strategic assessment of building adaptive reuse opportunities in Hong Kong', *Building and Environment*, vol. 43, no. 10, pp. 1709–18.

Lani, N.H.M., Syafiuddin, A., Yusop, Z. & bin Mat Amin, M.Z. 2018, 'Performance of small and large scales stormwater harvesting systems in commercial buildings under different reliability and future water tariff scenarios', *Science of the Total Environment*, vol. 636, pp. 1171–9.

Larsen, V.G., Tollin, N., Sattrup, P.A., Birkved, M. & Holmboe, T. 2022, 'What are the challenges in assessing circular economy for the built environment? A literature review on integrating LCA, LCC and S-LCA in life cycle sustainability assessment, LCSA', *Journal of Building Engineering*, vol. 50, pp. 104203.

Latham, D. 2016, Creative reuse of buildings: Volume One, Routledge.

Lazar, N. & Chithra, K. 2020, 'A comprehensive literature review on development of Building Sustainability Assessment Systems', *Journal of Building Engineering*, vol. 32, pp. 101450.

Lecamwasam, L., Wilson, J. & Chokolich, D. 2012, *Guide to best practice maintenance & operation of HVAC systems for energy efficiency*, Department of Climate Change and Energy Efficiency.

Lee, J.-Y., Wargocki, P., Chan, Y.-H., Chen, L. & Tham, K.-W. 2020, 'How does indoor environmental quality in green refurbished office buildings compare with the one in new certified buildings?', *Building and Environment*, vol. 171, pp. 106677.

Lee, J., Shepley, M.M. & Choi, J. 2019, 'Exploring the effects of a building retrofit to improve energy performance and sustainability: A case study of Korean public buildings', *Journal of Building Engineering*, vol. 25, pp. 100822.

Lei, H., Li, L., Yang, W., Bian, Y. & Li, C.-Q. 2021, 'An analytical review on application of life cycle assessment in circular economy for built environment', *Journal of Building Engineering*, vol. 44, pp. 103374.

Lenzen, M. 2008, 'Errors in conventional and Input-Output-based Life-Cycle inventories', *Journal of Industrial Ecology*, vol. 4, no. 4, pp. 127–48.

Lester, S. 1999, *An introduction to phenomenological research*, Stan Lester Developments, Taunton UK.

Li, C., Ding, K. & Runeson, K. 2018, 'A model for assessing the social impacts of building upgrades in China', 6th CIB International Conference: Smart and Sustainable Built Environment.

Li, D., Chen, H., Hui, E.C., Zhang, J. & Li, Q. 2013, 'A methodology for estimating the lifecycle carbon efficiency of a residential building', *Building and Environment*, vol. 59, pp. 448– 55.

Li, D., Xu, X., Chen, C.-f. & Menassa, C. 2019, 'Understanding energy-saving behaviors in the American workplace: A unified theory of motivation, opportunity, and ability', *Energy Research & Social Science*, vol. 51, pp. 198–209.

Li, M. 2015, 'Research of building life cycle energy consumption calculation based on BIM', Master dissertation thesis, Qingdao Technological University.

Li, M., Cao, J., Xiong, M., Li, J., Feng, X. & Meng, F. 2018, 'Different responses of cooling energy consumption in office buildings to climatic change in major climate zones of China', *Energy and Buildings*, vol. 173, pp. 38–44.

Li, X., Chen, S., Li, H., Lou, Y. & Li, J. 2020, 'Multi-dimensional analysis of air-conditioning energy use for energy-saving management in university teaching buildings', *Building and Environment*, vol. 185, pp. 107246.

Liang, J., Qiu, Y., Ruddell, B.L., Dalrymple, M., Earl, S., Castelazo, A. & James, T. 2018, 'Do energy retrofits work? Evidence from commercial and residential buildings in Phoenix', *Journal of Environmental Economics and Management*, pp. 1–18.

Likert, R. 1974, 'A method of constructing an attitude scale', *Scaling: A sourcebook for behavioral scientists*, pp. 233–43.

Lippiatt, B. 1998, 'Building for environmental and economic sustainability (BEES)', Construction and the environment, CIB World Congress, Gävle, Sweden, June.

Lippiatt, B.C. 1999, 'Selecting cost-effective green building products: BEES approach', *Journal of Construction Engineering and Management*, vol. 125, no. 6, pp. 448–55.

Liu, G., Li, X., Tan, Y. & Zhang, G. 2020, 'Building green retrofit in China: Policies, barriers and recommendations', *Energy Policy*, vol. 139, pp. 111356.

Liu, G., Tan, Y. & Li, X., 2020. 'China's policies of building green retrofit: A state-of-the-art overview', *Building and Environment*, vol. 169, pp. 106554.

Liu, J. 2014, 'Developing a life cycle assessment model for measuring sustainable performance of buildings in China', Doctoral dissertation thesis, University of Technology Sydney.

Liu, J., Ding, G. & Samali, B. 2013, 'Building sustainable score (BSS)-A hybrid process approach for sustainable building assessment in China', *Journal of Power and Energy Engineering*, vol. 1, pp. 58–62.

Liu, K. & Leng, J. 2021, 'Quantified CO2-related indicators for green building rating systems in China', *Indoor and Built Environment*, vol. 30, no. 6, pp. 763–76.

Liu, S. & Qian, S. 2019, 'Evaluation of social life-cycle performance of buildings: Theoretical framework and impact assessment approach', *Journal of Cleaner Production*, vol. 213, pp. 792–807.

Liu, S. & Wang, J. 2019, 'Climatic adaptability design strategy of residential zones based on Climate Consultant and Ecotect analysis – Taking Weidong New Town community as an example', *IOP Conference Series: Earth and Environmental Science*, vol. 384, IOP Publishing, pp. 012017.

Liu, Z., Zhou, Q., Tian, Z., He, B.-j. & Jin, G. 2019, 'A comprehensive analysis on definitions, development, and policies of nearly zero energy buildings in China', *Renewable and Sustainable Energy Reviews*, vol. 114, pp. 109314.

Llatas, C., Soust-Verdaguer, B. & Passer, A. 2020, 'Implementing life cycle sustainability assessment during design stages in Building Information Modelling: From systematic literature review to a methodological approach', *Building and Environment*, p. 107164.

Lohmann, J.R. 1988, 'The IRR, NPV and the fallacy of the reinvestment rate assumptions', *The Engineering Economist*, vol. 33, no. 4, pp. 303–30.

Lu, Y., Li, P., Lee, Y.P. & Song, X. 2021, 'An integrated decision-making framework for existing building retrofits based on energy simulation and cost-benefit analysis', *Journal of Building Engineering*, vol. 43, pp. 103200.

Lueddeckens, S., Saling, P. & Guenther, E. 2020, 'Temporal issues in life cycle assessment— a systematic review', *The International Journal of Life Cycle Assessment*, pp. 1–17.

Luo, M., Wang, Z., Brager, G., Cao, B. & Zhu, Y. 2018, 'Indoor climate experience, migration, and thermal comfort expectation in buildings', *Building and Environment*, vol. 141, pp. 262–72.

Luo, X., Lu, J. & Ge, J. 2021, 'Green and energy-saving reform technology research in traditional houses-taking Luo's house in Hangzhou as an example', *Journal of Asian Architecture and Building Engineering*, vol. 20, no. 2, pp. 165–78.

Ma, J.H., Lee, J.K. & Cha, S.H. 2022, 'Effects of lighting CCT and illuminance on visual perception and task performance in immersive virtual environments', *Building and Environment*, vol. 209, pp. 108678.

Ma, Z., Cooper, P., Daly, D. & Ledo, L. 2012, 'Existing building retrofits: Methodology and state-of-the-art', *Energy and Buildings*, vol. 55, pp. 889–902.

Mahmoud, S., Zayed, T. & Fahmy, M. 2019, 'Development of sustainability assessment tool for existing buildings', *Sustainable Cities and Society*, vol. 44, pp. 99–119.

Majeau-Bettez, G., Strømman, A.H. & Hertwich, E.G. 2011, 'Evaluation of process-and inputoutput-based life cycle inventory data with regard to truncation and aggregation issues', *Environmental Science & Technology*, vol. 45, no. 23, pp. 10170–7.

Maltz, E., Bi, H.H. & Bateman, M. 2018, 'Benchmarking sustainability performance: the next step in building sustainable business models', *Journal of Public Affairs*, vol. 18, no. 3, p. e1606.

Manzo, S. & Salling, K.B. 2016, 'Integrating life-cycle assessment into transport cost-benefit analysis', *Transportation Research Procedia*, vol. 14, pp. 273–82.

Marzouk, M., Azab, S. & Metawie, M. 2018, 'BIM-based approach for optimizing life cycle costs of sustainable buildings', *Journal of Cleaner Production*, vol. 188, pp. 217–26.

Mason, J. 2018, *Qualitative researching*, 3rd edn, Sage, London.

Mateus, R. & Bragança, L. 2011, 'Sustainability assessment and rating of buildings: Developing the methodology SBTool PT–H', *Building and Environment*, vol. 46, no. 10, pp. 1962–71.

Mathiyazhagan, K., Gnanavelbabu, A. & Prabhuraj, B.L. 2018, 'A sustainable assessment model for material selection in construction industries perspective using hybrid MCDM approaches', *Journal of Advances in Management Research*, vol. 16, no.2, pp. 234–59.

Mattoni, B., Guattari, C., Evangelisti, L., Bisegna, F., Gori, P. & Asdrubali, F. 2018, 'Critical review and methodological approach to evaluate the differences among international green building rating tools', *Renewable and Sustainable Energy Reviews*, vol. 82, pp. 950–60.

Maynard, M. & June, P. 1994, Researching women's lives from a feminist perspective, Taylor&Francis, London.

Meex, E., Hollberg, A., Knapen, E., Hildebrand, L. & Verbeeck, G. 2018, 'Requirements for applying LCA-based environmental impact assessment tools in the early stages of building design', *Building and Environment*, vol. 133, pp. 228–36.

Menadue, V.R. 2014, 'Green commercial office buildings: environmental performance and user perceptions', Doctoral dissertation thesis, University of Adelaide.

Menassa, C.C. 2011, 'Evaluating sustainable retrofits in existing buildings under uncertainty', *Energy and Buildings*, vol. 43, no. 12, pp. 3576–83.

Menassa, C.C. & Baer, B. 2014, 'A framework to assess the role of stakeholders in sustainable building retrofit decisions', *Sustainable Cities and Society*, vol. 10, pp. 207–21.

Merlet, Y., Rouchier, S., Arnaud, J., Cellier, N. & Woloszyn, M. 2021, 'Integration of phasing on multi-objective optimization of building stock energy retrofit', *Energy and Buildings*, pp. 111776.

Mertler, C.A. 2018, Introduction to educational research, Sage Publications.

Metallidou, C.K., Psannis, K.E. & Egyptiadou, E.A. 2020, 'Energy efficiency in smart buildings: IoT approaches', *IEEE Access*, vol. 8, pp. 63679–99.

Miah, J., Koh, S. & Stone, D. 2017, 'A hybridised framework combining integrated methods for environmental life cycle assessment and life cycle costing', *Journal of Cleaner Production*, vol. 168, pp. 846–66.

Mickaityte, A., Zavadskas, E.K., Kaklauskas, A. & Tupenaite, L. 2008, 'The concept model of sustainable buildings refurbishment', *International Journal of Strategic Property Management*, vol. 12, no. 1, pp. 53–68.

Mikulić, D., Bakarić, I.R. & Slijepčević, S. 2016, 'The economic impact of energy saving retrofits of residential and public buildings in Croatia', *Energy Policy*, vol. 96, pp. 630–44.

Miller, E. & Buys, L. 2008, 'Retrofitting commercial office buildings for sustainability: Tenants' perspectives', *Journal of Property Investment & Finance*, vol. 26, no. 6, pp. 552–61.

Miraj, P., Berawi, M.A. & Utami, S.R. 2021, 'Economic feasibility of green office building: combining life cycle cost analysis and cost-benefit evaluation', *Building Research & Information*, vol. 49, no. 6, pp. 624–38.

Mohammad, N.H. 2021, 'Multi-criteria decision making for environmental monitoring applications using fuzzy optimization algorithm', *Turkish Journal of Computer and Mathematics Education (TURCOMAT)*, vol. 12, no. 7, pp. 3278–85.

MOHURD, Ministry of Housing and Urban-Rural Development of the People's Republic of China. 2010, Code for Design of Sound Insulation of Civil Buildings.

MOHURD, Ministry of Housing and Urban-Rural Development of the People's Republic of China. 2013, Standard for Daylight design of Buildings.

MOHURD, Ministry of Housing and Urban-Rural Development of the People's Republic of China. 2014, *Assessment Standard for Green Building*, China Architecture & Building Press, China.

MOHURD, Ministry of Housing and Urban-Rural Development of the People's Republic of China. 2015, *GB50189 - 2015 Design standard for energy efficiency of public buildings*, China.

MOHURD, Ministry of Housing and Urban-Rural Development of the People's Republic of China. 2019, GB/T50378-2019 Assessment Standard for Green Building, China Architecture & Building Press, China.

Moslehi, S. & Reddy, T.A. 2019, 'A new quantitative life cycle sustainability assessment framework: Application to integrated energy systems', *Applied Energy*, vol. 239, pp. 482–93.

Motawa, I. & Carter, K. 2013, 'Sustainable BIM-based evaluation of buildings', *Procedia-Social and Behavioral Sciences*, vol. 74, pp. 419–28.

Motesharrei, S., Rivas, J., Kalnay, E., Asrar, G.R., Busalacchi, A.J., Cahalan, R.F., Cane, M.A., Colwell, R.R., Feng, K. & Franklin, R.S. 2016, 'Modeling sustainability: population, inequality, consumption, and bidirectional coupling of the Earth and Human Systems', *National Science Review*, vol. 3, no. 4, pp. 470–94.

Munarim, U. & Ghisi, E. 2016, 'Environmental feasibility of heritage buildings rehabilitation', *Renewable and Sustainable Energy Reviews*, vol. 58, pp. 235–49.

Munier, N., Hontoria, E. & Jiménez-Sáez, F. 2019, *Strategic approach in multi-criteria decision making – A practical guide for complex scenario*, vol. 275, Springer.

Murphy, J.P. 1990, Pragmatism from Peirce to Davidson Boulder, CO: Westview.

Naderifar, M., Goli, H. & Ghaljaie, F.J.S.i.D.o.M.E. 2017, 'Snowball sampling: A purposeful method of sampling in qualitative research', *Strides in development of medical education*, vol. 14, no. 3, pp. 1–6.

Newsham, G.R., Birt, B.J., Arsenault, C., Thompson, A.J., Veitch, J.A., Mancini, S., Galasiu, A.D., Gover, B.N., Macdonald, I.A. & Burns, G.J. 2013, 'Do 'green'buildings have better indoor environments? New evidence', *Building Research & Information*, vol. 41, no. 4, pp. 415–34.

Newton, P., Hampson, K. & Drogemuller, R. 2009, *Technology, design and process innovation in the built environment*, vol. 10, Routledge.

Nguyen, D.M. 2017, 'Maintenance practices-improving sustainability performance of existing office buildings: an Australian case study', Doctoral dissertation thesis, University of Technology Sydney.

National Health and Medical Research Council & Australian Research Council, Universities Australia 2018, *Australian code for the responsible conduct of research*, National Health and Medical Research Council

Niemelä, T., Levy, K., Kosonen, R. & Jokisalo, J. 2017, 'Cost-optimal renovation solutions to maximize environmental performance, indoor thermal conditions and productivity of office buildings in cold climate', *Sustainable Cities Society*, vol. 32, pp. 417–34.

Nijkamp, P., Rietveld, P. & Voogd, H. 1990, *Multicriteria evaluation in physical planning*, North-Holland, New York.
NIST, National Institute of Standards and Technology 2020, *BEES*, viewed 13 September 2022, https://www.nist.gov/services-resources/software/bees>.

Nowak, M. 2010, 'Trade-off analysis in discrete decision making problems under risk', *New developments in multiple objective and goal programming*, Springer, pp. 103–15.

Olsen, W. 2012, Data collection: Key debates and methods in social research, Sage, London.

Omann, I. & Spangenberg, J.H. 2002, 'Assessing social sustainability', *Biennial Conference of the International Society for Ecological Economics*, vol. 7, pp. 304.

Onat, N.C., Kucukvar, M. & Tatari, O. 2014, 'Integrating triple bottom line input–output analysis into life cycle sustainability assessment framework: the case for US buildings', *The International Journal of Life Cycle Assessment*, vol. 19, no. 8, pp. 1488–505.

Ongpenga, J.M.C., Rabea, B.I.B., Razonb, L.F., Avisob, K.B. & Tanb, R.R. 2020, 'Multicriteria decision analysis for energy retrofit in buildings', *Chemical Engineering*, vol. 81.

Oregi, X., Hernandez, P. & Hernandez, R. 2017, 'Analysis of life-cycle boundaries for environmental and economic assessment of building energy refurbishment projects', *Energy and Buildings*, vol. 136, pp. 12–25.

Ornaghi, C. & Van Beveren, I. 2011, 'Using proxy variables to control for unobservables when estimating productivity: A sensitivity analysis', *Available at SSRN 1984573*.

Orum, A.M., Feagin, J.R. & Sjoberg, G. 1991, 'Introduction: The nature of the case study', *A case for the case study*, pp. 1–26.

Ossadnik, W., Schinke, S. & Kaspar, R.H., 2016, 'Group aggregation techniques for analytic hierarchy process and analytic network process: A comparative analysis', *Group Decision and Negotiation*, vol. 25, no. 2, pp.421–457.

Ostermeyer, Y., Wallbaum, H. & Reuter, F. 2013, 'Multidimensional Pareto optimization as an approach for site-specific building refurbishment solutions applicable for life cycle sustainability assessment', *The International Journal of Life Cycle Assessment*, vol. 18, no. 9, pp. 1762–79.

Pang, Z., O'Neill, Z., Li, Y. & Niu, F. 2020, 'The role of sensitivity analysis in the building performance analysis: A critical review', *Energy and Buildings*, vol. 209, pp. 109659.

Park, C. 2016, Contemporary engineering economics, Global edition.

Park, J., Loftness, V., Aziz, A. & Wang, T.-H. 2021, 'Strategies to achieve optimum visual quality for maximum occupant satisfaction: Field study findings in office buildings', *Building and Environment*, vol. 195, pp. 107458.

Parker, C., Scott, S. & Geddes, A. 2019, 'Snowball sampling', SAGE research methods foundations.

Pauer, E., Wohner, B. & Tacker, M. 2020, 'The influence of database selection on environmental impact results. Life cycle assessment of packaging using gabi, ecoinvent 3.6, and the environmental footprint database', *Sustainability*, vol. 12, no. 23, p. 9948.

Penny, J. 2015, Add smart shading to solar strategies - Strike a balance between daylighting vs. solar heat gain, *Buildings*.

Peri, G., Traverso, M., Finkbeiner, M. & Rizzo, G. 2012, 'The cost of green roofs disposal in a life cycle perspective: Covering the gap', *Energy*, vol. 48, no. 1, pp. 406-14.

Phillips, R., Troup, L., Fannon, D. & Eckelman, M.J. 2020, 'Triple bottom line sustainability assessment of window-to-wall ratio in US office buildings', *Building and Environment*, vol. 182, pp. 107057.

Piccardo, C., Dodoo, A. & Gustavsson, L. 2020, 'Retrofitting a building to passive house level: A life cycle carbon balance', *Energy and Buildings*, vol. 223, pp. 110135.

Pombo, O., Rivela, B. & Neila, J. 2016, 'The challenge of sustainable building renovation: assessment of current criteria and future outlook', *Journal of Cleaner Production*, vol. 123, pp. 88–100.

Pomponi, F. & Moncaster, A. 2017, 'Circular economy for the built environment: A research framework', *Journal of Cleaner Production*, vol. 143, pp. 710–8.

Power, A. 2010, 'Housing and sustainability: demolition or refurbishment?', *Proceedings of the Institution of Civil Engineers-Urban Design and Planning*, vol. 163, no. 4, pp. 205–16.

Praseeda, K., Reddy, B.V. & Mani, M. 2015, 'Embodied energy assessment of building materials in India using process and input–output analysis', *Energy and Buildings*, vol. 86, pp. 677–86.

Purvis, B., Mao, Y. & Robinson, D. 2019, 'Three pillars of sustainability: in search of conceptual origins', *Sustainability Science*, vol. 14, no. 3, pp. 681–95.

Rasmussen, F.N., Zimmermann, R.K., Kanafani, K., Andersen, C. and Birgisdóttir, H., 2020, 'The choice of reference study period in building LCA–case-based analysis and arguments', In IOP Conference Series: Earth and Environmental Science, vol. 588, no. 3, pp. 032029

Razavi, S. & Gupta, H.V. 2015, 'What do we mean by sensitivity analysis? The need for comprehensive characterization of "global" sensitivity in Earth and Environmental systems models', *Water Resources Research*, vol. 51, no. 5, pp. 3070–92.

Rehm, M. & Ade, R. 2013, 'Construction costs comparison between 'green' and conventional office buildings', *Building Research & Information*, vol. 41, no. 2, pp. 198–208.

Remøy, H.T. & Wilkinson, S.J. 2012, 'Office building conversion and sustainable adaptation: a comparative study', *Property Management*, vol. 30, no. 3, pp. 218–31.

Rey, E. 2004, 'Office building retrofitting strategies: multicriteria approach of an architectural and technical issue', *Energy and Buildings*, vol. 36, no. 4, pp. 367–72.

RICS, the Royal Institution of Chartered Surveyors 2018, 'Recommended guidelines for building component lifespans in whole building life cycle assessment'.

Riyanto, I., Margatama, L., Hakim, H. & Hindarto, D.E. 2018, 'Motion sensor application on building lighting installation for energy saving and carbon reduction joint crediting mechanism', *Applied System Innovation*, vol. 1, no. 3, p. 23.

Rodrigues, C. & Freire, F. 2017, 'Building retrofit addressing occupancy: An integrated cost and environmental life-cycle analysis', *Energy and Buildings*, vol. 140, pp. 388-98.

Rodrigues, V., Martins, A.A., Nunes, M.I., Quintas, A., Mata, T.M. & Caetano, N.S. 2018, 'LCA of constructing an industrial building: Focus on embodied carbon and energy', *Energy Procedia*, vol. 153, pp. 420–5.

Romero, C. 2001, 'Extended lexicographic goal programming: a unifying approach', *Omega*, vol. 29, no. 1, pp. 63-71.

Saade, M.R.M., Guest, G. & Amor, B. 2020, 'Comparative whole building LCAs: How far are our expectations from the documented evidence?', *Building and Environment*, vol. 167, p. 106449.

Saaty, T.L. 1980, *The analytic hierarchy process: Planning, priorisetting, resource allocation*, McGraw-Hill, New York.

Saaty, T.L. 1990, 'How to make a decision: the analytic hierarchy process', *European Journal* of Operational Research, vol. 48, no. 1, pp. 9–26.

Saaty, T.L. 2008, 'Decision making with the analytic hierarchy process', *International Journal of Services Sciences*, vol. 1, no. 1, pp. 83–98.

Sadollah, A., Nasir, M. & Geem, Z.W. 2020, 'Sustainability and optimization: from conceptual fundamentals to applications', *Sustainability*, vol. 12, no. 5, p. 2027.

Salehian, S., Ismail, M.A. & Ariffin, A.R.M. 2020, 'Assessment on embodied energy of nonload bearing walls for office buildings', *Buildings*, vol. 10, no. 4, p. 79.

Saltelli, A., Aleksankina, K., Becker, W., Fennell, P., Ferretti, F., Holst, N., Li, S. & Wu, Q. 2019, 'Why so many published sensitivity analyses are false: A systematic review of sensitivity analysis practices', *Environmental Modelling & Software*, vol. 114, pp. 29–39.

Samandar, M.S. 2015, 'Construction Process Sustainability Index (CPSI): An Integrated Assessment Framework', Master dissertation thesis, North Carolina State University.

SAMR, State Administration for Market Regulation & SAC, Standardisation Administration of People's Republic of China 2020, General Rules for Calculation of the Comprehensive Energy Consumption, China.

Sanders, P. 1982, 'Phenomenology: A new way of viewing organizatinoal research', *The Academy of Management Review*, vol. 7, no. 3, pp. 353–60.

Santos, P., Pereira, A.C., Gervásio, H., Bettencourt, A. & Mateus, D. 2017, 'Assessment of health and comfort criteria in a life cycle social context: Application to buildings for higher education', *Building and Environment*, vol. 123, pp. 625–48.

Santos, R., Costa, A.A., Silvestre, J.D. & Pyl, L. 2019, 'Informetric analysis and review of literature on the role of BIM in sustainable construction', *Automation in Construction*, vol. 103, pp. 221–34.

Sartori, T., Drogemuller, R., Omrani, S. & Lamari, F. 2021, 'A schematic framework for life cycle assessment (LCA) and green building rating system (GBRS)', *Journal of Building Engineering*, vol. 38, pp. 102180.

Savitz, A. 2013, The triple bottom line: how today's best-run companies are achieving economic, social and environmental success-and how you can too, John Wiley & Sons.

Scotland, J. 2012, 'Exploring the philosophical underpinnings of research: Relating ontology and epistemology to the methodology and methods of the scientific, interpretive, and critical research paradigms', *English Language Teaching*, vol. 5, no. 9, pp. 9–16.

SDSN, Sustainable Development Solutions Network & FEEM, Fondazione Eni Enrico Mattei 2019, *Roadmap to 2050: A manual for nations to decarbonize by mid-century*, Development Solutions Network and Fondazione Eni Enrico Mattei.

Sedgwick, P. 2013, 'Snowball sampling', British Medical Journal, vol. 347.

Serghides, D., Dimitriou, S., Kyprianou, I. & Papanicolas, C.J.E.P. 2017, 'The adaptive comfort factor in evaluating the energy performance of office buildings in the Mediterranean coastal cities', *Energy Procedia*, vol. 134, pp. 683–91.

Shafique, M., Kim, R. & Rafiq, M. 2018, 'Green roof benefits, opportunities and challenges– A review', *Renewable and Sustainable Energy Reviews*, vol. 90, pp. 757–73.

Shaikh, P.H., Shaikh, F., Sahito, A.A., Uqaili, M.A. & Umrani, Z. 2017, Cost-effective energy efficient building retrofitting – Materials, technologies, optimization and case studies, Woodhead publishing.

Shan, M. & Hwang, B.-g. 2018, 'Green building rating systems: Global reviews of practices and research efforts', *Sustainable Cities and Society*, vol. 39, pp. 172–80.

Shen, C., Zhao, K. & Ge, J. 2020, 'An overview of the green building performance database', *Journal of Engineering*, vol. 2020.

Shrubsole, C., Hamilton, I., Zimmermann, N., Papachristos, G., Broyd, T., Burman, E., Mumovic, D., Zhu, Y., Lin, B. & Davies, M. 2019, 'Bridging the gap: The need for a systems thinking approach in understanding and addressing energy and environmental performance in buildings', *Indoor Built Environment*, vol. 28, no. 1, pp. 100–17.

Silva, D.A.L., Nunes, A.O., Piekarski, C.M., da Silva Moris, V.A., de Souza, L.S.M. & Rodrigues, T.O. 2019, 'Why using different Life Cycle Assessment software tools can generate different results for the same product system? A cause–effect analysis of the problem', *Sustainable Production and Consumption*, vol. 20, pp. 304–15.

Simko, T. & Moore, T. 2021, 'Optimal window designs for Australian houses', *Energy and Buildings*, vol. 250, pp. 111300.

Simonen, K. 2014, Life cycle assessment, Routledge.

Simpeh, E.K., Pillay, J.-P.G., Ndihokubwayo, R. & Nalumu, D.J. 2021, 'Improving energy efficiency of HVAC systems in buildings: A review of best practices', *International Journal of Building Pathology and Adaptation*, vol. 40, no. 2, pp. 165–82.

Singleton Jr, R. & Straits, B.C. 2018, *Approaches to social research*, 6th edn, Oxford University Press, New York.

Skulmoski, G.J., Hartman, F.T. & Krahn, J. 2007, 'The Delphi method for graduate research', *Journal of Information Technology Education: Research*, vol. 6, no. 1, pp. 1–21.

Slaper, T.F. & Hall, T.J. 2011, 'The triple bottom line: What is it and how does it work', *Indiana Business Review*, vol. 86, no. 1, pp. 4–8.

Solanki, S.K., Rastogi, R. & Paul, V. 2022, 'Cost analysis of functional retrofitting measures in buildings', *Journal of the Institution of Engineers (India): Series A*, vol. 103, no. 3, pp. 725–32.

Solla, M., Ismail, L.H. & Yunus, R. 2016, 'Investigation on the potential of integrating BIM into green building assessment tools', *ARPN Journal of Engineering and Applied Sciences*, vol. 11, no. 4, pp. 2412–8.

Song, X., Ye, C., Li, H., Wang, X. & Ma, W. 2017, 'Field study on energy economic assessment of office buildings envelope retrofitting in southern China', *Sustainable Cities and Society*, vol. 28, pp. 154–61.

Sousa, V., Silva, C.M. & Meireles, I. 2019, 'Performance of water efficiency measures in commercial buildings', *Resources, Conservation and Recycling*, vol. 143, pp. 251–9.

Soust-Verdaguer, B., Galeana, I.B., Llatas, C., Montes, M., Hoxha, E. & Passer, A. 2022, 'How to conduct consistent environmental, economic, and social assessment during the building design process. A BIM-based Life Cycle Sustainability Assessment method', *Journal of Building Engineering*, vol. 45, pp. 103516.

Souza, H.H.d.S., Evangelista, P.P.d.A., Medeiros, D.L., Albertí, J., Fullana-i-Palmer, P., Boncz, M.Á., Kiperstok, A. & Gonçalves, J.P. 2021, 'Functional unit influence on building life cycle assessment', *The International Journal of Life Cycle Assessment*, vol. 26, no. 3, pp. 435–54.

Stanitsas, M. & Kirytopoulos, K. 2021, 'Investigating the significance of sustainability indicators for promoting sustainable construction project management', *International Journal of Construction Management*, pp. 1–26.

Stansbury, J. & Mittelsdorf, A.M. 2001, 'Economic and environmental analysis of retrofitting a large office building with energy-efficient lighting systems', *Environmental Management*, vol. 27, no. 6, pp. 909–18.

Stazi, F., Naspi, F. & D'Orazio, M. 2017, 'A literature review on driving factors and contextual events influencing occupants' behaviours in buildings', *Building and Environment*, vol. 118, pp. 40–66.

Steele, K., Carmel, Y., Cross, J. & Wilcox, C. 2009, 'Uses and misuses of multicriteria decision analysis (MCDA) in environmental decision making', *Risk Analysis: An International Journal*, vol. 29, no. 1, pp. 26–33.

Sterner, E. 2000, 'Life-cycle costing and its use in the Swedish building sector', *Building Research & Information*, vol. 28, no. 5–6, pp. 387–93.

Stocker, T.F., Qin, D., Plattner, G.-K., Alexander, L.V., Allen, S.K., Bindoff, N.L., Bréon, F.-M., Church, J.A., Cubasch, U. & Emori, S. 2013, 'Technical summary', *Climate change 2013: the physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*, Cambridge University Press, UK, pp. 33–115.

Sun, N., Cui, Y. & Jiang, Y. 2018, 'Lighting and ventilation-based building sun-shading design and simulation case in cold regions', *Energy Procedia*, vol. 152, pp. 462–9.

Suzer, O. 2019, 'Analyzing the compliance and correlation of LEED and BREEAM by conducting a criteria-based comparative analysis and evaluating dual-certified projects', *Building and Environment*, vol. 147, pp. 158–70.

Tavakolan, M., Mostafazadeh, F., Eirdmousa, S.J., Safari, A. & Mirzaei, K. 2022, 'A parallel computing simulation-based multi-objective optimization framework for economic analysis of building energy retrofit: A case study in Iran', *Journal of Building Engineering*, vol. 45, pp. 103485.

Taylor, S.J., Bogdan, R. & DeVault, M. 2015, *Introduction to qualitative research methods: A guidebook and resource*, 4th edn, John Wiley & Sons, New Jersey.

Thatcher, A. 2013, 'Defining Human Factors for Sustainable Development', HFSD definition working paper. University of the Witwatersrand.

Thomas, R.M. 2003, *Blending qualitative and quantitative research methods in theses and dissertations*, Corwin Press, Thousand Oaks, California.

Thomsen, A. & Van der Flier, K. 2011, 'Understanding obsolescence: a conceptual model for buildings', *Building Research & Information*, vol. 39, no. 4, pp. 352–62.

Toosi, H.A., Lavagna, M., Leonforte, F., Del Pero, C. & Aste, N. 2020, 'Life cycle sustainability assessment in building energy retrofitting; a review', *Sustainable Cities and Society*, vol. 60, pp. 102248.

Treloar, G.J. 1997, 'Extracting embodied energy paths from input–output tables: towards an input–output-based hybrid energy analysis method', *Economic Systems Research*, vol. 9, no. 4, pp. 375–91.

Tung, Y.-K. 1992, 'Probability-distribution for benefit cost ratio and net benefit', *Journal of Water Resources Planning and Management-asce*, vol. 118, no. 2, pp. 133–50.

Tweed, C. & Sutherland, M. 2007, 'Built cultural heritage and sustainable urban development', *Landscape and Urban Planning*, vol. 83, no. 1, pp. 62–9.

Tzortzopoulos, P., Ma, L., Soliman Junior, J. & Koskela, L. 2019, 'Evaluating social housing retrofit options to support clients' decision making—SIMPLER BIM protocol', *Sustainability*, vol. 11, no. 9, pp. 2507.

UAD, Architecture Design and Research Institute of Zhejiang University, ZIAD, Zhejiang Province Institute of Architecture Design and Research & CMA, China Meteological Bureau. 2007, Design Standard for Energy Efficiency of Public Buildings, JHPress, Zhejiang, China.

Udomsap, A.D. & Hallinger, P. 2020, 'A bibliometric review of research on sustainable construction, 1994–2018', *Journal of Cleaner Production*, vol. 254, p. 120073.

UN, United Nations 2015, Transforming our world: the 2030 Agenda for Sustainable Development, United Nations.

UNEP/SETAC 2009, *Guidelines for social life cycle assessment of products*, United Nations Environment Programme, Society of Environmental Toxicology Chemistry, Belgium.

UNEP/SETAC 2011, Towards a life cycle sustainability assessment: making informed choices on products, UNEP/SETAC Life Cycle Initiative, Paris.

UNEP/SETAC 2020, Guidelines for social life cycle assessment of products and organizations 2020, United Nations Environment Programme.

United Nations Environmental Programme 2021, 2021 Global status report for buildings and construction: Towards a zero-emission, efficient and resilient buildings and construction Sector, Nairobi.

Vagias, W.M. 2006, 'Likert-type scale response anchors', Clemson International Institute for Tourism & Research Development, Department of Parks, Recreation and Tourism Management. Clemson University.

Valdes-Vasquez, R. & Klotz, L.E. 2012, 'Social sustainability considerations during planning and design: framework of processes for construction projects', *Journal of Construction Engineering and Management*, vol. 139, no. 1, pp. 80–9.

Valladares-Rendón, L., Schmid, G. & Lo, S.-L. 2017, 'Review on energy savings by solar control techniques and optimal building orientation for the strategic placement of façade shading systems', *Energy and Buildings*, vol. 140, pp. 458–79.

Van Renterghem, T., 2018. *Green roofs for acoustic insulation and noise reduction.* 'In Nature based strategies for urban and building sustainability', Butterworth-Heinemann, pp. 167-179.

Vanderbei, J.R. 2020, *Linear programming: Foundations and extensions*, Fifth edn, vol. 285, Springer.

Varma, C.S. & Palaniappan, S. 2019, 'Comparision of green building rating schemes used in North America, Europe and Asia', *Habitat International*, vol. 89, p. 101989.

Vilches, A., Garcia-Martinez, A. & Sanchez-Montañes, B. 2017, 'Life cycle assessment (LCA) of building refurbishment: A literature review', *Energy and Buildings*, vol. 135, pp. 286–301.

Vilnītis, M., Lapsa, V. & Veinbergs, A. 2019, 'Sustainable construction success indicators', *IOP Conference Series: Materials Science and Engineering*, vol. 660, IOP Publishing, pp. 012041.

Visentin, C., da Silva Trentin, A.W., Braun, A.B. & Thomé, A. 2020, 'Life cycle sustainability assessment: A systematic literature review through the application perspective, indicators, and methodologies', *Journal of Cleaner Production*, vol. 270, p. 122509.

von der Fehr, A., Sølberg, J. & Bruun, J. 2018, 'Validation of networks derived from snowball sampling of municipal science education actors', *International Journal of Research & Method in Education*, vol. 41, no. 1, pp. 38–52.

Wagiman, K.R., Abdullah, M.N., Hassan, M.Y., Radzi, N.H.M. & Kwang, T.C. 2020, 'Lighting system control techniques in commercial buildings: Current trends and future directions', *Journal of Building Engineering*, vol. 31, pp. 101342.

Walliman, N. 2016, Social research methods: The essentials, 2nd edn, Sage.

Walter, E. 2005, 'Retrofit', Cambridge Advanced Learners' Dictionary, Cambridge University Press, viewed 12 October 2022, https://dictionary.cambridge.org/dictionary/english/retrofit>.

Wang, D., Qi, T., Liu, Y., Wang, Y., Fan, J., Wang, Y. & Du, H. 2020, 'A method for evaluating both shading and power generation effects of rooftop solar PV panels for different climate zones of China', *Solar Energy*, vol. 205, pp. 432–45.

Wang, H. & Zhai, Z.J. 2016, 'Advances in building simulation and computational techniques: A review between 1987 and 2014', *Energy and Buildings*, vol. 128, pp. 319–35.

Wang, J. 2009, 'Calculation and analysis of life cycle CO2 emission of Chinese urban residential communities', Tsinghua University, Beijing.

Wang, J., Zhong, H., Yang, Z., Wang, M., Kammen, D.M., Liu, Z., Ma, Z., Xia, Q. & Kang, C. 2020, 'Exploring the trade-offs between electric heating policy and carbon mitigation in China', *Nature Communications*, vol. 11, no. 1, pp. 1–11.

Watson, K.J., Evans, J., Karvonen, A. & Whitley, T. 2016, 'Capturing the social value of buildings: The promise of Social Return on Investment (SROI)', *Building and Environment*, vol. 103, pp. 289–301.

Wei, W. & He, L.-Y. 2017, 'China building energy consumption: definitions and measures from an operational perspective', *Energies*, vol. 10, no. 5, p. 582.

Wilkinson, S.J. 2012, 'Analysing sustainable retrofit potential in premium office buildings', *Structural Survey*, vol. 30, no. 5, pp. 398–410.

Wilkinson, S.J. & Remoy, H. 2017, 'Adaptive reuse of Sydney offices and sustainability', *Sustainable Buildings*, vol. 2, pp. 6.

Wilkinson, S.J. 2014, 'How buildings learn: Adaptation of low grade commercial buildings for sustainability in Melbourne', *Facilities*, vol. 32, no. 7/8, pp. 382–95.

Wong, J.K.W. & Zhou, J. 2015, 'Enhancing environmental sustainability over building life cycles through green BIM: A review', *Automation in Construction*, vol. 57, pp. 156–65.

Woo, J., Wilsmann, J. & Kang, D. 2010, 'Use of as-built building information modeling', Construction Research Congress 2010: Innovation for Reshaping Construction Practice, pp. 538–48.

Wood, M., Daly, J., Miller, J. & Roper, M. 1999, 'Multi-method research: An empirical investigation of object-oriented technology', *Journal of Systems and Software*, vol. 48, no. 1, pp. 13–26.

Wu, G., Qiang, G., Zuo, J., Zhao, X. & Chang, R. 2018, 'What are the key indicators of mega sustainable construction projects?—a stakeholder-network perspective', *Sustainability*, vol. 10, no. 8, pp. 2939.

Wu, W., Yang, H., Chew, D., Hou, Y. & Li, Q. 2014, 'A real-time recording model of key indicators for energy consumption and carbon emissions of sustainable buildings', *Sensors*, vol. 14, no. 5, pp. 8465–84.

Wu, Z., Wang, B. & Xia, X. 2016, 'Large-scale building energy efficiency retrofit: Concept, model and control', *Energy*, vol. 109, pp. 456–65.

Wulf, C., Werker, J., Ball, C., Zapp, P. & Kuckshinrichs, W. 2019, 'Review of sustainability assessment approaches based on life cycles', *Sustainability*, vol. 11, no. 20, p. 5717.

Xu, G., Schwarz, P. & Yang, H. 2020, 'Adjusting energy consumption structure to achieve China's CO2 emissions peak', *Renewable and Sustainable Energy Reviews*, vol. 122, p. 109737.

Xu, L., Qu, J., Han, J., Zeng, J. & Li, H. 2021, 'Distribution and evolutionary in household energy-related CO2 emissions (HCEs) based on Chinese north–south demarcation', *Energy Reports*, vol. 7, pp. 6973–82.

Xu, P.P., Chan, E.H. & Qian, Q.K. 2012, 'Key performance indicators (KPI) for the sustainability of building energy efficiency retrofit (BEER) in hotel buildings in China', *Facilities*, vol. 30, no.9, pp. 432–48.

Xu, X., Xu, P., Zhu, J., Liu, J. & Xue, Q. 2022, 'How to minimize the embodied environmental impact of green building envelope? An automatic optimization method', *Environmental Impact Assessment Review*, vol. 93, pp. 106732.

Yan, H., Liu, Q., Zhang, H., Wang, H., Li, H. & Yang, L. 2019, 'Difference in the thermal response of the occupants living in northern and southern China', *Energy and Buildings*, vol. 204, pp. 109475.

Yang, J., Sun, Y., Qiu, X. & Zhang, G. 2020, 'Research on indoor environmental performance of green office buildings in Shanghai based on field investigation', *IOP Conference Series: Earth and Environmental Science*, vol. 531, IOP Publishing, pp. 012023.

Yildiz, B., Bilbao, J.I. & Sproul, A.B. 2017, 'A review and analysis of regression and machine learning models on commercial building electricity load forecasting', *Renewable and Sustainable Energy Reviews*, vol. 73, pp. 1104–22.

Yin, H. 2011, 'Decision support for building renovation strategies', *In Proceedings of the* ASCE International Workshop on Computing in Civil Engineering, pp. 834–841.

Yin, R.K. 2009, *Case study research: Design and methods*, 4th edn, vol. 5, Sage, London and Singapore.

Yolles, M. 2018, 'Sustainability development: part 2-exploring the dimensions of sustainability development', *International Journal of Markets and Business Systems*, vol. 3, no. 3, pp. 257–75.

Younis, A., Ebead, U. & Judd, S. 2018, 'Life cycle cost analysis of structural concrete using seawater, recycled concrete aggregate, and GFRP reinforcement', *Construction and Building Materials*, vol. 175, pp. 152–60.

Yu, C.-S. 2002, 'A GP-AHP method for solving group decision-making fuzzy AHP problems', *Computers Operations Research*, vol. 29, no. 14, pp. 1969–2001.

Yu, H. & Tong, X. 2021, 'Producer vs. local government: The locational strategy for end-oflife photovoltaic modules recycling in Zhejiang province', *Resources, Conservation and Recycling*, vol. 169, pp. 105484.

Yu, W., Li, B., Yang, X. & Wang, Q. 2015, 'A development of a rating method and weighting system for green store buildings in China', *Renewable Energy*, vol. 73, pp. 123–9.

Yuan, J., Nian, V. & Su, B. 2019, 'Evaluation of cost-effective building retrofit strategies through soft-linking a metamodel-based Bayesian method and a life cycle cost assessment method', *Applied Energy*, vol. 253, pp. 113573.

Yung, E.H. & Chan, E.H. 2012, 'Implementation challenges to the adaptive reuse of heritage buildings: Towards the goals of sustainable, low carbon cities', *Habitat International*, vol. 36, no. 3, pp. 352–61.

Zamagni, A., Pesonen, H.-L. & Swarr, T. 2013, 'From LCA to Life Cycle Sustainability Assessment: Concept, practice and future directions', *The International Journal of Life Cycle Assessment*, vol. 18, no. 9, pp. 1637–41.

Zarghami, E. & Fatourehchi, D. 2020, 'Comparative analysis of rating systems in developing and developed countries: A systematic review and a future agenda towards a region-based sustainability assessment', *Journal of Cleaner Production*, vol. 254, p. 120024.

Zavadskas, E.K., Turskis, Z. & Kildienė, S. 2014, 'State of art surveys of overviews on MCDM/MADM methods', *Technological and Economic Development of Economy*, vol. 20, no. 1, pp. 165–79.

Zeitz, A., Griffin, C. & Dusicka, P. 2019, 'Comparing the embodied carbon and energy of a mass timber structure system to typical steel and concrete alternatives for parking garages', *Energy and Buildings*, vol. 199, pp. 126–33.

Zhang, H., Hewage, K., Karunathilake, H., Feng, H. & Sadiq, R. 2021, 'Research on policy strategies for implementing energy retrofits in the residential buildings', *Journal of Building Engineering*, vol. 43, pp. 103161.

Zhang, X. & Wang, F. 2016, 'Assessment of embodied carbon emissions for building construction in China: Comparative case studies using alternative methods', *Energy and Buildings*, vol. 130, pp. 330–40.

Zhang, Y. 2017, 'Taking the time characteristic into account of life cycle assessment: method and application for buildings', *Sustainability*, vol. 9, no. 6, p. 922.

Zheng, D., Yu, L. & Wang, L. 2019, 'A techno-economic-risk decision-making methodology for large-scale building energy efficiency retrofit using Monte Carlo simulation', *Energy*, vol. 189, pp. 116169.

Zheng, H., Song, M. & Shen, Z. 2021, 'The evolution of renewable energy and its impact on carbon reduction in China', *Energy*, vol. 237, pp. 121639.

Zhu, W., Feng, W., Li, X. & Zhang, Z. 2020, 'Analysis of the embodied carbon dioxide in the building sector: A case of China', *Journal of Cleaner Production*, vol. 269, pp. 122438.

Zolghadr-Asli, B., Bozorg-Haddad, O., Enayati, M. & Chu, X. 2021, 'A review of 20-year applications of multi-attribute decision-making in environmental and water resources planning and management', *Environment, Development and Sustainability*, vol. 23, no. 10, pp. 14379–404.

ZPBS, Zhejiang Provincial Bureau of Statistics & SOTNBS, Survey Office of The National Bureau of Statistics in Zhejiang 2021, *Zhejiang Statistical Yearbook 2021*, China Statistics Press, Beijing, China.

Zuo, J. & Zhao, Z.-Y. 2014, 'Green building research–current status and future agenda: A review', *Renewable and Sustainable Energy Reviews*, vol. 30, pp. 271–81.

Zuo, J., Pullen, S., Rameezdeen, R., Bennetts, H., Wang, Y., Mao, G., Zhou, Z., Du, H. & Duan, H. 2017, 'Green building evaluation from a life-cycle perspective in Australia: A critical review', *Renewable and Sustainable Energy Reviews*, vol. 70, pp. 358–68.